UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	§	Chapter 11
	§	
Seahawk Drilling Inc.,	§	Case No.: 11
-	§	
Seahawk Mexico Holdings LLC,	§	Case No.: 11
	§	
Seahawk Drilling Management LLC,	§	Case No.: 11
	§	
Seahawk Offshore Management LLC,	§	Case No.: 11
Ç	§	
Energy Supply International LLC,	§	Case No.: 11
	§	
Seahawk Drilling LLC,	Š	Case No.: 11-
2	§	
Seahawk Global Holdings LLC,	§	Case No.: 11
	§	
Seahawk Drilling USA LLC,	8	Case No.: 11
-,	§	
Debtors.	§	Motion for Joint Administration Pending

DEBTORS' EMERGENCY MOTION, PURSUANT TO SECTIONS 105, 363 AND 365 OF THE BANKRUPTCY CODE AND RULES 2002, 6004, 6006, 9007, AND 9014 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE, FOR AN ORDER (i) SCHEDULING AN EXPEDITED HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS; (ii) SETTING DEADLINES FOR FILING OBJECTIONS TO (1) THE PROPOSED SALE AND (2) THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND THE PAYMENT OF CURE CLAIMS IN CONNECTION THEREWITH; AND (iii) APPROVING THE FORM AND MANNER OF NOTICE THEREOF

NOTICE UNDER BLR 9013(b) AND 9013(i)

THIS MOTION SEEKS AN ORDER THAT MAY ADVERSELY AFFECT YOU. IF YOU OPPOSE THE MOTION, YOU SHOULD IMMEDIATELY CONTACT THE MOVING PARTY TO RESOLVE THE DISPUTE. IF YOU AND THE MOVING PARTY CANNOT AGREE, YOU MUST FILE A RESPONSE AND SEND A COPY TO THE MOVING PARTY. YOU MUST

1120089110212000000000011

FILE AND SERVE YOUR RESPONSE WITHIN 21 DAYS OF THE DATE THIS WAS SERVED ON YOU. YOUR RESPONSE MUST STATE WHY THE MOTION SHOULD NOT BE GRANTED. IF YOU DO NOT FILE A TIMELY RESPONSE, THE RELIEF MAY BE GRANTED WITHOUT FURTHER NOTICE TO YOU. IF YOU OPPOSE THE MOTION AND HAVE NOT REACHED AN AGREEMENT, YOU MUST ATTEND THE HEARING. UNLESS THE PARTIES AGREE OTHERWISE, THE COURT MAY CONSIDER EVIDENCE AT THE HEARING AND MAY DECIDE THE MOTION AT THE HEARING.

EMERGENCY RELIEF HAS BEEN REQUESTED. IF THE COURT CONSIDERS THE MOTION ON AN EMERGENCY BASIS, THEN YOU WILL HAVE LESS THAN 21 DAYS TO ANSWER. IF YOU OBJECT TO THE REQUESTED RELIEF OR IF YOU BELIEVE THAT THE EMERGENCY CONSIDERATION IS NOT WARRANTED; YOU SHOULD FILE AN IMMEDIATE RESPONSE.

REPRESENTED PARTIES SHOULD ACT THROUGH THEIR ATTORNEY.

TO THE HONORABLE UNITED STATES BANKRUPTCY JUDGE:

Seahawk Drilling, Inc. ("Seahawk" of the "Company") and its above-captioned affiliated debtors (collectively, the "Debtors"), file this Emergency Motion Pursuant To Sections 105, 363 and 365 of the Bankruptcy Code And Rules 2002, 6004, 6006, 9007, and 9014 of the Federal Rules Of Bankruptcy Procedure, for an Order (i) Scheduling an Expedited Hearing To Approve The Sale of Substantially All of the Debtors' Assets; (ii) Setting Deadlines for Filing Objections To (1) the Proposed Sale and (2) the Proposed Assumption and Assignment of Executory Contracts and the Payment of Cure Claims in Connection Therewith; and (iii) Approving the Form and Manner of Notice Thereof (the "Sale Motion"). In support of this Sale Motion, the Debtors respectfully state:

JURISDICTION AND PROCEDURAL STATUS

- 1. This Court has jurisdiction over this Sale Motion pursuant to 28 U.S.C. §§ 157 and 1334. Venue of the Debtors' chapter 11 cases in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409. This is a core proceeding pursuant to 28 U.S.C. § 157(b).
- 2. On the date hereof (the "<u>Petition Date</u>"), the Debtors filed their respective voluntary petitions for relief under title 11 of the United States Code (the "<u>Bankruptcy Code</u>"). Contemporaneously herewith, the Debtors have sought joint administration of these chapter 11 cases.
- 3. The Debtors continue to operate their businesses and manage their properties as debtors-in-possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee, examiner, or creditors' committee has been appointed in these cases.

FACTUAL BACKGROUND

A. Events Leading To The Debtors' Petitions

- 1. Seahawk is a Delaware corporation formed in 2008 and headquartered in Houston, Texas, with operations throughout the U.S. Gulf of Mexico. Each of the other debtors is a direct or indirect subsidiary of Seahawk.¹ As of the Petition Date, Seahawk's stock is publicly traded on the NASDAQ Exchange under the ticker symbol, "HAWK."
- 2. Seahawk operates a jackup rig business that provides contract drilling services to the oil and natural gas exploration and production industry throughout the U.S. Gulf of Mexico.²

¹ The debtor-subsidiaries of Seahawk are: Seahawk Drilling LLC, Energy Supply International LLC, Seahawk Global Holdings LLC, Seahawk Mexico Holdings LLC, Seahawk Drilling Management LLC, Seahawk Offshore Management LLC and Seahawk Drilling USA LLC. The subsidiaries of Seahawk that are incorporated in Mexico or have branches in Mexico are not a part of this chapter 11 proceeding.

² Jackup rigs are mobile, self-elevating drilling platforms equipped with legs that can be lowered to the ocean floor until a foundation is established to support the drilling platform. The Debtors' rigs work in water depths up to 300 feet and can drill to a depth of 25,000 feet.

The Debtors contract with their customers on a dayrate basis to provide rigs and drilling crews. The Debtors' fleet of mobile offshore drilling rigs consists of twenty (20) shallow water jackup rigs. Three (3) of the Debtors' six (6) rigs in Texas are located at Matagorda Island 721 in Aransas County. Two rigs are located at Sabine Pass, Texas, and one rig is located at High Island 176 in Galveston County. The remainder of Seahawk's rigs are currently located in Louisiana.

- 3. Seahawk is a former subsidiary of Pride International, Inc. ("<u>Pride</u>"). On August 4, 2009, the Board of Directors of Pride approved a plan to separate Pride into two independent, publicly-traded companies (the "<u>Spin-Off</u>"). The separation occurred through the distribution to Pride stockholders of all of the shares of common stock of Seahawk that held, directly or indirectly, the assets and liabilities of Pride's jackup rig business.
- 4. On August 24, 2009 (the "Spin-Off Date"), each Pride stockholder received one-fifteenth (1/15) of a share of Seahawk's common stock for each share of Pride common stock held at the close of business on August 14, 2009. After the Spin-Off, Seahawk became an independent public company. Pride no longer retains any ownership interest in Seahawk, although the two companies are parties to various agreements entered into in connection with the Spin-Off (collectively, the "Pride Agreements").
- 5. Since mid-2008, the demand for drilling services has declined dramatically, principally as a result of the global financial crisis, declining prices of crude oil and natural gas and deteriorating worldwide economic conditions. The decline in the United States jackup rig market since 2009 has been one of the sharpest downturns for domestic jackup rig activity over the past thirty (30) years. In addition, the regulatory and financial uncertainties regarding a

former customer, PEMEX, have had a significant effect on Seahawk's business.³ Finally, from August 2009 to March 2010, the Debtors' active rig count averaged five (5) to seven (7) working rigs at any given time.

- 6. On April 20, 2010, the demand for offshore drilling services in the Gulf of Mexico was further negatively impacted by the Macondo well blowout, prompting, among other things, the United States Government to issue a moratorium on all U.S. offshore drilling. On May 26, 2010, the moratorium on Gulf of Mexico drilling in waters less than 500 feet deep was lifted. Subsequently, in November 2010, the deep water moratorium was lifted. However, since the Macondo incident, notwithstanding the termination of the Government-imposed drilling prohibitions, the Debtors' customers are experiencing significant delays in the issuance of drilling permits and very few new drilling permits have been issued. In addition, there is an increasingly uncertain regulatory and cost environment which continues to adversely affect the Debtors' business.
- 7. As a result of all of these factors, Debtors' active rig count declined to three (3) working rigs during October 2010. While there has been some recent marginal improvement in market conditions (seven (7) rigs are currently working), the combined impact of all of these events, together with negative cash flows throughout 2009 and 2010, has severely stressed and exhausted the Debtors' liquidity and even the current operating level results continue to produce negative cash flows for the Debtors. Seahawk's Board of Directors (the "Board") and its management have closely monitored the impact of these conditions and evaluated potential alternatives to address and improve these circumstances with the assistance of various professionals and outside advisors, including Alvarez & Marsal North America, LLC ("A&M").

³ Pemex Exploración y Producción ("<u>PEMEX</u>") is the exploration and production subsidiary of the national oil 90303148.17

- 8. In November 2010, Seahawk publicly announced that its Board had initiated a process to explore and consider possible strategic alternatives, including, but not limited to transactions involving additional funding, recapitalizations, sales of assets, or a sale or merger of Seahawk. Seahawk's Board, through its finance committee (the "Finance Committee"), with regular and substantial Board and independent directors' participation, supervised the process and were managed by Simmons & Company International ("Simmons") and other advisors.
- 9. In February 2011, after evaluating the indications of interest and potential offers received from a number of interested parties, the Board authorized the Debtors to enter into an Asset Purchase Agreement (the "APA") with Hercules Offshore, Inc. ("Hercules"). The executed APA contemplates the acquisition by Hercules or one or more of its subsidiaries ("Purchaser") of substantially all of the assets and jackup rigs of the Debtors (the "Purchased Assets") through a sale pursuant to section 363 of the Bankruptcy Code (the "Transaction"). The aggregate consideration for the Purchased Assets (defined in the APA as the "Base Aggregate Consideration") is (a) 22,321,425 shares of Hercules Common Stock (the "Hercules Shares") plus (b) cash in an amount equal to \$25,000,012 (the "Cash Payment"), subject to certain adjustments. Using the closing stock price of Hercules' stock as of February 10, 2011, the Base Aggregate Consideration would be valued at approximately \$105 million before any adjustments. The Base Aggregate Consideration is to be payable at closing by Purchaser to Sellers.
- 10. The Debtors have sought chapter 11 protection in order to protect and preserve their going concern value and to facilitate a prompt sale of substantially all of their assets to Hercules for the benefit of all stakeholders. The Debtors have determined in the prudent exercise

company of Mexico.

⁴ The summary description of the APA set forth herein is qualified in its entirety by reference to the terms of the APA itself.

of their business judgment that the proposed course of action is the best alternative to ensure that the maximum value of Seahawk's assets will inure to the benefit of all of their creditors and interestholders.

11. Additional information regarding the Debtors' business, capital structure, and the circumstances and events leading to these chapter 11 cases is contained in the Declaration of Randall D. Stilley in Support of First Day Motions, filed contemporaneously herewith and incorporated herein by reference.

B. The Proposed Sale

- 12. The Debtors filed these chapter 11 bankruptcy cases to effectuate the terms and provisions of the APA by and between the Debtors and Hercules (collectively, the "<u>Purchaser</u>") under a sale pursuant to section 363 of the Bankruptcy Code. A true and correct copy of the executed APA is attached hereto as <u>Exhibit A</u> and incorporated herein for all purposes.
- 13. One of the conditions precedent to the closing of the APA is that this Court enter an order approving the APA. The Debtors have concluded that a chapter 11 filing designed to facilitate a sale of the Debtors' assets to the Purchaser through a section 363 sale would most efficiently consummate the sale and is necessary to prevent continued operating losses and diminution of the Debtors' enterprise value. Under the APA, the Debtors also intend to assume and assign virtually all of their executory contracts, but not unexpired leases, to the Purchaser.
- 14. The Debtors, in consultation with Simmons, and its Board have concluded that the Purchaser's offer was the highest and best offer received. In the opinion of Simmons and the Board, further marketing efforts would not likely result in any offers superior to the Hercules

APA. The Declaration of Simmons & Company International is attached hereto as **Exhibit 1** and incorporated herein by reference for all purposes.

- 15. Pursuant to the terms and conditions of (i) that certain Debtor-In-Possession Credit Agreement (the "DIP Credit Agreement") by and between Seahawk and D.E. Shaw Direct Capital Portfolios, L.L.C. and certain lenders thereto (the "DIP Lenders") and (ii) that certain proposed Interim Order Pursuant to 11 U.S.C. §§ 105, 361, 362, 363, and 364 of the Bankruptcy Code (I) Authorizing the Debtors to Obtain Postpetition Financing and Use Cash Collateral and (II) Granting Liens, Security Interests and Superpriority Claims (the "DIP Order"), any sale of all or substantially all of the Debtors' assets is prohibited unless the sale generates sufficient proceeds to pay all DIP Obligations (as defined in the DIP Order), all DIP Obligations are indefeasibly paid in full in cash, and the commitments under the DIP Credit Agreement are terminated. As a result, the APA provides for a cash portion of the Base Aggregate Consideration (as defined in the APA) sufficient to pay in full in cash the DIP Obligations. Concurrently with the closing of the sale, the cash portion paid by Hercules to the Debtors will be used to indefeasibly pay the DIP Obligations and the commitments to make loans under the DIP Credit Agreement will terminate. Hercules Shares received by the Debtors will be held by an escrow agent to be distributed to holders of allowed claims against and interests in the Debtors pursuant a confirmed chapter 11 plan or pursuant to further orders of this Court.
- 16. The Debtors believe that because it is anticipated that the purchase price will fully satisfy all claims against the Debtors and Hercules' offer is the highest and best offer the Debtors have been able to obtain, the APA is in the best interests of the Debtors and all parties-in-interest, and accordingly, should be approved.

THE NEED FOR AN EXPEDITED SALE

17. As set forth above, and as further detailed in the First Day Affidavit of Randall D. Stilley filed contemporaneously herewith, the Debtors have been substantially affected by the downturn in the natural gas market, the moratorium on new offshore oil and gas drilling permits and general economic conditions. The Debtors have suffered operating losses prior to the Petition Date and likewise have projected that they will continue to suffer operating losses throughout 2011. Every day the Debtors continue their operations, and even during the pendency of these bankruptcy cases, the Debtors assets continue to experience a further deterioration in value.

18. Furthermore, an expedited sale of the Debtors' assets is a mandatory condition of the APA. Without an expedited sale, the APA will not be consummated and the Debtors' estate and all the Debtors' creditors and shareholders will lose a significant opportunity to achieve what the Debtors believe is the highest and best recovery on their claims and interests.

RELIEF REQUESTED

19. The Debtors request that, at the initial hearing on this Sale Motion (the "Preliminary Sale Hearing"), the Court enter an order substantially in the form attached hereto as Exhibit C, that: (1) schedules an expedited hearing to consider final approval of the Sale Motion; (2) sets deadlines for filing (a) objections to the Sale Motion, and (b) objections to the proposed assumption and assignment of executory contracts in connection with the sale and the payment of cure claims in connection therewith; and (3) approves the form and manner of notice of the final hearing to approve the sale ("Final Sale Hearing").⁵

A. REQUEST FOR AN EXPEDITED HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS.

⁵ The Debtors have filed a separate motion seeking approval of the Termination Fee payable to Purchaser under certain conditions.

- 20. The Debtors submit that the sale of the Purchased Assets is essential to preserve the going concern value of Debtors' business. The Debtors assert that each day the Purchased Assets are subject to the market and the bankruptcy process will cause further losses in operating revenue, diminish the value of the Purchased Assets and threaten the viability of projected creditor and stakeholder recoveries. Accordingly, from the Debtors' perspective, the APA represents a reasonable exercise of the Board's and management's business judgment and an expedited sale is in the best interests of the Debtors and all of their economic claim and interestholders.
- 21. The Debtors and Simmons have expended substantial time and resources soliciting potential purchasers of the Debtors' assets pre-petition and accordingly are of the opinion that the APA with the Purchaser represents the best terms the Debtors can reasonably anticipate to obtain and that the process they have implemented is comprehensive, fulsome and appropriate in every respect.

KEY TERMS OF THE APA

22. A summary of the pertinent provisions of the APA are as follows⁶:

Seller (APA p.1)	The Debtors
Purchasers	Hercules Offshore, Inc. and SD Drilling LLC, a Delaware limited liability
(APA p.1)	company ("Newco")
Purchase Price	Subject to the other terms of this Agreement, the aggregate consideration for
(APA § 2.4)	the Purchased Assets shall consist of (a) 22,321,425 shares of Hercules
	Common Stock (the "Hercules Shares") and (b) cash in an amount equal to
	\$25,000,012 (the "Cash Payment" and, together with the Hercules Shares,
	the "Aggregate Consideration")

The summary of the APA is intended only to assist the Court and parties-in-interest in understanding the key aspects of the transaction. In the event of any conflict between this summary and the APA, the APA shall control. The APA can be amended by the parties pursuant to a written amendment, provided that the amendment does not materially reduce the consideration provided to the Debtors' estates. Any capitalized terms not defined herein shall have the meaning set forth in the APA.

Purchased Assets (APA § 2.1(a)– (n))

All of the Debtors' right, title, and interest in and to all of the property, including but not limited to:

Rigs. The jackup Rigs listed on <u>Schedule 2.9</u>, each of which is located in the United States of America (the "*Transferred Rigs*");

Contracts. All of the interests, rights, Claims, and benefits arising or accruing to any of Sellers under any Contract (including any Lease) to which a Seller is a party or has or may acquire a benefit and that relate to the Purchased Assets or Business (the "Assigned Contracts"), including the Assigned Contracts listed in Schedule 2.3(a), to the extent Sellers' interest in such Contract is transferable;

Equipment. All furniture, equipment, computers, computer equipment, machinery, tools, hand tools, spare parts, test equipment, supplies, inventory, office supplies, telephones, and all other tangible personal property of every kind and description insofar as any of the foregoing relates to the operation of the Purchased Assets or Business (the "Equipment"), including the Equipment listed in Schedule 2.1(c);

Rolling Stock. All automobiles, vans, trucks, trailers, and other motorized and similar vehicles and stock of every kind, whether owned or leased, used in the operation of the Purchased Assets or Business, including as listed in Schedule 2.1(d);

Additional Tangible Assets. All other Tangible Personal Property of every kind used in the operation of the Purchased Assets or Business, together with any rights appurtenant thereto, including any express or implied warranty by the manufacturer, vendor, or lessor of any such Tangible Personal Property;

Accounts and Notes Receivable. All trade accounts receivable, notes receivable, and other rights of Sellers to payment from customers and other third parties and other amounts due from customers and other third parties or that become due, including all assets constituting working capital, that relate to or arise from the operation of the Purchased Assets or Business;

Insurance Benefits. All insurance benefits arising from or related to the Purchased Assets, Business, and Assumed Liabilities;

Cash. All cash and cash equivalents, securities, money on deposit with banks, certificates of deposit, and similar short-term investments (if any) of Sellers:

Prepaid Deposits and Expenses. Any deposits and prepaid expenses (paid

to or by Sellers), claims for refunds, and rights of set off related to the Purchased Assets or Business;

Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities against third parties relating or attributable to the Business, the Purchased Assets, or the Assumed Liabilities, whether choate or inchoate, known or unknown, contingent or non-contingent;

Permits. Any and all Permits, including pending applications or filings therefor and renewals thereof, of every kind under which any of Sellers has or may acquire any benefits or rights, or by which any of Sellers or any of the Purchased Assets may be subject or bound, or that relate to or are used or held for use in the operation of the Purchased Assets or Business (collectively, the "Assigned Permits");

Intangible Assets. Any and all customer and supplier relationships, and other Intangible Personal Property of every kind (other than the Computer Software, Trademark Intellectual Property and Website Intellectual Property, and uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names that constitute Intellectual Property) that relate to or are used in the operation of the Purchased Assets or Business, together with all rights appurtenant thereto (the "Assigned Intangible Assets");

Books and Records. Original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets and photocopies of the other Books and Records (collectively, the "Transferred Books and Records");

Other Assets. To the extent not otherwise enumerated in this <u>Section 2.1</u> all other tangible or intangible assets, rights, privileges, benefits, Claims, and interests of a Seller, whether real, personal or mixed, of every kind and description and wherever located, that relate to, used in or held for use in the operation of the Purchased Assets or Business.

Excluded Assets (APA § 2.2(a)– (o))

Notwithstanding anything to the contrary in <u>Section 2.1</u> or elsewhere in this Agreement or other Transaction Documents, the Purchased Assets shall not include:

Transaction Rights. All rights of Sellers under this Agreement and the other Transaction Documents, and all cash and non-cash consideration (including the Hercules Shares and the Cash Payment) payable or deliverable to, or on behalf of, Sellers by Purchasers pursuant hereto and thereto;

Rigs. Any Rig that is deemed to be an Excluded Asset pursuant to <u>Section 2.9</u>, together with all of the Equipment located on each such Rig;

Contracts. All rights of Sellers under any Contract that is not an Assigned Contract, including those Contracts listed in <u>Schedule 2.2(c)</u>;

Permits. Any and all rights of Sellers under any Permit that is not an Assigned Permit;

Insurance Benefits. Any and all insurance benefits arising from or relating to (i) Excluded Assets and Retained Liabilities, (ii) Claims made by Sellers prior to the Closing with respect to the Purchased Assets and Business, and (iii) the matters set forth in <u>Schedule 2.2(e)</u>;

Books and Records. The original copies of all of the Books and Records, other than the original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets (which shall be provided to Purchasers pursuant to Section 2.1(m) and of which Sellers may retain photocopies), and any Books and Records of Sellers (i) that are not permitted to be transferred to Purchasers under applicable Law, (ii) that constitute charters, bylaws, limited liability company agreements, minute books, stock transfer records, and other records related to the corporate governance of Sellers, and (iii) all other books and records of Sellers that do not relate primarily to the Purchased Assets, Transferred Employees, or Business (collectively, the "Retained Books and Records");

Third Party Property. Any improvements, equipment, inventory and any other tangible personal property located at the Leased Real Property or on the Transferred Rigs as of the Closing Date that are not owned by or leased to Sellers; and

Equity Interests. Any Equity Interest in any Seller or any direct or indirect subsidiary of a Seller;

Bankruptcy Claims. Any and all Claims or rights of Sellers arising under chapter 5 of the Bankruptcy Code;

Pride Claims. Any and all Claims or rights of Sellers or any Affiliate thereof against Pride or any Affiliate thereof;

Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities of Sellers, whether choate or inchoate, known or unknown, contingent or non-contingent with respect to the matters set forth in Schedule 2.2(k);

Leased Real Property. All leasehold interests in the real property listed in Schedule 2.2(1) (the "Leased Real Property"), together with all Real Property Rights related thereto and all Improvements and Tangible Personal

Property located at, under, or on such Leased Real Properties or otherwise related thereto;

Intellectual Property. The Intellectual Property constituting: (i) Sellers' software programs, tools, kits, and any content or related documentation or third party or open source code embedded therein, either locally stored on Sellers' computers or remotely accessed by Sellers, that is owned or made available to Sellers and is used in, held for use in, or is necessary for the operation of the Purchased Assets or Business or otherwise, and any upgrades, updates, releases, fixes, enhancements, or modifications thereto, and all written materials and specifications applicable thereto (collectively, the "Computer Software"); (ii) the name "Seahawk Drilling" and the mark "Seahawk Drilling" and any other registered or unregistered trademarks, trade names, service marks, domain names, and email addresses, including the term "Seahawk", used or held for use in the Business (collectively, the "Trademarks"), and any and all goodwill associated with the Business embodied in the Trademarks, and any and all rights of Sellers with respect to the Trademarks (collectively, the "Trademark Intellectual Property"); (iii) Sellers' internet website located at www.seahawkdrilling.com and any derivations thereof (collectively, the "Website") and all intellectual property rights that may exist or arise in connection with the Website, including all Underlying Technology (collectively, the "Website Intellectual Property"); and; (iv) uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names.

Goodwill. Any goodwill and going concern value related to the Business; and

Accounts and Notes Receivable. The trade accounts receivable, notes receivable, and other rights of Sellers to payment from, and other amounts due or that become due, including all assets constituting working capital, that relate to Blake International USA Rigs, LLC or any Affiliate thereof.

Assigned Contracts (APA § 2.3)

Transfer of Contracts and Permits. At least three Business Days prior to the Closing Date, Purchasers may designate any Contracts that would otherwise be an Assigned Contract as a Contract to be excluded from the Purchased Assets for purposes of Section 2.1(b). Any such designation by Purchasers shall not reduce Purchaser's assumption of any working capital liabilities pursuant to Section 2.7(a), unless both (i) such working capital liability was incurred pursuant to such designated Contract and (ii) such working capital liability is not attributable to goods and services provided to the Business prior to the Closing, in which case such working capital liability shall not be assumed under Section 2.7(a).

Notwithstanding anything herein to the contrary, this Agreement shall not constitute an assignment to Purchasers of (and Purchasers shall not assume

any rights or obligations under) any Contract or Permit if an attempted assignment thereof, without the Consent of a Governmental Authority or other third party thereto, would constitute a breach of the Contract or Permit, unless such Consent has been received or such assignment is otherwise authorized and permitted by the Transfer Order or other Order of the Bankruptcy Court. The Transfer Order shall provide that any Assigned Contracts that are executory contracts shall be assumed by Sellers and assigned to Purchasers at the Closing, and Sellers shall pay all pre-Closing cure amounts under Assigned Contracts required to be so paid by the Bankruptcy Court in connection with such assumptions by Sellers (the "Cure Amounts"). The Sellers shall use their commercially reasonable efforts to advise the Purchasers promptly in writing with respect to any Contract or Permit that Sellers' Know or have substantial reason to believe shall not or may not be subject to assignment to Purchasers hereunder. Without in any way limiting Sellers' obligations to obtain all Consents necessary for the sale, transfer, assignment and delivery of the Assigned Contracts and Assigned Permits and the other Purchased Assets to Purchasers hereunder, if any such Consent is not obtained or if such assignment is not permitted irrespective of Consent and the Closing hereunder is consummated, Sellers shall use commercially reasonable efforts to continue to promptly seek Consent or other approval for the sale, transfer, assignment and delivery thereof; provided, that in no circumstance shall any such obligation of Sellers extend beyond the date that Sellers' Plan of Reorganization is confirmed by the Bankruptcy Court and has become effective.

Assumed Liabilities (APA § 2.7(a)– (e))

The Assumed Liabilities include:

Working Capital Liabilities. The current accounts payable of, and accruals for good and services received by, Sellers as of the Closing Date attributable to the Purchased Assets; provided that, for purposes of clarity, such working capital liabilities shall exclude accounts payable to Blake International USA Rigs, LLC or its Affiliates. Nothing in this Section 2.7(a) shall be deemed to effect the assumption of any Contract that is not otherwise an Assigned Contract, but the assumption of liabilities under this Section 2.7(a) shall be subject to Section 2.3(a).

Contracts. The Liabilities of Sellers under the terms of any Assigned Contract to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the closing;

Permits. The Liabilities of Sellers under the terms of any Assigned Permit to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the Closing;

Claims relating to Assumed Liabilities. In the event Claims are filed in the

Sellers' Bankruptcy Cases on account on the Assumed Liabilities, Sellers shall provide Purchasers written notice of such filing within five (5) Business Days after their filing. Purchasers shall have the right to appear, defend and settle those Claims in the Bankruptcy Court. Notwithstanding the foregoing, Sellers shall be solely responsible for any Claims that exceed the amount of current liabilities assumed under Section 2.7(a) and included in the post-Closing adjustment process under Section 2.10; and

Performance Bond. Purchasers shall either assume Sellers' \$2.0 performance bond in favor of the Texas general land office relating to the Seahawk 2008 Rig, or otherwise post their own performance bond such that such \$2.0 performance bond may be terminated by Sellers.

Retained Liabilities (APA § 2.8(a)– (b))

The Excluded Liabilities include:

Subject to the terms of the Transfer Order, Sellers shall retain and be solely liable for and hereby expressly agree to retain any and all of their respective Liabilities not expressly assumed by Purchasers in Section 2.7 (collectively, the "Retained Liabilities"), regardless of whether any such Retained Liability is disclosed herein or in any Schedule hereto, whether known or unknown, absolute or contingent, liquidated or unliquidated, whether due or to become due, and whether Claims with respect thereto are asserted before or after the Closing Date. Notwithstanding the preceding sentence, Retained Liabilities shall include the following:

- *Transaction Liabilities*. Any and all Liabilities of Sellers under the Transaction Documents;
- *Excluded Assets*. Any and all Liabilities of Sellers with respect to the Excluded Assets;
- Purchased Assets and Pre-Closing Business. Except as provided in Sections 2.7(a), 2.7(b), and 2.7(c), all Claims and Liabilities arising out of Sellers' ownership, operation, use, or maintenance of the Purchased Assets or Business, as well as any matters arising from events occurring, conditions existing, or costs accruing prior to the Closing and Taxes;
- *Contracts*. Except as provided in <u>Sections 2.7(a)</u> and <u>2.7(b)</u>, any and all Liabilities of Seller, including Cure Amounts, that relate to the Assigned Contracts to the extent any such Liability is performable or otherwise attributable to the period prior to the Closing;
- *Permits*. Except as provided in <u>Sections 2.7(a)</u> and <u>2.7(c)</u>, any and all Liabilities of Sellers that relate to the Assigned Permits to the extent any such Liability is performable or otherwise attributable to the period

prior to the Closing;

- Taxes. Any and all Liabilities of Sellers for Taxes that relate to (A) the ownership, operation, use, or maintenance of the Purchased Assets or Business prior to the Closing Date, or (B) any sales, use, transfer, or other similar Taxes imposed as a result of the consummation of the Contemplated Transactions or performance of the Transaction Documents;
- Legal Proceedings. Any and all Liabilities of Sellers that relate to any Proceeding (A) involving the Purchased Assets or Business, including warranty, personal injury, breach of contract, failure to perform, infringement, noncompliance with Law, and tort Claims, that is (1) pending or threatened as of the Closing, or (2) commenced after the Closing but that arises out of or relate to any event, omission, or occurrence happening as of or prior to the Closing, or (B) set forth in Schedule 2.8(a)(vii);
- Pride Claims. Any and all Liabilities of Sellers under agreements entered into among Seahawk Parent and Pride International, Inc. ("Pride") at the time of Seahawk Parent's spin-off from Pride effective August 4, 2009, including (i) any obligation to indemnify Pride and its Affiliates for Mexican tax liabilities associated with Seahawk Parent's Mexican subsidiaries under that certain Tax Sharing Agreement dated August 4, 2009, between Seahawk Parent and Pride, part of which tax liabilities are supported by letters of credit from Pride, and any settlement with respect to such matters, and (ii) any and all Liabilities of Sellers (if any) to Pride and its Affiliates under such agreements resulting from the Contemplated Transactions;
- *Pride Wyoming*. Any and all Liabilities of Sellers relating to the loss of the Pride Wyoming Rig, including salvage and wreckage removal costs and any third party damage Claims and any continuing Liability under applicable Law relating to the Pride Wyoming Rig;
- FCPA Investigation. Any and all Liabilities of Sellers, including for purposes of clarity any and all fines and penalties (if any), of Sellers with respect to violations, alleged violations, and investigations for alleged potential or actual violations by Sellers and any of their respective Affiliates of the U.S. Foreign Corrupt Practices Act;
- Environmental Liabilities. Any and all Liabilities of Sellers with respect to any violation of Law including those arising from (A) the release, threatened release, presence, treatment, storage, disposal (including disposal at off site locations), handling, transportation or arrangement for transportation of hazardous substances prior to the Closing, (B) any failure of Sellers to comply in any respect with

Environmental, Health, and Safety Laws prior to the Closing, and (C) any facts, events, or circumstances in existence prior to the Closing that give rise to Liabilities pursuant to Environmental, Health, and Safety Laws;

- Employee Benefit Plans. Any and all Liabilities of Sellers, fiduciaries or ERISA Affiliate under any Employee Benefit Plan maintained or contributed to by any Seller, fiduciary or any ERISA Affiliate or with respect to which any Seller, fiduciary or ERISA Affiliate has any Liability, whether prior to, on, or after the Closing, including any (a) failure to comply with all applicable Laws, (b) liability with respect to audits, inquiries, Proceedings, or Claims with any Governmental Authority with respect to any Employee Benefit Plan, (c) participation in any "multiemployer" plan (within the meaning of Section 3(37) of ERISA, whether or not governed by the provisions of ERISA) or withdrawal liability with respect to any multiemployer plan, (d) required employer contributions with respect to the Employee Benefit Plans, (e) accumulated funding deficiency, (f) Lien under ERISA or Section 412 of the Code, or (g) the termination of, or intent to terminate, any Plan;
- *Employees*. Any and all Liabilities of any nature of Sellers or any fiduciary or ERISA Affiliate to Seller Employees, including Liabilities with respect to (a) any Contract, plan or policy, (b) wages, withholdings, overtime pay, minimum wage, employment Tax, vacation, sick pay, bonuses, severance pay, retirement, or other compensation, (c) benefits under Employee Benefit Plans, (d) the Worker Adjustment and Retraining Notification Act (WARN) of August 4, 1988 or equivalent state or local statutory or regulatory requirements, (e) any collective bargaining agreement or obligation or requirement under the National Labor Relations Act, (f) reporting, filing, hiring or other employment obligations with the Office of Federal Contract Compliance Programs, (g) all immigration related obligations, including all requirements of the Immigration Reform and Control Act of 1986, (h) any governmental or administrative proceeding for the enforcement of labor and employment laws and regulations, and (i) all other employment and employment related federal, state and local statutes, regulations, administrative requirements, common laws, and public policies;
- *Intercompany Liabilities*. Any and all Liabilities of Sellers for intercompany advances, charges, or accounts payable of any kind or nature;
- Bankruptcy Claims. Any and all Claims filed against Sellers in their respective Bankruptcy Cases, except for the Assumed Liabilities, as

	1'	
	limited in Section 2.7(a);	
	• <i>Broker Fees</i> . Any and all fees, commissions, and other compensation to any broker, finder, or agent retained by any Seller as contemplated by <u>Section 4.22</u> ;	
	The Retained Liabilities shall constitute Claims and alleged Claims in the Sellers' Bankruptcy Cases; <i>provided</i> , <i>however</i> , that nothing herein shall create any rights in favor of the holders of such Claims and alleged Claims or create any priority right of payment.	
Releases (APA §9.9)	In consideration for payment of the Cash Payment, issuance of the Hercules Shares and assumption of the Assumed Liabilities by Purchasers, Sellers shall cause the Plan of Reorganization to contain a mutual release provision by Sellers in favor of Purchasers and by Purchasers in favor of Sellers on terms mutually acceptable to the Parties.	
Closing (APA § 3.1)	Provided that this Agreement shall not have been earlier terminated pursuant to Section 11.1, and further provided that all of the conditions set forth in Sections 10.1 and 10.2 to the obligations of the Parties to consummate the Contemplated Transactions (other than conditions with respect to actions each Party will take at the Closing itself) shall have been satisfied or waived, the closing of the Contemplated Transactions (the "Closing") shall take place at the offices of Fulbright & Jaworski L.L.P. at 1301 McKinney, Suite 5100, Houston, Texas 77010 commencing at 11:00 a.m. Central Time on the date that is no earlier than the first Business Day that the Transfer Order has become a Final Order, but no later than the third (3rd) Business Day following such date, or such other date or location as Purchasers and Sellers may mutually agree to in writing (the "Closing Date").	
Termination Provisions (APA § 11.1(a)– (d))	 Mutual Consent. Purchasers and Sellers may terminate this Agreement as to all Parties by mutual written consent at any time prior to the Closing by written instrument authorized by the respective Boards of Directors of Seahawk Parent and Hercules Parent. By Purchasers. Purchasers may terminate this Agreement by giving written notice to Sellers at any time prior to the Closing in the event that (i) Sellers have breached any representation, warranty, or covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.1(a) and 10.1(b) shall not have been satisfied, or cannot be satisfied by August 15, 2011 (the "Termination Date"); provided, that Purchasers shall have provided written notification to Sellers of such breach and the breach shall 	

have continued without cure for a period of ten (10) days after delivery of the notice of such breach; (ii) the Transfer Order shall not have been entered by the Bankruptcy Court and become a Final Order by one hundred twenty (120) days after the date hereof; (iii) any of the Bankruptcy Cases are converted to cases pursuant to Chapter 7 of the Bankruptcy Code; (iv) a Chapter 7 or Chapter 11 Trustee has been appointed; (v) an Examiner with expanded powers has been appointed in any of the Bankruptcy Cases of Sellers; (vi) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement; (vii) the DIP Financing is not approved within twenty-one (21) days of the Petition Date; (viii) Sellers are in breach or default in any material respect of, the DIP Credit Documents, DIP Order or DIP Budget after five days notice by any Person and opportunity to cure; or (ix) the Termination Fee Order shall not have been entered by the Bankruptcy Court within twenty-five (25) days of the date hereof or shall not have become a Final Order by the eleventh day after such entry of the Termination Fee Order. At their sole and absolute discretion, Purchasers have the right to waive termination or agree to extend any deadlines under this Section11.1(b).

- By Sellers. Sellers may terminate this Agreement by giving written notice to Purchasers at any time prior to the Closing in the event that (i) any Purchaser has breached any representation, warranty, or covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.2(a) and 10.2(b) shall not have been satisfied, or cannot be satisfied by the Termination Date; provided, that Sellers shall have provided written notification to Purchasers of such breach and the breach shall have continued without cure for a period of ten (10) days after delivery of the notice of breach; or (ii) in accordance with the terms and subject to the conditions of Section 6.5; or (iii) Purchasers' failure to timely obtain the Consent as contemplated by Section 6.9.
- By Either Party. Sellers or Purchasers may terminate this Agreement by giving written notice to the other Party if (i) any court of competent jurisdiction or any other Governmental Authority in a suit instituted by a third party or a Governmental Authority shall have issued an Order or shall have taken any other action prior to the Termination Date permanently enjoining, restraining, or otherwise prohibiting the Contemplated Transactions or a material portion thereof, (ii) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement or denies approval of this Agreement or the Contemplated Transactions, or (iii) the Closing has not occurred by the Termination Date, provided, that the Party electing to terminate shall not have caused such failure to close.

- 23. At the Preliminary Sale Hearing, the Debtors request that the Court schedule an expedited hearing on the Sale Motion.
- B. REQUEST FOR THE SETTING OF DEADLINES FOR FILING OBJECTIONS TO (i) THE PROPOSED SALE AND (ii) TO THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND THE PAYMENT OF CURE CLAIMS IN CONNECTION THEREWITH
- 24. The Debtors request that, at the Preliminary Sale Hearing, the Court order that any objections to the Sale Motion (a "Sale Objection") set forth in writing with particularity the factual and legal basis for the objection and be filed by a party with standing and served on counsel for the Debtors, counsel for the Purchaser and all other parties entitled to notice no later than 12:00 p.m. (CST) on the third Business Day prior to the Sale Hearing Date (the "Objection Deadline").
- 25. The Debtors further request the Court to order that, unless an objection to the sale is timely filed and served by the Objection Deadline, all interested parties who have received actual or constructive notice thereof be deemed to have waived and released any right to assert a Sale Objection and shall be deemed to have otherwise consented to the proposed sale and shall be forever barred and estopped from asserting any Sale Objection against the Debtors, or the Purchaser.

ASSUMPTION AND ASSIGNMENT OF ASSIGNED CONTRACTS

26. As part of the sale, the Debtors desire to assume and assign the executory contracts set forth on **Exhibit B** hereto (Schedule 2.1(b) to the APA)(the "**Assigned Contracts**") to the Purchaser. Pursuant to this Sale Motion, the Debtors seek authority pursuant to sections 105(a) and 365 of the Bankruptcy Code to (i) assume and assign the Assigned Contracts to the Purchaser, and (ii) execute and deliver to the Purchaser, or its assignee, such documents or other instruments

as may be necessary by the Purchaser to assign and transfer the Assigned Contracts at the Final Sale Hearing. The Debtors have set forth on **Exhibit B** the cure amounts they believe to be payable to each of the counterparties under the Assigned Contracts.

- 27. The Debtors request that the Court order, at the Preliminary Sale Hearing, that any objections relating to the proposed assumption and assignment of executory contracts (including, but not limited to, any objections relating to the validity of the cure amount as determined by the Debtors or to otherwise assert that any amounts, defaults, conditions, or pecuniary losses must be cured or satisfied under any of the Assigned Contracts as of the date of the Sale Hearing) (a "Cure Objection") be filed and served so as to be actually received by counsel for both the Debtors and the Purchaser no later than the Objection Deadline.
- 28. The Debtors further request that the Court order that, unless a Cure Objection is timely filed and served by a party to an Assigned Contract by the Objection Deadline, all interested parties who have received actual or constructive notice thereof be deemed to have waived and released any right to assert a Cure Objection and to have otherwise consented to the assignment of the Assigned Contracts with payment of the cure amounts set forth on **Exhibit B** hereto and shall be forever barred and estopped from asserting or claiming against Debtors, the Purchaser or any assignee that any additional amounts are due or that a default exists with respect to such Assumed Contract.
- C. APPROVAL OF THE FORM AND MANNER OF NOTICE OF THE TERMINATION FEES, SALE HEARING AND OF THE OBJECTION DEADLINES.
- 29. At the Preliminary Sale Hearing, the Debtors request approval of the form and manner of notice of the Sale Hearing and of the Objection Deadlines to be served on all parties-in-interest affected by the Sale Motion.

30. The Debtors believe that the form of sale hearing notice (the "Sale Hearing Notice") attached hereto as Exhibit D provides due and adequate notice of the Final Sale Hearing and of the proposed assumption and assignment of the Assigned Contracts and the payment of cure amounts in connection therewith.

D. APPROVAL OF THE DISPOSITION OF SALE PROCEEDS.

- 31. Finally, at the Sale Hearing, the Debtors request, pursuant to sections 105, 363, and 365 of the Bankruptcy Code and Bankruptcy Rules 2002, 6003, 6004, 6006, and 9006, that, at the Final Sale Hearing, the Court will enter an order (the "**Transfer Order**"):
 - (i) approving the APA by and between the Debtors and the Purchaser, substantially in the form attached hereto as Exhibit A,
 - (ii) authorizing the sale of the Debtors' assets under the APA free and clear of all liens, claims, encumbrances, and other interests;
 - (iii) authorizing the assumption and assignment of substantially all of the Debtors' executory contracts, identified in the attached Exhibit B, to the Purchaser;
 - (iv) (i) authorizing the Debtors to use the cash portion of the Aggregate Consideration to pay the DIP Obligations (the "DIP Loan Payment"), (ii) directing the Debtors to make the DIP Loan Payment concurrently with the Closing; and (iii) stating that, upon receipt of the DIP Loan Payment, (i) the DIP Lenders are authorized to apply the DIP Loan Payment to the DIP Obligations, (ii) such payment shall constitute indefeasible payment of the DIP Obligations so paid and (iii) any commitments of the DIP Lenders to make any loans to the Debtors are terminated;
 - (v) authorizing the Debtors to place the Hercules Shares received as the remainder of the Aggregate Consideration with an Escrow Agent in an account to be held in trust pursuant to the an escrow agreement and to be distributed to holders of allowed claims and interests of the Debtors pursuant a confirmed chapter 11 plan or pursuant to further orders of this Court; and
 - (vi) granting other related and appropriate relief.

BASIS FOR RELIEF

A. Approval of the Sale of Substantially All of the Debtors' Assets.

- 32. Section 363(b)(1) of the Bankruptcy Code provides, in relevant part, that "[t]he trustee, after notice and a hearing, may use, sell, or lease, other than in the ordinary course of business, property of the estate." 11 U.S.C. § 363(b)(1).
- Although section 363 of the Bankruptcy Code does not set forth a standard for 33. determining when it is appropriate for a court to authorize the sale or disposition of a debtor's assets, in applying this section, courts have required that it be based upon the debtor's business judgment. In re Continental Air Lines, Inc., 780 F.2d 1223, 1226 (5th Cir. 1986) (A debtor must demonstrate "some articulated business justification for using, selling, or leasing property outside of the ordinary course of business."); Comm. Of Equity Security Holders v. Lionel Corp., 722 F.2d 1063, 1071 (2d Cir. 1983) (A sale of a substantial part of a Chapter 11 estate other than in the ordinary course of business may be conducted if a good business reason exists to support it.). "Where the debtor articulates a reasonable basis for its business decisions (as distinct from a decision made arbitrarily or capriciously), courts will generally not entertain objections to the debtor's conduct." Comm. of Asbestos-Related Litigation v. Johns-Manville Corp. (In re Johns-Manville Corp.), 60 B.R. 612, 616 (Bankr. S.D.N.Y. 1986). When a valid business justification exists, the law vests the debtor's decision to use property out of the ordinary course of business with a strong presumption that "in making a business decision the directors of a corporation acted on an informed basis, in good faith and in the honest belief that the action taken was in the best interest of the company." Official Comm. of Subordinated Bondholders v. Integrated Res., Inc. (In re Integrated Res., Inc.), 147 B.R. 650, 656 (S.D.N.Y. 1992) (quoting Smith v. Van Gorkom, 488 A. 2d 858, 872 (Del. 1985)).

- 34. Moreover, section 105(a) of the Bankruptcy Code grants this Court broad authority under its equitable powers to fashion any order or decree that would preserve or protect the value of the debtor's assets. *See* 11 U.S.C. § 105(a); *see also, Adelphia Commc'ns Corp. v. Rigas (In re Adelphia Commc'ns Crop.)*, 2003 WL 21297258, at *4 (S.D.N.Y. June 4, 2003) ("Section 105 of Title 11 provides the bankruptcy courts with a broad range of equitable powers over cases within its jurisdiction.").
- 35. Courts look to various factors to determine whether to approve a motion under section 363(b), such as: (i) whether a sound business reason exists for the proposed transaction; (ii) whether fair and reasonable consideration is provided; (iii) whether the transaction has been proposed and negotiated in good faith; and (iv) whether adequate and reasonable notice is provided. *See In re Condere*, 228 B.R. 615, 626 (Bankr. S.D. Miss. 1998); *See In re Delaware & Hudson Ry. Co.*, 124 B.R. 169, 176 (D. Del. 1991). The Debtors respectfully submit that the proposed sale of the Purchased Assets fits squarely within the parameters of the sound business judgment test articulated above.

1. A Sound Business Reason Exists For The Sale Of The Purchased Assets

36. The Debtors' decision to sell property out of the ordinary course of business enjoys a strong presumption "that in making a business decision the Debtors acted on an informed basis, in good faith and in an honest belief that the action taken was in the best interests of the company." *In re Integrated Res., Inc.*, 147 B.R. 650, 656 (S.D.N.Y. 1992). Therefore, parties objecting to the Debtors' proposed sale must make a showing of "bad faith, self-interest, or gross negligence." *Integrated Res.*, 147 B.R. at 656.

- 37. The Debtors have determined, in their considered business judgment, that the sale of the Purchased Assets pursuant to the APA is appropriate and in the best interest of, and represents the best means of preserving and maximizing value for, the Debtors' estates.
- 38. As discussed above, the Debtors have faced pre-petition monthly operating losses for an extended period of time. The Debtors project continued losses if a sale is not effectuated. Without additional funding, the Debtors anticipate that they will soon run out of cash to operate their business. Because of the lack of capital, continued operation of the Debtors' business without a sale of the Purchased Assets is not an option.
- 39. To avoid a chapter 7, which likely would result in unsecured creditors and equity interests receiving little or no recovery or distribution, subject to approval of the Court, the Debtors have agreed to sell the Purchased Assets to the Purchaser pursuant to section 363 of the Bankruptcy Code.
- 40. The sale to the Purchaser will permit the Debtors to distribute cash and Hercules Common Stock generated from the sale to creditors and interestholders pursuant to a chapter 11 plan or pursuant to further orders of this Court. Accordingly, the proposed disposition of assets is born of the Debtors' economic circumstances, and not "bad faith, self-interest or gross negligence."
- 41. The Debtors note that while the sale of the Purchased Assets is not part of its ordinary business, the proposed sale will not substantially impact (and, in fact, will facilitate a plan of liquidation) or significantly alter the rights of the Debtors' creditors. Therefore, the Debtors submit that the proposed sale of the Purchased Assets in no way constitutes a *sub rosa* plan or falls outside the scope of Bankruptcy Code § 363. *C.f. Continental Airline, Inc. v. Braniff Airways, Inc.* (*In re Braniff Airways, Inc.*), 700 F.2d 935. 939 (5th Cir. 1983) (noting that the

proposed sale pursuant to section 363 would require "significant restructuring of the rights of Braniff creditors").

2. The Purchaser Is Paying Fair And Reasonable Consideration For The Purchased Assets

42. The aggregate consideration for the Purchased Assets is approximately One Hundred Five Million Dollars (\$105,000,000). Given the difficulty the Debtors have faced trying to sell the Purchased Assets, the Debtors believe that the purchase price is fair and reasonable consideration for the Purchased Assets and is the result of extensive arm's length negotiations between the respective representatives of the Debtors and the Purchaser. The purchase price offered by the Purchaser represents the highest and best bid for the Purchased Assets received to date.

3. The Sale Of The Purchased Assets Has Been Proposed And Negotiated In Good Faith

- 43. The Purchaser should be afforded all protections under Bankruptcy Code Section 363(m) as a good faith purchaser. Section 363(m) of the Bankruptcy Code provides that "the reversal or modification on appeal of an authorization under subsection (b) or (c) of this section of a sale or lease of property does not affect the validity of a sale or lease under such authorization to an entity that purchased or leased such property in good faith." 11 U.S.C. § 363(m). Courts generally determine that parties have acted in good faith with respect to a proposed sale if the purchase price is adequate and reasonable and the terms of the sale are disclosed fully. *See In re Abbotts Dairies of Pa., Inc.*, 788 F.2d 143, 149-50 (3d Cir. 1986).
- 44. The Debtors submit that the APA is not tainted by self-dealing, collusion or manipulation as it was proposed and negotiated in good faith and executed by the Debtors and the Purchaser in an arm's length transaction, and the Debtors and the Purchaser have fully disclosed all material information.

- 45. Similarly, the Purchaser's proposed consideration for the Purchased Assets was made in good faith. Moreover, the Purchaser was not preferred and did not receive special treatment. Though Randall D. Stilley was formerly employed as the Chief Executive Officer of Hercules, his former relationship did not influence Simmons' solicitation process or the negotiation or consummation of the APA. All aspects of Simmons' solicitations and the negotiation and consummation of the APA were comported in good faith and pursuant to reasonable business judgment.
- 46. Accordingly, the Debtors assert that it is appropriate that the Purchaser receives the protection of a good faith finding by the Court in accordance with section 363(m).
 - 4. The Court Should Authorize The Sale Of The Purchased Assets To The Purchaser Free And Clear Of All Liens, Claims, Encumbrances, And Interests
- 47. The Debtors request that the sale and transfer of the Purchased Assets be approved free and clear of all interests, liens, claims, encumbrances and other interests, other than those specifically assumed by the Purchaser. Such relief is consistent with the provisions of section 363(f).
- 48. The Court may, pursuant to section 363(f) of the Bankruptcy Code, authorize a debtor-in-possession to sell property free and clear of any interest, lien, claim or interest of another entity in such property if any of the following circumstances pertain:
 - (i) applicable non-bankruptcy law permits sale of such property free and clear of such interest;
 - (ii) such entity consents;
 - (iii) such interest is a lien and the price at which such property is to be sold is greater than the aggregate value of all liens on such property;
 - (iv) such interest is in bona fide dispute; or
 - (v) such entity could be compelled, in a legal or equitable proceeding, to accept a money satisfaction of such interest.

11 U.S.C. § 363(f).

- 49. As indicated by the use of the disjunctive term "or," satisfaction of any of the five requirements listed in section 363(f) is sufficient to permit the sale of assets free and clear of all interests, liens, claims, encumbrances, pledges, mortgages, security interests, charges, options and other interests, including any right of setoff. *See*, *e.g.*, *In re C-Power Prods.*, *Inc.*, 230 B.R. 800 (Bankr. N.D. Tex. 1998); *In re Elliott*, 94 B.R. 343, 345 (E.D. Pa. 1988) (stating that section 363(f) is written in the disjunctive; the court may approve a sale "free and clear" provided that at least one of the subsections is met).
- Assets as of the Petition Date are Natixis, New York Branch, as Administrative Agent for the Lenders (the "<u>Pre-Petition Lenders</u>"), and the U.S. Internal Revenue Service. First, the price at which the Purchased Assets is to be sold is greater than the aggregate value of all liens on such property pursuant to section 363(f)(iii) of the Bankruptcy Code. Second, it is anticipated that the DIP Lenders will be substituted in the place of the Pre-Petition Lenders with senior liens on the Purchased Assets as of the Closing Date.
- 51. Therefore, the Debtors submit, to the extent section 363(f) is applicable, at least one requirement of section 363(f)(iii) has been satisfied and the Purchased Assets may be sold and transferred free and clear of all liens, claims, and encumbrances.

5. Approval Of The APA

52. Courts routinely approve asset purchase agreements in connection with the sale of a debtor's assets. *See In re Arlco, Inc.*, 239 B.R. 261, 265 (Bankr. S.D.N.Y. 1999). In this case, the Purchaser's APA was the subject of intense arm's-length negotiations between representatives of the Debtors and the Purchaser, and the Debtors submit that the terms and conditions of the

APA are the best that could be obtained under the circumstances, and that entry into the APA is a sound exercise of the Debtors' business judgment.

- 6. Assumption And Assignment Of The Assigned Contracts Should Be <u>Authorized</u>
- 53. The Debtors intend to assume and assign the Assigned Contracts to the Purchaser. Under section 365 of the Bankruptcy Code, a debtor-in-possession may assume an executory contract "subject to the court's approval" if (a) the debtor-in-possession cures, or provides adequate assurance of a prompt cure, of any default under such contract or lease; (b) compensates, or provides adequate assurance that prompt compensation will be paid to any non-debtor party to such contract or lease for any pecuniary loss to such party resulting from such default; and (c) provides adequate assurance of future performance under such contract. *See* 11 U.S.C. § 365(b)(1).
- 54. Courts have consistently applied the business judgment test in determining whether to approve a debtor-in-possession's decision to assume an executory contract. *See Richmond Leasing Co. v. Capital Bank, N.A.*, 762 F.2d 1303, 1312 (5th Cir. 1985). The business judgment test is satisfied, when the debtor-in-possession has determined in good faith that assumption of a contract will benefit the estate.
- 55. The Debtors have determined, in the exercise of their sound business judgment, that the assumption and assignment of the Assigned Contracts will benefit the Debtors' estates. The assumption and assignment of the Assigned Contracts is essential to the deal between the Debtors and the Purchaser. Further, the assumption and assignment of the Assigned Contracts will significantly limit administrative expense and rejection damage claims.
- 56. The Debtors also propose to cure defaults and provide adequate assurance of future performance of the Assigned Contracts. The cure amounts that the Debtors believe to be owed in 90303148.17

connection with the assumption and assignment of the Assigned Contracts are set forth on the attached **Exhibit B**. The payment of any cure amounts related to the Assigned Contracts will be in final satisfaction of all obligations to cure defaults and compensate the non-debtor counterparties for any pecuniary losses under section 365(b)(1) of the Bankruptcy Code.

57. At the Final Sale Hearing to approve the sale, the Debtors will provided evidence that illustrates the Purchaser's financial credibility, experience in the industry, and willingness and ability to perform under the Assigned Contracts. The meaning of "adequate assurance of future performance" depends on the facts and circumstances of each case, but should be given "practical, pragmatic construction." *In re Sanshoe Worldwide Corp.*, 139 B.R. 585, 592 (S.D.N.Y. 1992). When an executory contract or unexpired lease is to be assumed and assigned, adequate assurance may be provided by, among other things, demonstrating the financial health of the assignee and its experience and ability in managing the type of enterprise or property assigned. *See In re Bygaph*, 56 B.R. 596, 605-06 (Bankr. S.D.N.Y. 1986).

BANKRUPTCY RULES 6004(H) AND 6006(D) SHOULD BE WAIVED

- 58. Bankruptcy Rule 6004(h) provides that an "order authorizing the use, sale, or lease of property ... is stayed until the expiration of 14 days after the entry of the order, unless the court orders otherwise." Fed. R .Bank. P. 6004(h). Bankruptcy Rule 6006(d) provides that an "order authorizing the trustee to assign an executory contract or unexpired lease ... is stayed until the expiration of 14 days after the entry of an order, unless the court orders otherwise." Fed R. Bank. P. 6006(d).
- 59. The Debtors request that any order entered pursuant to this Sale Motion authorizing the consummation of a transaction that is deemed a sale of assets and/or an assignment of an executory contract be effective immediately by providing that the 14-day stay

under Rules 6004 or 6006, as the case may be, is inapplicable, so that they may proceed to close on the transaction as expeditiously as possible and within the time frames contemplated by the Debtors and the Purchaser.

NOTICE OF THE SALE MOTION

- 60. In accordance with applicable rules, including the United States Bankruptcy Court for the Southern District of Texas Procedures for Complex Chapter 11 Bankruptcy Cases, notice of this Motion has been provided by overnight delivery and e-mail or facsimile to: (a) Natixis, New York Branch, as Administrative Agent for the Lenders; (b) the 30 largest unsecured creditors of the Debtors on a consolidated basis; (c) Pride International, Inc.; (d) Hercules Offshore, Inc.; (e) D.E. Shaw Direct Capital Portfolios, L.L.C.; (f) the U.S. Trustee's Office; (g) the United States Attorney's Office; (h) the Securities and Exchange Commission; (i) the Internal Revenue Service; and (j) parties to executory contracts proposed to be assumed and assigned. The Debtors believe that the notice provided for herein is fair and adequate and no other or further notice is necessary.
- 61. In addition, as soon as approved by the Court, the Debtors will immediately serve the Sale Hearing Notice, substantially in the form of the attached **Exhibit D**, by standard first-class mail, e-mail, or facsimile, to all known parties in interest in the Debtors' bankruptcy cases and on all other parties ordered by the Court.

CONCLUSION

WHEREFORE, PREMISES CONSIDERED, the Debtors respectfully request (A) at the preliminary hearing on the Sale Motion, entry of an order (1) scheduling an expedited hearing to consider final approval of the Sale Motion; (2) setting deadlines and requirements for parties in interest to file (a) objections to the Sale Motion, and (b) objections to the proposed assumption and assignment of executory contracts in connection with the sale and the payment of cure claims

in connection therewith; and (3) approving the form and manner of notice of the Final Sale Hearing; and (B) at the Final Sale Hearing on the Sale Motion, entry of an order (1) approving the Debtors' sale of the Purchased Assets to the Purchaser free and clear of liens, claims, encumbrances and interests to the fullest extent of section 363 of the Bankruptcy Code; (2) approving the proposed assumption and assignment by the Debtors to purchaser of the Assumed Contracts and payment of all cure claims in connection therewith pursuant to section 365 of the Bankruptcy Code; (3) approving the Debtors' use of the cash portion of the consideration received from Purchaser at closing to repay all outstanding principal, interest and fees due under the DIP Loan as of the Closing Date; (4) approving the transfer by Hercules of the Hercules Shares at closing to an escrow agent to be distributed by the escrow agent pursuant to a confirmed chapter 11 plan or pursuant to further orders of this Court; and (C) all such other and further relief as may be just and proper both at law and in equity.

Dated: February 11, 2011

Respectfully submitted,

FULBRIGHT & JAWORSKI L.L.P.

By: _/s/ Berry D. Spears_

Berry D. Spears
State Bar No. 18893300
Johnathan C. Bolton
State Bar No. 24025260
1301 McKinney, Suite 5100
Houston, Texas 77010-3095
Telephone: (713) 651-5151
Facsimile: (713) 651-5246
bspears@fulbright.com
jbolton@fulbright.com

and

JORDAN, HYDEN, WOMBLE, CULBRETH & HOLZER P.C.

Shelby A. Jordan
State Bar No. 11016700
Nathanial Peter Holzer
State Bar No. 00793971
500 N. Shoreline Blvd., Suite 900
Corpus Christi, Texas 78401-0341
Telephone: (361) 653-6624

Facsimile: (361) 888-5555 sjordan@jhwclaw.com pholzer@jhwclaw.com

PROPOSED ATTORNEYS FOR THE DEBTORS AND DEBTORS-IN-POSSESSION

EXHIBIT A

The Asset Purchase Agreement

ASSET PURCHASE AGREEMENT

By and Among

SEAHAWK DRILLING, INC.,
SEAHAWK GLOBAL HOLDINGS LLC,
SEAHAWK MEXICO HOLDINGS LLC,
SEAHAWK DRILLING MANAGEMENT LLC,
SEAHAWK DRILLING LLC,
SEAHAWK OFFSHORE MANAGEMENT LLC,
ENERGY SUPPLY INTERNATIONAL LLC,
and
SEAHAWK DRILLING USA, LLC

(Sellers)

SD DRILLING LLC

and

HERCULES OFFSHORE, INC.

(Purchasers)

FEBRUARY 11, 2011

TABLE OF CONTENTS

		Page
ARTIC	LE I DEFINITIONS AND INTERPRETATION	2
1.1	Definitions and Interpretation	
ARTIC	LE II PURCHASE AND SALE	
2.1	Purchase and Sale of Purchased Assets	2
2.2	Excluded Assets	3
2.3	Transfer of Contracts and Permits	
2.4	Aggregate Consideration	6
2.5	Pre-Closing Adjustment to Base Aggregate Consideration	6
2.6	Payment of Aggregate Consideration	
2.7	Assumption of Liabilities	
2.8	Retained Liabilities	
2.9	Casualty Losses	
2.10	Post-Closing Adjustment.	11
ARTICI	LE III CLOSING	12
3.1	Closing	12
3.2	Risk of Loss	
3.3	Deliveries of Sellers.	
3.4	Deliveries of Purchasers	14
ARTICI	LE IV REPRESENTATIONS AND WARRANTIES OF SELLERS	16
4.1	Organization	
4.2	Power and Authority	
4.3	Authorization	
4.4	Noncontravention	
4.5	Assets	
4.6	Financial Statements	
4.7	Absence of Undisclosed Liabilities	
4.8	Legal Compliance	
4.9	Litigation; Proceedings	
4.10	Permits	
4.11	Rigs	
4.12	Employees	
4.13	Employee Benefit Plans	20
4.14	Environmental Matters	20
4.15	Tax Matters	20
4.16	Absence of Certain Developments	21
4.17	Contracts	
4.18	Customers and Suppliers	
4.19	Accounts Receivable	
4.20	Cash Management	
4 21	DIP Financing	22

TABLE OF CONTENTS

(continued)

		Page
4.22	Broker Fees	22
4.23	No Preferential Purchase Rights	
	LE V REPRESENTATIONS AND WARRANTIES OF PURCHASERS	
5.1	Organization	23
5.2	Power and Authority	
5.3	Authorization	
5.4	Noncontravention	24
5.5	Consents	24
5.6	Financing	24
5.7	Litigation	24
5.8	Hercules Shares	24
5.9	Hercules Capitalization	25
5.10	Financial Statements	25
5.11	SEC Filings and Exchange Listing Compliance	25
5.12	No Material Adverse Change	25
5.13	Broker Fees	
5.14	"AS IS, WHERE IS	26
ARTICI	LE VI PRE-CLOSING COVENANTS	26
6.1	General	26
6.2	Notices and Consents	26
6.3	Bankruptcy	26
6.4	Preservation of Purchased Assets and Business	27
6.5	No Shop	28
6.6	DIP Matters	29
6.7	Other Information	30
6.8	Public Announcement	30
6.9	Lender Consent	30
6.10	Notices of Certain Events	30
6.11	Employee Data	30
ARTICI	LE VII EMPLOYEE MATTERS	31
7.1	Announcement	31
7.2	Offers of Employment	
7.3	Terms of Employment and Transition	
7.4	Employee Benefits	
7.5	Cooperation	32
7.6	Miscellaneous	32
ARTICI	LE VIII TAX MATTERS	32
8.1	Sales and Transfer Taxes	32
8.2	Purchase Consideration Allocation	
8.3	Tax Allocation	

TABLE OF CONTENTS (continued)

		Page
ADTICI	LE IX POST-CLOSING COVENANTS	22
AKIICI		
9.1	Books and Records	
9.2	Computer Software	
9.3	Monies Collected	
9.4	Survivability	
9.5	Other Information	
9.6	Public Announcement	
9.7	Trademarks	
9.8	Hercules Shares	
9.9	Releases	
9.10	Office Equipment	35
ARTICI	LE X CONDITIONS TO CLOSING	35
10.1	Conditions Precedent to Obligation of Purchasers	35
10.2	Conditions Precedent to Obligations of Sellers	36
ARTICI	LE XI TERMINATION	37
11.1	Termination of Agreement	37
11.2	Effect of Termination	
ARTICI	LE XII MISCELLANEOUS	39
12.1	Waiver	39
12.2	Notices	39
12.3	Further Assurances	
12.4	Expenses	41
12.5	Successors and Assigns	
12.6	Third Party Beneficiaries	41
12.7	Time of the Essence	41
12.8	Assignment	41
12.9	Governing Law; Venue	
12.10	,	
12.11	O ,	
12.12	-	
12.13	Certain Limitations	42

SCHEDULES

Schedule	Name	
0.1.1.1.1	D. C. idiana and Indonesia	
Schedule 1.1	Definitions and Interpretation	
Schedule 2.1(a)	Transferred Rigs	
Schedule 2.1(b)	Assigned Contracts	
Schedule 2.1(c)	Transferred Equipment	
Schedule 2.1(d)	Transferred Rolling Stock	
Schedule 2.2(c)	Excluded Contracts	
Schedule 2.2(e)	Excluded Insurance Benefits	
Schedule 2.2(k)	Excluded Claims and Warranties	
Schedule 2.2(l)	Excluded Leased Real Property	
Schedule 2.8(a)(vii)	Excluded Legal Proceedings	
Schedule 2.9	Transferred Rig Casualties	
Schedule 4.4	Noncontravention	
Schedule 4.5	Asset Title, Maintenance, and Condition	
Schedule 4.6	Financial Statements	
Schedule 4.7	Undisclosed Liabilities	
Schedule 4.8	Legal Compliance	
Schedule 4.9	Litigation Proceedings	
Schedule 4.10	Permits	
Schedule 4.11(a)	Rig Documentation and Maintenance	
Schedule 4.12	Employee and Labor Matters	
Schedule 4.13	Employee Benefits	
Schedule 4.14	Environmental Matters	
Schedule 4.15	Tax Matters	
Schedule 4.16	Certain Developments	
Schedule 5.12	Adverse Changes of Purchasers	
Schedule 10.1(g)	Consents	

Note: the inclusion of any matter on a Schedule shall not affect the treatment or characterization of such matter in accordance with <u>Article II</u>. In the event of any ambiguity, the provisions of <u>Article II</u> shall prevail.

EXHIBITS

<u>Exhibit</u>	Name
Exhibit 3.3(a)	Form of Master Bill of Sale
Exhibit 3.3(b)	Form of Assignment and Assumption of Contracts
Exhibit 3.3(c)	Form of Assignment and Assumption of Intangible Assets
Exhibit 3.3(g)	Form of Business Capabilities Certificate
Exhibit 4.21	Initial DIP Budget
Exhibit 6.3(a)	Form of Sale Motion
Exhibit 6.3(b)	Form of Termination Fee Motion
Exhibit 6.3(c)	Form of Transfer Order [To be provided to the Bankruptcy
Exhibit 6.3(d)	-
Exhibit 6.3(d)	Court] Form of Termination Fee Order

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "Agreement") dated as of February 11, 2011, is by and among SEAHAWK DRILLING, INC., a Delaware corporation ("Seahawk Parent"), SEAHAWK GLOBAL HOLDINGS LLC, a Delaware limited liability company ("Seahawk Global Holdings"), SEAHAWK MEXICO HOLDINGS LLC, a Delaware limited liability company ("Seahawk Mexico Holdings"), SEAHAWK DRILLING MANAGEMENT LLC, a Delaware limited liability company ("Seahawk Drilling Management"), SEAHAWK DRILLING LLC, a Delaware limited liability company ("Seahawk Drilling"), SEAHAWK OFFSHORE MANAGEMENT LLC, a Delaware limited liability company ("Seahawk Offshore Management"), ENERGY SUPPLY INTERNATIONAL LLC, a Delaware limited liability company ("Energy Supply International"), SEAHAWK DRILLING USA LLC, a Delaware limited liability company ("Seahawk USA"), SD DRILLING LLC, a Delaware limited liability company ("Newco"), and Hercules Offshore, Inc., a Delaware corporation ("Hercules Parent"). Seahawk Parent, Seahawk Global Holdings, Seahawk Mexico Holdings, Seahawk Drilling Management, Seahawk Drilling, Seahawk Offshore Management, Energy Supply International, and Seahawk USA may be referred to in this Agreement collectively as the "Sellers" and individually as a "Seller." Newco and Hercules Parent may be referred to in this Agreement collectively as the "Purchasers" and individually as a "Purchaser." Sellers and Purchasers may be referred to in this Agreement collectively as the "Parties" and individually as a "Party."

RECITALS

WHEREAS, Sellers are engaged in the business of owning and operating jackup rigs located in the United States (the "Rigs") that provide contract offshore drilling services to the oil and natural gas exploration and production industry in the Gulf of Mexico (the "Business");

WHEREAS, Sellers have filed or will file voluntary petitions (the "Petitions") for relief commencing bankruptcy cases (the "Bankruptcy Cases") under Chapter 11 of title 11 of the United States Bankruptcy Code, 11 U.S.C. §§101 et seq. (the "Bankruptcy Code") in the United States Bankruptcy Court for the Southern District of Texas, Corpus Christi Division (the "Bankruptcy Court") or such other court as the Parties may mutually agree;

WHEREAS, subject to the approval of the Bankruptcy Court, Sellers each desire to sell to Purchasers, and Purchasers desire to acquire from Sellers, all or substantially all of the property, assets, rights, and privileges of Sellers related to, used in, or otherwise associated with the operation of the Business on the terms and subject to the conditions set forth in this Agreement, pursuant to the Transfer Order, and in accordance with the Bankruptcy Code and for the purpose of facilitating the confirmation of a Plan of Reorganization;

NOW THEREFORE, in consideration of the premises and mutual agreements, benefits, representations, warranties, and covenants of the Parties contained herein, and other good and valuable consideration the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound subject to the terms and conditions hereof and approval of the Bankruptcy Court, the Parties each agree as follows:

ARTICLE I DEFINITIONS AND INTERPRETATION

1.1 Definitions and Interpretation. Unless otherwise expressly provided to the contrary in this Agreement (a) capitalized terms used herein shall have the meanings set forth in Section 1.1 of Schedule 1.1, unless the context otherwise requires and (b) this Agreement shall be interpreted in accordance with the provisions set forth in Section 1.2 of Schedule 1.1.

ARTICLE II PURCHASE AND SALE

- 2.1 Purchase and Sale of Purchased Assets. Upon the terms and subject to the conditions set forth in this Agreement and the Transfer Order, at the Closing, Sellers shall sell, convey, assign, transfer, and deliver to Purchasers, and Purchasers shall purchase, acquire, and accept from each Seller, free and clear of any interest in such property as provided by 11 U.S.C. §363(f), all of Sellers' right, title, and interest in and to all of the property, assets, rights, and privileges of Sellers related to, used in, held for use in, or otherwise associated with the operation of the Business, whether real, personal, or mixed, tangible or intangible, of every kind and description and wherever located, whether identified in the Transfer Order Schedules, whether accrued, contingent or otherwise, other than the Excluded Assets (collectively, the "Purchased Assets"), including the following:
- (a) Rigs. Subject to Section 2.9, the jackup Rigs listed on Schedule 2.1(a), each of which is located in the United States of America (the "Transferred Rigs");
- (b) Contracts. All of the interests, rights, Claims, and benefits arising or accruing to any of Sellers under any Contracts to which a Seller is a party or has or may acquire a benefit and that relate to the Purchased Assets or Business, including the Contracts listed in Schedule 2.1(b), to the extent Sellers' interest in any such Contract is assignable, but subject to the rights of Purchasers not to assume specified Contracts as contemplated by Section 2.3(a) (after taking into account all of the foregoing, the "Assigned Contracts");
- (c) Equipment. All furniture, equipment, computers, computer equipment, machinery, tools, hand tools, spare parts, test equipment, supplies, inventory, office supplies, telephones, and all other tangible personal property of every kind and description insofar as any of the foregoing relates to the operation of the Purchased Assets or Business (the "Equipment"), including the Equipment listed in Schedule 2.1(c).
- (d) Rolling Stock. All automobiles, vans, trucks, trailers, and other motorized and similar vehicles and stock of every kind, whether owned or leased, used in the operation of the Purchased Assets or Business, including as listed in Schedule 2.1(d);
- (e) Additional Tangible Assets. All other Tangible Personal Property of every kind used in the operation of the Purchased Assets or Business, together with any rights appurtenant thereto, including any express or implied warranty by the manufacturer, vendor, or lessor of any such Tangible Personal Property;

- (f) Accounts and Notes Receivable. All trade accounts receivable, notes receivable, and other rights of Sellers to payment from customers and other third parties and other amounts due from customers and other third parties or that become due, including all assets constituting working capital, that relate to or arise from the operation of the Purchased Assets or Business;
- (g) Insurance Benefits. All insurance benefits arising from or related to the Purchased Assets, Business, and Assumed Liabilities;
- (h) Cash. All cash and cash equivalents, securities, money on deposit with banks, certificates of deposit, and short-term investments (if any) of Sellers;
- (i) Prepaid Deposits and Expenses. Any deposits and prepaid expenses (paid to or by Sellers), Claims for refunds, and rights of set off related to the Purchased Assets or Business;
- (j) Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities against third parties relating or attributable to the Business, the Purchased Assets, or the Assumed Liabilities, whether choate or inchoate, known or unknown, contingent or non-contingent;
- (k) *Permits*. Any and all Permits, including pending applications or filings therefor and renewals thereof, of every kind under which any of Sellers has or may acquire any benefits or rights, or by which any of Sellers or any of the Purchased Assets may be subject or bound, or that relate to or are used or held for use in the operation of the Purchased Assets or Business (collectively, the "Assigned Permits");
- (l) Intangible Assets. Any and all customer and supplier relationships, and other Intangible Personal Property of every kind (other than the Computer Software, Trademark Intellectual Property and Website Intellectual Property, and uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names that constitute Intellectual Property) that relate to or are used in the operation of the Purchased Assets or Business, together with all rights appurtenant thereto (the "Assigned Intangible Assets");
- (m) Books and Records. Original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets and photocopies of the other Books and Records (collectively, the "Transferred Books and Records");
- (n) Other Assets. To the extent not otherwise enumerated in this Section 2.1, all other tangible or intangible assets, rights, privileges, benefits, Claims, and interests of a Seller, whether real, personal or mixed, of every kind and description and wherever located, that relate to, used in or held for use in the operation of the Purchased Assets or Business.
- **2.2 Excluded Assets**. Notwithstanding anything to the contrary in <u>Section 2.1</u> or elsewhere in this Agreement or other Transaction Documents, the Purchased Assets shall not include, and Purchasers will not acquire any interest in or purchase the following assets which shall remain the property of the applicable Seller (collectively, the "*Excluded Assets*"):

- (a) Transaction Rights. All rights of Sellers under this Agreement and the other Transaction Documents, and all cash and non-cash consideration (including the Hercules Shares and the Cash Payment) payable or deliverable to, or on behalf of, Sellers by Purchasers pursuant hereto and thereto;
- (b) Rigs. Any Rig that is deemed to be an Excluded Asset pursuant to Section 2.9, together with all of the Equipment located on each such Rig;
- (c) Contracts. Any and all of the interests, rights, Claims, and benefits arising or accruing to any of Sellers under any Contract (including any Lease) that is not an Assigned Contract, including those Contracts listed in Schedule 2.2(c);
- (d) Permits. Any and all rights of Sellers under any Permit that is not an Assigned Permit;
- (e) Insurance Benefits. Any and all insurance benefits arising from or relating to (i) Excluded Assets and Retained Liabilities, (ii) Claims made by Sellers prior to the Closing with respect to the Purchased Assets and Business, and (iii) the matters set forth in Schedule 2.2(e);
- (f) Books and Records. The original copies of all of the Books and Records, other than the original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets (which shall be provided to Purchasers pursuant to Section 2.1(m) and of which Sellers may retain photocopies), and any Books and Records of Sellers (i) that are not permitted to be transferred to Purchasers under applicable Law, (ii) that constitute charters, bylaws, limited liability company agreements, minute books, stock transfer records, and other records related to the corporate governance of Sellers, and (iii) all other books and records of Sellers that do not relate primarily to the Purchased Assets, Transferred Employees, or Business (collectively, the "Retained Books and Records");
- (g) Third Party Property. Any improvements, equipment, inventory and any other tangible personal property located at the Leased Real Property or on the Transferred Rigs as of the Closing Date that are not owned by or leased to Sellers; and
- (h) Equity Interests. Any Equity Interest in any Seller or any direct or indirect subsidiary of a Seller;
- (i) Bankruptcy Claims. Any and all Claims or rights of Sellers arising under chapter 5 of the Bankruptcy Code;
- (j) Pride Claims. Any and all Claims or rights of Sellers or any Affiliate thereof against Pride or any Affiliate thereof;
- (k) Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities of Sellers, whether choate or inchoate, known or unknown, contingent or noncontingent with respect to the matters set forth in Schedule 2.2(k);

- (l) Leased Real Property. All leasehold interests in the real property listed in Schedule 2.2(l) (the "Leased Real Property"), together with all Real Property Rights related thereto and all Improvements and Tangible Personal Property located at, under, or on such Leased Real Properties or otherwise related thereto;
 - (m) Intellectual Property. The Intellectual Property constituting:
- (i) Sellers' software programs, tools, kits, and any content or related documentation or third party or open source code embedded therein, either locally stored on Sellers' computers or remotely accessed by Sellers, that is owned or made available to Sellers and is used in, held for use in, or is necessary for the operation of the Purchased Assets or Business or otherwise, and any upgrades, updates, releases, fixes, enhancements, or modifications thereto, and all written materials and specifications applicable thereto (collectively, the "Computer Software");
- (ii) the name "Seahawk Drilling" and the mark "Seahawk Drilling" and any other registered or unregistered trademarks, trade names, service marks, domain names, and email addresses, including the term "Seahawk", used or held for use in the Business (collectively, the "*Trademarks*"), and any and all goodwill associated with the Business embodied in the Trademarks, and any and all rights of Sellers with respect to the Trademarks (collectively, the "*Trademark Intellectual Property*");
- (iii) Sellers' internet website located at <u>www.seahawkdrilling.com</u> and any derivations thereof (collectively, the "Website") and all intellectual property rights that may exist or arise in connection with the Website, including all Underlying Technology (collectively, the "Website Intellectual Property"); and
- (iv) uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names.
- (n) Goodwill. Any goodwill and going concern value related to the Business; and
- (o) Accounts and Notes Receivable. The trade accounts receivable, notes receivable, and other rights of Sellers to payment from, and other amounts due or that become due, including all assets constituting working capital, that relate to Blake International USA Rigs, LLC or any Affiliate thereof.

2.3 Transfer of Contracts and Permits.

(a) At least three Business Days prior to the Closing Date, Purchasers may designate any Contracts that would otherwise be an Assigned Contract as a Contract to be excluded from the Purchased Assets for purposes of Section 2.1(b). Any such designation by Purchasers shall not reduce Purchaser's assumption of any working capital liabilities pursuant to Section 2.7(a), unless both (i) such working capital liability was incurred pursuant to such designated Contract and (ii) such working capital liability is not attributable to goods and services provided to the Business prior to the Closing, in which case such working capital liability shall not be assumed under Section 2.7(a).

- Notwithstanding anything herein to the contrary, this Agreement shall not (b) constitute an assignment to Purchasers of (and Purchasers shall not assume any rights or obligations under) any Contract or Permit if an attempted assignment thereof, without the Consent of a Governmental Authority or other third party thereto, would constitute a breach of the Contract or Permit, unless such Consent has been received or such assignment is otherwise authorized and permitted by the Transfer Order or other Order of the Bankruptcy Court. The Transfer Order shall provide that any Assigned Contracts that are executory contracts shall be assumed by Sellers and assigned to Purchasers at the Closing, and Sellers shall pay all pre-Closing cure amounts under Assigned Contracts required to be so paid by the Bankruptcy Court in connection with such assumptions by Sellers (the "Cure Amounts"). The Sellers shall use their commercially reasonable efforts to advise the Purchasers promptly in writing with respect to any Contract or Permit that Sellers' Know or have substantial reason to believe shall not or may not be subject to assignment to Purchasers hereunder. Without in any way limiting Sellers' obligations to obtain all Consents necessary for the sale, transfer, assignment and delivery of the Assigned Contracts and Assigned Permits and the other Purchased Assets to Purchasers hereunder, if any such Consent is not obtained or if such assignment is not permitted irrespective of Consent and the Closing hereunder is consummated, Sellers shall use commercially reasonable efforts to continue to promptly seek Consent or other approval for the sale, transfer, assignment and delivery thereof; provided, that in no circumstance shall any such obligation of Sellers extend beyond the date that Sellers' Plan of Reorganization is confirmed by the Bankruptcy Court and has become effective.
- 2.4 Aggregate Consideration. Subject to the other terms of this Agreement, the aggregate consideration for the Purchased Assets shall consist of (a) 22,321,425 shares of Hercules Common Stock (the "Hercules Shares") and (b) cash in an amount equal to \$25,000,012 (the "Cash Payment" and, together with the Hercules Shares, the "Aggregate Consideration").

2.5 Pre-Closing Adjustment to Base Aggregate Consideration.

- (a) Not more than six (6) Business Days nor less than three (3) Business Days prior to the Closing Date, Sellers shall deliver to Purchasers a certificate of an authorized officer setting forth Sellers' good faith estimate, as of the Closing Date, of Net Working Capital (the "Estimated Net Working Capital"). Such statement shall include separate estimates, as of the Closing Date, for (1) cash and cash equivalents included in the Purchased Assets, (2) the amount of outstanding accounts receivable included in the Purchased Assets and (3) the amount of current liabilities to be assumed pursuant to Section 2.7(a).
- (b) If the Estimated Net Working Capital is greater than \$4,000,000 (the "Net Working Capital Target"), then the number of Hercules Shares shall be increased by the number of shares (rounded to the nearest whole share) of Hercules Common Stock equal to the quotient of (A) such excess, divided by (B) 3.36. If the Net Working Capital Target is greater than the Estimated Net Working Capital, then the number of Hercules Shares shall be decreased by the number of shares (rounded to the nearest whole share) of Hercules Common Stock equal to the quotient of (A) such excess, divided by (B) 3.36.

- (c) At the Closing, the Aggregate Consideration (and its components) payable by Purchasers pursuant to Section 2.4 shall be further adjusted as follows:
- (i) if the outstanding principal amount of the DIP Loan and accrued interest thereon as of the Closing Date (the "Closing DIP Amount") is greater than \$25,000,012 (the "Original DIP Amount"), then (1) the Cash Payment shall be increased dollar-for-dollar by the amount of the difference, up to, and in any event not more than, \$20,000,000, between the Closing DIP Amount less the Original DIP Amount (such allowed difference, the "Excess DIP Amount") and (2) the number of Hercules Shares shall be reduced by the number of shares (rounded to the nearest whole share) of Hercules Common Stock equal to the quotient of (A) the Excess DIP Amount, divided by (B) 3.36;
- (ii) the number of Hercules Shares shall be reduced by the number of shares (rounded to the nearest whole share) of Hercules Common Stock equal to the quotient of (A) the aggregate amount, if any, of Rig Loss Value and Rig Repair Costs claimed under <u>Section</u> 2.9, divided by (B) 3.36; and
- (iii) with respect to COBRA coverage for Seller Employees, (1) the Aggregate Consideration shall be reduced by \$1,200,000.00 and (2) either, as determined by Purchasers in their sole discretion, (A) the Cash Payment shall be reduced by \$1,200,000.00 or (B) the number of Hercules Shares shall be reduced by the number of shares (rounded to the nearest whole share) of Hercules Common Stock equal to \$1,200,000 divided by 3.36.
- 2.6 Payment of Aggregate Consideration. At the Closing and against delivery of the items specified in Section 3.3, Purchasers shall pay and deliver to Sellers the Aggregate Consideration, as adjusted, as consideration for all of the Purchased Assets.
- **2.7** Assumption of Liabilities. Upon the terms and subject to the conditions of this Agreement and the Transfer Order, at the Closing, Sellers agree to assign and transfer to Purchasers, and Purchasers agree to accept, assume, pay, perform, discharge, and satisfy, each of the following Liabilities to the extent not previously paid, performed, discharged or satisfied (collectively, the "Assumed Liabilities"):
- (a) Working Capital Liabilities. The current accounts payable of, and accruals for goods and services received by, Sellers as of the Closing Date attributable to the Purchased Assets; provided that, for purposes of clarity, such working capital liabilities shall exclude accounts payable to Blake International USA Rigs, LLC or its Affiliates. Nothing in this Section 2.7(a) shall be deemed to effect the assumption of any Contract that is not otherwise an Assigned Contract, but the assumption of liabilities under this Section 2.7(a) shall be subject to Section 2.3(a).
- (b) Contracts. The Liabilities of Sellers under the terms of any Assigned Contract to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the Closing;
- (c) *Permits*. The Liabilities of Sellers under the terms of any Assigned Permit to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the Closing;

- (d) Claims relating to Assumed Liabilities. In the event Claims are filed in the Sellers' Bankruptcy Cases on account on the Assumed Liabilities, Sellers shall provide Purchasers written notice of such filing within five (5) Business Days after their filing. Purchasers shall have the right to appear, defend and settle those Claims in the Bankruptcy Court. Notwithstanding the foregoing, Sellers shall be solely responsible for any Claims that exceed the amount of current liabilities assumed under Section 2.7(a) and included in the statement of Final Closing Net Working Capital; and
- (e) Performance Bond. Purchasers shall either assume Sellers' \$2.0 performance bond in favor of the Texas general land office relating to the Seahawk 2008 Rig, or otherwise post their own performance bond such that such \$2.0 performance bond may be terminated by Sellers.

2.8 Retained Liabilities.

- (a) Subject to the terms of the Transfer Order, Sellers shall retain and be solely liable for and hereby expressly agree to retain any and all of their respective Liabilities not expressly assumed by Purchasers in Section 2.7 (collectively, the "Retained Liabilities"), regardless of whether any such Retained Liability is disclosed herein or in any Schedule hereto, whether known or unknown, absolute or contingent, liquidated or unliquidated, whether due or to become due, and whether Claims with respect thereto are asserted before or after the Closing Date. Notwithstanding the preceding sentence, Retained Liabilities shall include the following:
- (i) Transaction Liabilities. Any and all Liabilities of Sellers under the Transaction Documents;
- (ii) Excluded Assets. Any and all Liabilities of Sellers with respect to the Excluded Assets;
- (iii) Purchased Assets and Pre-Closing Business. Except as provided in Sections 2.7(a), 2.7(b), and 2.7(c), all Claims and Liabilities arising out of Sellers' ownership, operation, use, or maintenance of the Purchased Assets or Business, as well as any matters arising from events occurring, conditions existing, or costs accruing prior to the Closing and Taxes;
- (iv) Contracts. Except as provided in Sections 2.7(a) and 2.7(b), any and all Liabilities of Seller, including Cure Amounts, that relate to the Assigned Contracts to the extent any such Liability is performable or otherwise attributable to the period prior to the Closing;
- (v) *Permits*. Except as provided in <u>Sections 2.7(a)</u> and <u>2.7(c)</u>, any and all Liabilities of Sellers that relate to the Assigned Permits to the extent any such Liability is performable or otherwise attributable to the period prior to the Closing;
- (vi) Taxes. Any and all Liabilities of Sellers for Taxes that relate to (A) the ownership, operation, use, or maintenance of the Purchased Assets or Business prior to the Closing Date, or (B) any sales, use, transfer, or other similar Taxes imposed as a result of the consummation of the Contemplated Transactions or performance of the Transaction Documents;

- (vii) Legal Proceedings. Any and all Liabilities of Sellers that relate to any Proceeding (A) involving the Purchased Assets or Business, including warranty, personal injury, breach of contract, failure to perform, infringement, noncompliance with Law, and tort Claims, that is (1) pending or threatened as of the Closing, or (2) commenced after the Closing but that arises out of or relate to any event, omission, or occurrence happening as of or prior to the Closing, or (B) set forth in Schedule 2.8(a)(vii);
- (viii) *Pride Claims*. Any and all Liabilities of Sellers under agreements entered into among Seahawk Parent and Pride International, Inc. ("*Pride*") at the time of Seahawk Parent's spin-off from Pride effective August 4, 2009, including (i) any obligation to indemnify Pride and its Affiliates for Mexican tax liabilities associated with Seahawk Parent's Mexican subsidiaries under that certain Tax Sharing Agreement dated August 4, 2009, between Seahawk Parent and Pride, part of which tax liabilities are supported by letters of credit from Pride, and any settlement with respect to such matters, and (ii) any and all Liabilities of Sellers (if any) to Pride and its Affiliates under such agreements resulting from the Contemplated Transactions;
- (ix) *Pride Wyoming*. Any and all Liabilities of Sellers relating to the loss of the Pride Wyoming Rig, including salvage and wreckage removal costs and any third party damage Claims and any continuing Liability under applicable Law relating to the Pride Wyoming Rig;
- (x) FCPA Investigation. Any and all Liabilities of Sellers, including for purposes of clarity any and all fines and penalties (if any), of Sellers with respect to violations, alleged violations, and investigations for alleged potential or actual violations by Sellers and any of their respective Affiliates of the U.S. Foreign Corrupt Practices Act;
- (xi) Environmental Liabilities. Any and all Liabilities of Sellers with respect to any violation of Law including those arising from (A) the release, threatened release, presence, treatment, storage, disposal (including disposal at off site locations), handling, transportation or arrangement for transportation of hazardous substances prior to the Closing, (B) any failure of Sellers to comply in any respect with Environmental, Health, and Safety Laws prior to the Closing, and (C) any facts, events, or circumstances in existence prior to the Closing that give rise to Liabilities pursuant to Environmental, Health, and Safety Laws;
- (xii) Employee Benefit Plans. Any and all Liabilities of Sellers, fiduciaries or ERISA Affiliate under any Employee Benefit Plan maintained or contributed to by any Seller, fiduciary or any ERISA Affiliate or with respect to which any Seller, fiduciary or ERISA Affiliate has any Liability, whether prior to, on, or after the Closing, including any (a) failure to comply with all applicable Laws, (b) liability with respect to audits, inquiries, Proceedings, or Claims with any Governmental Authority with respect to any Employee Benefit Plan, (c) participation in any "multiemployer" plan (within the meaning of Section 3(37) of ERISA, whether or not governed by the provisions of ERISA) or withdrawal liability with respect to any multiemployer plan, (d) required employer contributions with respect to the Employee Benefit Plans, (e) accumulated funding deficiency, (f) Lien under ERISA or Section 412 of the Code, or (g) the termination of, or intent to terminate, any Plan;

- (xiii) Employees. Any and all Liabilities of any nature of Sellers or any fiduciary or ERISA Affiliate to Seller Employees, including Liabilities with respect to (a) any Contract, plan or policy, (b) wages, withholdings, overtime pay, minimum wage, employment Tax, vacation, sick pay, bonuses, severance pay, retirement, or other compensation, (c) benefits under Employee Benefit Plans, (d) the Worker Adjustment and Retraining Notification Act (WARN) of August 4, 1988 or equivalent state or local statutory or regulatory requirements, (e) any collective bargaining agreement or obligation or requirement under the National Labor Relations Act, (f) reporting, filing, hiring or other employment obligations with the Office of Federal Contract Compliance Programs, (g) all immigration related obligations, including all requirements of the Immigration Reform and Control Act of 1986, (h) any governmental or administrative proceeding for the enforcement of labor and employment laws and regulations, and (i) all other employment and employment related federal, state and local statutes, regulations, administrative requirements, common laws, and public policies;
- (xiv) Intercompany Liabilities. Any and all Liabilities of Sellers for intercompany advances, charges, or accounts payable of any kind or nature;
- (xv) Bankruptcy Claims. Any and all Claims filed against Sellers in their respective Bankruptcy Cases, except for the Assumed Liabilities, as limited in Section 2.7(a);
- (xvi) Broker Fees. Any and all fees, commissions, and other compensation to any broker, finder, or agent retained by any Seller as contemplated by Section 4.22.
- (b) The Retained Liabilities shall constitute Claims and alleged Claims in the Sellers' Bankruptcy Cases; *provided*, *however*, that nothing herein shall create any rights in favor of the holders of such Claims and alleged Claims or create any priority right of payment.

2.9 Casualty Losses.

If, between the date of this Agreement and the Closing, there is an actual casualty loss to any Transferred Rig in which the cost to recover, salvage, and repair such Transferred Rig to the state of condition and repair existing for such Transferred Rig as of the date of this Agreement (collectively, the "Rig Repair Costs") exceeds the casualty loss value ascribed to such Rig in Schedule 2.9 (the "Casualty Loss Amount"), or any such loss by seizure, forced sale or other involuntary transfer (any of such events, as determined by a qualified, independent marine surveyor based in Louisiana or Texas, and reasonably acceptable to Sellers and Purchasers, a "Casualty Loss"), then, if so directed by Purchaser in its sole discretion, the subject Transferred Rig shall be deemed an Excluded Asset and shall not be sold to Purchasers hereunder, and the Aggregate Consideration shall be reduced by an amount equal to the rig loss value ascribed to such Transferred Rig in Schedule 2.9 (the "Rig Loss Value"), and such reduction shall be effected through a reduction in the number of Hercules Shares in accordance with Section 2.5(c)(ii). If, between the date of this Agreement and the Closing, any Rig is damaged (other than ordinary wear and tear), but such damage is less than the Casualty Loss Amount, then the number of Hercules Shares shall be reduced in accordance with Section 2.5(c)(ii); provided, however, that for the avoidance of doubt in no event shall any such reduction in the number of Hercules Shares be made for ordinary wear and tear occurring prior to the Closing. The Purchasers and the Sellers shall cooperate with each other and work together to

determine the amount of any such Rig Repair Costs. If the Parties are unable to agree on such Rig Repair Costs within five Business Days after Sellers notify the Purchasers of the damage to the Rig, the Parties shall submit the issue to GL Noble Denton or, if such firm is unable or unwilling to serve, to another Person mutually acceptable to the Parties (the "Appraiser"). Sellers and Purchasers shall each submit to the Appraiser their proposed Rig Repair Costs together with support calculations and documentation with respect thereto. The Appraiser shall, within 15 days following receipt of both proposals, advise the Parties as to its determination of the Rig Repair Costs. The final value of such costs of recovery, salvage and repairs as determined in this Section 2.9 shall be the "Rig Repair Costs" for purposes of this Agreement. The costs of retaining the Appraiser shall be borne equally by Sellers and Purchasers. Sellers shall provide Purchasers with prompt notice of any such casualty loss or damage.

(b) The provisions of this <u>Section 2.9</u> shall be applied with respect to all casualty losses or damage suffered by any Transferred Rig and, notwithstanding any other provision of this Agreement, the Parties shall be required to proceed with the Closing regardless of any such casualty loss or damage to any such Transferred Rig, subject to the other terms and conditions for the Closing, *provided*, further that in no event shall any casualty loss or damage to which this <u>Section 2.9</u> applies constitute a breach of any other provision of this Agreement (including the provisions in <u>Article VI</u>) by Sellers or be aggregated with any other actions, omissions, or failures of Sellers in the determination of any breach of this Agreement by Sellers (including the provisions in <u>Article VI</u>), and for the avoidance of doubt no such casualty loss or damage shall in any way be considered in the determination of a termination of this Agreement under <u>Section 11.1</u>.

2.10 Post-Closing Adjustment.

- (a) Not more than twenty (20) days after the Closing Date, Purchasers shall deliver to Sellers a certificate of an authorized officer setting forth Purchasers' calculation, as of the Closing Date, of the Net Working Capital (the "Proposed Closing Net Working Capital"). Such statement shall include separate line items, as of the Closing, for (i) cash and cash equivalents included in the Purchased Assets, (ii) the amount of outstanding accounts receivable included in the Purchased Assets, and (iii) the amount of Assumed Liabilities described in Section 2.7(a).
- (b) If within ten (10) days following delivery of the Proposed Closing Net Working Capital calculation Sellers have not given Purchasers written notice of their objection to the Proposed Closing Net Working Capital calculation (which notice shall state the basis of Sellers' objection(s)), then the Proposed Closing Net Working Capital calculated by Purchasers (or any portion of the calculation to which Sellers do not object) shall constitute the "Final Closing Net Working Capital," shall be binding and conclusive on the Parties.
- (c) If Sellers give Purchasers timely notice of objection, and if Sellers and Purchasers fail to resolve the issues outstanding with respect to the Proposed Closing Net Working Capital within ten (10) days of Purchasers' receipt of Sellers' objection notice, Sellers and Purchasers shall submit the issues remaining in dispute to the Houston office of Deloitte LLP (the "Independent Accountants") for resolution. If for any reason the Houston office of Deloitte LLP is unwilling to act as the Independent Accountants, the Independent Accounts shall

be such other recognized national or regional independent accounting firm mutually acceptable to Purchasers and Sellers.

- (d) If issues are submitted to the Independent Accountants for resolution, (1) Seller and Purchasers shall furnish or cause to be furnished to the Independent Accountants such work papers and other documents and information relating to the disputed issues as the Independent Accountants may request and are available to that Party or its agents and shall be afforded the opportunity to present to the Independent Accountants any material relating to the disputed issues and to discuss the issues with the Independent Accountants; and (2) the determination by the Independent Accountants, as set forth in a notice to be delivered to both Seller and Purchasers within twenty (20) days of the submission to the Independent Accountants of the issues remaining in dispute, shall constitute the "Final Closing Net Working Capital," shall be final, binding and conclusive on the Parties and shall be used in computing the Adjustment Amount. The costs and fees related to such determination by the Independent Accountants with respect to the terms and conditions of such Independent Accountants' engagement, will be shared equally by Purchasers and Sellers.
- Working Capital then within five (5) Business Days of such Final Closing Net Working Capital being provided to the Purchasers, the Purchasers shall issue to Sellers the number of shares (rounded to the nearest whole share) of Hercules Common Stock equal to the quotient of (A) such excess, divided by (B) 3.36, provided that, in no event shall Purchasers be required to issue more than an aggregate of 22,321,425 shares of Hercules Common Stock pursuant to this Agreement, and if Purchasers would otherwise be required, but for this proviso, to issue more shares, then such additional amount owed to Sellers shall be paid in cash in an amount equal to the number of shares exceeding 22,321,425 multiplied by \$3.36. If the Estimated Net Working Capital is greater than the Final Closing Net Working Capital, then the Sellers shall return to Purchasers for cancellation the number of Hercules Shares (rounded to the nearest whole share) equal to the quotient of (A) such excess, divided by (B) 3.36.

ARTICLE III CLOSING

- 3.1 Closing. Provided that this Agreement shall not have been earlier terminated pursuant to Section 11.1, and further provided that all of the conditions set forth in Sections 10.1 and 10.2 to the obligations of the Parties to consummate the Contemplated Transactions (other than conditions with respect to actions each Party will take at the Closing itself) shall have been satisfied or waived, the closing of the Contemplated Transactions (the "Closing") shall take place at the offices of Fulbright & Jaworski L.L.P. at 1301 McKinney, Suite 5100, Houston, Texas 77010 commencing at 11:00 a.m. Central Time on the date that is no earlier than the first Business Day that the Transfer Order has become a Final Order, but no later than the third (3rd) Business Day following such date, or such other date or location as Purchasers and Sellers may mutually agree to in writing (the "Closing Date").
- 3.2 Risk of Loss. The risk of damage, destruction, or other casualty loss to or of the Purchased Assets shall remain with Sellers from and after the execution of this Agreement until

11:59 p.m. on Closing Date, at which time Sellers shall place Purchasers in possession of the Purchased Assets. From and after the Closing, all risk of damage, destruction, or other casualty loss to or of the Purchased Assets shall be borne solely by Purchasers and to the fullest extent permitted by Law, Purchasers agree jointly and severally to indemnify, defend and hold Sellers and their respective officers, directors, equity owners, and agents harmless from against any and all Claims and pay any and all Damages (including for personal injury, property damage or loss, and third party suits) attributable to the Purchased Assets and arising from events first occurring or conditions first existing from and after the Closing.

- 3.3 Deliveries of Sellers. At the Closing, Sellers will deliver (or cause to be delivered) to Purchasers each of the following items:
- (a) Master Bills of Sale. One or more Master Bills of Sale, each substantially in the form of Exhibit 3.3(a), duly executed by the appropriate Seller or Sellers and dated as of the Closing Date;
- (b) Assignments and Assumptions of Contracts. One or more Assignments and Assumptions of Contracts, each substantially in the form of Exhibit 3.3(b), duly executed by the appropriate Seller or Sellers and dated as of the Closing Date;
- (c) Assignments and Assumptions of Intangible Assets. One or more Assignments and Assumptions of Intangible Assets, each substantially in the form of Exhibit 3.3(c), duly executed by the appropriate Seller or Sellers and dated as of the Closing Date;
- (d) Maritime Documents. Any and all certificates, documents or instruments required by the National Vessel Documentation Center, United States Coast Guard, or Vanuatu Maritime Services Ltd. and/or the Office of the Deputy Commissioner of Maritime Affairs, The Republic of Vanuatu, to transfer title of the Transferred Rigs from Sellers to Purchasers, including recordable bills of sale, mortgage releases, class certificates, and requests for permission to sell Transferred Rigs.
- (e) Officer's Certificates. A certificate from Seahawk Parent duly executed by an authorized officer thereof certifying as to the fulfillment of each condition specified in Sections 10.1(a) and 10.1(b) and dated as of the Closing Date;
- (f) Bankruptcy Order. A certified copy of a Transfer Order confirming the Bankruptcy Court's approval of the Contemplated Transactions and this Agreement;
- (g) Business Capabilities Certificate. A certificate substantially in the form of Exhibit 3.3(g), duly executed by an authorized officer of Seahawk Parent and dated as of the Closing Date;
- (h) Corporate Authorizations. Copies of the resolutions of each Seller, certified by the Secretary or Assistant Secretary thereof as being correct and complete and then in full force and effect, authorizing the execution, delivery and performance of this Agreement and the Transaction Documents to which such Seller is a party and the consummation of the

Contemplated Transactions (in each case to the extent required by such Seller's Organizational Documents); and

- (i) Miscellaneous. Any and all other documents, instruments, or agreements contemplated by this Agreement or as are necessary or appropriate or reasonably requested by Purchasers to fully consummate the Contemplated Transactions, in each case in form and substance reasonably satisfactory to Purchasers, duly executed, and dated as of the Closing Date.
- 3.4 Deliveries of Purchasers. At the Closing, Purchasers will deliver (or cause to be delivered) to Sellers each of the following items:
 - (a) Aggregate Consideration.
- (i) Cash Payment. The Cash Payment, against delivery of the items specified in <u>Section 3.3</u> and in accordance with <u>Section 2.6</u>;
- (ii) Hercules Share Certificates. Certificates, dated as of the Closing Date, registered in the name of any one or more of Sellers representing, or a book entry advice statement showing Sellers as the registered holders of, the aggregate number of the Hercules Shares; such Hercules Shares shall be distributed to Escrow Agent, who shall hold and distribute the Hercules Shares in accordance with Orders of the Bankruptcy Court, such shares to have a legend as follows:

THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION. NEITHER THESE SECURITIES NOR ANY INTEREST OR PARTICIPATION HEREIN MAY BE REOFFERED, SOLD, ASSIGNED, TRANSFERRED, DISTRIBUTED, PLEDGED, ENCUMBERED OR OTHERWISE DISPOSED OF IN THE ABSENCE OF SUCH REGISTRATION UNLESS SUCH TRANSACTION IS EXEMPT FROM, OR NOT SUBJECT TO, SUCH REGISTRATION.

In addition, until the amount of the Final Net Working Capital has been determined and paid in accordance with Section 2.10, such shares shall also have a legend as follows (provided that, after Purchasers have delivered the certificate setting forth the Proposed Closing Net Working Capital, Sellers may request removal of such legend, and Purchasers shall deliver shares to Sellers with such legend removed, on the number of Hercules Shares that exceeds the quotient of (x) the amount that the Aggregate Consideration would be reduced pursuant to Section 2.10 if the Proposed Closing Net Working Capital were the Final Closing Net Working Capital, divided by (y) \$3.36):

THE SALE, ASSIGNMENT, PLEDGE, ENCUMBRANCE OR OTHER TRANSFER OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE IS RESTRICTED BY THE TERMS OF, AND THE HOLDER HEREOF IS SUBJECT TO CERTAIN OTHER OBLIGATIONS PURSUANT TO, THE PROVISIONS OF AN ASSET PURCHASE AGREEMENT, DATED AS OF FEBRUARY 11, 2011, AMONG SEAHAWK DRILLING INC., CERTAIN OF ITS AFFILIATES, SD DRILLING LLC AND HERCULES OFFSHORE, INC., AS THE SAME MAY BE AMENDED IN ACCORDANCE WITH ITS TERMS. A COPY OF SUCH ASSET PURCHASE AGREEMENT IS ON FILE AT THE PRINCIPAL OFFICES OF HERCULES OFFSHORE INC.

- (b) Maritime Documents. Any and all certificates, documents or instruments required by the National Vessel Documentation Center, United States Coast Guard, or Vanuatu Maritime Services Ltd. and/or the Office of the Deputy Commissioner of Maritime Affairs, The Republic of Vanuatu, to reregister the Transferred Rigs in the name of the Purchaser, including applicable forms (CG-1258, and Vanuatu forms A-1, A-3, A-10 and A-21), Power of Attorney or Officer's certificate, good standing certificate, certificates of insurance, IMO company number (to the extent required), advice of radio accounting authority of coverage of the Transferred Rigs, ISM & ISPS certificates (to the extent required), and payment of re-registration fees.
- (c) Officer's Certificates. A certificate of each Purchaser duly executed by an authorized officer thereof certifying as to the fulfillment of each condition specified in Sections 10.2(a) and 10.2(b) and dated as of the Closing Date;
- (d) Corporate Authorizations. Copies of the resolutions of each Purchaser, certified by the Secretary or Assistant Secretary thereof as being correct and complete and then in full force and effect, authorizing the execution of this Agreement and the Transaction Documents to which it is a party and the consummation of the Contemplated Transactions and the Transaction Documents to which it is a party (in each case to the extent required by such Purchaser's Organizational Documents);
- (e) Certificates. With respect to each Purchaser, Certificates of Existence and Good Standing (or equivalent) issued by the jurisdiction of incorporation, formation, or organization of such Purchaser and dated within ten (10) days of the Closing Date;
- (f) Business Capabilities Certificate. A certificate substantially in the form of Exhibit 3.3(g), duly executed by an authorized officer of Hercules Parent and dated as of the Closing Date; and

(g) Miscellaneous. Any and all other documents, instruments, or agreements contemplated by this Agreement or as are necessary or appropriate or reasonably requested by Sellers to fully consummate the Contemplated Transactions, in each case in form and substance reasonably satisfactory to Sellers, duly executed, and dated as of the Closing Date.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF SELLERS

As a material inducement to Purchasers to enter into this Agreement and consummate the Contemplated Transactions, Sellers, jointly and severally, represent and warrant to Purchasers that the statements contained in this <u>Article IV</u> are correct and complete as of the date of this Agreement, and will be correct and complete as of the Closing (unless any such representation or warranty speaks to an earlier date, in which case the statements contained in such representation or warranty will be correct and complete as of such date) as though made then and as though the Closing were substituted for the date of this Agreement throughout this <u>Article IV</u>; *provided*, that each of the representations below is subject to and qualified by the filing of the Petitions, the entry of the Transfer Order, and applicable Bankruptcy Law:

4.1 Organization.

- (a) Seahawk Parent is a corporation, duly organized, validly existing, and in good standing under the Laws of the State of Delaware.
- (b) Each of the Seahawk LLCs is a limited liability company duly organized, validly existing, and in good standing under the Laws of the State of Delaware.
- (c) Each Seller is duly qualified or licensed to conduct its business as a foreign entity and is in good standing under the Laws of each jurisdiction where such qualification or license is required, except where the failure to be so qualified as would not, individually or in the aggregate, have a Material Adverse Effect.

4.2 Power and Authority.

- (a) Each Seller has the requisite corporate or limited liability company, as applicable, power and authority necessary to own, lease, or operate its properties and assets and carry on its business as presently conducted. Sellers have the corporate or equivalent authorization necessary to file their respective Bankruptcy Cases. Upon the filing of the Petitions, each Seller intends to operate the Business as a debtor-in-possession subject to the authority of the Bankruptcy Court.
- (b) Each Seller has the requisite corporate or limited liability company, as applicable, power and authority to execute and deliver this Agreement, the Transaction Documents and the other documents contemplated hereby to which such Seller is a party, to perform its obligations hereunder and thereunder, and to consummate the Contemplated Transactions.
- 4.3 Authorization. Each Seller has taken all corporate or limited liability company, as applicable, actions necessary to authorize the execution and delivery of this Agreement, the

Transaction Documents and the other documents contemplated hereby to which such Seller is a party, the performance of such Seller's obligations hereunder and thereunder, and the consummation of the Contemplated Transactions. Subject to the entry of the Transfer Order, this Agreement, the Transaction Documents, and the other documents contemplated hereby to which each Seller is a party have been duly authorized, approved, executed, and delivered by such Seller. Subject to the entry of the Transfer Order, this Agreement constitutes and, as of the Closing, the Transaction Documents and the other documents required to be executed and delivered by each Seller at the Closing will each constitute, a valid and legally binding obligation of such Seller and, assuming the due authorization, execution, and delivery thereof by the other parties hereto and thereto, enforceable against such Seller in accordance with its terms and conditions.

- Noncontravention. Except as set forth on Schedule 4.4, neither the execution 4.4 and delivery by any Seller of this Agreement, the Transaction Documents or any other documents contemplated hereby to which any of Sellers is a party, nor the performance by Sellers of their obligations hereunder or thereunder, nor the consummation by Sellers of the Contemplated Transactions, will (a) violate any provision of the Organizational Documents of any Seller, (b) violate any Permit, Law, Order, or other restriction of any Governmental Authority to which any Seller or any of the Purchased Assets is subject or bound, (c) conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify, or cancel or require any notice under, or result in the creation of any Lien upon any of the Purchased Assets under any Assigned Contract to which any Seller is a party or by which any Seller or any of the Purchased Assets is subject or bound, or (d) subject to the entry of the Transfer Order and compliance with the HSR Act and any requirements under the securities Laws or the rules of any securities exchange, require any Seller to give any notice to, make any filing with, or obtain any Consent of any Governmental Authority or other third party, in each case in clauses (b), (c), and (d) except as would not, individually or in the aggregate, have a Material Adverse Effect.
- 4.5 Assets. Sellers own good and marketable title, free and clear of all Liens other than Permitted Encumbrances, to all of the properties and assets, including Purchased Assets, used in the conduct of the Business, including the Transferred Rigs, except for leased properties, leased assets, and third party property contemplated in Section 2.2(g). Immediately after the Closing, Purchasers shall have good and marketable title to all of the Purchased Assets, free and clear of all Liens to the fullest extent permitted by Bankruptcy Law. Except as set forth in Schedule 4.5, the Purchased Assets have been maintained in accordance with past practice, are in sufficient operating condition and repair (subject to normal wear and tear) suitable (i) with respect to the Rigs, for the performance of the Contracts under which they are currently or most recently leased, and (ii) with respect to the other Purchased Assets, for the purposes for which they are presently used.

4.6 Financial Statements.

(a) Sellers have delivered to Purchasers, and attached hereto as <u>Schedule 4.6</u> is, the consolidated unaudited balance sheet of Sellers as of December 31, 2010 (the "*Latest Balance Sheet*") and the related income statement for Sellers for the fiscal period ended December 31, 2010 (collectively, the "*Financial Statements*"). The Financial Statements

(including the notes thereto, if any) have been prepared in accordance with GAAP throughout the periods covered thereby, present fairly in all material respects the financial condition of Sellers as of such date and the results of operations of Sellers for such period and are consistent with the books and records of Sellers, except for non-cash charges that may be made in accordance with GAAP.

- As of the date of this Agreement, (i) Seahawk Parent has timely made all SEC filings required to be made, (ii) all SEC Filings filed by Seahawk Parent, at the time filed, complied as to form in all material respects with the applicable requirements of the Securities Act and the Exchange Act and the Sarbanes-Oxley Act of 2002, as applicable, and the rules and regulations of the SEC thereunder, (iii) no such SEC Report, at the time described above, contained any untrue statement of a material fact or omitted to state any material fact required to be stated therein in order to make the statements contained therein, in the light of the circumstances under which they were made, not misleading, and (iv) all financial statements contained or incorporated by reference in such SEC Reports complied as to form when filed in all material respects with the rules and regulations of the SEC with respect thereto, were prepared in accordance with GAAP applied on a consistent basis throughout the periods involved (except as may be indicated in the notes thereto) and fairly present in all material respects the financial condition of Seahawk Parent and its consolidated subsidiaries at and as of the respective dates thereof and the consolidated results of operations and changes in cash flows for the periods indicated (subject, in the case of unaudited financial statements, to normal year-end audit adjustments consistent with prior periods). The SEC Reports included all certifications required to be included therein pursuant to Section 13a-14(a) and Section 13a-14(b) of the Exchange Act and the statements made by such certifications are true and correct.
- 4.7 Absence of Undisclosed Liabilities. Sellers do not have, and as of the Closing, will not have, any obligation or Liability (in any case, whether known or unknown, asserted or unasserted, absolute or contingent, accrued or unaccrued, liquidated or unliquidated or due or to become due), other than: (i) Liabilities reflected on the face of the Latest Balance Sheet, (ii) Liabilities incurred in the Ordinary Course of Business (none of which is a Liability resulting from, arising out of, relating to, in the nature of, or caused by any breach of contract, breach of warranty, tort, infringement, violation of Law, environmental matter, claim or lawsuit or indebtedness for borrowed money), (iii) other Liabilities expressly disclosed (both by description and amount) on Schedule 4.7 or as specifically described in the SEC Reports filed before the date of this Agreement or (iv) non-cash charges in accordance with GAAP.
- 4.8 Legal Compliance. Except as set forth on Schedule 4.8 or as specifically described in the SEC Reports of Seahawk Parent filed before the date of this Agreement, the Business is in material compliance with all Laws applicable to the Business or any of the Purchased Assets. Except as disclosed on Schedule 4.8, no Proceeding or Claim has been received by any Seller or filed, commenced or, to the Knowledge of Sellers, threatened against any Seller, in each case with respect to the Business and the Purchased Assets, alleging a violation of or liability or potential responsibility under any such Law that has not been duly cured and for which there is no remaining Liability.
- 4.9 Litigation; Proceedings. Except as set forth on Schedule 4.9 or as specifically described in the SEC Reports filed before the date of this Agreement, there are no material

Proceedings or Claims pending or, to Seller's Knowledge, threatened against or affecting Sellers and relating to the Business or the Purchased Assets, or to which the Purchased Assets may be bound or affected, at Law or in equity, or before or by any Governmental Authority; none of Sellers are subject to any Order with respect to the Business or the Purchased Assets.

4.10 Permits. Schedule 4.10 contains a complete and accurate listing and summary description of all material Permits used by Sellers in the conduct of the Business. Except as indicated on Schedule 4.10, Sellers own or possess all right, title and interest in and to all of the Permits that are necessary to own and operate the Purchased Assets and to conduct the Business as presently conducted. Sellers are and have been in compliance with the material terms and conditions of all such Permits. No Claim has been made against any Seller alleging that such Seller has breached or violated any of the terms or conditions of any Permit in such manner (a) as would permit any other Person to cancel, terminate or materially amend any Permit or (ii) is reasonably likely to result in a penalty or fee. Sellers have taken all necessary action to maintain such Permits. No loss or expiration of any Permit is pending, or to Sellers' Knowledge, threatened other than expiration by time in accordance with the terms thereof. The consummation of the Contemplated Transactions will not, except as disclosed on Schedule 4.10, require any Consent or renewal with respect to any Permit. The Assigned Permits constitute all of the Permits necessary to own and operate the Purchased Assets and to conduct the Business as presently conducted.

4.11 Rigs.

- (a) The Transferred Rigs listed in <u>Schedule 2.1(a)</u> are the only Rigs used or held for use in the Business. At the Closing, the Rigs shall be free and clear of all Liens. Except as disclosed on <u>Schedule 4.11(a)</u>, the Transferred Rigs are duly documented under the laws and flag of their respective flag state, solely in the name of one of the Sellers, and each Transferred Rig carries valid and current classification documentation and certificates, without recommendation. Except as disclosed on <u>Schedule 4.11(a)</u>, all major repairs and maintenance have been performed and completed on each operating Transferred Rig in accordance with any applicable rules or requirements of the respective classification society.
- (b) Each operating Transferred Rig is in satisfactory operating condition in all material respects for use under its applicable drilling Contract. Each Transferred Rig that is cold or warm stacked or otherwise not operating is safely afloat, or securely and safely jacked up. Prior to the date of this Agreement, Sellers have delivered to Purchasers (i) to the best of Sellers' Knowledge, a complete list of Equipment identified by site, whether at a Rig or Sellers' yard, at which such Equipment is located as of the date hereof, and (ii) copies of the maintenance records for each of the Rigs, and such records are accurate and complete in all material respects.
- 4.12 Employees. Except for the list of Claims set forth on Schedule 4.12, Sellers have complied in all material respects with all Laws relating to the employment of labor, including provisions thereof relating to wages, hours, equal opportunity, collective bargaining and the payment of social security and other taxes, and it does not have any material labor relations problems (including any union organization activities, threatened or actual strikes or work stoppages or material grievances). There are no unfair labor practice charges or complaints pending or, to the Knowledge of Sellers, threatened against any Seller or the Business that would

have a Material Adverse Effect. No Seller is a party to, bound by, or negotiating any collective bargaining agreement or other labor union contract; and there are no current or, to the Knowledge of Sellers, threatened organizational campaigns, petitions or other unionization activities.

- Employee Benefit Plans. No Employee Benefit Plan is or ever was subject to 4.13 Title IV of ERISA or to the funding requirements of Section 412 of the Code or Section 302 of ERISA. Sellers have no obligation to contribute to or any Liability or potential Liability (including actual or potential withdrawal Liability) with respect to any "multiemployer pension plan" (as defined in Section 3(37) of ERISA) or with respect to any employee benefit plan of the type described in Section 4063 or 4064 of ERISA or in Section 413(c) of the Code. With respect to each of the Employee Benefit Plans, all required or discretionary (in accordance with historical practices) payments, premiums, contributions, reimbursements and accruals for all periods ending prior to or as of the Closing Date shall have been made or properly accrued for on the books and records of the Company (and in such case will be subsequently made) and there is no unfunded Liability which is not reflected on the face of the Latest Balance Sheet (rather than the notes thereto). Except as set forth on Schedule 4.13, no Employee Benefit Plan provides health, medical, accident, life insurance or other "welfare-type" benefits with respect to current or former Seller Employees beyond their retirement or other termination of service other than coverage mandated by Section 4980B of the Code or other applicable state continuation coverage Law.
- 4.14 Environmental Matters. Except as set forth on Schedule 4.14, Sellers have not handled or disposed of any substance, arranged for the disposal of any substance, exposed any employee or other individual to any substance or condition, or owned or operated the Business or any property or facility (and no such property or facility is contaminated with such substance) so as to give rise to any Liability or corrective or remedial obligation under any Environmental, Health and Safety Laws as would have a Material Adverse Effect. Except as set forth on Schedule 4.14, Sellers and their predecessors are and have been in compliance with and have complied with all Environmental, Health and Safety Laws, and no Proceeding or Claim has been filed or commenced against Sellers alleging any failure to so comply, in each case as would have a Material Adverse Effect. Except as set forth on Schedule 4.14, Sellers have not, either expressly or by operation of law, assumed or undertaken any Liability of any other Person under any Environmental, Health and Safety Laws.
- 4.15 Tax Matters. Except as set forth on Schedule 4.15, (a) Sellers have timely filed (or joined in the filing of) all Tax Returns required by applicable Law to be filed by Sellers (taking into account any extensions of time within which to file); (b) all such Tax Returns were true, correct and complete in all respects; all Taxes owing by Sellers (whether or not shown on any Tax Return) have been paid; (c) any Liability of Sellers for Taxes not yet due and payable, or that is being contested in good faith in appropriate proceedings, has been adequately reserved for on the financial statements of Sellers, as the case may be, in accordance with GAAP; (d) there is no Proceeding or Claim concerning any Seller Taxes either claimed or raised by any Tax authority in writing; (e) no written Claim has been made by any Tax authority in a jurisdiction where Sellers do not currently file a Tax Return that it is or may be subject to any Tax by such jurisdiction, nor has any such assertion been threatened or proposed in writing; (f) Sellers do not have any outstanding request for any extension of time within which to pay Taxes or file any Tax

- Returns; (g) there are no outstanding waivers or extensions of any applicable statute of limitations for the assessment or collection of any Taxes; (h) none of Sellers is a "foreign person" within the meaning of Section 1445 of the Code; (i) no Seller is a party to, or is bound by, any Tax allocation, Tax indemnity, Tax sharing, or similar agreement or arrangement that imposes or could impose liability on Sellers for the Taxes of another Person (other than any other Seller); (j) Sellers do not have any material liability for the Taxes of another Person (other than any other Seller), consolidated group or combined group under Treasury Regulation Section 1.1502-6 or any similar provision of applicable Law, including as a transferee or successor, (k) Sellers have withheld and paid all material Taxes required to be withheld by Sellers in connection with any amounts paid or owing to any employee, creditor, independent contractor or other third party (other than any other Seller); and (l) no material liens for Taxes exist with respect to Sellers' assets, except for statutory liens for Taxes not yet due and payable or that are being contested in good faith and reserved for in accordance with GAAP.
- **4.16 Absence of Certain Developments**. Since December 31, 2010, except as set forth on Schedule 4.16 or as described in Seahawk Parent's SEC Filings:
- (a) Sellers have not sold, leased, transferred or assigned any of their assets, tangible or intangible, other than in the Ordinary Course of Business;
- (b) Sellers have not entered into any Contract outside the Ordinary Course of Business, except for Contracts entered into in connection with Sellers' strategic sale or restructuring process (excluding contingent sales agreements);
- (c) no party (including any Seller) has accelerated, terminated, modified or canceled any material Contract to which any Seller is a party or by which any Seller is bound and, to Sellers' the Knowledge, no party intends to take any such action;
- (d) Sellers have not suffered or imposed any Lien (other than any Permitted Encumbrances) upon any of its assets, tangible or intangible;
- (e) Sellers have not canceled, compromised, waived, or released any right or Claim outside the Ordinary Course of Business;
- (f) Sellers have not experienced any material damage, destruction, or loss (whether or not covered by insurance) to their properties or the Purchased Assets;
- (g) to Sellers' Knowledge, there has not been any other occurrence, event, incident, action, failure to act or transaction outside the Ordinary Course of Business involving the Business or the Purchased Assets; and
 - (h) no Seller has committed to do any of the foregoing.

4.17 Contracts.

Sellers have delivered to Purchasers a correct and complete copy of each Assigned Contract. With respect to each such Assigned Contract: (i) the Assigned Contract is legal, valid binding, enforceable, and in full force and effect; (ii) the Assigned Contract will continue to be

legal, valid, binding, enforceable, and in full force and effect on identical terms following consummation of the Contemplated Transactions; (iii) no Seller is, and to Sellers' Knowledge, no other party to the Assigned Contract is, in breach or default, and no event has occurred that with notice or the lapse of time would constitute a breach or default, or permit termination, modification, or acceleration under the Assigned Contract; and (iv) no party has repudiated any provision of the Assigned Contract.

- 4.18 Customers and Suppliers. Since September 30, 2010, no material supplier of Sellers has indicated that it shall stop, or materially decrease the rate of, supplying, products to Sellers for the Business and no material customer has indicated that it shall stop or decrease the rate of, purchasing services of the Business. With respect to each material customer, since September 30, 2010, there has been no material change, and no such customer has requested or indicated that it may request a material change, in the terms or prices at which such customer purchases, services from the Business.
- 4.19 Accounts Receivable. All accounts receivable of Sellers are reflected properly on their books and records, are valid receivables subject to no setoffs or counterclaims, and are current and collectible, subject only to the reserve for bad debts set forth on the face of the Latest Balance Sheet (rather than in any notes thereto) as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of Sellers.
- 4.20 Cash Management. Since the date of the Latest Balance Sheet, Sellers have not:
 (a) accelerated the collection of or discounted any billed or unbilled accounts receivable or deferred revenue or otherwise accelerated cash collections of any type (including by factoring), or (b) offered any special sales or incentive programs outside the Ordinary Course of Business.
- **4.21 DIP Financing**. As of the date of this Agreement, Sellers have obtained and executed a commitment letter and term sheet from a group of lenders for DIP Financing in an original principal amount of up to \$35.0 million, and have provided Purchasers with true and correct copies of the DIP Credit Documents and the related Initial DIP Budget in form approved by such DIP lenders.
- 4.22 Broker Fees. Sellers are solely liable for any and all fees, commissions, or other compensations to any broker, finder, or agent retained by any Seller with respect to the Contemplated Transactions, including to Simmons & Company and Alvarez & Marsal.
- 4.23 No Preferential Purchase Rights There are no preferential purchase rights, options or other similar rights in any Person, not a party to this Agreement, to purchase or acquire any interest in the Purchased Assets, in whole or in part.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF PURCHASERS

As a material inducement to Sellers to enter into this Agreement and to consummate the Contemplated Transactions, Purchasers represent and warrant to Sellers that the statements contained in this Article V are correct and complete as of the date of this Agreement, and will be correct and complete as of the Closing Date (unless any such representation or warranty speaks to an earlier date in which case, the statements contained in such representation or warranty will

be correct and complete as of such date) as though made then and as though the Closing Date were substituted for the date of this Agreement throughout this Article V:

5.1 Organization.

- (a) Hercules Parent is a corporation duly incorporated, validly existing, and in good standing under the Laws of the State of Delaware.
- (b) Newco is a limited liability company duly organized, validly existing, and in good standing under the Laws of the State of Delaware, and is a wholly-owned subsidiary of Hercules Parent.
- (c) Each Purchaser is duly qualified or licensed to conduct its business as a foreign entity and is in good standing under the Laws of each jurisdiction where such qualification or license is required, except where the failure to be so qualified would not individually or in the aggregate have a material adverse effect on the ability of Purchasers to perform their obligations under this Agreement or the Transaction Documents or to consummate the Contemplated Transactions.

5.2 Power and Authority.

- (a) Each Purchaser has the requisite corporate or limited liability company, as applicable, power and authority necessary to own, lease, or operate its properties and assets and carry on its business as presently conducted.
- (b) Each Purchaser has the requisite corporate or limited liability company, as applicable, power and authority to execute and deliver this Agreement, the Transaction Documents and the other documents contemplated hereby to which such Purchaser is a party, to perform its obligations hereunder and thereunder, and to consummate the Contemplated Transactions.
- Authorization. Each Purchaser has taken all corporate or limited liability 5.3 company, as applicable, actions necessary to authorize the execution and delivery of this Agreement, the Transaction Documents and the other documents contemplated hereby to which such Purchaser is a party, the performance of such Purchaser's obligations hereunder and thereunder, and the consummation of the Contemplated Transactions. This Agreement, the Transaction Documents and the other documents contemplated hereby to which such Purchaser is a party have been duly authorized, approved, executed, and delivered by such Purchaser. Subject to the entry of a Transfer Order, this Agreement constitutes and, as of the Closing, the Transaction Documents and other documents required to be executed and delivered by such Purchaser at the Closing will each constitute, a valid and legally binding obligation of such Purchaser and, assuming the due authorization, execution, and delivery thereof by the other parties hereto and thereto, enforceable against such Purchaser in accordance with its terms and conditions, except as enforceability thereof may be limited by bankruptcy, insolvency, reorganization, or other Laws affecting the enforcement of creditors' rights generally and the availability of equitable remedies.

- Noncontravention. Neither the execution and delivery by Purchasers of this 5.4 Agreement, the Transaction Documents or any other documents contemplated hereby to which Purchasers are a party, nor the performance by Purchasers of their obligations hereunder or thereunder, nor the consummation by Purchasers of the Contemplated Transactions, will (a) violate any provision of the Organizational Documents of any Purchaser or any Permit, Law, Order, or other restriction of any Governmental Authority to which any Purchaser or its assets are subject or bound, (b) conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify, or cancel, or require any notice under any material Contract or material Lien to which any Purchaser is a party or by which any Purchaser or its assets is subject or bound, (c) provide any party other than Sellers with the right to exercise any right of first refusal to purchase or other right to purchase the Hercules Shares, or (d) subject to the entry of a Transfer Order and except for (i) applicable notices, filings and Consents as may be required under the HSR Act, and (ii) any filings with the SEC or Nasdaq required to be made, require any Purchaser to give any notice to, make any filing with, or obtain any Consent of any Governmental Authority or other third party, in each case in all of the clauses above except as would not, individually or in the aggregate, materially adversely affect the ability of such Purchaser to consummate the Contemplated Transactions or perform its obligations under this Agreement.
- 5.5 Consents. Subject to the entry of a Transfer Order and compliance with the HSR Act, no Consent of any Governmental Authority is required by Purchasers in connection with the execution, delivery, and performance of this Agreement by Purchasers and the consummation of the Contemplated Transactions by Purchasers.
- 5.6 Financing. Purchasers will have at the Closing sufficient funds to timely and fully pay the Cash Payment (in accordance with Section 2.6) and consummate the Contemplated Transactions in accordance with the terms hereof, and there is no financing contingency of any nature with respect to Purchasers' obligations to pay the Cash Payment or otherwise effect the Closing of the Contemplated Transactions, subject to the Consent of the requisite lenders required under Purchasers' Credit Facility.
- 5.7 Litigation. As of the date of this Agreement, there is no Claim, Proceeding or Order pending or, to the Knowledge of Purchasers, threatened against Purchasers or their Subsidiaries, or to which Purchasers or their Subsidiaries are otherwise a party relating to this Agreement or the Contemplated Transactions that would have a material adverse effect on the ability of Purchasers to perform their obligations under this Agreement or the Transaction Documents or to consummate the Contemplated Transactions.
- 5.8 Hercules Shares. Hercules Parent has taken all corporate action necessary to authorize the issuance and delivery of the Hercules Shares. The Hercules Shares when issued in accordance with the provisions of this Agreement will be validly issued, fully paid, and nonassessable, free and clear of all Liens (except for restrictions on transfer imposed by applicable federal or state securities Laws). None of the Hercules Shares issued pursuant to this Agreement will, upon issuance, be subject to any preemptive rights, rights of first refusal, or other rights to purchase the Hercules Shares (whether in favor of Purchasers or any other Person), except as provided under Hercules Parent's charter.

- 5.9 Hercules Capitalization. The capital stock of Hercules Parent consists solely of 200,000,000 shares of common stock, par value \$0.01 per share (the "Hercules Common Stock"), of which 114,784,465 shares are issued and outstanding as of January 31, 2011, and 50,000,000 shares of preferred stock, par value \$0.01 per share, of which zero shares are issued and outstanding as of the date of this Agreement.
- 5.10 Financial Statements. The financial statements of Hercules Parent and its Subsidiaries included in each SEC Filing (collectively, the "Hercules Financial Statements"): (i) have been prepared in accordance with GAAP applied on a consistent basis throughout the periods covered thereby (except as may be otherwise indicated in such financial statements or the notes thereto or, in the case of unaudited interim statements, to the extent they may exclude footnotes or may be condensed or summary statements), (ii) fairly present in all material respects the consolidated financial position and results of operations of Hercules Parent and its Subsidiaries as of the dates and for the periods indicated (subject, in the case of unaudited statements, to normal year-end audit adjustments), and (iii) are consistent with the books and records of Hercules Parent and its Subsidiaries.
- 5.11 SEC Filings and Exchange Listing Compliance. Since January 1, 2009, each SEC Filing of Hercules Parent, when filed by Hercules Parent with the SEC, complied in all material respects with the requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act of 2002, as applicable, and the rules and regulations of the SEC thereunder applicable to SEC Filings. None of such SEC Filings of Hercules Parent (including any Hercules Financial Statements or schedules included or incorporated by reference therein) contained, as of the respective dates thereof, any untrue statement of a material fact or omitted to state any material fact required to be stated therein or necessary to make the statements therein not misleading in light of the circumstances under which they were made. Hercules Parent filed in a timely manner all documents that Hercules Parent was required to file under the Exchange Act during the twelve (12) months preceding the date of this Agreement. As of the date hereof, Hercules Parent is in compliance in all material respects with the listing requirements of NASDAQ and Hercules Parent has not received written notice from NASDAQ regarding any failure to so comply.
- **5.12** No Material Adverse Change. Except as set forth on <u>Schedule 5.12</u> or as identified and described in the SEC Filings, since September 30, 2010, there has not been:
- (a) any declaration or payment of any dividend, or any authorization or payment of any distribution, on any shares of capital stock of Hercules Parent, or any redemption or repurchase of any securities of Hercules Parent;
 - (b) any amendment to the Organizational Documents of Hercules Parent; or
- (c) any event outside the Ordinary Course of Business that has had or would reasonably be expected to have a Material Adverse Effect on Purchasers.
- 5.13 Broker Fees. Purchasers are solely liable for any and all Liability to pay any fees, commissions, or other compensations to any broker, finder, or agent retained by any Purchaser with respect to the Contemplated Transactions.

5.14 "AS IS, WHERE IS." It is understood that Purchasers take the Purchased Assets "as is, where is." Except as specifically set forth in this Agreement or the Transaction Documents, each of the Purchasers explicitly acknowledges that Sellers make no representations or warranties, express or implied, with respect to the Purchased Assets or the Business. Each Purchaser acknowledges that it has conducted its own due diligence and has made such investigations as it has deemed appropriate and such other inquiries as it has deemed necessary or desirable to satisfy itself as to the condition, operations, and prospects of the Purchased Assets and the Business.

ARTICLE VI PRE-CLOSING COVENANTS

The Parties agree as follows with respect to the Interim Period:

- 6.1 General. Each Party agrees to use all commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper, or advisable under applicable Law or otherwise to consummate, make effective, and comply with all of the terms of this Agreement and the Contemplated Transactions, including (a) providing all information and making all filings necessary in connection herewith and therewith, and (b) satisfying, but not waiving, the conditions precedent set forth in Article X.
- Notices and Consents. As promptly as practicable following the date hereof, each Party will give any notices to, make any filings with, and use all commercially reasonable efforts to obtain the Consents of third parties and Governmental Authorities required to consummate, make effective, and comply with all of the terms of this Agreement and the Contemplated Transactions, and will use all commercially reasonable efforts to agree jointly on a method to overcome any objections by any third party or Governmental Authority to this Agreement or the Contemplated Transactions. Subject to applicable Law, the Parties shall cooperate with each other in exchanging information and assistance in connection with obtaining any Consents of third parties and Governmental Authorities and shall promptly provide to the other Parties or their representatives copies of all filings made with any third party or Governmental Authority with respect to this Agreement or the Contemplated Transactions. Without limiting the foregoing, the Parties shall cooperate fully with each other Party to cause to be filed as promptly as possible, but in no event later than ten (10) days following the date hereof, with the Federal Trade Commission and the Department of Justice the notification and report forms required under the HSR Act in connection with the Contemplated Transactions, to seek early termination (applicable to a bankruptcy situation) of the applicable waiting period thereunder, and to promptly provide supplemental information that may be requested in connection with such filings or make any further filings related thereto. Any and all fees incurred and owing to the applicable Governmental Authority or Governmental Authorities in connection with filings hereunder pursuant to the HSR Act shall be borne by Purchasers.
- 6.3 Bankruptcy. Within three days of the date hereof, Sellers will file the Petitions with the Bankruptcy Court. On the Petition Date, Sellers shall file with the Bankruptcy Court a Sale Motion, substantially in the form attached hereto as Exhibit 6.3(a), a Termination Fee Motion, substantially in the form attached hereto as Exhibit 6.3(b), and a motion to approve DIP Financing (with the Initial DIP Budget), each reasonably acceptable to Purchasers and consistent

in all material respects with the terms of this Agreement, and shall seek entry of the Transfer Order by the Bankruptcy Court, in the form attached hereto as Exhibit 6.3(c), approving this Agreement and the Contemplated Transactions, and a Termination Fee Order substantially in the form attached hereto as Exhibit 6.3(d), approving the payment of the Termination Fee and Purchaser's Reimbursable Expenses. Sellers shall use commercially reasonable efforts to obtain entry of any Orders of the Bankruptcy Court necessary to approve and implement this Agreement and the Contemplated Transactions, provided, that Sellers' efforts shall comply with any procedures approved by the Bankruptcy Court or imposed by the Bankruptcy Code. Sellers shall use their best efforts to provide Purchasers with copies of all material proposed pleadings, motions, orders and notices prepared by or on behalf of the Sellers relating to the Purchased Assets or this Agreement prior to the filing thereof in the Bankruptcy Court and to allow Purchasers to provide reasonable comments for incorporation into same if time permits.

- Preservation of Purchased Assets and Business. Purchasers acknowledge that, upon the filing of the Petitions, Sellers will be conducting their Business as debtors-in-possession pursuant to the Bankruptcy Cases, pursuant to Sections 1107 and 1108 of the Bankruptcy Code and accordingly, there may occur changes to the Business that are not in the Ordinary Course of Business of Sellers. Notwithstanding the foregoing, except as otherwise provided by Orders of the Bankruptcy Court and Bankruptcy Law, Sellers shall use commercially reasonable efforts to (a) preserve Sellers' current Business and the Purchased Assets substantially intact, including Sellers' present operations, customer and supplier relationships, physical facilities, and working conditions, (b) maintain the Purchased Assets in good working order and condition, normal wear excepted, and continue all required or desirable repairs and maintenance of the Purchased Assets, and (c) timely obtain, maintain in full force and effect, and comply with all provisions of all Permits required in connection with the operation of the Purchased Assets or Business. Sellers shall promptly inform Purchasers of any actions by Sellers contemplated herein requiring the approval of the Bankruptcy Court. Without limiting the generality of the foregoing, from the date hereof through the Closing Date, Sellers shall not, without the prior written consent of Purchasers, take any of the following actions (in each case except as otherwise ordered by the Bankruptcy Court):
- (a) conduct the cash management customs and practices of the Business (including the collection of receivables and payment of payables) and inventory management customs and practices (including purchasing inventory) of the Business other than in the Ordinary Course of Business;
- (b) make any commitment for capital expenditures with respect to the Business, other than (i) in the Ordinary Course of Business to service new or existing customers, (ii) pursuant to existing commitments or business plans that have previously been provided to Purchasers in writing, and (iii) to repair or maintain any assets, properties or facilities;
- (c) take any action or omit to take any action, the taking or omission of which, could reasonably be expected to have a Material Adverse Effect or to Sellers' Knowledge violate in any material respect this Agreement;
- (d) enter into any Contract that would be an Assigned Contract (i) out of the Ordinary Course of Business, or (2) restricting in any material way the conduct of the Business;

- (e) settle or compromise any litigation, Claim, administrative or regulatory proceeding relating to the Business that would increase the Assumed Liabilities;
- (f) sell, transfer or otherwise dispose of or cause a Lien to exist on the Purchased Assets, other than a Permitted Encumbrance or pursuant to the DIP Credit Documents, in the Ordinary Course of Business or pursuant to an existing obligation or arrangement; or
 - (g) authorize or enter into an agreement to do any of the foregoing.

6.5 No Shop.

- During the Interim Period, none of Sellers, or any agent or advisor to any (a) Seller, or any Affiliate of any Seller shall, directly or indirectly, through any officer, director, employee, agent, professional, advisor, other representative or otherwise, (A) solicit, initiate, knowingly encourage or knowingly facilitate any proposal or offer from any third party (other than Purchasers or any Affiliate of Purchasers) relating to any acquisition, divestiture, business combination, or reorganization or similar transaction involving any portion of the Business, the Purchased Assets or the Equity Interests of Sellers, whether structured as a financing or refinancing or otherwise (a "Competing Transaction"), (B) enter into discussions or negotiations regarding a Competing Transaction, (C) furnish any information with respect to, enter into any agreement or understanding with respect to, otherwise assist or participate in, or facilitate in any other manner any Competing Transaction (including executing any confidentiality agreement with any other third party with respect to a Competing Transaction), (D) waive any rights under any existing standstill or waiver agreements, or (E) seek or support Bankruptcy Court approval of a motion regarding bid procedures or expense reimbursement or break up or termination fee for the benefit of any other party with respect to a Competing Transaction or take any action inconsistent in any way with this Agreement (any actions described in clauses (A) through (E) are "Prohibited Transactions"). Notwithstanding anything in the foregoing to the contrary, prior to the entry of the Transfer Order, the Board of Directors of Seahawk Parent, with the assistance of outside advisors, may review and consider proposals and offers for a Competing Transaction provided such proposals and offers were not submitted in violation of clause (A) above.
- (b) Notwithstanding the foregoing provisions, in the event Sellers receive an unsolicited bona fide offer or proposal for a Competing Transaction prior to entry of the Transfer Order and Seahawk Parent's Board of Directors concludes in good faith (after consultation with its outside financial and legal advisors) that such offer or proposal constitutes or is reasonably likely to result in a Superior Proposal, then, prior to the entry of the Transfer Order, Sellers may, and may permit their subsidiaries and representatives to, take any Prohibited Transaction described in subclauses (B) or (C) of clause (a) above (other than enter into any agreement); provided that (x) prior to providing any nonpublic information permitted to be provided pursuant to this sentence, Sellers shall have entered into a confidentiality agreement with such third party on customary terms and which in any event is no less favorable to Sellers than the confidentiality agreement entered into with Hercules Parent in connection with the Contemplated Transactions, and (y) concurrently with providing such information, Sellers shall also furnish to Purchasers a copy of any confidential data or information being furnished to any third party pursuant to this Section to the extent not previously furnished to Purchasers.

- (c) Sellers shall promptly (within 24 hours) advise Purchasers orally and in writing following receipt of (1) any offer or proposal for a Competing Transaction or indication by any Person that it is considering making an offer or proposal relating to a Competing Transaction, (2) any request for nonpublic information relating to Sellers or access to the properties, books or records of Sellers, other than requests in the Ordinary Course of Business and unrelated to an offer or proposal relating to a Competing Transaction, or (3) any inquiry or request for discussions or negotiations regarding an offer or proposal relating to a Competing Transaction. Sellers shall promptly (within 24 hours) provide Purchasers with a copy (if in writing) and summary of the related material terms of any such offer or proposal or request (including the identity of the Person making or considering such offer or proposal or making such request and shall keep Purchasers apprised of any material developments and discussions on a reasonably current basis (and in any event within 24 hours).
- (d) Notwithstanding anything herein to the contrary, prior to entering into an agreement in connection with any Competing Transaction, (A) Sellers shall provide prior written notice to Purchasers, at least 48 hours in advance (the "Notice Period"), of its intention to take such action with respect to such Competing Transaction, specifying the material terms and conditions of any such Competing Transaction (including the identity of the party proposing to effect such Competing Transaction) and furnishing to Purchasers a copy of the relevant proposed transaction agreements with the party proposing to effect such Competing Transaction and other material documents and (B) during the Notice Period, and in any event prior to taking such action, Sellers shall negotiate, and shall cause its financial and legal advisors to negotiate, with Purchasers in good faith (to the extent Purchasers desire to negotiate) to make such adjustments in the terms and conditions of this Agreement so that such proposal or offer for a Competing Transaction ceases to constitute a Superior Proposal.
- (e) At any time prior to the Transfer Order, Sellers may terminate this Agreement and concurrently with such termination, upon payment to Purchasers of the Termination Fee and Purchasers' Reimbursable Expenses pursuant to Section 11.2, enter into a definitive agreement with respect to a Superior Proposal if the Board of Directors of Seahawk Parent determines in good faith, after consultation with Seahawk Parent's outside legal counsel, that in light of such Superior Proposal such action is required in order for Seahawk Parent's Board of Directors to comply with its fiduciary obligations under applicable Law, provided that Sellers have otherwise strictly complied with all of their obligations in this Section 6.5. Such agreement to pay the Termination Fee and Purchaser's Reimbursable Expenses is, in part, based on Sellers' recognizing Purchasers' expenditure of time, energy, and resources in connection with this Agreement. The payment of the Termination Fee and Expense Reimbursement shall be governed by the provisions of this Agreement and any related Bankruptcy Court Orders. Except as may be negotiated as part of a Superior Proposal, under no circumstances will a break-up fee, expense reimbursement or other similar bid protections be provided by Sellers to any potential bidder or bidders for any portion of the Business or the Purchased Assets, other than Purchasers.
- 6.6 DIP Matters. A copy of the Initial DIP Budget, as approved by the DIP lenders, is attached as Exhibit 4.21. Sellers shall immediately notify Purchasers of any and all notifications, whether written or oral, received thereby regarding noncompliance with the DIP Credit Agreements, DIP Order or DIP Budget, or breach or default by Sellers thereunder. Each Seller agrees that it shall be, at all times from and after the date of the DIP Order, in compliance

in all material respects with the terms of the DIP Credit Agreements to which it is a party, DIP Budget and DIP Order subject to any and all grace and cure periods provided therein. Sellers shall not amend or replace the DIP Budget, create or amend any other budget with respect to the DIP Financing, or increase the principal amount of the facility under the DIP Credit Agreement in excess of \$45.0 million (including amendments related to collateral and security matters), and in any event Sellers agree not to seek any modification or amendment to the DIP Budget with any disbursement line item constituting an amount that exceeds the disbursement line item of the prior week in the DIP Budget by 25% of such disbursement line item.

- 6.7 Other Information. Sellers shall promptly provide such information, including all financial information pertaining to the Purchased Assets and the Business, as reasonably requested by Purchaser in connection with its reporting obligations under Securities Act and the Exchange Act.
- each other the opportunity to review and comment upon any press release or other public statement prior to the issuance of such press release or other public statement relating to this Agreement or the Contemplated Transactions, shall coordinate the timing of any such press release or other public statement, and shall not issue any such press release or other public statement prior to such consultation except as such Party in its good faith judgment may be required by applicable Law or by obligations pursuant to any listing agreement with any national securities exchange. Hercules Parent and Seahawk Parent agree that they will be issuing separate press releases announcing the execution and delivery of this Agreement.
- **6.9** Lender Consent. Within 30 days of the date hereof, Purchasers shall obtain the requisite Consent of the lenders under the Purchasers' Credit Facility that is necessary in connection with the consummation of the Contemplated Transactions.
 - 6.10 Notices of Certain Events. Sellers shall promptly notify Purchaser of:
- (a) any written communication or written notice from any Person alleging that the consent of such Person is or may be required in connection with the consummation of the Transactions;
- (b) any material written communication from any Governmental Authority in connection with or relating to the Transactions; and
 - (c) any Material Adverse Effect on the Purchased Assets.
- 6.11 Employee Data. Sellers shall provide Purchasers with such employee data or other information as may be reasonably requested by Purchasers or required to carry out the arrangements described in this Agreement except as prohibited by applicable Law. At least thirty (30) days prior to Closing Date, Sellers shall provide Purchasers with a listing in hardcopy and electronic spreadsheet format of Transferred Employees with date of hire; service date for benefit plan purposes; wage or salary rate; the amount of any bonus paid in 2010, and the amount paid, accrued, or expected to be paid by Sellers in 2011; the rate at which vacation time and paid leave is accrued; and each Transferred Employee's employment status, including whether such Transferred Employee is currently or is reasonably expected to be, on any form of leave of

absence (other than vacation) or any form of disability leave on or prior to Closing Date. If, prior to Closing, any Transferred Employee's status changes to any form of leave of absence or disability leave status, Sellers shall notify Purchaser as soon as practicable.

ARTICLE VII EMPLOYEE MATTERS

- 7.1 Announcement. Sellers and Purchasers shall jointly determine a date on which to announce the Contemplated Transaction to the Seller Employees. Purchasers acknowledge that Sellers retain the right to terminate or otherwise alter the terms of employment of any or all Seller Employees prior to the Closing Date.
- 7.2 Offers of Employment. Purchasers or any Affiliate of Purchasers shall be entitled, but not obligated, to make offers of employment to such of the Seller Employees as Purchasers or such Affiliate of Purchasers shall determine and on such terms as Purchasers or such Affiliate determines (each, an "Offer of Employment"). In connection therewith, Sellers shall permit Purchasers and their representatives and Affiliates to have access, at reasonable times and upon reasonable notice, to all Seller Employees for conducting interviews and evaluations to assess whether Purchasers or any Affiliates thereof desires to extend Offers of Employment to any of the Seller Employees. In making hiring determinations pursuant to this Section 7.2, Purchasers and Affiliates thereof shall comply in all material respects with applicable Laws with respect to employment offers and hiring matters and shall conduct such interviews and evaluations in such a manner as not to unreasonably interfere with Sellers' operation of the Business.
- 7.3 Terms of Employment and Transition. In the event Purchasers or any Affiliate thereof extends an Offer of Employment to a Seller Employee, and such Seller Employee accepts employment with Purchasers or such Affiliate (each, a "Transferred Employee"), employment of such Transferred Employee by Purchasers or such Affiliate shall be effective as of the Closing Date and shall be on the terms of the Offer of Employment applicable to such Transferred Employee which shall include the continued at-will status of such Transferred Employee unless expressly stated otherwise.

7.4 Employee Benefits.

- (a) Sellers shall be Liable for the payment of all wages, compensation, and other remuneration due as of the Closing Date to each Seller Employee, and Sellers shall be Liable for all payments or contributions required to be made to any Employee Benefit Plans of Sellers.
- (b) Effective as of Closing and thereafter, Purchasers (or an Affiliate thereof) shall provide, or cause to be provided, under Purchaser's plans or otherwise, continuation health coverage with respect to Seller Employees (and their eligible dependents) as required under, and in accordance with, COBRA and in accordance with Sellers' agreements and plans listed on Schedule 4.13; provided, however, that Sellers (or an Affiliate thereof) shall, at their sole expense, be solely responsible for providing any and all notices and/or elections required by COBRA, and by Sellers' agreements, policies, plans or practices.

- Cooperation. To the extent that Purchasers or an Affiliate thereof hire any 7.5 employees of any Seller in the United States, such Seller shall cooperate with Purchasers or such Affiliate and Purchasers' or such Affiliate's agents to provide Purchasers or such Affiliate the adequate payroll tax records required by federal and state agencies necessary for Purchasers or such Affiliate to optimize federal and state payroll tax law relating to transactions such as the Contemplated Transactions. Such records shall include, but not be limited to an executed release form granting permission to Purchasers or such Affiliate and their agents to obtain quarterly payroll data from all states within which the Business was conducted by Sellers. In addition, Sellers will provide Purchasers or such Affiliate and their agents with (i) signatures (or notarized signatures) necessary to grant permission for Purchasers or such Affiliate to file for transfers of experience of payroll tax accounts in states which require a signed release by the such Seller, (ii) the most recent Annual 940 Report (including Schedule A), and most recent years "tax rate notices" received from individual state agencies, and (iii) copies of all quarterly wage detail reports filed with individual state agencies in the calendar year through the Closing Date. If any Seller used an outside payroll tax administrator, then such Seller hereby grants Purchasers or such Affiliate, as applicable, permission to have access to relevant successor-in-interest reports from the payroll vendor, such as state tax rate notices, state quarterly contribution reports, W2s and federal recap reports such as 940 and 941, and will provide Purchasers or such Affiliate with a contact person at such payroll vendor.
- 7.6 Miscellaneous. Notwithstanding any provision to the contrary in this Agreement, nothing expressed or implied in this Agreement shall confer upon any Seller Employee (including any Transferred Employee) any rights or remedies of any nature or kind whatsoever, including any right to employment or continued employment for any specified period by Purchasers or any Affiliate thereof, or restrict in any way the right of Purchasers or any such Affiliate to terminate employment of any Transferred Employee.

ARTICLE VIII TAX MATTERS

- 8.1 Sales and Transfer Taxes. The Parties contemplate that the Contemplated Transactions will qualify as an exempt isolated or occasional sale pursuant to La. Rev. Stat. § 47:301 and Tex. Tax Code §151.304 respectively. Pursuant to La. Rev. Stat. §47:301, Sellers represent they have not regularly engaged in the business of, or held themselves out to be in the business of, selling tangible personal property similar in nature to the Purchased Assets. Subject to the terms of this Agreement, the requirements under 34 TAC §3.316 that Sellers shall convey, transfer and deliver, and Purchasers shall purchase and acquire, all of Sellers "operating assets" associated with the Business, as that term is defined under subsection (d)(3) of 34 TAC §3.316, have been met. Upon the Closing, Sellers agree to complete and sign a Texas Comptroller of Public Accounts Form 01-917 "Statement of Occasional Sale" indicating Sellers' acknowledgement that the Contemplated Transactions should qualify as an exempt occasional sale pursuant to 34 TAC §3.316 and Sellers agree to provide a copy thereof to Purchasers.
- 8.2 Purchase Consideration Allocation. Sellers and Purchasers shall negotiate in good faith prior to the Closing to agree upon an allocation of the Aggregate Consideration among the Purchased Assets (the "Purchase Consideration Allocation"). In the event the Parties are unable to finalize the Purchase Consideration Allocation prior to the Closing then the

Parties shall attempt to finalize the Purchase Consideration Allocation within sixty (60) days after the Closing Date, provided, however, the Parties shall not be obligated to reach an agreement. If an agreement is reached, the Parties shall treat and report (and, if necessary, to cause each of their respective Affiliates to so treat and report) the sale and purchase of the Purchased Assets for all federal, state and local Tax purposes in a manner consistent with the agreed Purchase Consideration Allocation and shall not take any position on their respective Tax Returns that is inconsistent with such Purchase Consideration Allocation. Without limiting the generality of the preceding sentence, the Purchase Consideration Allocation will be reflected in Form 8594 that will be filed by Sellers and Purchasers in accordance with Section 1060 of the Code and in any other filings under the Code. The Parties recognize that the Purchase Consideration Allocation shall not include Purchasers' acquisition expenses and that Purchasers will allocate such expenses appropriately.

8.3 Tax Allocation. For any ad valorem or similar property Taxes where the applicable Tax period begins before the Closing Date and ends after the Closing Date (the "Straddle Period"), such Taxes shall be allocated between the pre-Closing and post-Closing portion of the Straddle Period as described herein. The amount of such Taxes for the Straddle Period allocated to the portion of the period ending on the Closing Date shall be the total of such Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of days from the beginning of such Straddle Period to and including the Closing Date and the denominator of which is the total number of days in the entire Straddle Period. The balance of such taxes shall be allocated to the portion of the Straddle Period beginning after the Closing Date.

ARTICLE IX POST-CLOSING COVENANTS

The Parties agree as follows with respect to the period following the Closing Date:

- 9.1 Books and Records. At the Closing, or such subsequent date as the Parties may mutually agree, Sellers shall deliver, or cause to be delivered, to Purchasers the Transferred Books and Records, provided that such delivery shall not include any Retained Books and Records and shall be at Sellers' sole cost and expense.
- 9.2 Computer Software. Sellers agree to use commercially reasonable efforts to provide Purchasers with a sublicense with respect to any Computer Software (in accordance with the terms of any primary license relating thereto) reasonably requested by Purchasers as being necessary in the operation of the Purchased Assets, or in the alternative to provide reasonable access to Purchasers to such information relating to the Purchased Assets that is embodied in such Computer Software.

- 9.3 Monies Collected. To the extent any payments of trade accounts receivable, credit card receipts or other monies that are for the account of Sellers are received by any Purchaser, such Purchaser shall, within three (3) Business Days of receipt thereof, advance to Sellers the amount of any such payments together with any documentation received by such Purchaser in connection therewith. To the extent any payments of trade accounts receivable, credit card receipts, or other monies that are for the account of Purchasers are received by a Seller, such Seller shall, within three (3) Business Days of receipt thereof, advance to Purchasers the amount of any such payments together with any documentation received by such Seller in connection therewith.
- 9.4 Survivability. The representations and warranties and pre-Closing covenants and obligations of Sellers and of Purchasers in <u>Article IV</u>, <u>Article V</u> and <u>Article VI</u> of this Agreement shall not survive the Closing.
- 9.5 Other Information. Sellers shall promptly provide such information, including all financial information pertaining to the Purchased Assets and the Business, as reasonably requested by Purchasers in connection with Hercules Parent's reporting obligations under Securities Act and the Exchange Act.
- 9.6 Public Announcement. Sellers and Purchasers shall consult with and provide each other the opportunity to review and comment upon any press release or other public statement prior to the issuance of such press release or other public statement relating to this Agreement or the Contemplated Transactions, shall coordinate the timing of any such press release or other public statement, and shall not issue any such press release or other public statement prior to such consultation except as such Party in its good faith judgment may be required by applicable Law or by obligations pursuant to any listing agreement with any national securities exchange.
- 7.7 Trademarks. Purchasers agree to remove any and all logos and other Trademarks of Sellers on any of the Purchased Assets within a reasonable amount of time not to exceed 120 days from the date of Closing; provided, that (a) with respect to Transferred Rigs that are non-operating as of the Closing, Purchasers shall not be obligated to comply with this Section 9.7 until such time as actions are commenced to prepare any such Transferred Rig to become operating, at which time Purchasers shall have 90 days after such preparations are first commenced to comply with this Section 9.7, and (b) with respect to all of the Transferred Rigs, Purchasers may utilize the rig name for purposes of effecting the transfer of title of such Transferred Rig to Purchasers in accordance herewith.
- 9.8 Hercules Shares. Provided that Sellers follow in all material respects the precedents set forth in the applicable SEC no-action letters acknowledging an exemption from the registration requirements of the Securities Act involving sales under Section 363 of the Bankruptcy Code and subsequent plans of reorganization, such as the sale contemplated by the Parties in this Agreement, Purchasers shall use all commercially reasonable efforts to ensure that any Hercules Shares distributed on account of claims or interests of Sellers under the Plan of Reorganization are not restricted, as that term is defined for purposes of Rule 144 of the Securities Act of 1933, or may be resold without restriction (other than with respect to third parties that may be Affiliates or underwriters) pursuant to a registration statement. Sellers agree

not to distribute, transfer or assign, or permit to be distributed, transferred or assigned, any of (a) the Hercules Shares containing the last legend specified in Section 3.4(a) prior to the determination of the Final Closing Net Working Capital and (b) any of the Hercules Shares other than pursuant to, and in accordance with the Plan of Reorganization.

- 9.9 Releases. In consideration for payment of the Cash Payment, issuance of the Hercules Shares and assumption of the Assumed Liabilities by Purchasers, Sellers shall cause the Plan of Reorganization to contain a mutual release provision by Sellers in favor of Purchasers and by Purchasers in favor of Sellers on terms mutually acceptable to the Parties.
- 9.10 Office Equipment. For 120 days after the Closing, or for such longer period as may be agreed by Purchasers, Purchasers shall permit Sellers to continue to utilize the Purchased Assets consisting of office equipment, furniture, computers, computer equipment, office supplies, telephones, and similar equipment at the Leased Real Properties at no cost to Sellers.

ARTICLE X CONDITIONS TO CLOSING

- 10.1 Conditions Precedent to Obligation of Purchasers. The obligations of Purchasers to consummate the Contemplated Transactions and take any other action required to be taken by Purchasers at the Closing or thereafter is subject to the satisfaction, at or prior to the Closing, of each of the following conditions, any of which may be waived in writing by Purchasers in whole or in part:
- (a) Accuracy of Representations and Warranties. The representations and warranties of Sellers set forth in this Agreement shall have been and be true and correct in all respects (it being understood that, for purposes of determining the accuracy of such representations and warranties, all materiality qualifications contained in such representations and warranties shall be disregarded) as of the date hereof and as of the Closing Date, as though made on and as of the Closing Date (except to the extent representations and warranties speak as of a specified date, which representation and warranties shall have been true and correct as of such date), except for failures that would not, individually or in the aggregate, have a material adverse effect on the Purchased Assets, Business, or the ability of Sellers, in each case taken as a whole, to perform their obligations under this Agreement or consummate the Contemplated Transactions.
- (b) Compliance with Obligations. Sellers must have performed and complied with all of their respective covenants and obligations required by this Agreement to be performed or complied with at or prior to the Closing in all material respects.
- (c) Transfer Order. The Transfer Order (a) shall have been entered by the Bankruptcy Court in form and substance reasonably acceptable to Sellers and Purchasers and consistent in all material respects with the terms of this Agreement, (i) approving this Agreement and the Contemplated Transactions, and (ii) finding that notice of the hearing concerning approval of the Contemplated Transactions was given in accordance with the Bankruptcy Code and constitutes such notice as is appropriate under the particular circumstances and that Purchasers are "good faith" purchasers entitled to the protections afforded by Section 363(m) of

the Bankruptcy Code, and (iii) providing for the vesting of the Purchased Assets in the Purchasers, free and clear of all interest in such property as provided by 11 U.S.C. §363(f), and (b) shall have become a Final Order.

- (d) *DIP Matters*. Sellers shall have obtained expedited approval of and be in compliance in all material respects with the DIP Financing, DIP Order and DIP Budget.
- (e) No Legal Proceedings. There shall be no Claim, Proceeding, or Order pending against any Seller (excluding such by or at the direction of Purchasers or any Affiliates thereof) or against any Purchaser (excluding such by or at the direction of Sellers or any Affiliates thereof) by or before any Governmental Authority that would reasonably be expected to have the effect of or seek to challenge, restrain, prohibit, invalidate, interfere with, or collect Damages arising out of, the Contemplated Transactions.
- (f) HSR Waiting Period. Any applicable waiting period under the HSR Act shall have expired or been terminated.
- (g) Consents. All third party Consents set forth in Schedule 10.1(g), after giving effect to the Transfer Order, shall have been received by Sellers and delivered to Purchasers and such Consents shall be on terms reasonably acceptable to Purchasers and in full force and effect as of the Closing.
- (h) Closing Deliveries. Sellers shall have delivered, or caused to be delivered, at the Closing each item described in <u>Section 3.3</u>.
- (i) Consent of Lenders. Purchasers shall have obtained, in accordance with Section 6.9, the requisite Consent of the lenders under the Purchasers' Credit Facility that is necessary in connection with the consummation of the Contemplated Transactions.
- 10.2 Conditions Precedent to Obligations of Sellers. The obligation of Sellers to consummate the Contemplated Transactions and take any other action required to be taken by Sellers at the Closing or thereafter is subject to the satisfaction, at or prior to the Closing, of each of the following conditions, any of which may be waived by Sellers in whole or in part:
- (a) Accuracy of Representations and Warranties. The representations and warranties of Purchasers set forth in this Agreement shall have been and be true and correct in all respects (it being understood that, for purposes of determining the accuracy of such representations and warranties, all materiality qualifications contained in such representations and warranties shall be disregarded) as of the date hereof and as of the Closing Date, as though made on and as of the Closing Date (except to the extent representations and warranties that speak as of a specified date, such representation and warranties shall have been true and correct as of such date), except for failures that would not, individually or in the aggregate, have a material adverse effect on the ability of Purchasers to perform their respective obligations under this Agreement or consummate the Contemplated Transactions.
- (b) Compliance with Obligations. Purchasers must have performed and complied with all of their respective covenants and obligations required by this Agreement to be performed or complied with at or prior to the Closing in all material respects.

- Bankruptcy Court in form and substance reasonably acceptable to Sellers and Purchasers and consistent in all material respects with the terms of this Agreement, (i) approving this Agreement and the Contemplated Transactions, and (ii) finding that notice of the hearing concerning approval of the Contemplated Transactions was given in accordance with the Bankruptcy Code and constitutes such notice as is appropriate under the particular circumstances and that Purchasers are "good faith" purchasers entitled to the protections afforded by Section 363(m) of the Bankruptcy Code, and (iii) providing for the vesting of the Purchased Assets in the Purchasers, free and clear of all interest in such property as provided by 11 U.S.C. §363(f), and (b) shall have become a Final Order.
- (d) No Legal Proceedings. There shall be no Claim, Proceeding, or Order pending against any Purchaser (excluding such by or at the direction of Sellers) by or before any Governmental Authority that may have the effect of or seek to challenge, restrain, prohibit, invalidate, interfere with, or collect Damages arising out of, the Contemplated Transactions.
- (e) HSR Waiting Period. Any applicable waiting period under the HSR Act shall have expired or been terminated.
- (f) Closing Deliveries. Purchasers shall have delivered, or caused to be delivered, at the Closing each item described in <u>Section 3.4</u>.

ARTICLE XI TERMINATION

- 11.1 Termination of Agreement. The Parties may terminate this Agreement as provided below:
- (a) Mutual Consent. Purchasers and Sellers may terminate this Agreement as to all Parties by mutual written consent at any time prior to the Closing by written instrument authorized by the respective Boards of Directors of Seahawk Parent and Hercules Parent.
- (b) By Purchasers. Purchasers may terminate this Agreement by giving written notice to Sellers at any time prior to the Closing in the event that (i) Sellers have breached any representation, warranty, or covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.1(a) and 10.1(b) shall not have been satisfied, or cannot be satisfied by August 15, 2011 (the "Termination Date"); provided, that Purchasers shall have provided written notification to Sellers of such breach and the breach shall have continued without cure for a period of ten (10) days after delivery of the notice of such breach; (ii) the Transfer Order shall not have been entered by the Bankruptcy Court and become a Final Order by one hundred twenty (120) days after the date hereof; (iii) any of the Bankruptcy Cases are converted to cases pursuant to Chapter 7 of the Bankruptcy Code; (iv) a Chapter 7 or Chapter 11 Trustee has been appointed; (v) an Examiner with expanded powers has been appointed in any of the Bankruptcy Cases of Sellers; (vi) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement; (vii) the DIP Financing is not approved within twenty-one (21) days of the Petition Date; (viii) Sellers are in breach or default in any material respect of, the DIP Credit Documents, DIP Order or DIP Budget after five days notice by any

Person and opportunity to cure; or (ix) the Termination Fee Order shall not have been entered by the Bankruptcy Court within twenty-five (25) days of the date hereof or shall not have become a Final Order by the eleventh day after such entry of the Termination Fee Order. At their sole and absolute discretion, Purchasers have the right to waive termination or agree to extend any deadlines under this Section 11.1(b).

- (c) By Sellers. Sellers may terminate this Agreement by giving written notice to Purchasers at any time prior to the Closing in the event that (i) any Purchaser has breached any representation, warranty, or covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.2(a) and 10.2(b) shall not have been satisfied, or cannot be satisfied by the Termination Date; provided, that Sellers shall have provided written notification to Purchasers of such breach and the breach shall have continued without cure for a period of ten (10) days after delivery of the notice of breach; or (ii) in accordance with the terms and subject to the conditions of Section 6.5; or (iii) Purchasers' failure to timely obtain the Consent as contemplated by Section 6.9.
- (d) By Either Party. Sellers or Purchasers may terminate this Agreement by giving written notice to the other Party if (i) any court of competent jurisdiction or any other Governmental Authority in a suit instituted by a third party or a Governmental Authority shall have issued an Order or shall have taken any other action prior to the Termination Date permanently enjoining, restraining, or otherwise prohibiting the Contemplated Transactions or a material portion thereof, (ii) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement or denies approval of this Agreement or the Contemplated Transactions, or (iii) the Closing has not occurred by the Termination Date, provided, that the Party electing to terminate shall not have caused such failure to close.

11.2 Effect of Termination.

- (a) If either Party terminates this Agreement pursuant to Section 11.1, all rights and obligations of the Parties under this Agreement shall terminate; provided, that the rights and obligations of the Parties under this Section 11.2 (Effect of Termination), any provisions regarding the interpretation or enforcement of this Agreement, and Article XII (Miscellaneous) will survive any such termination.
- (b) Notwithstanding any other provision of this Agreement, in no event shall Sellers be entitled to recover more than their actual direct out-of-pocket Damages caused by any breach by Purchasers of this Agreement, and in no event more than \$2,000,000 in the aggregate, inclusive of any interest, and such aggregate dollar limitation shall be a limitation on any remedy to which Sellers may be entitled, at Law or in equity, including the right to enforce any provision of this Agreement by an Order of specific performance.
- (c) If Sellers terminate this Agreement pursuant to Section 11.1(c)(iii), Purchasers shall pay to Sellers up to an aggregate of \$1,000,000 of the Sellers' Reimbursable Expenses, which aggregate payment shall be made by wire transfer of immediately available funds by the second (2nd) Business Day following such termination. Purchasers' payment of such Sellers' Reimbursable Expenses pursuant to this Section 11.2(c) shall be the sole and exclusive remedy of Sellers with respect to the occurrence giving rise to such payment.

(d) If (i) Sellers terminate this Agreement pursuant to Section 11.1(c)(ii) or (ii) Purchasers terminate this Agreement pursuant to Section 11.1(b)(i) as a result of a breach by Sellers of Section 6.5 or Section 6.6 or (iii) Purchasers terminate this Agreement pursuant to Section 11.1(b)(ii)-(ix), then Sellers shall pay to Purchasers an aggregate amount equal to the Termination Fee and Purchasers' Reimbursable Expenses, which aggregate payment shall be treated in accordance with the Order entered approving the Termination Fee Motion, or if no such Order has been entered, then by wire transfer of immediately available funds no later than the second (2nd) Business Day following such termination of this Agreement. Sellers' payment of the Termination Fee and Purchasers' Reimbursable Expenses pursuant to this Section 11.2(d) shall be the sole and exclusive remedy of Purchasers with respect to the occurrences giving rise to such payment.

ARTICLE XII MISCELLANEOUS

- 12.1 Waiver. Any term or provision of this Agreement may be waived in writing at any time by the Party which is entitled to the benefits thereof. The failure of either Party at any time or times to require performance of any provision hereof shall in no manner affect such Party's right at a later time to enforce the same. No waiver by any Party of a condition or of the breach of any term, covenant, representation or warranty contained in this Agreement, whether by conduct or otherwise, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such condition or breach or a waiver of any other condition or of the breach of any other term, covenant, representation, or warranty of this Agreement.
- 12.2 Notices. All notices, requests, demands, Claims, Consents, or other communications required or authorized hereunder shall be in writing and shall be deemed to have been duly given by the applicable Party if personally delivered, sent by facsimile with receipt acknowledged, sent by a recognized commercial overnight delivery service which guarantees next Business Day delivery, sent by U.S. registered or certified mail return receipt requested and postage prepaid, or otherwise actually received by the other Party at the address of the intended recipient set forth below:

If to Sellers:

Seahawk Drilling, Inc. 5 Greenway Plaza, Suite 2700 Houston, Texas 77046 Attention: General Counsel

Fax: 713-369-7312

With copies to (which shall not constitute notice):

Fulbright & Jaworski L.L.P. 1301 McKinney, Suite 5100 Houston, Texas 77010 Attention: P. Kevin Trautner

Fax: 713-651-5246

and

Asset Purchase Agreement Page 39 of 42 Fulbright & Jaworski L.L.P. 1301 McKinney, Suite 5100 Houston, Texas 77010 Attention: Berry D. Spears

Fax: 713-651-5246

If to Purchasers:

Hercules Offshore, Inc.

9 Greenway Plaza, Suite 2200

Houston, Texas 77046 Attention: General Counsel

Fax: 713-350-5109

With copies to (which shall not constitute notice):

Andrews Kurth LLP 600 Travis, Suite 4200 Houston, Texas 77001

Attention: Melinda H. Brunger

Fax: 713-238-7235

and

Thompson & Knight
333 Clay Street, Suite 3300
Houston, Texas 77002

Attention: Rhett G. Campbell

Fax: 832-397-8260

All such notices and communications shall be deemed to have been received if personally delivered, at the time delivered by hand; if mailed, three (3) Business Days after being deposited in the mail; if faxed, upon confirmation of receipt if the confirmation is between 9:00 a.m. and 5:00 p.m. local time of the recipient on a Business Day, otherwise on the first Business Day following confirmation of receipt; and, if sent by overnight air courier, on the next Business Day after timely delivery to the courier.

Either Party may change the address to which notices, requests, Claims, Consents, and other communications hereunder are to be delivered by giving the other Party prior written notice thereof in the manner herein set forth in this <u>Section 12.2</u>.

12.3 Further Assurances. Each of the Parties hereby agrees that after Closing it will execute and deliver such additional documents and will use commercially reasonable efforts to take or cause to be taken such further action as may be necessary or desirable, or as the other Party may reasonably request, to close and make effective the Contemplated Transactions. After the Closing, each Party, at the request of the other Party, and without additional consideration, shall execute and deliver, from time to time, such additional documents of conveyance and transfer as may be necessary to accomplish the orderly transfer of the Purchased Assets and Business to Purchasers in the manner contemplated in this Agreement

- 12.4 Expenses. Except as otherwise expressly provided in this Agreement, each of the Parties shall pay all costs and expenses incurred or to be incurred by it and its advisors and representatives in connection with any negotiations respecting this Agreement and Contemplated Transactions, including preparation of documents, obtaining any necessary regulatory approvals, and the consummation of the other transactions contemplated hereby.
- 12.5 Successors and Assigns. This Agreement, and all rights and powers granted hereby, will bind and inure to the benefit of the Parties and their respective successors and permitted assigns.
- 12.6 Third Party Beneficiaries. This Agreement and any agreement contained, expressed, or implied herein, shall not confer any rights or remedies upon any Person other than the Parties and their respective successors and permitted assigns.
- 12.7 Time of the Essence. Time is of the essence in the performance of all covenants and obligations under this Agreement.
- 12.8 Assignment. Neither this Agreement nor any Party's rights, interests, or liabilities hereunder may be assigned, transferred, conveyed, or pledged by operation of law or otherwise; provided, that Purchasers may transfer and assign prior to the Closing all or any portion of its rights and liabilities pursuant to this Agreement to an Affiliate thereof but (i) Purchasers shall not be relieved of their obligations hereunder as a result of such assignment, and (ii) to the extent any such assignment by Purchasers relates to the assignment by Sellers of an executory contract or unexpired Lease and occurs prior to the Closing such that, at the Closing, this Agreement will provide for Sellers' assignment of such executory contract or unexpired Lease to a Person other than Purchasers, such assignment by Sellers shall be subject to all applicable provisions of the Bankruptcy Code.
- 12.9 Governing Law; Venue. THIS AGREEMENT, THE TRANSACTION DOCUMENTS, AND THE LEGAL RELATIONS BETWEEN THE PARTIES WITH RESPECT TO THIS AGREEMENT, SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS WITHOUT REGARD TO RULES CONCERNING CONFLICTS OF LAWS. Purchasers and Sellers agree that the Bankruptcy Court shall have exclusive jurisdiction over all disputes and other matters relating to (a) the interpretation and enforcement of this Agreement or any ancillary document executed pursuant hereto, and (b) the Purchased Assets and Assumed Liabilities, and Purchasers expressly consents to and agrees not to contest such exclusive jurisdiction. All Claims and Proceedings brought, arising out of, or related to the Contemplated Transactions shall be brought in the Bankruptcy Court, and the Bankruptcy Court shall retain jurisdiction to determine any and all such Claims and Proceedings. The Parties waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the bringing of any such Claim or Proceeding in such jurisdiction.
- 12.10 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction.

- 12.11 Entire Agreement; Amendment. This Agreement (including any documents referred to in this Agreement) constitutes the entire agreement between the Parties with respect to the Contemplated Transactions and supersedes any prior understandings, negotiations, statements, discussions, correspondence, offers, agreements, or representations by the Parties, written or oral, relating in any way to the subject matter of this Agreement and the Contemplated Transactions. No modification, amendment, or supplement of any provision of this Agreement shall be valid unless the same shall be in writing and signed by the Parties. Without limiting the generality of the preceding sentence, no conditions, usage of trade, course of dealing or performance, understanding or agreement purporting to modify, vary, explain or supplement the terms or conditions of this Agreement will be binding unless hereafter made in writing and signed by the Party to be bound, and no modification will be effected by the acknowledgment or acceptance of documents containing terms or conditions at variance with or in addition to those set forth in this Agreement, except as otherwise specifically agreed to by the Parties in writing.
- 12.12 Counterparts. This Agreement may be executed by Purchasers and Sellers in one or more counterparts, each of which shall be deemed an original instrument, but all of which together shall constitute one and the same instrument. A facsimile transmission of a signed copy of this Agreement, shall be effective as a valid and binding agreement between the Parties for all purposes.
- 12.13 Certain Limitations. NOTWITHSTANDING ANY OTHER PROVISION OF THIS AGREEMENT (OR ANY OTHER AGREEMENT RELATED HERETO) TO THE CONTRARY, IN NO EVENT SHALL ANY PARTY BE LIABLE (OR ENTITLED TO RECOVER) UNDER, OR IN RESPECT OF, THIS AGREEMENT FOR EXEMPLARY, SPECIAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date and year first above written.

SELLERS:

SEAHAWK DRILLING, INC.

Name: <u>Randall D. Stilley</u>
Title: President and CEO

SEAHAWK GLOBAL HOLDINGS LLC

By: SEAHAWK DRILLING, INC., its sole member

By: Kandall & Stilley
Name: Kandall D. Stilley
Title: President and CEO

SEAHAWK DRILLING LLC

By: SEAHAWK GLOBAL HOLDINGS LLC, its sole member

By: SEAHAWK DRILLING, INC., its sole member

Name: Randy D Stilley
Title: President and (E)

SEAHAWK DRILLING USA LLC

By: SEAHAWK GLOBAL HOLDINGS LLC, its sole member

By: SEAHAWK DRILLING, INC., its sole member

Name: Kandall D. Stiller Title: Mesident and (D)

SEAHAWK DRILLING MANAGEMENT LLC

By: SEAHAWK GLOBAL HOLDINGS LLC, its sole member

By: SEAHAWK DRILLING, INC., its sole member

Name: Randall D. Stilley
Title: Wesident and LED

SEAHAWK MEXICO HOLDINGS LLC

By: SEAHAWK GLOBAL HOLDINGS LLC, its sole member

By: SEAHAWK DRILLING, INC., its sole member

Name: Randall D. Stilley
Title: President and CEO

ENERGY SUPPLY INTERNATIONAL LLC

By: SEAHAWK GLOBAL HOLDINGS LLC, its sole member

By: SEAHAWK DRILLING, INC., its sole member

Name: Randall D

Title: President and Cl

SEAHAWK OFFSHORE MANAGEMENT LLC

By: SEAHAWK GLOBAL HOLDINGS LLC, its sole member

By: SEAHAWK DRILLING, INC., its

sole member

Name: Randall D St Title: President and

PURCHASERS:

HERCULES OFFSHORE, INC.

John T Rync

Chief Executive Officer and President

SIL DRILLING VILC

By:_

James W. Noe Vice President

SCHEDULE 1.1 DEFINITIONS AND INTERPRETATIONS

Definitions. Subject to <u>Section 1.2</u> below, the following terms shall have the following definitions when used in the Agreement:

- "Affiliate" means a Person that directly, or indirectly through one or more intermediaries, controls or is controlled by or is under common control with another Person, As used in this definition, "control" means the possession, directly or indirectly, of the power to direct or cause the direction of management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. Purchasers, on the one hand, and Sellers, on the other hand, will not be deemed to be Affiliates of each other.
 - "Aggregate Consideration" has the meaning set forth in Section 2.4.
 - "Agreement" has the meaning set forth in the Preamble.
 - "Appraiser" has the meaning set forth in Section 2.9.
 - "Assigned Contracts" has the meaning set forth in Section 2.1(b).
 - "Assigned Intangible Assets" has the meaning set forth in Section 2.1(1).
 - "Assigned Permits" has the meaning set forth in Section 2.1(k).
 - "Assumed Liabilities" has the meaning set forth in Section 2.7.
 - "Bankruptcy Cases" has the meaning set forth in the Recitals above.
 - "Bankruptcy Code" has the meaning set forth in the Recitals above.
 - "Bankruptcy Court" has the meaning set forth in the Recitals above.
- "Bankruptcy Law" means the Bankruptcy Code and case Law applying and interpreting the Bankruptcy Code.
- "Books and Records" means the books, files, records, information, and data (including all Tax Returns and related workpapers and electronic data to the extent transferable) relating to the Purchased Assets, Transferred Employees, and Business that are in the possession of Sellers, including data regarding all accounts receivable and accounts payable in such form as reasonably requested by Purchaser so that such data can be transferred to Purchasers' accounting system from Sellers' accounting system.
 - "Business" has the meaning set forth in the Recitals above.
- "Business Day" means any day other than a Saturday, Sunday, or other day on which commercial banks in Houston, Texas are authorized or required by Law to close.
 - "Cash Payment" has the meaning set forth in Section 2.4.

- "Claim" means any demand, claim (including, without limitation, as defined in Section 101(5) of the Bankruptcy Code), cause of action, suit, demand, judgment, complaint, notice of noncompliance or violation, or other assertion of Liability.
 - "Closing" has the meaning set forth in Section 3.1.
 - "Closing Date" has the meaning set forth in Section 3.1.
 - "Closing DIP Amount" has the meaning set forth in Section 2.5(c)(i).
- "COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.
 - "Code" means the Internal Revenue Code of 1986, as amended.
 - "Competing Transaction" has the meaning set forth in Section 6.5.
 - "Computer Software" has the meaning set forth in Section 2.2(m).
- "Consent" means any necessary ratification by, notification requirement to, filing or registration with, or consent, waiver, approval, permit, or other authorization from, a Governmental Authority or other third party.
- "Contemplated Transactions" means all of the transactions contemplated by this Agreement, including: (a) the sale of the Purchased Assets to Purchasers as approved by the Transfer Order; (b) the execution and delivery of the Transaction Documents; and (c) the performance by the Parties of their respective covenants and obligations under this Agreement.
- "Contract" means any contract, agreement, lease, license, instrument, commitment, or other obligation or arrangement (whether written or oral), and any liability, cost, or expense of whatever kind or nature relating to the foregoing, that is binding on a Person or any part of its property under applicable Law.
 - "Cure Amounts" has the meaning set forth in Section 2.3.
- "Damages" means any and all damages, losses, liabilities, payments, obligations, penalties, fines, assessments, charges, costs, Taxes, disbursements or expenses (including interest, awards, judgments, settlements, costs of redemption, fees, reasonable disbursements and expenses of attorneys, accountants and other professional advisors and of expert witnesses and costs of investigation and preparation of any kind or nature whatsoever) and court costs.
- "DIP Budget" means the Initial DIP Budget and any amendment or replacement thereto approved by the DIP lenders, and the Court.
- "DIP Credit Documents" means the documentation evidencing the DIP Financing, including a credit agreement, security agreements, mortgages, collateral assignments and other related security instruments.

- "DIP Financing" means any debtor-in-possession financing arrangement entered into by Sellers in connection with the Bankruptcy Cases, including the DIP Credit Documents, DIP Order, and DIP Budget.
- "DIP Order" means any interim or Final Orders entered by the Bankruptcy Court in connection with the DIP Financing.
- "Employee Benefit Plan" means any (a) employee welfare benefit plans or employee pension benefit plans as defined in sections 3(1) and 3(2) of ERISA, including plans that provide retirement income or results in deferrals of income by employees for periods extending to their terminations of employment or beyond, and plans that provides medical, surgical or hospital care benefits or benefits in the event of sickness, accident, disability, death or unemployment and (b) other employee benefit agreements or arrangements that are not ERISA plans, including any deferred compensation plans, incentive plans, bonus plans or arrangements, stock option plans, stock purchase plans, stock award plans, golden parachute agreements, severance pay plans, dependent care plans, cafeteria plans, employee assistance programs, scholarship programs, retention incentive agreements, noncompetition agreements, vacation policies and, or other similar plans, agreement or arrangements that (i) are maintained by or required to be contributed to Sellers or any Affiliates thereof or with respect to which Sellers may have any Liability, (ii) have been approved by Sellers or any Affiliates thereof but are not yet effective for the benefit of Seller Employees or their beneficiaries, or (iii) that were previously maintained by Sellers or any Affiliates thereof and with respect to which Sellers or any Affiliates thereof may have any Liability, contingent or otherwise.

"Energy Supply International" has the meaning set forth in the Preamble.

"Environmental, Health and Safety Laws" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, the Resource Conservation and Recovery Act of 1976, the Occupational Safety and Health Act of 1970, each as amended, together with all other laws, rules and regulations of federal, state, local and foreign governments (and all agencies thereof) and other requirements having the force or effect of law relating to or imposing liability or standards of conduct concerning pollution or protection of the environment, public health and safety, or employee health and safety, and all judgments, orders and decrees of federal, state, local and foreign governments (and all agencies thereof) having the force and effect of law issued or promulgated thereunder, and all related common law theories.

"Equipment" has the meaning set forth in Section 2.1(c).

"Equity Interest" means (a) with respect to a corporation, any and all shares of capital stock of such corporation, (b) with respect to a partnership, limited liability company, trust, or similar Person, any and all units, interests, or other partnership/limited liability company interests, and (c) any other direct or indirect equity ownership or participation in a Person.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Escrow Agent" means The Bank of New York Mellon Trust Company, N.A.

"Excess DIP Amount" has the meaning set forth in Section 2.5(c)(i).

- "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Excluded Assets" has the meaning set forth in Section 2.2.
- "Final Order" means an Order by the Bankruptcy Court, (a) with respect to which the deadline for appeal provided by applicable statute, rule or regulation has passed and the subject Order has not been reversed, stayed, enjoined, set aside, annulled, or suspended, within the deadline, if any, provided by statute, rule or regulation, (b) with respect to which no request for stay, motion or petition for reconsideration, application or request for review, or notice of appeal or other judicial petition for review that is filed within such period is pending, and (c) as to which the deadlines, if any, for filing any such request, motion, petition, application, appeal or notice have expired; provided, however, that no Order shall fail to be a Final Order solely because of the possibility that a motion pursuant to Rule 60 of the Federal Rules of Civil Procedure or a motion under section 1144 of the Bankruptcy Code may be filed with respect to such Order.
- "GAAP" means generally accepted accounting principles, as recognized by the U.S. Financial Accounting Standards Board, consistently applied.
- "Governmental Authority" means the United States and any foreign, state, county, city, or other political subdivision and any department, commission, ministry, board, bureau, agency, commission, officer, official, court, tribunal, arbitrator, board or bureau or other instrumentality thereof or any quasi-governmental or private body exercising any regulatory, administrative, taxing, importing, exporting, or other governmental or quasi-governmental function and any self-regulatory organization, such as a securities exchange.
- "Governmental Order" means any Order, directive, writ, judgment, injunction, decree, stipulation, determination, or award by or with any Governmental Authority.
- "Hercules Common Stock" means the common stock, par value \$0.01 per share, of Hercules Parent.
 - "Hercules Financial Statements" has the meaning set forth in Section 5.10.
 - "Hercules Shares" has the meaning set forth in Section 2.4.
- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended and in effect from time to time.
- "Improvements" means all buildings, structures, fixtures, paving, curbing, trees, shrubs, plants, and other improvements of every kind and nature presently situated on, in, or under, or hereafter erected or installed or used in, on, or about or in connection with the ownership, use or operation of the Real Properties.
- "Initial DIP Budget" means the budget referenced in <u>Section 4.21</u> and attached as Exhibit 4.21 to this Agreement.

"Intangible Personal Property" means proprietary rights, business names, business logos, slogans, processes, Intellectual Property, franchises, licenses, technology, computer software, source codes, operating rights, plans and specifications, third party guaranties and warranties, rights with respect to telephone and facsimile numbers and email addresses or listings, SAP server code, data and content, and all other intangible rights and property, and any work product, knowledge, creative works, and other proprietary data related thereto.

"Intellectual Property" means all copyrights, patents, trademarks, trade names, service marks, mask works, domain names, email addresses, uniform resource locator addresses, trade dress, logos, slogans, symbols, corporate names, rights of publicity and privacy, inventions, design rights, drawings, trade secrets, customer and supplier lists, pricing and marketing plans, policies and strategies, methods of operation, royalties, secret processes, formulae, rights in know-how and all applications, registrations, renewals and other rights relating to the foregoing (whether or not any registration or filing has been made with respect thereto) used primarily in connection with the Business, including the Computer Software, Trademark Intellectual Property and the Website Intellectual Property.

"Interim Period" means the period of time from the date of this Agreement until the earlier of the Closing or the Termination Date.

"Knowledge" "Know" "Knowing" means in the case of Sellers, the actual knowledge of the Seller Designees after reasonable inquiry but without independent investigation and, in the case of Purchasers, the actual knowledge of the Purchasers Designees after reasonable inquiry but without independent investigation.

"Law" means all applicable local, state, federal, tribal, and foreign laws (whether statutory or common) and rules, regulations, rules, tariffs and regulatory authorizations, codes (including the Code), and ordinances promulgated thereunder, as well as case law, judgments, orders, consent orders, or decrees.

"Lease" means any lease or sublease, including any amendments thereto, of any Purchased Asset.

"Leased Real Property" has the meaning set forth in Section 2.2(1).

"Liabilities" means any and all debts, liabilities and obligations, whether accrued or fixed, absolute or contingent, matured or unmatured, or determined or determinable, including those arising under any Law or Governmental Order and those arising under any contract, agreement, arrangement, commitment or undertaking.

"Lien" means any charge against or interest in property to secure payment of a debt or performance of an obligation such as a Claim, debt, mortgage, indenture, hypothecation, encumbrance, Liability, lien, right of redemption, security interest, mechanic's or materialman's lien, judgment lien, assessment, special assessment, title defect, restriction, reservation, reversion, Contract, or right or interest of any third party, whether absolute or contingent, and the filing of or agreement to give any financing statement under the Uniform Commercial Code of any jurisdiction.

"Material Adverse Effect" with respect to any Person means any change, event, development, circumstance or effect (each, an "Effect") that, individually or taken together with all other related Effects, is, or would be reasonably expected to be, materially adverse (i) to the condition (financial or otherwise), assets, liabilities (taken together), business or results of operations of such Person and its subsidiaries, taken as a whole, but excluding the Effect (A) resulting from general economic conditions that does not disproportionately affect such Person in any material respect, (B) affecting companies in the industry in which it conducts its business generally, provided that such Effect does not disproportionately affect such Person in any material respect, or (C) resulting from the announcement or performance of this Agreement or the transactions contemplated hereby or (ii) on the ability of such Person to perform its obligations under this Agreement and the Transaction Documents or to consummate the Contemplated Transactions.

"Net Working Capital" as of a given date shall mean (1) the aggregate current assets included in the Purchased Assets as of that date, minus (2) the aggregate current liabilities included in the Assumed Liabilities as of the same date. Net Working Capital, including any accruals or reserves relating to the components thereof, shall be determined in accordance with GAAP, but shall not include any Excluded Assets or Retained Liabilities.

"Newco" has the meaning set forth in the Preamble.

"Notice Period" has the meaning set forth in Section 6.5(d).

"Offer of Employment" has the meaning set forth in Section 7.2.

"Order" means any order, ruling, decision, verdict, decree, charge, award (including arbitration awards), judgment, injunction, directive or other similar determination or finding by, before, or under the supervision of any Governmental Authority, or any arbitrator, board of arbitration, or similar entity including any regulatory or administrative Proceeding.

"Ordinary Course of Business" means, with respect to any Person, the ordinary course of business of such Person consistent with past custom and practice (including with respect to quantity and frequency).

"Organizational Documents" means the articles of incorporation, certificate of incorporation, charter, bylaws, articles or certificate of formation, regulations, operating or company agreement, certificate of limited partnership, partnership agreement, and all other similar documents, instruments or certificates executed, adopted, or filed in connection with the creation, formation, or organization of a Person, including any amendments thereto.

"Original DIP Amount" has the meaning set forth in Section 2.5(c)(i).

"Hercules Parent" has the meaning set forth in the Preamble above.

"Party" or "Parties" has the meaning as set forth in the Preamble.

"Performance Bond" means that certain \$2.0 million performance bond in favor of the Texas General Land Office.

- "Permits" means all licenses, permits, concessions, franchises, certificates, consents, exemptions, approvals, variances, registrations, or authorizations of any Governmental Authority, necessary or desirable in connection with, related to, or associated with the Purchased Assets or the conduct of the Business.
- "Permitted Encumbrances" means, with respect to the Purchased Assets, any of the following matters:
- (a) all easements, rights-of-way, and restrictive covenants affecting the Real Properties which individually or in the aggregate could not be reasonably expected to interfere with the operation of the Purchased Assets or Business by Purchasers; and
- (b) Liens that are excused or unenforceable from and after the Closing Date as a result of the filing of the Petitions or the applicability of the Bankruptcy Code.
- "Person" means any individual, firm, association, incorporated or unincorporated organization, partnership, business, trust, estate, joint stock company, joint venture, club, syndicate, company, corporation, Governmental Authority, or other legal entity.
- "Petition Date" means the date on which Sellers file the Petitions commencing the Bankruptcy Cases.
 - "Petitions" has the meaning set forth in the Recitals above.
- "Plan of Reorganization" means a plan of reorganization or liquidation to be filed by Sellers in the Bankruptcy Court.
 - "Pride" has the meaning set forth in Section 2.8(a)(viii).
- "Proceeding" means any action, suit, hearing, investigation, audit, examination, arbitration, litigation, review or other proceeding, at law or in equity, before any Governmental Authority, including any regulatory or administrative proceeding.
 - "Prohibited Transactions" has the meaning set forth in Section 6.5(a).
 - "Purchase Consideration Allocation" has the meaning set forth in Section 8.2.
 - "Purchased Assets" has the meaning set forth in Section 2.1.
 - "Purchaser" and "Purchasers" has the meaning set forth in the Preamble.
 - "Purchasers Designees" means John Rynd, Jim Noe, Stephen Butz and Troy Carson.
- "Purchasers' Reimbursable Expenses" means, with respect to Purchasers, the reasonable and documented out-of-pocket fees and expenses incurred by Purchasers, prior to the termination of this Agreement, in connection with the Contemplated Transactions, the Bankruptcy Cases, and the preparation, negotiation, prosecution, and performance of this

Agreement, including all reasonable fees and expenses of counsel, investment banking firms, financial advisors, accountants, and consultants to Purchasers in connection therewith.

"Real Property Rights" all easements, servitudes, rights-of-way, privileges, licenses, and appurtenances related or pertaining to the Leased Real Property, including all right, title and interest, if any, in and to (a) any land in the bed of any street, road or avenue open or proposed in front of or adjoining the Leased Real Property; (b) any rights-of-way, rights of ingress or egress or other interests in, on, or to, any land, highway, street, road, or avenue, open or proposed, in, on, or across, in front of, abutting or adjoining the Leased Real Property, and any awards made, or to be made in lieu thereof, and in and to any unpaid awards for damage thereto by reason of a change of grade of any such highway, street, road, or avenue; (c) any easement across, adjacent or appurtenant to the Leased Real Property, existing or abandoned; (d) all sewage treatment capacity and water capacity and other utility capacity to serve the Leased Real Property and Improvements; (e) all oil, gas, and other minerals in, on or under, and that may be produced from the Leased Real Property; (f) all water in, under or that may be produced from the Leased Real Property, and all wells, water rights, permits and historical water usage pertaining to or associated with the Leased Real Property; (g) any land adjacent or contiguous to, or part of the Leased Real Property, whether those lands are owned or claimed by deed, limitations or otherwise, and whether or not they are located inside or outside the description given herein, or whether or not they are held under fence by Sellers or whether or not they are located on any survey; and (h) any reversionary rights attributable to the Leased Real Property.

"Retained Books and Records" has the meaning set forth in Section 2.2(f).

"Retained Liabilities" has the meaning set forth in Section 2.8.

"Rig Loss Value" has the meaning set forth in Section 2.9.

"Rig Repair Costs" has the meaning set forth in Section 2.9.

"Rigs" has the meaning set forth in the Recitals.

"Sale Motion" means a motion filed by Sellers seeking approval of the Bankruptcy Court of this Agreement and the Contemplated Transactions.

"Seahawk Drilling" has the meaning set forth in the Preamble.

"Seahawk Drilling Management" has the meaning set forth in the Preamble.

"Seahawk Global Holdings" has the meaning set forth in the Preamble.

"Seahawk LLCs" means, collectively, Seahawk Global Holdings, Seahawk Mexico Holdings, Seahawk Drilling Management, Seahawk Drilling, Seahawk Offshore Management, Energy Supply International, and Seahawk USA.

"Seahawk Mexico Holdings" has the meaning set forth in the Preamble.

"Seahawk Offshore Management" has the meaning set forth in the Preamble.

- "Seahawk Parent" has the meaning set forth in the Preamble.
- "Seahawk USA" has the meaning set forth in the Preamble.
- "SEC" means the Securities and Exchange Commission of the United States of America or any other U.S. federal agency at the time administering the Securities Act.
- "SEC Filings" means, with respect to any Person, the forms, reports, schedules, statements and other documents filed with the SEC by such Person pursuant to the Securities Act and the Exchange Act.
 - "Securities Act" means the Securities Act of 1933, as amended.
 - "Seller" and "Sellers" have the meanings set forth in the Preamble.
- "Seller Employees" mean all employees of Sellers or any of their Affiliates and, for purposes of continuation coverage under COBRA, all former employees of Sellers or any of their Affiliates who have elected to receive COBRA continuation coverage under a Seller group health plan.
- "Seller Designees" means Randall Stilley, James Easter, Kurt Hoffman, Alejandro Cestero and Raymond Gonzales.
- "Sellers' Reimbursable Expenses" means, with respect to Sellers, the reasonable and documented out-of-pocket fees and expenses incurred by Sellers, prior to the termination of this Agreement, in connection with the Contemplated Transactions, the administration of this Agreement in the Bankruptcy Cases, and the preparation, negotiation, prosecution, and performance of this Agreement, including all reasonable fees and expenses of counsel, investment banking firms, financial advisors, accountants, and consultants to Sellers in connection therewith which fees and expenses shall be subject to approval by the Bankruptcy Court.
 - "Straddle Period" has the meaning set forth in Section 8.3.
- "Superior Proposal" means any bona fide written proposal or offer with respect to a Competing Transaction made by a third party (A) to acquire, directly or indirectly, the Business for consideration consisting of cash and/or securities (with a value of at least \$105,000,000) and the assumption of substantially all liabilities required to be assumed by Purchaser under this Agreement, and (B) that is otherwise on terms that the Board of Directors of Seahawk Parent determines in its reasonable good faith judgment (after consultation with its financial advisor and outside counsel), taking into account, among other things, all legal, financial and other aspects of the proposal or offer (1) if consummated, would result in a transaction that is more favorable, from a financial point of view, to the Sellers' stakeholders than the Contemplated Transactions and (2) is reasonably capable of being promptly consummated including with respect to receipt of all required regulatory approvals.
- "Tangible Personal Property" means furniture, fixtures, furnishings, racks, cabinets, hardware, tools, machinery, instruments, measuring and other devices, appliances, supplies,

communications devices and systems, computers, servers, and other computer equipment, hardware and systems, electrical equipment and systems, construction, repair, and maintenance equipment, materials, spare parts, and any other items of tangible personal property.

"Tax" means any federal, state, local or foreign income tax, ad valorem tax, excise tax, sales tax, use tax, value added tax, alternative or add-on minimum tax, franchise tax, real or personal property tax, transfer tax, gross receipts tax, escheat or unclaimed property tax, or other tax, assessment, duty, fee, levy or other governmental charge of any kind whatsoever, together with and including any and all interest, fines, penalties, assessments and additions to tax resulting from, relating to, or incurred in connection with any such tax or any contest or dispute thereof.

"Tax Returns" means all returns, declarations, reports, statements and other documents required to be filed in respect of Taxes.

"Termination Date" has the meaning set forth in Section 11.1(b).

"Termination Fee" means \$3,000,000.

"Termination Fee Motion" means the motion to be filed by Sellers pursuant to Section 6.3 seeking to approve the payment of the Termination Fee and Reimbursable Expenses to Purchasers as an administrative expense of the Bankruptcy Cases.

"Termination Fee Order" means an Order, in form reasonably acceptable to Purchasers, issued by the Bankruptcy Court approving the payment by Sellers to Purchasers of the Termination Fee and Reimbursable Expenses in accordance with Section 11.2(c) as an administrative expense of the Bankruptcy Cases.

"Total Loss" has the meaning set forth in Section 2.9.

"Trademark Intellectual Property" has the meaning set forth in Section 2.2(m)(ii).

"Trademarks" has the meaning set forth in Section 2.2(m)(ii).

"Transferred Books and Records" has the meaning set forth in Section 2.1(m).

"Transaction Documents" means the agreements listed in <u>Sections 3.3</u> and <u>3.4</u> and any other agreements or documents executed in connection with or as required under this Agreement.

"Transfer Order" means a Final Order issued by the Bankruptcy Court approving the Sale Motion, this Agreement and the Contemplated Transactions in form attached as Exhibit 6.3(c) and approved by Purchaser.

"Transfer Order Schedules" means the schedules of assets and liabilities to be filed by Sellers in the Bankruptcy Court.

"Transferred Employee" has the meaning set forth in Section 7.3.

"Transferred Rig Documents" means any and all class and flag documentation (originals and any existing copies); all existing drawings (hard and electronic); any third party survey or gauging reports; any manuals for foreign equipment that are in Sellers' possession; all shop and maintenance records, including for all well control and lifting/ hoisting equipment; Permits listed on Schedule 4.10; and other documents primarily relating to Transferred Rigs.

"Transferred Rigs" has the meaning set forth in Section 2.1(a).

"Website" has the meaning set forth in Section 2.2(m)(iii).

"Website Intellectual Property" has the meaning set forth in Section 2.2(m)(iii).

"Underlying Technology" means any and all technical information, software, specifications, drawings, records, shared drive and other computer files, work product, works of authorship, or other creative works or ideas knowledge, know-how, trade-secrets, invention disclosures, or other data including works subject to copyright protection and mask works associated with Sellers' website or operation thereof, including related e-mail.

Interpretations. Unless expressly provided for elsewhere in this Agreement, this Agreement shall be interpreted in accordance with the following provisions:

All references herein to Articles, Sections, Exhibits and Schedules are to Articles and Sections of and Exhibits and Schedules attached to and forming part of this Agreement, unless the contrary is specifically stated;

- (a) The headings of the Articles, Sections and subsections of this Agreement and the headings contained in the Exhibits and Schedules hereto are inserted for convenience of reference only and shall not in any way define or affect the meaning, construction, or scope of any of the provisions hereof or thereof;
- (b) a defined term has its defined meaning throughout this Agreement and each Exhibit and Schedule to this Agreement, regardless of whether it appears before or after the place where it is defined;
- (c) In the event of any conflict between the main body of this Agreement and the Exhibits and Schedules hereto, the provisions of the main body of this Agreement shall prevail;
- (d) Except where specifically stated otherwise, any reference to any statute, regulation, rule, or agreement shall be a reference to the same as amended, supplemented or reenacted from time to time;
- (e) Whenever the words "include," "including," or "includes" appear in this Agreement, they shall be read as if followed by the words "without limitation" or words having similar import;
- (f) Whenever the context may require, any pronoun used in this Agreement shall include the corresponding masculine, feminine, or neuter forms, and the singular form of nouns, pronouns and verbs shall include the plural and vice versa;

- (g) A reference to any agreement or document (including a reference to this Agreement) is to the agreement or document as amended, varied, supplemented, novated or replaced, from time to time, except to the extent prohibited by this Agreement or that other agreement or document;
- (h) A reference to any party to this Agreement or another agreement or document includes the party's permitted successors and assigns;
- (i) A reference to a writing includes a facsimile transmission of it and any means of reproducing of its words in a tangible and permanently visible form;
- (j) A reference to a statute, regulation or law shall include any amendment thereof or any successor thereto and any rules and regulations promulgated thereunder;
- (k) The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement;
- (l) Unless otherwise specified, all references to a specific time of day in this Agreement shall be based upon Central Standard Time or Central Daylight Savings Time, as applicable on the date in question.
- (m) References to "\$" or to "dollars" shall mean the lawful currency of the United States of America.
- (n) No action shall be required of the Parties except on a Business Day and in the event an action is required on a day which is not a Business Day, such action shall be required to be performed on the next succeeding day which is a Business Day.
- (o) All references to "day" or "days" shall mean calendar days unless specified as a "Business Day."
- (p) Time periods within or following which any payment is to be made or act is to be done shall be calculated by excluding the day on which the time period commences and including the day on which the time period ends and by extending the period to the next Business Day following if the last day of the time period is not a Business Day.

SCHEDULE 2.1(a) TRANSFERRED RIGS

U.S. Flagged Vessels

- 1. Seahawk 2003, Official Number 641802
- 2. Seahawk 2500, Official Number 638626
- 3. Seahawk 2502, Official Number 638954
- 4. Seahawk 2504, Official Number 567686
- 5. Seahawk 2602, Official Number 652045

Vanuatu Flagged Vessels

- 6. Seahawk 800, Official Number, 1831
- 7. Seahawk 2000, Official Number 1520
- 8. Seahawk 2001, Official Number 1542
- 9. Seahawk 2002, Official Number 1521
- 10. Seahawk 2004, Official Number 1463
- 11. Seahawk 2005, Official Number 1489
- 12. Seahawk 2006, Official Number 921
- 13. Seahawk 2007, Official Number 1721
- 14. Seahawk 2008, Official Number 1225
- 15. Seahawk 2501, Official Number 1826
- 16. Seahawk 2503, Official Number 1828
- 17. Seahawk 2505, Official Number 1829
- 18. Seahawk 2600, Official Number 1488
- 19. Seahawk 2601, Official Number 1827
- 20. Seahawk 3000, Official Number 1830

SCHEDULE 2.1(b) ASSIGNED CONTRACTS

See attached

Debion	Contact Counterpany	Description of Contract of Contract of Leading Contract of Leaders
Seahawk Drilling LLC	ARENA OFFSHORE	SEAHAWK 2602 (PRIDE MISSOURI) DRILLING AGREEMENT CONTRACT START DATE:12/03/2010
Seahawk Drilling LLC	ARENA OFFSHORE	SEAHAWK 3000 (PRIDE TEXAS) DRILLING SERVICES AGREEMENT CONTRACT START DATE:7/28/2010
Seahawk Drilling LLC	BADGER OIL	SEAHAWK 3000 (PRIDE TEXAS) DRILLING SERVICES AGREEMENT CONTRACT START DATE:6/28/2010
Seahawk Drilling LLC	BRISTOW US LLC	CONTRACTOR IS TO PROVIDE AIRCRAFTS, EQUIPMENT AND SERVICES TO THE COMPANY CONTRACT START DATE:09/13/2010
Seahawk Drilling LLC	CASTEX OFFSHORE, INC.	SEAHAWK 2007 (PRIDE NEW MEXICO) CONTRACT START DATE:12/09/2010
Seahawk Drilling LLC	CHEVRON U.S.A.	SEAHAWK 2601 (PRIDE KANSAS) CONTRACT START DATE:11/13/2010
Seahawk Drilling LLC	ENERGY XXI	SEAHAWK 3000 (PRIDE TEXAS) DRILLING SERVICES AGREEMENT CONTRACT START DATE:07/05/2010
Seahawk Drilling LLC	ENI	SEAHAWK 2001 (PRIDE ARKANSAS) DRILLING SERVICES AGREEMENT CONTRACT START DATE:11/22/2010
Seahawk Drilling LLC	ESSAR OILFIELD SERVICES INDIA	CONTINGENT SALE OF 2502 CONTRACT START DATE:09/28/2010
Seahawk Drilling LLC	ESSAR OILFIELD SERVICES INDIA LTD AND BNY MELLON, N.A.	ESCROW AGREEMENT FOR SALE OF 2505 CONTRACT START DATE:09/28/2010
Seahawk Drilling LLC	JONES, ROSE MARY W.	WEST HALBERT & HALBERT PASTURES CONTRACT START DATE:02/19/2010
Seahawk Drilling LLC	KEPPEL AMFELS, INC.	VESSEL REPAIR & MOD AGREEMENT - SEAHAWK 3000 REPAIRS AT YARD CONTRACT START DATE:07/30/2010
Seahawk Drilling LLC	KEPPEL AMFELS, INC.	VESSEL REPAIR & MOD AGREEMENT - SEAHAWK 3000 REPAIRS ONBOARD CONTRACT START DATE:1/105/2010
Seahawk Drilling LLC	NABORS DRILLING INTERNATIONAL LIMITED	SALE OF CEMENTING UNITS LOCATED IN TEXAS CONTRACT START DATE:03/18/2010
Seahawk Drilling LLC	NATIXIS	MORTGAGE FOR 2000, 2001, 2002, 2004, 2005, 2007, 2501, 2503, 2505, 2600, 2601, 3000, 800 CONTRACT START DATE:09(03/2009
Seahawk Drilling LLC	NATIXIS	MORTGAGE FOR 2504 AND 2602 CONTRACT START DATE:09(03/2009
Seahawk Drilling LLC	NATIXIS	JOINDER AGREEMENT CONTRACT START DATE:09(03/2009
Seahawk Drilling LLC	NATIXIS	ASSIGNMENT OF INSURANCE CONTRACT START DATE:09(03/2009
Seahawk Drilling LLC	NATKIS	SEAHAWK 2001 VESSEL; CREATES SECURITY INTEREST ONLY-ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS UNDER THE CONTRACT CONTRACT CONTRACT START DATE:10/29/2009
Seahawk Drilling LLC	NATIXIS	SEAHAWK 2501 VESSEL; CREATES SECURITY INTEREST ONLY-ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS UNDER THE CONTRACT CONTRACT CONTRACT CONTRACT START DATE:09/29/2009
Seahawk Drilling LLC	NATIXIS	SEAHAWK 2503 VESSEL; CREATES SECURITY INTEREST ONLY-ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS UNDER THE CONTRACT CONTRACT START DATE:09/29/2009
Seahawk Drilling LLC	NATIXIS	SEAHAWK 2505 VESSEL, CREATES SECURITY INTEREST ONLY-ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS UNDER THE CONTRACT CONTRACT START DATE:09/29/2009
Seahawk Drilling LLC	NATIXIS	SEAHAWK 3000 VESSEL; CREATES SECURITY INTEREST ONLY-ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS UNDER THE CONTRACT CONTRACT START DATE:09/29/2009
Seahawk Drilling LLC	OILFIELD EQUIPMENT SALES, INC.	PURCHASE AGREEMENT FOR EQUIPMENT BILL OF SALE CONTRACT START DATE: 12/31/2011
Seahawk Drilling LLC	PENINSULA DRILLING LLC	TRANSFER OF SEAHAWK 2503 FROM PRIDE LOUISIANA TO SEAHAWK DRILLING LLC CONTRACT START DATE:09/29/2009
Seahawk Drilling LLC	SEAHAWK DRILLING DE MEXICO LLC	TRANSFER OF SEAHAWK 2001 FROM PRIDE ARKANSAS TO SEAHAWK DRILLING LLC CONTRACT START DATE:10/29/2009
Seahawk Drilling LLC	TARPON OFFSHORE VENTURES	SEAHAWK 2600 (PRIDE ALASKA) CONTRACT START DATE:11/22/2010
Seahawk Drilling LLC	WALTER OIL AND GAS	SEAHAWK 2004 (PRIDE MISSISSIPPI) CONTRACT START DATE:12/03/2010
Seahawk Drilling, Inc.	A & L REPAIR SERVICES, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	ACCOUNTEMPS	TEMPORARY EMPLOYMENT AGENCY AGREEMENT AND CONDITIONS OF ASSIGNIMENT CONTRACT START DATE: 04/05/2010
Seahawk Drilling, Inc.	ACE TRANSPORTATION, LLC	TRANSPORTATION SERVICES CONTRACT START DATE:09/29/2009
Seahawk Drilling, Inc.	ACEVEDA, FELICIA	CONTRACT EMPLOYMENT AGREEMENT CONTRACT START DATE:11/17/2010
Seahawk Drilling, Inc.	ACME TRUCK LINE, INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT

Seahawk Drilling, Inc.	ACME TRUCK LINE, INC.	TRUCKING SERVICES AGREEMENT AND RELATED AMENDMENT CONTRACT START DATE:09/15/2010
Seahawk Drilling, Inc.	AFP ALARM & DETECTION	FIRE SUPPRESSION SYSTEM ALARMS CONTRACT START DATE:09/07/2010
Seahawk Drilling, Inc.	AGGREKO, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	ALLIS-CHALMERS ENERGY, INC.	PROTECTS INFORMATION PROVIDED FOR PURPOSES OF EVALUATING SEAHAWK'S MEXICO CEMENTING DIVISION FOR A POTENTIAL TRANSACTION CONTRACT START DATE:01/15/2010
Seahawk Drilling, Inc.	ALVAREZ & MARSAL BUSINESS CONSULTING, LLC	SERVICES TRANSITIONING IT DUTIES FROM PRIDE TO SEAHAWK CONTRACT START DATE:01/22/2009
Seahawk Drilling, Inc.	AMERICAN EXPRESS TRAVEL RELATED SERVICES CO., INC.	ENGAGEMENT OF AXP TO PROVIDE THE COMPANY WITH TRAVEL AND RELATED SERVICES CONTRACT START DATE:10/19/2009
Seahawk Drilling, Inc.	AMERICAN POLLUTION CONTROL	SPILL RESPONSE NORM CONTRACT START DATE-07/08/2009
Seahawk Drilling, Inc.	ANGEL AIR REPAIR & SPECIALTY COMPANY	PNEUMATIC REPAIRS, SALES AND SERVICE AND RELATED AMENDMENTS CONTRACT START DATE:06(0)/12009
Seahawk Drilling, Inc.	AP MOLLER	RECIPROCAL INDEMNITY AGREEMENT CONTRACT START DATE:11/22/2001
Seahawk Drilling, Inc.	APA FABRICATION, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	ARAMARK US OFFSHORE SERVICES, LLC	RICEBASED CATERING SERVICES CONTRACT START DATE-0245040
Seahawk Drilling, Inc.	ARGO PARTNERS	ASSIGNMENT OF RIGHTS TO CLAIM NOS. LS-11072-003 IN THE UNITED CAPITOL INSURANCE CO. CASE
Seahawk Drilling, Inc.	AT&T	VOICE/DATA STANDARD SERVICE AT THE BROUSSARD LOCATION CONTRACT START DATE:01/28/2010
Seahawk Drilling, Inc.	AT&T	WIRELESS SERVICES CONTRACT START DATE:05/08/2009
Seahawk Drilling, inc.	AXON ENERGY PRODUCTS AS	PARENT OF CHURCH SERVICES & DRILING CONTROLS BOP CERTIFICATIONS CONTRACT START DATE-66/16/2010
Seahawk Drilling, Inc.	BAKER ENERGY	PROVISION OF ONGOING TECHNICAL SUPPORT FOR MAXIMO CONTRACT START DATE-08/28/2008
Seahawk Drilling, Inc.	BAKERIMO SERVICES, INC.	MAXIMO SUPPORT SERVICES AGREEMENT CONTRACT START DATE-06/30/2009
Seahawk Drilling, Inc.	BASSOE OFFSHORE, INC.	BASSOE APPOINTED AS SOLE REPRESENTATIVE TO MARKET THE SEAHAWK 800 RIG CONTRACT START DATE:11(02)2010
Seahawk Drilling, Inc.	BASTION TECHNOLOGIES, INC.	ENGINEERING DESIGN & ANALYSIS CONTRACT START DATE:08/12/2010
Seahawk Drilling, Inc.	BAYOU BLACK ELECTRIC SUPPLY, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	BEACON MARITIME, INC.	FABRICATION & WELDING CONTRACT START DATE:08/22/2009
Seahawk Drilling, Inc.	BEACON MARITIME, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	BECERRA, CHRIS	CONTRACT EMPLOYMENT AGREEMENT CONTRACT START DATE:09/14/2009
Seahawk Drilling, Inc.	BLAKE DRILLING & WORKOVER COMPANY INC.	INDEMNITY AGREEMENTS MUTUAL INDEMNITY AND WAIVER OF RECOURSE CONTRACT START DATES:01/14/2008
Seahawk Drilling, Inc.	BLUE CROSS AND BLUE SHIELD OF TEXAS	BCBS TO ACT AS CLAIMS ADMINISTRATOR IN PROVIDING ADMINISTRATIVE SERVICES, INCLUDING THE DISCLOSURE OF PERSONAL HEALTH INFORMATION; ALSO SETS FORTH TERMS FOR THE COMPLIANCE OF BOTH PARTIES TO THE HEALTH INSIRANCE PORTABILITY AND ACCOUNTABIL!
Seahawk Drilling, Inc.	BLUEWATER RUBBER & GASKET CO.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	BOARDVANTAGE	PURCHASE AGREEMENT FOR EXTENDED SERVICES CONTRACT START DATE:06/30/2009
Seahawk Drilling, Inc.	BOHEN CRANE AND EQUIPMENT REPAIR, LLC	ENGAGEMENT OF BOHEN CRANE TO ACT AS STORAGE/BROKER TO SEAHAWK FOR A POSSIBLE SALE OF TWO CRANES CONTRACT START DATE:08/07/2010
Seahawk Drilling, Inc.	BOLLINGER FOURCHON, LLC	SHIPYARD REPAIRS CONTRACT START DATE:06/29/2009
Seahawk Drilling, Inc.	BRACEWELL & GIULIANI	LEGAL SERVICES CONTRACT START DATE:06/17/2009
Seahawk Drilling, Inc.	BRACEWELL & GIULIANI	TRADEMARK & IP COUNSELING CONTRACT START DATE:06/30/2009
		The state of the s

Seahawk Drilling, Inc.	BRIDGEWAY SEARCH GROUP	FEE ARKANGEMEN FOX EMPLOYEE PLACEMEN SERVICES CONTRACT START DATE:01/25/2010
Seahawk Drilling, Inc.	BUSINESSSOLVER, INC.	HR BENEFITS CONSULTING SERVICES AGREEMENT CONTRACT START DATE:07/07/2009
Seahawk Drilling, Inc.	C&G BOATS, INC.	RIG TOWING SERVICES CONTRACT START DATE:01/06/2010
Seahawk Drilling, Inc.	C.V. NIFFRAY	AGENT CONTRACT CONTRACT START DATE:12/29/1998
Seahawk Drilling, Inc.	CAD CONTROL SYSTEMS	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	CAMERON EQUIPMENT	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	CAPGEMINI U.S., LLC	MASTER SERVICE AGREEMENT CONTRACT START DATE:01/15/2010
Seahawk Drilling, Inc.	CAPGEMINI U.S., LLC	SERVER HOST CONTRACT START DATE:03/19/2010
Seahawk Drilling, Inc.	CAPGEMINI U.S., LLC	SAP SUPPORT CONTRACT START DATE:05/07/2010
Seahawk Drilling, Inc.	CAPGEMINI U.S., LLC	SAP INSTALLATION & TRAINING CONTRACT START DATE:02/15/2010
Seahawk Drilling, Inc.	CHAMBERLAIN HRDLICKA	LEGAL SERVICES CONTRACT START DATE:04/15/2010
Seahawk Drilling, Inc.	CHARTER SUPPLY CO.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	CHUBB GROUP OF INSURANCE COMPANIES	SURETY BONDS CONTRACT START DATE:08/17/2009
Seahawk Drilling, Inc.	CHURCH ENERGY SERVICES LTD.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	CINTAS FIRST AID & SAFETY	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	OLEANBLAST I & II, LLC	SANDBLASTING PAINTING CONTRACT START DATE:07/10/2009
Seahawk Drilling, Inc.	COGENT COMMUNICATIONS, INC.	TELECOMMUNICATIONS SERVICES AGREEMENT CONTRACT START DATE:02/19/2010
Seahawk Drilling, Inc.	COMCAST	COMMIERCIAL SERVICE ORDER FOR BUSINESS CLASS TELEVISION CONTRACT START DATE:02/25/2010
Seahawk Drilling, Inc.	COMPLIANCE INSPECTION SERVICES INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	COMPRESSED AIR SYSTEMS, LLC	MASTER SERVICE AGREEMENT CONTRACT START DATE:02/15/2010
Seahawk Drilling, Inc.	COMPUCYCLE, INC.	PROVISION OF CERTAIN ELECTRONICS RECYCLING AND ELECTRONIC ASSET MANAGEMENT SERVICES TO SEAHAWK CONTRACT START DATE:08/31/2010
Seahawk Drilling, Inc.	COMPUCYCLE, INC.	TECHNOLOGICAL SERVICES AGREEMENT CONTRACT START DATE:08/31/2010
Seahawk Drilling, Inc.	COPPER STATE RUBBER - SPECIALTIES CO.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	COX BUSINESS	INTERNET SERVICES FOR THE BROUSSARD, LA FIELD OFFICE CONTRACT START DATE:12/01/2009
Seahawk Drilling, Inc.	CROSBY TUGS, LLC	RIG TOWING SERVICES CONTRACT START DATE:01/22/2001
Seahawk Drilling, Inc.	DAHILL (XEROX)	MAINTENANCE SERVICE FOR ALL XEROX COPIERS CONTRACT START DATE:11/20/2009
Seahawk Drilling, Inc.	DELL, INC.	CONFIDENTIAL INFORMATION OBTAINED BY SEAHAWK IS ONLY TO BE USED IN CONNECTION WITH ITS BUSINESS RELATIONSHIP WITH DELL, WHICH INCLUDES DESIGN, TESTING, MANUFACTURING, SALE AND SUPPORT OF DELL PRODUCTS AND/OR SERVICES THAT INTEROPERATE WITH
Seahawk Drilling, Inc.	DELOITTE TAX, LLP	TAX ADVISORY SERVICES CONTRACT START DATE:09/14/2009
Seahawk Drilling, Inc.	DETECTION & SUPPRESSION INTERNATIONAL, LTD	INSPECTION AND TESTING OF FIKE CHEETAH XI - 50 PT FM 200 SYSTEM - LEVEL 28 FIRE SUPPRESSION SYSTEM TWO TIMES PER AGREEMENT PERIOD CONTRACT START DATE:05/01/2010
Seahawk Drilling, Inc.	DISCOVERY BENEFITS, INC.	DISCOVERY BENEFITS TO ASSIST IN THE ESTABLISHMENT AND OPERATION OF SEAHAWKS CAFETERIA PLAN, HEALTH FSA AND DEPENDENT CARE FSA CONTRACT START DATE:01/01/2010
Seahawk Drilling, Inc.	DOLPHIN SCAFFOLD SERVICES, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT CONTRACT START DATE:01/27/2011
Seahawk Drilling, Inc.	DOLPHIN SERVICES, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	DOLPHIN TOWING	MASTER TIME CHARTER AGREEMENT CONTRACT START DATE:01/01/2000

Seahawk Drilling, inc.	DON ABNEY, INC.	SERVICE CON RACI MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	EAST PARK RADIATOR & BATTERY SHOP, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	EDISON, MCDOWELL & HETHERINGTON LLP	LEGAL SERVICES AGREEMENT
Seahawk Drilling, Inc.	EMDSI - HUNT FKA HUNT ENGINE, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	ENCORE BANK	\$5,000,000 NOTE CONTRACT START DATE:09/03/2009
Seahawk Drilling, Inc.	ENCORE BANK, N.A.	REVOLVING CREDIT AGREEMENT CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	ENVIRONMENTAL SAFETY AND HEALTH	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	ERNST & YOUNG, LLP	CONSULTING SERVICES AGREEMENT CONTRACT START DATE:10/27/2009
Seahawk Drilling, Inc.	ESSAR OILFIELD SERVICES INDIA LTD	TCA CONTRACT START DATE:09/28/2010
Seahawk Drilling, Inc.	FIANZAS MONTERREY, S.A.	GUARANTEE OF OBLIGATIONS AS TO THE BONDS ISSUED CONTRACT START DATE:08/13/2009
Seahawk Drilling, Inc.	FIRE AND SAFETY SPECIALISTS, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	FOSTER QUAN LLP	LEGAL SERVICES CONTRACT START DATE:01/18/2010
Seahawk Drilling, Inc.	FULBRIGHT & JAWORSKI, LLP	LEGAL SERVICES CONTRACT START DATE:12/24/2008
Seahawk Drilling, Inc.	GALLOWAY JOHNSON TOMPKINS BURR & SMITH	LEGAL SERVICES CONTRACT START DATE:01/28/2010
Seahawk Drilling, Inc.	GARDERE WYNNE SEWELL, LLP	LEGAL SERVICES CONTRACT START DATE:01/19/2010
Seahawk Drilling, Inc.	GARDERE, ARENA Y ASOCIADOS, S.C.	LEGAL SERVICES CONTRACT START DATE:01/19/2010
Seahawk Drilling, Inc.	GARZA TELLO Y ASOCIADOS, S.C.	LEGAL SERVICES CONTRACT START DATE:01/18/2010
Seahawk Drilling, Inc.	GEC CAPITAL GROUP, LP	RELATED TO USE AND DISCLOSURE OF EVALUATION MATERIAL RECEIVED BY GEC CAPITAL FOR PURPOSES OF EVALUATING A PROPOSED INVESTMENT IN THE COMPANY CONTRACT START DATE:03/16/2010
Seahawk Drilling, Inc.	GENERAL MARINE LEASING, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	GENERAL SERVICES, CO.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	GEOPHYSICAL DATA MANAGEMENT, INC.	SURVEY WORK CONTRACT START DATE:09/13/2010
Seahawk Drilling, Inc.	GERMAN, OSCAR A.	EMPLOYMENT AGREEMENT CONTRACT START DATE:11/01/2008
Seahawk Drilling, Inc.	GERMAN, OSCAR A.	SEPARATION AGREEMENT CONTRACT START DATE:01/29/2010
Seahawk Drilling, Inc.	GLOBAL CATHODIC PROTECTION, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	GLOBAL COMPLIANCE SERVICES, INC.	BRIGHTLINE - ONLINE AND IN-PERSON TRAINING CONTRACT START DATE:08/12/2009
Seahawk Drilling, Inc.	GLOBAL COMPLIANCE SERVICES, INC.	GCS TO PROVIDE SEAHAWK AN ETHICS AND COMPLIANCE HOTLINE, AS WELL AS A WEB-BASED CASE MANAGEMENT PROGRAM CONTRACT START DATE:08/10/2009
Seahawk Drilling, Inc.	GOWAN, INC.	MAINTENANCE SERVICE AGREEMENT CONTRACT START DATE-12/14/2009
Seahawk Drilling, Inc.	GRUPO HEGEMONIA SA DE CV	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	GULF COAST INTERNATIONAL, LLC	BRAKE REPAIR SCR REPAIR & MANUFACTURE CONTRACT START DATE:06/02/2009
Seahawk Drilling, Inc.	GULF COAST MARINE & ASSOCIATES, INC.	SERVICES TO INCLUDE PERFORMANCE OF RIG MOVEMENT, RIG REPAIR, SURVEY AND PROJECT MANAGEMENT AS REQUESTED FROM TIME TO TIME CONTRACT START DATE:06/02/2009
Seahawk Drilling, Inc.	GULF COPPER & MANUFACTURING CO.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	HANNON HYDRAULICS	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	HB RENTALS, LC	SERVICE CONTRACT MASTER SERVICE AGREEMENT

		TABLETALANDE CENTRALE
Seahawk Drilling, Inc.	HILLER OFFSHORE SERVICES, INC.	CONTRACT START DATE:08/15/2009
Seahawk Drilling, Inc.	HITECH INTEGRATED SOLUTIONS, INC.	FIRE SAFETY EQUIPMENT CONTRACT START DATE:09/01/2009
Seahawk Drilling, Inc.	HOUMA ARMATURE WORKS & SUPPLY, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	HOUMA FILTER AND SUPPLY, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	HSE SUPPLIES (PTY) LTD.	MASTER SERVICE CONTRACT CONTRACT START DATE:11/18/2008
Seahawk Drilling, Inc.	IKON FINANCIAL SERVICES	LEASE OF PRINTER AND OTHER SERVICES PROVIDED CONTRACT START DATE: 10/27/2009
Seahawk Drilling, Inc.	IKON FINANCIAL SERVICES	FINANCIAL SERVICES AGREEMENT CONTRACT START DATE: 10/27/2009
Seahawk Drilling, Inc.	INFOSTAT SYSTEMS, INC.	SEAHAWK WAS GRANTED A LICENSE TO USE THE RIMDRILL SOFTWARE PROGRAM. CONTRACT START DATE:08/01/2009
Seahawk Drilling, Inc.	INTEGRICERT LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	INTERCALL, INC.	AUDIO CONFERENCING SERVICES CONTRACT START DATE:06/01/2010
Seahawk Drilling, Inc.	INTERNATIONAL SATELLITE SERVICES, INC.	SATELLITE TELEPHONE SERVICE CONTRACT START DATE:08/30/2010
Seahawk Drilling, Inc.	JORDAN, HYDEN, WOMBLE CULBRETH & HOLZER	ENGAGEMENT LETTER FEE AGREEEMENT CONTRACT START DATE: 01/11/2011
Seahawk Drilling, Inc.	KAHANEK ADVISORY SERVICES	FINANCIAL ADVISORY SERVICES AGREEMENT CONTRACT START DATE:08/14/2010
Seahawk Drilling, Inc.	KEAN MILLER HAWTHORNE D'ARMOND MCCOWAN & JARMAN LLP	LEGAL SERVICES CONTRACT START DATE:04/14/2010
Seahawk Drilling, Inc.	KEANE, INC.	PROVIDES SCOPE OF PROJECT AND SCHEDULE OF SERVICES CONTRACT START DATE:12/13/2010
Seahawk Drilling, Inc.	KEANE, INC.	FINANCIAL ADVISORY SERVICES AGREEMENT CONTRACT START DATE:11/03/2010
Seahawk Drilling, Inc.	KEM SUPPLY HOUSE, INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	KPMG	AUDIT SERVICES AGREEMENT CONTRACT START DATE:07/23/2009
Seahawk Drilling, Inc.	KUEHNE + NAGEL, INC.	LOGISTICS SERVICES AGREEMENT CONTRACT START DATE:09/02/2009
Seahawk Drilling, Inc.	KURTZMAN CARSON CONSULTANTS LLC	SERVICE AGREEMENT FOR NOTICING AND SOLICITATION PURPOSES CONTRACT START DATE: 12/15/2010
Seahawk Drilling, Inc.	L&L RIG SERVICES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	LABORDE AND NEUNER	LEGAL SERVICES CONTRACT START DATE:01/20/2010
Seahawk Drilling, Inc.	LAFAYETTE STEEL ERECTOR, INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	LOADMASTER DERRICK & EQUIPMENT, INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	LOCKE LORD BISSELL & LIDDELL, ILP	LEGAL SERVICES AGREEMENT CONTRACT START DATE:09/29/2010
Seahawk Drilling, Inc.	LOCKE LORD BISSELL & LIDDELL, LLP	PROVIDES GUIDELINES IN CONNECTION WITH THE FIRM'S LEGAL REPRESENTATION FOR 2010 CONTRACT START DATE:10/19/2010
Seahawk Drilling, Inc.	LOF INTERNATIONAL HUMAN RESOURCES SOLUTIONS, INC.	HUMAN RESOURCES SERVICES AGREEMENT CONTRACT START DATE:03/16/2010
Seahawk Drilling, Inc.	LONGNECKER & ASSOCIATES	PROVIDES DETAILS OF ENGAGEMENT CONTRACT START DATE:04/29/2009
Seahawk Drilling, inc.	LOUISIANA ELECTRIC RIG SERVICE, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	LOUISIANA ENVIRONMENTAL MONITORING INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	LOUISIANA INTERNATIONAL MARINE, LLC	RIG TOWING SERVICES CONTRACT START DATE:06/29/2009
Seahawk Drilling, Inc.	LOUISIANA MACHINERY	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	МАР LTD.	SERVICE CONTRACT MASTER SERVICE AGREEMENT

		אסיים בויים ביים ביים ביים ביים ביים ביים
Seahawk Drilling, Inc.	MAGNUM CONSTRUCTION SERVICES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	MANIFOLD VALVE SERVICES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	MANZ, STEVEN A.	RELEASE AGREEMENT CONTRACT START DATE:05/07/2010
Seahawk Drilling, Inc.	MARINE SYSTEMS, INC.	DIESEL ENGINE REPAIR CONTRACT START DATE:11/02/2009
Seahawk Drilling, Inc.	MARITIME INDUSTRIAL SERVICES CO. LIMITED	SECRECY AGREEMENT CONTRACT START DATE:09/24/2009
Seahawk Drilling, Inc.	MARTIN, DISIERE, JEFFERSON & WISDOM LLP	LEGAL SERVICES CONTRACT START DATE:01/19/2010
Seahawk Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	AUTOMATIC SERVICES PROVIDED CONTRACT START DATE:1124/2009
Seahawk Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	PROVIDES DETAILED OUTLINE OF ALL OF THE ADMINISTRATIVE SERVICES THAT ARE TO BE PERFORMED UNDER THE AGREEMENT CONTRACT START DATE:01/01/2010
Seahawk Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	DISCLOSURE AND ACKNOWLEDGEMENT RELATED TO AGENT COMMISSIONS UNDER THE BENEFITS PLAN CONTRACT START DATE:01/01/2010
Seahawk Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	INVESTMENT SERVICES AGREEMENT CONTRACT START DATE:01/01/2010
Seahawk Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	PRESCRIBES GUIDELINES FOR SELECTING, MONITORING & EVALUATING INVESTMENT OPTIONS UNDER THE PLAN CONTRACT START DATE:04/30/2010
Seahawk Drilling, Inc.	MASTER RIG INTERNATIONAL LLC	STORAGE SERVICES AGREEMENT AND RELATED AMENDMENTS CONTRACT START DATE: 02/09/2011
Seahawk Drilling, Inc.	MATTHEWS-DANIEL COMPANY	MARINE WARRANTY SERVICES CONTRACT START DATE:08/05/2009
Seahawk Drilling, Inc.	MCGRIFF, SEIBELS & WILLIAMS OF TEXAS, INC.	BROKERAGE SERVICE AGREEMENT FOR INSURANCE POLICIES CONTRACT START DATE:08/24/2010
Seahawk Drilling, Inc.	MCGRIFF, SEIBELS & WILLIAMS OF TEXAS, INC.	BROKERAGE SERVICE AGREEMENT FOR INSURANCE POLICIES CONTRACT START DATE:06/30/2010
Seahawk Drilling, Inc.	MELLON INVESTOR SERVICES LLC	PRIDE COMMON STOCK DISTRIBUTION AT TIME OF SPIN AND THE RELATED RIGHTS OF HOLDERS OF RECORD CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	MELLON INVESTOR SERVICES LLC	PLAN ADMINISTRATOR AND AMENDMENT INCLUDING ADMINSTRATION OF 10B-5 PLANS CONTRACT START DATE:07/01/2009
Seahawk Drilling, Inc.	MELLON INVESTOR SERVICES LLC	SERVICE AGREEMENT FOR TRANSFER AGENT SERVICES CONTRACT START DATE:06/10/2009
Seahawk Drilling, Inc.	MELLON INVESTOR SERVICES LLC	SOLICIT PROXIES FOR 2010 ANNUAL MEETING CONTRACT START DATE:03/11/2010
Seahawk Drilling, Inc.	MELLON INVESTOR SERVICES LLC	INTENT TO ENTER INTO ADMINISTRATIVE AGENT AGREEMENT CONTRACT START DATE:05/01/2009
Seahawk Drilling, Inc.	MICROSOFT LICENSING, GP	USE OF MICROSOFT PRODCUTS CONTRACT START DATE:06/129/2009
Seahawk Drilling, Inc.	MICROSOFT LICENSING, GP	USE OF MICROSOFT PRODCUTS CONTRACT START DATE:08/23/2009
Seahawk Drilling, Inc.	MICROSOFT LICENSING, GP	SIGNATURE AGREEMENT FOR THE OTHER MICROSOFT AGREEMENTS AND AMENDMENT TO ENTERPRISE AGREEMENT CONTRACT START DATE:08/29/2009
Seahawk Drilling, Inc.	MONTCO OFFSHORE, INC.	MASTER TIME CHARTER AGREEMENT CONTRACT START DATE:08/08/2006
Seahawk Drilling, Inc.	MS LEGAL SEARCH, LLC	LEGAL HEADHUNTER ENGAGEMENT CONTRACT CONTRACT START DATE:01/29/2010
Seahawk Drilling, Inc.	NANCE INTERNATIONAL, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	NATIONAL OILFIELD SERVICES, LLC	STRUCTURAL PIPING REPLACEMENT CONTRACT START DATE:12/14/2009
Seahawk Drilling, Inc.	NATIONAL OILWELL VARCO, LP	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	NATIXIS	FINANCIAL AGREEMENT REVOLVING CREDIT AGREEMENT, INCLUDING 1ST, 2ND, 3RD, AND 4TH AMENDMENTS TO AGREEMENT CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	NATIXIS	WAIVER UNDER THE CREDIT AGREEMENT CONTRACT START DATE:03/23/2010
Seahawk Drilling, Inc.	NATIXIS	\$11,000,000 NOTE CONTRACT START DATE,09/03/2009
Seahawk Drilling, Inc.	NATIXIS	SECURITY AGREEMENT CONTRACT START DATE:09/03/2009
Seahawk Drilling, Inc.	NATIXIS	PLEDGE AGREEMENT CONTRACT START DATE:09/03/2009

Seahawk Drilling, Inc.	NATIXIS	ASSIGNMENT OF INSURANCE CONTRACT START DATE;09/29/2009
Seahawk Drilling, Inc.	NATIXIS	NATIXIS IS GRANTED CONTROL OVER EACH OF THE GRANTORS ACCOUNTS CONTRACT START DATE:09/03/2009
Seahawk Drilling, Inc.	NATIXIS	WAIVES TERMS RELATING TO EXPIRATION CONTRACT START DATE:03/23/2010
Seahawk Drilling, Inc.	NATIXIS	FINANCIAL AGREEMENT GUARANTY ON REVOLVING CREDIT AGREEMENT, INCLUDING 1ST, 2ND, 3RD, AND 4TH AMENDMENTS TO AGREEMENT CONTRACT START DATE: 804/2009
Seahawk Drilling, Inc.	NAVIGIS, LLC	SOFTWARE LICENSE AGREEMENT CONTRACT START DATE:04/01/2010
Seahawk Drilling, Inc.	OCEAN MARINE CONTRACTORS, INC.	WELDING AND FABRICATION CONTRACT START DATE:08/15/2009
Seahawk Drilling, Inc.	OCEANWEATHER INC.	LICENSE AGREEMENT CONTRACT START DATE: 07/15/2009
Seahawk Drilling, Inc.	ODL, INC.	MASTER SERVICES AGREEMENT
Seahawk Drilling, Inc.	OES OILFIELD SERVICES	NDT LIFTING GEAR INSPECTION PROPPED OBJECT SURVEY CONTRACT START DATE:06/15/2009
Seahawk Drilling, Inc.	OFFSHORE EQUIPMENT SOLUTIONS	STORAGE SERVICES AGREEMENT AND RELATED AMENDMENTS CONTRACT START DATE: 02/09/2011
Seahawk Drilling, Inc.	OFFSHORE STAFFING SERVICES OF ACADIANA, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT CONTRACT START DATE:02/01/2011
Seahawk Drilling, Inc.	OFFSHORE TOWING, INC.	RIG TOWING SERVICES CONTRACT START DATE:06/22/2009
Seahawk Drilling, Inc.	OFFSHORE TOWING, INC.	TOWING AGREEMENT CONTRACT START DATE:02/19/2010
Seahawk Drilling, Inc.	OFFSHORE TOWING, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	OIL AND GAS RENTAL SERVICES INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	OILFIELD AUDIT SERVICES, INC.	RELEASE TO VIEW A RIG CONTRACT START DATE:08/25/2010
Seahawk Drilling, Inc.	OILFIELD BEARING INDUSTRIES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	OPACLE USA, INC.	ASSIGNMENT AND CERTIFICATION OF NON POSSESSION CONTRACT START DATE:11/13/2009
Seahawk Drilling, Inc.	ORACLE USA, INC.	LICENSE AND SERVICES AGREEMENT AND RELATED AMENDMENT TO ORDERING DOCUMENT CONTRACT START DATE:02/25/2010
Seahawk Drilling, Inc.	ORACLE USA, INC.	ASSIGNMENT AND CERTIFICATION OF NON POSSESSION CONTRACT START DATE:02/28/2010
Seahawk Drilling, Inc.	PARTNERS CONSULTING	IT CONSULTING AGREEMENT CONTRACT START DATE:06/01/2009
Seahawk Drilling, Inc.	PARTNERS CONSULTING	IT CONSULTING AGREEMENT CONTRACT START DATE:06/03/2009
Seahawk Drilling, Inc.	PAWS PUMPS, INC.	PUMPS ON SEAHAWK 2601 CONTRACT START DATE:09/20/2010
Seahawk Drilling, Inc.	PENINSULA DRILLING LLC	TRANSFER OF SEAHAWK 2501 FROM PRIDE CALIFORNIA TO SEAHAWK DRILLING LLC CONTRACT START DATE: 09/29/2009
Seahawk Drilling, Inc.	PETRON INDUSTRIES, INC.	MANUFACTURE AND RENTAL OF RIG MONITORING INSTRUMENTATION CONTRACT START DATE:05/27/2009
Seahawk Drilling, Inc.	PITNEY BOWES	LEASE AGREEMENT CONTRACT START DATE:05/22/2009
Seahawk Drilling, Inc.	PLATINUM EQUITY ADVISORS, LLC	THIRD PARTY PROFESSIONAL AGREEMENT TRANSACTIONAL AGREEMENT CONTRACT START DATE: 12/17/2010
Seahawk Drilling, Inc.	POWER PRODUCTS SERVICE CO., INC.	PUMP AND COMPRESSOR SALES AND SERVICE CONTRACT START DATE:05/27/2009
Seahawk Drilling, Inc.	PREFERRED ELECTRIC, INC.	ELECTRIC & HVAC CONSTRUCTION AND MAINTENANCE CONTRACT START DATE:04/02/2010

Seahawk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	TRANSITION SERVICES BETWEEN PRIDE AND SEAHAWK POST SPIN-OFF AND RELATED AMENDMENT OF SPECIAL PROVISIONS TO SCH, E OF AGREEMENT CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	TERMS AND CONDITIONS OF SEPARATION OF SDI FROM PII CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	ALLOCATION OF TAXES AMONG PARTIES AT DISTRIBUTION CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	CREDIT SUPPORT BY PRIDE FOR MEXICAN TAX OBLIGATIONS CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	AGREEMENT TO EFFECT ORDERLY TRANSITION OF EMPLOYEES, BENEFITS, ETC. CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	AGREEMENT CONCERNING PRIDE TN ISSUES CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	AGREEMENT CONCERNING PRIDE WISCONSIN ISSUES CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE TENNESEE LLC	ASSIGNMENT OF INTEREST IN THE CHARTERPARTY AND MODIFICATION OF TERMS RELATED TO PAYMENT FOR USE OF RIG CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIDE WISCONSIN LLC	ASSIGNIMENT & ASSUMPTION AGREEMENT AND AMENDMENT TO CHARTERPARTY AGREEMENT CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	PRIME ELECTRICAL SERVICES, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	PSC INDUSTRIAL OUTSOURCING, LP	NORM REMEDIATION, VACUUM SERVICES, SALT WATER INJECTION, DOCKSIDE CLEANING CONTRACT START DATE:07/27/2009
Seahawk Drilling, Inc.	QUALITY OIL TOOLS, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	REID DAVIS LLP	LEGAL SERVICES CONTRACT START DATE:09/08/2010
Seahawk Drilling, Inc.	RESOURCES GLOBAL PROFESSIONALS	JDE OPTIMIZATION CONTRACT START DATE:07/27/2009
Seahawk Drilling, Inc.	RIGNET, INC.	DATA CIRCUITS RIG COMUNICATION SERVICES AGREEMENT AND RELATED ADDENDA CONTRACT START DATE:04/01/2009
Seahawk Drilling, Inc.	RIMINI STREET, INC.	SUPPORT FOR ORACLE SYSTEMS (JD EDWARDS) CONTRACT START DATE:09/22/2010
Seahawk Drilling, Inc.	RL ELDRIDGE CONSTRUCTION, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	RLI INSURANCE COMPANY	INDEMNITY AGREEMENT FOR PERFORMANCE BONDS CONTRACT START DATE:08/23/2009
Seahawk Drilling, Inc.	ROSALES, EDGAR ALEXANDER	Separation Agreement CONTRACT START DATE:03/17/2010
Seahawk Drilling, Inc.	ROTOCRAFT LEASING COMPANY, LLC	HELICOPTER SERVICES FOR OFFSHORE CONTRACT START DATE:05/18/2010
Seahawk Drilling, Inc.	ROTORCRAFT LEASING COMPANY LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SAGE SOFTWARE, INC.	SAGE FAS PRODUCT CONTRACT START DATE:12/01/2010
Seahawk Drilling, Inc.	SAP AMERICA, INC.	SAP LICENSING AGREEMENT CONTRACT START DATE:12/18/2009
Seahawk Drilling, Inc.	SAP AMERICA, INC.	SAP TRAINING SOFTWARE CONTRACT START DATE:02/04/2010
Seahawk Drilling, Inc.	SAVAGE DESIGN	COMPANY WEBSITE DESIGN AND MAINTENANCE CONTRACT START DATE:02/12/2009
Seahawk Drilling, Inc.	SAVAGE DESIGN	FLEXSCAPE HOSTING AGREEMENT FOR WEBSITE CONTRACT START DATE:07/31/2009
Seahawk Drilling, Inc.	SCHLUMBERGER INFORMATION SOLUTIONS	SATELLITE CONNECTIVITY SERVICES FOR RIGS, CONTRACT #20245269 CONTRACT START DATE:07/20/2009
Seahawk Drilling, Inc.	SCHLUMBERGER INFORMATION SOLUTIONS	EQUIPMENT PURCHASE FOR SATELLITE CONNECTIVITY, CONTRACT #20245310 CONTRACT START DATE:08/09/2009
Seahawk Drilling, Inc.	SCHLUMBERGER INFORMATION SOLUTIONS	SATELLITE CONNECTIVITY SERVICES FOR RIGS, CONTRACT #20245479 CONTRACT START DATE:07/20/2009
Seahawk Drilling, Inc.	SEABROKERS LTD	REPRESENTATION/COMMISSION AGREEMENT FOR SALE OF 2000, 2002, 2502, 2503 CONTRACT START DATE:05/21/2010
Seahawk Drilling, Inc.	SEAHAWK DRILLING DE MEXICO LLC	TRANSFER OF SEAHAWK 2505 FROM PRIDE OKLAHOMA TO SEAHAWK DRILLING LLC CONTRACT START DATE: 09/29/2009
Seahawk Drilling, Inc.	SEAHAWK DRILLING DE MEXICO LLC	TRANSFER OF SEAHAWK 3000 FROM PRIDE OKLAHOMA TO SEAHAWK DRILLING LLC CONTRACT START DATE:09/29/2009
Seahawk Drilling, Inc.	SEAMAR DIVERS, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT

Seahawk Drilling, Inc.	SEATRAX SERVICES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SEATRAX, INC	STORAGE SERVICES AGREEMENT AND RELATED AMENDMENTS CONTRACT START DATE: 02/09/2011
Seahawk Drilling, Inc.	SECON	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SHER & BLACKWELL	LEGAL SERVICES CONTRACT START DATE:01/19/2010
Seahawk Drilling, Inc.	SHER GARNER CAHILL RICHTER KLEIN & HILBERT LLC	LEGAL SERVICES CONTRACT START DATE:01/19/2010
Seahawk Drilling, Inc.	SIGNAL INTERNATIONAL, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SINOR ENGINE COMPANY, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SIRIUS SOLUTIONS, LLLP	REVIEW AND DOCUMENTATION OF INTERNAL CONTROL OVER FINANCIAL REPORTING CONTRACT START DATE:04/28/2009
Seahawk Drilling, Inc.	SITRICK AND COMPANY	ENGAGEMENT LETTER SERVICES AGREEMENT CONTRACT START DATE:12/22/2010
Seahawk Drilling, Inc.	SMIT SALVAGE B.V.	TERMINATION NOTICE OF SERVICE AGREEMENT CONTRACT START DATE:09/14/2009
Seahawk Drilling, Inc.	SOLUTION CRANE SERVICESMEXICO S. DE RL DE CV	CRANE SERVICES AND SUPPLIES ON AND OFFSHORE CONTRACT START DATE:05/09/2009
Seahawk Drilling, Inc.	SOUTHERN SPECIALTY & SUPPLY, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SOUTHERN TECHNOLOGY & SERVICES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SOUTHWEST OILFIELD PRODUCTS, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	STUART & ASSOCIATES	LEGAL SERVICES CONTRACT START DATE:01/28/2010
Seahawk Drilling, Inc.	SUNGARD AVAILABILITY SERVICES, LP	VPN, internet, SAN hosting CONTRACT START DATE:08/01/2009
Seahawk Drilling, Inc.	SUPERIOR SUPPLY & STEEL	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	SURVIVAL SYSTEMS INTERNATIONAL, INC.	STORAGE SERVICES AGREEMENT AND RELATED AMENDMENTS CONTRACT START DATE: 02/09/2011
Seahawk Drilling, Inc.	SYNACTIVE, INC.	SAP CONSULTING SERVICES CONTRACT START DATE:02/26/2010
Seahawk Drilling, Inc.	SYNACTIVE, INC.	LICENSE TO USE GUIXT SOFTWARE CONTRACT START DATE:02/25/2010
Seahawk Drilling, Inc.	T BAKER SMITH, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	T3 ENERGY SERVICES	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	TALEN'S MARINE & FUEL, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	TELEMAR USA, LLC	REPAIR MARINE ELECTRONICS CONTRACT START DATE:08/11/2010
Seahawk Drilling, Inc.	TESCO, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	THE NACHER CORPORATION	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	THE STATE OF TEXAS - GENERAL LAND OFFICE	LAND LEASE FOR STORAGE OF SEAHAWK 2008 RIG CONTRACT START DATE:12/01/2009
Seahawk Drilling, Inc.	THE TRITANIUM COMPANY BV	SKILLS XP DATABASE CONTRACT START DATE:05/18/2009
Seahawk Drilling, Inc.	THE ULTIMATE SOFTWARE GROUP, INC.	COMPUTER INTEGRATION TRAINING AND CONSULTING SERVICES CONTRACT START DATE:04/08/2009
Seahawk Drilling, Inc.	THE ULTIMATE SOFTWARE GROUP, INC.	ULTIPRO SOFTWARE, AND AMENDMENTS RELATING TO DATA EXCHANGE SERVICES AND TESTING ENVIRONMENT SERVICES AGREEMENT CONTRACT START DATE:04/08/2009
Seahawk Drilling, Inc.	THOMSON FINANCIAL CORPORATE SERVICES	IR WEBSITE CONTRACT START DATE:02/25/2009
Seahawk Drilling, Inc.	THOMSON FINANCIAL CORPORATE SERVICES	THOMSON SOFTWARE CONTRACT START DATE:02/25/2009
Seahawk Drilling, Inc.	T-MOBILE USA, INC.	CELL PHONE SERVICE CONTRACT START DATE:08/24/2009

		EOV DI ICHTER MEMO GEOMETIE
Seahawk Drilling, Inc.	TODAY IN AMERICA	CONTRACT START DATE:07/08/2010
Seahawk Drilling, Inc.	TOTAL SAFETY U.S., INC	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	TRANSCONTINENTAL GAS PIPE LINE COMPANY, LLC	RELATED TO WYOMING CLEAN UP CONTRACT START DATE:07/20/2010
Seahawk Drilling, Inc.	TRANSCONTINENTAL GAS PIPE LINE COMPANY, LLC	SETTLEMENT AGREEMENT RE: WYOMING DAMAGES CONTRACT START DATE:07/29/2010
Seahawk Drilling, Inc.	TRIANGLE WASTE SOLUTIONS, LP	VACUUM AND CLEANING SERVICES CONTRACT START DATE:11/05/2009
Seahawk Drilling, Inc.	TRIDRILL, INC.	NDT / TUBULAR INSPECTION CONTRACT START DATE:06/15/2009
Seahawk Drilling, Inc.	UBS LOAN FINANCE LLC	REVOLVING CREDIT AGREEMENT CONTRACT START DATE:08/04/2009
Seahawk Drilling, Inc.	UHY ADVISORS TX, LLC	VULNERABILITY ASSESSMENT CONTRACT START DATE:11/22/2009
Seahawk Drilling, Inc.	UHY ADVISORS TX, LLC	IT PROCESS DOCUMENTATION SERVICES CONTRACT START DATE:03/25/2010
Seahawk Drilling, Inc.	UHY ADVISORS TX, LLC	INTERNAL AUDIT CONTRACT START DATE:02/27/2009
Seahawk Drilling, Inc.	UNITED MARINE SHIPYARD, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	US UNDERWATER SERVICES, LP	NDT UNDERWATER WELDING UNDING SERVICES CONTRACT START DATE:06/01/2009
Seahawk Drilling, Inc.	VENTURE TRANSPORT, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	VERIZON WIRELESS	ASSIGNMENT OF LIABILITY FROM PRIDE TO SEAHAWK CONTRACT START DATE:05/27/2009
Seahawk Drilling, Inc.	VERIZON WIRELESS	TELECOMMUNICATIONS SERVICES AGREEMENT CONTRACT START DATE:01/25/2010
Seahawk Drilling, Inc.	WADLEIGH OFFSHORE EQUIPMENT SOLUTIONS	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	WAKHARIYA & WAKHARIYA	ADVICE ON TECHNICAL SUPPORT DOCUMENTS CONTRACT START DATE:08/19/2010
Seahawk Drilling, Inc.	WALZEL TECHNICAL SERVICES, INC.	NDE, INSPECTIONS CONTRACT START DATE:06/02/2009
Seahawk Drilling, Inc.		LEGAL SERVICES CONTRACT START DATE:04/14/2010
Seahawk Drilling, Inc.	WAYPOINT BUSINESS SOLUTIONS, LP	CONSULTING SERVICES AGREEMENT
Seahawk Drilling, Inc.	WILD WELL CONTROL, INC.	MARINE SÁLVAGE SERVICES AND ADDENDUM RENEWING SERVICES AGREEMENT CONTRACT START DATE:06/01/2009
Seahawk Drilling, Inc.	WILL DUETT	TANK CLEANING AND COATING CONTRACT START DATE:06/23/2009
Seahawk Drilling, Inc.	WORKPLACE STAFFING SOLUTIONS, LLC	SERVICE CONTRACT MASTER SERVICE AGREEMENT CONTRACT START DATE:01/25/2011
Seahawk Drilling, Inc.	хенох	EQUIPMENT LEASE AGREEMENT CONTRACT START DATE:07/14/2006
Seahawk Drilling, Inc.	XEROX	EQUIPMENT LEASE AGREEMENT CONTRACT START DATE:01/24/2005
Seahawk Drilling, Inc.	XEROX	PRINTER AND FAX MACHINES CONTRACT START DATE:12/01/2009
Seahawk Drilling, Inc.	ZURICH AMERICAN INSURANCE COMPANY	INDEMNITY FOR BONDS RE: HACIENDA CONTRACT START DATE:06/23/2009
Seahawk Global Holdings LLC	NATIXIS	JOINDER AGREEMENT CONTRACT START DATE:09/03/2009
Seahawk Global Holdings LLC	NATIXIS	ASSIGNMENT OF INSURANCE CONTRACT START DATE:09/29/2009
Seahawk Global Holdings LLC	NATIXIS	FINANCIAL AGREEMENT GUARANTY ON REVOLVING CREDIT AGREEMENT, INCLUDING 1ST, 2ND, 3RD, AND 4TH AMENDMENTS TO AGREEMENT CONTRACT START DATE: 8/04/2009

SCHEDULE 2.1(c) TRANSFERRED EQUIPMENT

Vendor	Type of Equipment	City	State
Access Oil Tools (owned by Forum Oilfield Technologies)	Jaw Assemblies (Access Oil Tools)	Broussard	LA
A&L Repair Services, LLC	Pipe Spinner	Broussard	LA
Angel Air Repair & Specialty Co., Inc.	Lifting Equipment, Air Hoist, Man Rider Equipment.	New Iberia	LA
CAD Control Systems, Inc.	BOP Closing Units and Control Panels	Broussard	LA
Church Energy Services Ltd.	BOPs and Ram Blocks	Houston	TX
Don Abney, Inc.	Electrical Equipment, DC Motors, Generators	Patterson	LA
Evans Enterprises Inc.	Generator(s)	Oklahoma City	ОК
Global Mfg. of Acadiana, Inc.	Subs - IBOP Valves	Lafayette	LA
Globaltech Motor & Controls Inc.	Electric Motors	Pasadena	TX
Hannon Hydraulics, Inc.	Jacking and Hydraulic Equipment.	Houston	TX
HL Haley Enterprises	Motor-Generator Set (Letourneau)	Longview	TX
Master Rig International LLC	Drawworks System	Houston	TX
Master Rig International LLC	Drawworks System	Houston	TX
Max Access Inc.	Spider Basket	Houston	TX
National Oilwell Varco	Top Drive TDS 11	Houston	TX
National Oilwell Varco	Top Drive PS 500	New Iberia	LA
Wadleigh Industries Inc. DBA Offshore Equipment Solutions	Cranes, Boom Sections, Brakes, Clutch Shoes, etc.	Slidell	LA.
Power Products Service Co, Inc.	Deepwell Pumps	Morgan City	LA
Quality Oil Tools Inc.	Ball Valves, Gate Valves, Choke & Kill System Equipment	Jennings	LA
Clydeunion S&N Pumps, Inc.	Deepwell Pump(s)	Houston	TX
Seatrax, Inc.	Crane Boom Sections	Houston	TX
Survival Systems International, Inc.	Life Capsules	Kenner	LA
T3 Energy Services, Inc.	BOPs, Ram Blocks, BOP Accessories	Houma	LA
T3 Energy Services, Inc.	Ball Valves, Gate Valves, Choke & Kill System Equipment	Jennings	LA
T-3 Energy Services Mexico, SDRLVCV	BOPs	H. Cardenas	TAB, MX
S.T.S. (Parent Company, Energy Services Intl.)	Various types of Drilling Equipment	Houma	LA.
VMI Marine, Inc. (Parent Company Energy Services Intl.)	Letourneau Crane parts	Vicksburg	MS
MAD LTD	Plates and Life Boat Davit	Beaumont	TX
Integra Cert, LLC	Life Boats	New Iberia	LA
Houma Armature Works & Supply, Inc.	Traction Motor	Houma	LA
Bohen Crane & Equipment Repair, LLC	Crane Booms	La Marque	TX
Tri Drill Services, Inc.	Drill Collars and Pipe	Broussard	LA
Seahawk Drilling, LLC - Broussard, LA Yard	Equipment spares for rig fleet	Broussard	LA
Seahawk Drilling de Mexico, LLC, Carmen, Mexico Yard	Equipment spares for rig fleet	Ciudad del Carmen	Campeche, MX

SCHEDULE 2.1(d) TRANSFERRED ROLLING STOCK

Make/Model	Year	VIN#	Description Car, Truck, Other	Location	Owned or Leased
Chevrolet Express Van	2008	1GNFG154681222949	Van	Carmen, Campeche	Owned
GMC 1500 Sierra	2007	2GTEC13J071613547	Truck	Louisiana	Owned
GMC 1500 Sierra	2005	1GTEC19T75Z354732	Truck	Louisiana	Owned
24' x 8' Cargo Trailer	2002	47ZAB2A222X017189	Trailer	Houston, TX	Owned
20' Gooseneck	1998	None	Trailer	Houston, TX	Owned
BBQ Trailer	1996	TR180413	Trailer	Houston, TX	Owned
Ford F350	2006	1FTWW30P46EB78819	Truck	Houston, TX	Owned
Ford F350	2006	1FTWW30P46EB74810	Truck	Houston, TX	Owned
Utility Trailer	2006	4R7BU16286T069055	Trailer	Sonora, TX	Owned
Ford F350	1992	2FTJW36M2NCA10124	Truck	Sonora, TX	Owned
Ford F350	1992	2FTJW36M9NCA13215	Truck	Sonora, TX	Owned
Ford F350	1992	2FTJW36M8NCA16736	Truck	Sonora, TX	Owned
Ford F250	2006	1FTSW20P26EC63683	Truck	Houston, TX	Owned

SCHEDULE 2.2(c) EXCLUDED CONTRACTS

- 1. Third Amended and Restated Employment/Non-Competition/Confidentiality Agreement of Randall D. Stilley, effective as of June 25, 2010.
- 2. Employment/Non-Competition/Confidentiality Agreement of James R. Easter, effective as of May 10, 2010.
- 3. Employment/Non-Competition/Confidentiality Agreement of William C. Hoffman, effective as of August 31, 2009.
- 4. Second Amended and Restated Employment/Non-Competition/Confidentiality Agreement of Alejandro Cestero, effective as of August 26, 2009.
- 5. Employment/Non-Competition/Confidentiality Agreement of Raymond Gonzales, effective as of March 22, 2010.
- 6. Employment/Non-Competition/Confidentiality Agreement of Robert E. Moore, effective as of May 10, 2010.
- 7. Separation Agreement of Steven A. Manz, effective as of March 23, 2010, as amended.
- 8. Seahawk Drilling, Inc. 2009 Long Term Incentive Plan
- 9. Seahawk Drilling, Inc. Senior Management and Key Employee Incentive Program
- 10. Seahawk Drilling, Inc. Amended and Restated Change In Control Severance
- 11. Seahawk Drilling, Inc. Annual Incentive Plan (AIP)
- 12. Health and Welfare Benefits, as described in Seahawk Drilling, Inc.'s 2010 Employee Benefits Guide
- 13. The information in Schedule 2.2(l) Excluded Insurance Benefits is hereby incorporated by reference into this Schedule 2.2(c) as though set forth in its entirety in this schedule.
- 14. Any contracts excluded at the election of Purchasers in accordance with Section 2.3(a) of the Asset Purchase Agreement

SCHEDULE 2.2(e) EXCLUDED INSURANCE BENEFITS

- 1. Any and all insurance benefits and Claims made by Sellers prior to the Closing or arising from events known to Sellers and first occurring prior to the Closing and related to the Seahawk 3000.
- 2. Any and all insurance benefits and Claims arising from or related to the loss of the Pride Wyoming Rig.
- 3. Any and all insurance benefits and Claims arising from or related to the Oil Petroleum Act claim against British Petroleum.
- 4. Any and all insurance benefits and Claims arising from or related to all matters in the summary of contingent losses recognized under ASC 450 in accordance with GAAP (the "FASS Reports") copies of which are attached hereto as Annex 2.2(e).

SCHEDULE 2.2(k) EXCLUDED CLAIMS AND WARRANTIES

Any and all Claims and warranties arising from or related to each of the following:

- 1. Bankruptcy Claim due from Argo Partners related to the United Capitol Insurance Company claim (estimated net book value: \$28,875.00).
- 2. Chapter 11 Claim filed in the United States Bankruptcy Court in the Western District of Louisiana East Cameron Partners, LP Case No. 08-51207.
- 3. Tax Refund State of Louisiana (estimated net book value: \$185,000.00).
- 4. Tax Refund State of Texas (estimated net book value: \$10,000.00).
- 5. Contract dispute filed in the International Centre for Dispute Resolution (ICDR) Blake International USA Rigs, LLC Case No. 50 198 T 00410 10.
- 6. Claims and warranties related to the Seahawk 3000 Rig made by Sellers prior to the Closing or arising from events known to Sellers and first occurring prior to the Closing.
- 7. The Pride Wyoming Rig.
- 8. The Oil Petroleum Act Claim against British Petroleum.
- 9. Claims against Pride International, Inc. (estimated net book value: \$50,000,000 or more, plus additional causes of action)
- 10. Norman Churchwell v. Friede Goldman Ltd., et al; In the matter of the liquidation of United Capital Insurance Company (estimated net book value: \$35,000).
- 11. Motion Granting Summary Judgment in the Complaint of Pride Offshore, Inc., as owner of the J/U Pride Wyoming for exoneration from, or alternatively, limitation of liability, related to the Century Exploration Claim Civil Action No. H-08-3109.
- 12. The information in Schedule 2.2(e) Excluded Insurance Benefits is hereby incorporated by reference into this Schedule 2.2(k) as though set forth in its entirety in this schedule.

SCHEDULE 2.2(1) EXCLUDED LEASED REAL PROPERTY

- 1. Office Lease between MS Crescent Greenway Plaza SPV, LLC and Seahawk Drilling, Inc., for use of office space located on the 27th and 28th floors of 5 Greenway Plaza, Houston, TX 77090 effective as of July 27, 2009.
- 2. Sublease Agreement between Pogo Producing Company LLC and Seahawk Drilling, Inc., for use of office space located on the 27th and 28th floors of 5 Greenway Plaza, Houston, TX 77090, effective as of July 27, 2009.
- 3. Lease Agreement entered into by V.J. Caro, Jr. and Rhondalyn Caro and Seahawk Drilling LLC, for use of certain property located at 1198 Barrow Street, Houma, LA 70360, effective as of April 11, 2010.
- 4. Lease Agreement between Petro Real Estate Rentals, L.L.C. and Seahawk Drilling LLC, for use of the premises at 1320 Petroleum Parkway, Broussard, Louisiana 70518, effective as of October 23, 2009.
- 5. Hunting Lease Agreement between Rose Mary W. Jones and Seahawk Drilling LLC dated February 19, 2010.
- 6. Hunting Lease Agreement between Lesa Whitehead and Seahawk Drilling LLC dated February 19, 2010.
- 7. Hunting Lease Agreement between W. E. Whitehead and Seahawk Drilling LLC dated February 17, 2010.

SCHEDULE 2.8(a)(vii) EXCLUDED LEGAL PROCEEDINGS

All matters identified in the FAS 5 Reports.

The matters on this Schedule 2.8(a)(vii) shall be Retained Liabilities.

SCHEDULE 2.9 TRANSFERRED RIG CASUALTIES

RIG	CASUALTY LOSS AMOUNT	RIG LOSS VALUE
U.S. Flagged Vessels	10.149 10.149	
Seahawk 2003	\$500,000	\$1,250,000
Seahawk 2500	\$500,000	\$1,250,000
Seahawk 2502	\$500,000	\$1,250,000
Seahawk 2504	\$3,000,000	\$1,250,000
Seahawk 2602	\$3,000,000	\$13,000,000
Vanuatu Flagged Vessels		
Seahawk 800	\$500,000	\$1,000,000
Seahawk 2000	\$500,000	\$1,000,000
Seahawk 2001	\$3,000,000	\$7,000,000
Seahawk 2002	\$500,000	\$1,000,000
Seahawk 2004	\$3,000,000	\$7,000,000
Seahawk 2005	\$500,000	\$1,250,000
Seahawk 2006	\$500,000	\$1,250,000
Seahawk 2007	\$3,000,000	\$7,000,000
Seahawk 2008	\$500,000	\$1,000,000
Seahawk 2501	\$500,000	\$1,250,000
Seahawk 2503	\$500,000	\$1,000,000
Seahawk 2505	\$3,000,000	\$1,250,000
Seahawk 2600	\$3,000,000	\$13,000,000
Seahawk 2601	\$44,000,000	\$18,000,000
Seahawk 3000	\$44,000,000	\$20,000,000

SCHEDULE 4.4 NONCONTRAVENTION

- 1. Drilling Contract by and between Seahawk Drilling LLC and ENI for the Seahawk 2001, with a contract start date of February 18, 2011 through March 20, 2011.
- 2. Drilling Contract by and between Seahawk Drilling LLC and Walter Oil & Gas Company for the Seahawk 2004, with a contract start date of January 22, 2001 through March 12, 2011.
- 3. Drilling Contract by and between Seahawk Drilling LLC and Walter Oil & Gas Company for the Seahawk 2004, with a contract start date of March 13, 2011 through May 12, 2011.
- 4. Drilling Contract by and between Seahawk Drilling LLC and Castex Energy, Inc. for the Seahawk 2007, with a contract start date of December 9, 2010 through February 15, 2011.
- 5. Drilling Contract by and between Seahawk Drilling LLC and Tarpon Offshore Ventures for the Seahawk 2600, with a contract start date of November 22, 2010 through February 23, 2011.
- 6. Drilling Contract by and between Seahawk Drilling LLC and LLOG Exploration Company, L.L.C. for the Seahawk 2600, with a contract start date of February 24, 2011 through March 21, 2011.
- 7. Drilling Contract by and between Seahawk Drilling LLC and Chevron for the Seahawk 2601, with a contract start date of November 13, 2010 through February 5, 2011.
- 8. Drilling Contract by and between Seahawk Drilling LLC and Chevron for the Seahawk 2601, with a contract start date of February 6, 2011 through August 5, 2011.
- 9. Drilling Contract by and between Seahawk Drilling LLC and Arena Offshore LLC for the Seahawk 2602, with a contract start date of December 3, 2010 through February 14, 2011.
- 10. Drilling Contract by and between Seahawk Drilling LLC and Arena Offshore LLC for the Seahawk 2602, with a contract start date of February 15, 2011 through March 30, 2011.
- 11. Drilling Contract by and between Seahawk Drilling LLC and Badger Oil Corporation for the Seahawk 3000, with a contract start date of January 13, 2011 through February 11, 2011.
- 12. Drilling Contract by and between Seahawk Drilling LLC and Arena Offshore LLC for the Seahawk 3000, with a contract start date of February 12, 2011 through June 12, 2011.
- 13. The information in Schedule 2.1(b) Assigned Contracts is hereby incorporated by reference into this Schedule 4.4 as though set forth in its entirety in this schedule.

SCHEDULE 4.5 ASSET TITLE, MAINTENANCE AND CONDITION

The following maintenance has <u>not</u> been performed on the specified rigs:

- 1. Seahawk 800: Complete SPS/UWILD, minor steel repairs based on gauging to be performed, complete equipment inspections, minor repairs to drilling equipment, overhaul well control equipment and reinstall top drive.
- 2. Seahawk 2000: Complete SPS/UWILD, perform major steel work on the legs, mat, hull, sub structure, piping, and heliport; overhaul drilling, well control, and main power systems; complete equipment inspections; reinstall top drive.
- 3. Seahawk 2002: Complete SPS/UWILD, perform major steel work on the legs, mat, hull, sub structure, piping, and heliport; overhaul drilling, well control, and main power systems; complete equipment inspections; reinstall top drive.
- 4. Seahawk 2003: Complete UWILD, minor steel repairs based on gauging to be performed, repair Keel Kool system, refurbish choke manifold, complete equipment inspections, and reinstall top drive.
- 5. Seahawk 2005: Complete SPS/UWILD, perform major steel work on the legs/pin holes, mat, hull, substructure, piping, and heliport; complete equipment inspections; overhauls of drilling, well control and main power systems; reinstall top drive.
- 6. Seahawk 2006: Complete SPS/UWILD, minor steel repairs based on gauging to be performed, overhauls of some drilling equipment, complete equipment inspections, and install top drive from stock.
- 7. Seahawk 2008: Complete SPS/UWILD, perform major steel work on the legs/pin holes, mat, hull, sub structure, piping, and heliport; overhauls of drilling, well control and main power systems; complete equipment inspections; replace life capsules; refurbish top drive.
- 8. Seahawk 2500: Complete SPS/UWILD, steel work on substructure/tanks, repair jacking system, complete equipment inspections, and reinstall top drive.
- 9. Seahawk 2501: Complete UWILD and dry-dock, perform steel work on the legs/pin holes, decking, and mat; overhauls of well control and main power systems; complete equipment inspections.
- 10. Seahawk 2502: Complete UWILD, minor steel repairs, partial quarters repair, replace one crane, complete equipment inspections, reinstall top drive.
- 11. Seahawk 2503: Complete SPS/UWILD, perform major steel work on the legs/pin holes, mat, hull, sub structure, piping, and heliport; complete equipment inspections; overhauls of drilling, well control and main power systems; reinstall top drive.

- 12. Seahawk 2504: Complete UWILD, commence gauging, repair jacking system yokes, minor steel repairs to tanks and leg inserts, repair choke manifold, and complete equipment inspections.
- 13. Seahawk 2505: Complete SPS/UWILD, minor steel repairs based on gauging to be performed, complete equipment inspections, and overhauls of well control equipment.

SCHEDULE 4.6 FINANCIAL STATEMENTS

See attached

Operating Days Available Days Utilization Utilization Average Daily Drilling Revenues Average Daily Available Rig Operating Costs Shore-based Operating Costs	EBITDA	Income Tax Expense (Benefit) Income (Loss) from Continuing Operations	Other Income (Expense) Interest Income (Expense) Other Income (Expense) Income Before Income Taxes	Depreciation & Amortization General & Administrative (Gain) Loss on Sale of Fixed Assets Impairment of Fixed Assets Operating Income (Loss)	Costs and Expenses Operating Costs Wages & Burden Repairs & Maintenance / Supplies Rental Expenses Other Operating Costs Mob/Demob/Reimbursable Expense Other Direct Costs Total Operating Costs	Seahawk Drilling, Inc. Consolidated Income Statement For the year ended December 31, 2010 (Dollar amounts in thousands) (Unaudited) Operating Revenue Mob/Demob/Reimbursable Revenue Total Revenues
1831.8 7300.14 0.250926695 0.633450638 36834.94505 16646.23861 15551.9705	-104819.3402	-60848.36563 -97937.31041	-3588.27042 -538.00327 -158785.676	50378.06542 40448.47947 -4067.26019 28171.11765 -154659.4024	53674 22690 -246.74311 19254 15946.64854 10201.09457 121519	68686 13104 81790

SCHEDULE 4.7 UNDISCLOSED LIABILITIES

All matters identified in (a) the summary of unrecognized tax benefits recognized under ASC 450 in accordance with GAAP (the "FIN48 Reports") copies of which are attached hereto as Annex 4.7 and (b) the FAS5 Reports.

The matters on this Schedule 4.7 shall be Retained Liabilities.

SCHEDULE 4.8 LEGAL COMPLIANCE

All matters identified in the FAS5 Reports.

The matters on this Schedule 4.8 shall be Retained Liabilities.

SCHEDULE 4.9 LITIGATION PROCEEDINGS

All matters identified in the FAS5 Reports.

The matters on this Schedule 4.9 shall be Retained Liabilities.

SCHEDULE 4.10 PERMITS

Permits relating to each U.S. Flag Vessel listed in Schedule 2.1(a)

- 1. United States Coast Guard Certificate of Inspection
- 2. United States Coast Guard Certificate of Financial Responsibility (Water Pollution)
- 3. United States Coast Guard Certificate of Documentation
- 4. ABS Load Line Certificate
- 5. ABS Classification Certificate
- 6. ABS Cargo Gear (Cranes) / Statement of Fact
- 7. ABS International Tonnage Certificate
- 8. United States Coast Guard Shipboard Oil Pollution Emergency Plan
- 9. United States Coast Guard and ABS Operations Manual
- 10. EPA NPDES Vessel General Permit

Permits relating to each Vanuatu Flag Vessel listed in Schedule 2.1(a)

- 11. Certificate of Registry
- 12. Radio Station License
- 13. Safe Manning Certificate
- 14. ABS Mobile Offshore Drilling Unit Safety Certificate
- 15. ABS International Oil Pollution Prevention Certificate
- 16. ABS International Air Pollution Prevention Certificate
- 17. ABS International Sewage Prevention Certificate
- 18. ABS Cargo Gear (Cranes) / Statement of Fact
- 19. ABS International Tonnage Certificate
- 20. Certificate of Insurance (Bunker Insurance Certificate)

- 21. Shipboard Oil Pollution Emergency Plan Vanuatu or ABS
- 22. ABS Operations Manual
- 23. United States Coast Guard Certificate of Financial Responsibility (Water Pollution)
- 24. EPA NPDES Vessel General Permit

SCHEDULE 4.11(a) RIG DOCUMENTATION AND MAINTENANCE

The following rigs are cold stacked and are out of status:

- 1. Seahawk 800
- 2. Seahawk 2000
- 3. Seahawk 2002
- 4. Seahawk 2003
- 5. Seahawk 2005
- 6. Seahawk 2006
- 7. Seahawk 2008
- 8. Seahawk 2500
- 9. Seahawk 2501
- 10. Seahawk 2502
- 11. Seahawk 2503
- 12. Seahawk 2504
- 13. Seahawk 2505

SCHEDULE 4.12 EMPLOYEE AND LABOR MATTERS

See attached

			C	as	е	11	-2	00	89)	Do	DC	um	nei	nt	19	-2		-il	ed	in	Τ	XS	B
	Style of Case			RICHARD HILEMAN VS. SEAHAWK								BADI OCTAVIO AVILA V GOMPS, Y	MEXICO DRILLING LIMITED	LEANDRO PERALTA vGOMPS Y	MEXICO DRILLLING LIMITED Y	GILBERTO RAMOS MONTEJO V.	GOMPS Y MEXICO DRILLING	LIMITED Y PRIDE INTERNATIONAL,	Jesus Ortiz v. GOMPS, Mexico Drilling	Limited, Y Pride International, Inc.	Enrique Augusto de la Cruz Aragon v.	Gomps Y Pride Central America	DANIEL ADRIAN DIAZ VS. PRIDE	Ana Betsabe Morales Sandria vs.
Contingency	Category	Employment/Labor	Employment/Labor	Employment/Labor	Employment/Labor	Employment/Labor	Employment/Labor	Employment/Labor	Employment/Labor	Employment/Labor	Employment/Labor		Employment/Labor		Employment/Labor			Employment/Labor		Employment/Labor	and a same for the same party and the same party of the same party	Employment/Labor	Employment/Labor	Employment/Labor
	e Vessel Name	38064 OFFICE	40324 OFFICE	40557 OFFICE	38031 PRIDE MEXICO	39686 PRIDE MEXICO	39845 PRIDE MICHIGAN	37769 PRIDE SOUTH SEAS	39749 PRIDE UTAH	37795 RIG 1003E	12		13		68			68		68		89	85	30
	Accident Date	3806	4032	405	3803	3968	3984	3776	3974	3778	38012		39413		39489			39489		39489		39489	37785	38030
	First Name	CLAUDIA	MARIA	RICKY	SILVERIO MENDOZA	TRACY	DAVID	MELCHOR	JEREMIAH	JOSE	ISMAEL		BADI		LEANDRO			GILBERTO		JOSE		ENRIQUE AUGUSTO	DANIEL	ANA
	Last Name	CASTELLANOS	RODRIGUEZ	HILEMAN	SANTOS	JACKSON	NELSON	SANTOS	LARKINS	HERNANDEZ	VELAZQUEZ		AVILA		PERALTA		-	MONTEJO		ORTIZ	The second secon	ARAGÓN	DIAZ	SANDRIA
	Claim Number	042690-003473-01 CASTELLANOS	060799-010033-01 RODRIGUEZ	060799-010252-01 HILEMAN	042690-003472-01 SANTOS	042690-008137-01 JACKSON	042690-009417-01 NELSON	042690-003469-01 SANTOS	042690-008508-01 LARKINS	042690-003466-01 HERNANDEZ	042690-005634-01 VELAZQUEZ	The state of the s	042690-006874-01 AVILA	minimizer in the state of the s	042690-006875-01 PERALTA	WARRIE TO THE TOTAL THE TOTAL TO THE TOTAL TOTAL TO THE T		042690-006876-01 MONTEJO		042690-006877-01 ORTIZ	The second secon	042690-006878-01 ARAGÓN	060799-010157-01 DIAZ	060799-010158-01 SANDRIA

SCHEDULE 4.13 EMPLOYEE BENEFITS

- 1. Third Amended and Restated Employment/Non-Competition/Confidentiality Agreement of Randall D. Stilley, effective as of June 25, 2010.
- 2. Employment/Non-Competition/Confidentiality Agreement of James R. Easter, effective as of May 10, 2010.
- 3. Employment/Non-Competition/Confidentiality Agreement of William C. Hoffman, effective as of August 31, 2009.
- 4. Second Amended and Restated Employment/Non-Competition/Confidentiality Agreement of Alejandro Cestero, effective as of August 26, 2009.
- 5. Employment/Non-Competition/Confidentiality Agreement of Raymond Gonzales, effective as of March 22, 2010.
- 6. Employment/Non-Competition/Confidentiality Agreement of Robert E. Moore, effective as of May 10, 2010.
- 7. Separation Agreement of Steven A. Manz, effective as of March 23, 2010, as amended.
- 8. Seahawk Drilling, Inc. Change of Control Severance Plan.

The matters on this Schedule 4.13 shall be Retained Liabilities.

SCHEDULE 4.14 ENVIRONMENTAL MATTERS

None

SCHEDULE 4.15 TAX MATTERS

1. Tax Sharing Agreement between Seahawk Drilling, Inc. and Pride International, Inc. dated August 4, 2009

The matters on this Schedule 4.15 shall be Retained Liabilities.

SCHEDULE 4.16 CERTAIN DEVELOPMENTS

- 1. Letter Agreement between Seahawk Drilling, Inc. and Simmons & Company International dated October 7, 2010.
- 2. Agreements relating to the sale of the Seahawk 2505 to Essar, which will not close because Essar did not receive the bid.
- 3. Confidential Receipt, Release, Indemnity and Settlement Agreement between British Petroleum entities and Seahawk Drilling, Inc. effective as of January 14, 2011.
- 4. Additional Pride Consideration Agreement between Seahawk Drilling, Inc., Pride International, Inc., Petroleum Supply Company and British Petroleum entities effective as of January 14, 2011.
- 5. Agreements relating to debtor-in-possession financing entered into with DE Shaw.
- 6. Settlement Agreement between Seahawk Drilling, Inc. and Tennessee Gas Pipeline Company effective as of October 12, 2010.

SCHEDULE 5.12 ADVERSE CHANGES OF PURCHASERS

None

SCHEDULE 10.1(g) CONSENTS

None

ANNEX 2.2(e) FAS5 REPORTS

See attached (Parts 1 and 2)

se	11-2	2008	9	STEPHEN	Do	Cl	um	er	10#	19-3	File	ed §	in T	X\$I	В	on Q	2/	12/11	Pa asor	ıge	2 of
wak Bon Esstowel			CRAR	CRAM				MARTH, DISIERE, JEFFERSON AND WISDOM				BRACEWELL & GULLANI, LIP DANIELS		HGBERT		aworon			BURR & SHITH SHITH CRA ABODADOS VALDEZ	MAARTRE, DISIERE, JEFFERSON AND KROGER WYSDOM	
			M. P. BRACEWELL & GRUMM LIP			CBS ABOGADOS		MARTH, DISIER	EDISON, MCDOWELL & HETHERMOTON		ALEX CONTROL	- 1		FEHORE, NC. SHER GARKER.		ONE NO KRAH MILLER			E. Mr. BUTR & SMITH	MARTER, DISIEF	1
Brote of Gree			Robert Auton, et al. Vitages de Company, James B. Basall v. Phillos de Company, Inc.	War or Bridge, of M. v. Phillips 55 Company				:			WAVANGE STATE OF THE PARTY OF T			MACHAEL FRANCOIS VÉ, PRIDE OFFEHORE, INC. AND AGPÉR RISVRANCE COMPANY		RODHEY EVANS V. PRIOE OFFINORE, INC.			1946 Streberge vs. Pide Mernefors		
se Leest Counsel	This is the brandence of Contact Miss Services (CRS) at Sparal develor. CRS has have contact with Claimark Course fragefully operated sefement of 1614 sales. Claimarks coursel has defend any empores util rate the Publicys. His medical transferred is being offered for the default of the second services.	The state of the advances the new beat seams. We control to mention with loose control to mention on the control of the advances the advances of the advances of the advances the advances the advances the advances the advances the advances to the advances the adva	mediation at this time is remote. Please see claim 002210-01 reporting the stoke of the subset of this first.	Plane see claim 0022(9-0) regarding the status of the patients of the patients of the section of	Per Libra O'Stelly of Prides, the nubbranes suited between Price and 8P for 1944s. A request was submitted to Mary to secure a copy of the submitted to the secure of even eventual. The claim of by cheed to be rest factor to experience.	Seatherk wealth the meeting to set a data and how to relef a porord in the instituto Mesteano Del Sepuro Social Related seth the Feleshad Documents.	All. Redigious has retained course and school that the behavior as he had been discriminated against for version reason (houlding termination). SOI has estained aid whereon for the defense at files in mitter, A position attainment has been find with the ECO's. Available the ECO's.	response to determine what type of investigation they will conduct. The option for a mediation of this time has been taked out.	The archadon panel has been chosen. The telephonic probminery hashing will take place on Fabruary 7th at 250 pm.	Section County for the black of Order and with the grapher black of the black of Order and the black of the for any descriptor, best of the ways of the black of the for the place of the for the black of the black of the black of black of the black of the black of the black of the black of the clark of the black of the black of the black of The clark has been reflect frought the two process set by The clark has been reflect frought the two process set by The clark has been reflect frought the two process set by The clark has been reflect frought the two process set by The clark has been reflect frought the two process set by The clark has been reflect frought the two process set by The clark has been reflect the black of the clark of The clark has been reflect the black of The clark has been reflect to the clark of The clark of the black of the black of The black of the black of the black of The black of the black of the black of The black o	OGC For Mr. Course, (no. 19.1.07). We recoived five divests over the hardest behavior. We recoived five divests over the hardest behaviors company that were behavior and 1711, 421.00 from the "free five were behavior and property and 1711, 421.00 from the Trust paraging the finded Goldman Davierator, guide cheests were seed to Timenty. The divest five the Probed Golden Trust was made of the Priory Promodicula and Prioris Model Golden.	int. Will await the mentaining amount prior to obtaining the claim. Polestial Breach of Contract suit that by Mr. Hencen. Employment contract obligations may be nevel. Coursel	Leave of the condition	and FCE gray to see that Mark. Additional accordance to condition of the formation operating a Bobistatives and basis beginning a good of the condition of the conditional and the second and money about constitution. Additional accordance and the second accordance and the second accordance and the second accordance on the Annicket or as preventible and catherine.	thice per week. It seems that ha is receiving pallative care at the text. An ME may be required for further evaluation of MMI.	Mit. Evens and the coursel have solelated for a response to their wedienest demont. So it sufficiently was serviced to drive source allo sandle to see the sufficient response. The sasts is feromatic for definent, Seediment to be considered for the most thin relations on the second principary.	The ENT in Augusta, CA wanted to see Mr. Make back in B nootes for Austral seating following his movem in Ostober 2010. No requests have been reade for the hearing side of for the follow up with a first firm.	The tried for the refer was pushed hand from 12.71.2010 for 22.2010. Called and 18.8010 for 18.2010 for 18.8010 fo	could in proceed a 2000 March as a construction of the country of	12, Jackson field an application to proceed in forms a proportion of the proceed in forms a proportion of the proceed in forms a proceed. Currently there is no be and on which the process of the proces	Editament conference with Adigs Roy's and for 215 (2011). The control of the many and an end for 21 (2011) for answers were first in control of the control
Correc	FMEXICO		O GULFOFMEXICO UNIN	1				STATE OF TEXAS	ğ		ONPORATE OFFICE UNIN			PFMEXICO P41	Na.	000			0 0	2000	
Total Information Surveyore	ZZBIZ.81 GULF OF MEXICO		0 00150	0 5000	e dute	O MEXICO		0 STATE	24600 0		A400 0	0 CORPC		0 GULFS	o MEXICO	ě	ann o		O MEXICO		
Total Reimbursed By Province Undewrite to Date 10 Date	0		6-6	•					-			•			17 9987				0		
Posential Commission C	730000		00 000					85	a		00041	1		9	3000				310000		
Potential Loss or Danger of w or Danger in bard to	0		278480			0 105443.08		00091	CO19000 52 EAG		7870.1	3000		41000	0000	}			21000		
Part Part Part Part Part Part Part Part	*******		2 0		۰				2		0	•		\$17.100	1745.12				9		
Park Park	3 240115.76	1						-			0	0		102.03	8				*	,	
Advetin Other			0 0	0 0							0	0 02052:		0 21202	200		0 101000	<u></u>	4000 0 2000		
MU PA UM TOOM	100 EC 1010E		1 6	0 0	-			-	, ,					*			-		9		
Reserve Asserve Po	_		000000	00000							0	0005		00090					20000		
Profession Reserve	4E OLIVENS		0 0								0	2500		180000					00058		
Gross Adjusting Other	•		300000	100000							000001	0 00000	0 0000	410000 0 25000					000000		8000
Gross Reserve in Ga	, ,		30000	L			,				88	90000	0000	410000					31,000		
Proposition	2,000		5 5	1	1				1		4	\$	***	000001		 			0000	1	ž
7AF Sumos			Dogste USD	Possible USD							Manager NED	Probable USD	Porebbs USD	Probable USD	į	90	Probable USD		Probable USD		Use Osmonia
Contrigency			Personal Plur	Personal Paley		1-			Employment Called		Commercial Dispute	Connected Prepare	Employmentilebox	AGO Personal tract			FLORIE Personal Intern		CAUS Personal Plan	ACXXX Employment/also	LEXIC Engigements, abo
Pret Herre - Aceldons Date Vessel Herre	800		DISTRIBUTED HOUSE	DISTRIBUTED HOUNES	SOU GAM			-	200000		Oeneno Office.	ORZENIO OFFICE	BIMAN OFFICE	ORZEGZ PRIDE ARIZO		and	acrose serios		SOLIENS	8748	DAZAGA PAIDE MEXIC
Karra	•	моджоо	THE	PUTTON NAMETRA	импрод - мири	CASTELLANOS	Ropmoutz		BLAKE NTERNAT	SEAHAWK DRILL	SEVINAK DEEL	HLEWAN	FRANCOIS	POSTER MICHAEL			MILLER RODHEY	SAWBERGER	KIRK	MCKSON SIVERIO M	POWEU TRACY
Chair Number	- 5		De260002218-01	TTIN TO TAXABLE TO TAX	O POPULATION OF THE POPULATION				BUAN		\$5.00	HEE	PRAY	10-201900-006276		EVANS EVANS	MILL MILL	<u> </u>		AC 2800 000-17-01	900 PON

8

Seahwet Drilling, Inc. P.A.S. - S. Report GZ-01-2011

se 11-2	00	89	CROSSE	OÇ	um	ent (О	-3	Fi	led in TXS	PETER .	3 on () <mark>2</mark> /	12	/11	Page	3 of	60
KROGER	SMITH		клодел	188 188	CRAN	a was	CRAN	į			HUSERT	КОНЦИАК	VALDEZ		эюпоэя	вю поли		
1	GALLOWAY, JOHNSON, TOMPKINS, C. BURR & SMITH CES ABOGADOS		MARTH, DISGRE, JEFFERSON AND WEDOM	BRACEWELL & GALLMAILLE	PRYCEMENT & ORTHWE	ST INTINO FIRM	ATT INVITABLE TIEMSOVER		34.		IC. BIND SHER GARNER		Chranogados		אפראי אוור איני	PLUS W KEAN MALER		
WARN LUEN EARTH EAGUL 4 R. v. PRIOE	gregory moore v. Pride offshore, Inc.								Leonard Castille vs. POI DiBM. Seehank Drifting		Kelth Taylor v. Tesco, hr., Pride International, It Medos Orling, LTO.					AMENA OFFICIAL LE AND ST. PAUL SURPLUS PRES RESURANCE COMPARY VS. ESCHANKE PRES RESURANCE		
Democratic in the process of comparison and democratic community of the process of the JUNEAN IN the same of the process of the JUNEAN IN the same of the process of the JUNEAN IN the same of the process of the the shades of the process of the process of the tent of the process of the process of the tent of the process of the process of the tent to the same of the process of the tent to the tent of tent of the tent of tent of the tent of tent of tent of tent of tent of tent of tent	e authorser is paid. This file will dose once legal connes are and aid. The file possible of the resolution of the Project of several control possible to here received paid of several control to the paid of th	recent CT mystogram show belt the shall be backwards they belt good to the control of the contr	ame is no further novements by Mr. Landon on this scutter cooking to defense coursel. However, defense coursel Meed theil Markes coursely to re-file his conclision.	please of delote have been treatoned and released for Angle forbiding the legal, Orna at histories are most-rel- bones for paying and model to the model from denoting the claim will be closed.	The Wilmans other has angient of Fit Litt. The Enth-(geablands and main has settled in the amount of 13 Mai. The TOPE (CEP has oldered has settled to 15 Mai. The Cebring debit is controlly predicte a MICL. The judge set clearly debit is controlly provides a MICL. The judge set which on the Gentry MICL has near Maint. The judge's inferious and desembles too factor MICL than the property.	Claim for damages by Tarvesses Clas Phelin Company (C) Pauc) and aniwar to Phide Otheron, Inc.'s Company to Pauc) and aniwar to Phide Otheron, Inc.'s Company to Wyoning caused damage to their public before Wyoning caused damage to their popules damage of function Target Stephen Confer in Penal Confer and for Philosophy Located states in vicent in Carlo SECTOCOCOST POLICE Located states in vicent in Carlo SECTOCOCOST POLICE Located states and vicent in Carlo SECTOCOCOST POLICE Located States Section SECTOCOCOST POLICE Located Section	who Whe allowed by Judge Rosenthal on 08/04/2009.	Dougog educa as of OTTOMO in the potent Stevenier to the Prince Wayners, Desirett Connel for Waters in se- rectived or to the presented Connel for Waters in se- claim day Contrast (Connel for Marie Connel demanded demand by Contrast (Connel for Marie Total Contrast of demanded on the Contrast Connel for Marie Total Contrast of contrasts in the best in chicked to TAML In the contrasts in the best in chicked to the contrast of the contrasts in the best of the contrast o	Cadimir deposition in same in concentration of the conference	Lead of the filtering to the second the seco		A William has their compaint/hotics with the OMF in LA. profiles it is required with the Section of the Child has a performed in the Section of the Child has a performed in the Child has been about the Child nection all manners the last and of Decease. A Loud nection all manners the last and Section is confirmed to profit as the distribution of the profit of the Child has a Milkelium to confirme to the Child has a fit for interesting the Child has a section to the Child has a fit for its confidence to the Child has a fit for its confidence to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the Child has a fit for its receivere to the contract of the Child has a fit for its receivere to the child has a fit for the confidence to the contract of the Child has a fit for the child	is the is in the istage of reflet of proofs. The court has district that there are proofs periodic refer. The meeting to conclude to regularite the matters has not been set as	The time for the served COLA increases. The sweet will be seases from \$44,00 to \$170,00. An updated LE-208 will be falled with the DOL to show this increase. The DOL test not lowerided a LE-221.	In Bapters raths was transferred to Costabl find Environment Control of the Contr	A Licinous was restricted to have consent recented the bit markets. Estember incrychelege professe of the Bitmary file was reformed. Consent of preferred between statements of the professe of the between the professe of the professe of the bitmarkets. Consent considerate and files at 2601 professe between the consented and the professe of more and the professe of the site of professe of the professe of the professe of professe of the professe of professe of the professe of the professe professe of the the professe of the profes	This data is now hereafted to Contact field is softeness on Spanish returnation. M. Edit-banks attenting was in control of Contact fields in an infect on seath a must have definement. A U.S.A. would be required. Committy available an injected from Contact field to describe from the propriets has been include towards sentenced (1844) and propriets has been include towards sentenced (1844).	1. Hebert had a 6 day leyadest etcy at a tocal hospital for waterstak in Doesenbar. No is ast for additional traffic in about ay a two biodistays were discovered in the Ingest. His offs were changed. If these did not leystren the
16	25	2	i	5	8	<u> </u>	ž		i i	######################################	wc	D S	,	300	ş	8	55	
O GULF.OF MERICO	D QUIT OF MEXICO D MEXICO	O MEXICO	O OUIF OF MEXICO	ANDORS ON SOLUTION	STORMES & GULF OF MEXICO	o onit of wexico	O GULL OF MIXIOS				0 MEXICO	0 2 2 3 3 4	9	g quif of mexico	SXIII IO MINO	100	OKEN SO THE RESERVE	
	00	60.17.00		-	o o			,			17 SCEED	***************************************			9		8	
00002 0005 00002 0005	41073,43 \$22714.00	00052	20001	200007	000000		350000				000057		3	1351644 BE 1301644			8	
1887.04	0 0	o	H 0625	9500	3216256	•	٥		2		106213.4	•		27 64037	s an	80,242	8	
0	71.2448 8C.172780	9 25128	0	0 0 00	0 2009191						16 C0005 G	9		06247.17	1 × 1		1000	
14 14 14 14 14 14 14 14 14 14 14 14 14 1	0 11473 6	72,8411 0	0		0 41292		0				m occ	60			<u> </u>		8 292	
0 10002 04	0 800008	54705	9 6730 pe	1,000,000	79812847						144146.77	8 3 5		47,574			24 446478	1
000001	8000	9	00001		00005E		0		° [200002	. •		0005s	71 MANA 11 THE TAXABLE TO ANNOUNCE TO ANNOUNCE TO ANNUAL		SORCE	
0 0	00007 0	882		0000099	0005191		350000				197000 50000			2 0005	***************************************		97.00	
0 00	0 0 0001 000	٥	a				-				0008			9 0		<u> </u>		
000003 35	000028	20092		\$1/2	19855000		350000				450000			1351884.88	2001	1	8	
00000	1000000 \$20000			7450000	00099001		1000000				0	0005 0007		84 1481127				
neess usp, nus,	Proteable UED 10 Agrade MXN NA.	Procedor USD	2	WWW OSD equipment	Alvon USD NA	de la company					rotable USD	CSU ###		OSD MONOR	<u> </u>			
Property and property of the p			ppe	onmerchal Dispute	Openit Demage	on Desired					Proceed Printy Po		_					
COUTOR PRIDE MICHIES	OTOLOG PRIDE NEW A Personal INLY OCCASO PRIDE SOUTH EMPERMENTANO	1979-07 PRIDE TENNEPARONAL PLAY	102808 PRIDE UTAH E	O MOUM BOILE SOCIED	ON THE PRICE WYOM P	SHOW BORR	PRIDE WYDIA'S				4 35001 04 001010	Annual Manager		2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2				
PANID GOOT	GREGORY 075M	CALVIN 1975	3				98	An Control			жетн ото	144 198		KENNETH 0801				8
	SANTOS		LARKHS	WYOUNG REMO	THE WILLIAMS O	TENNESSEE OAG	ENBRIDGE 042990-009179-00		CASTILLE	TANOR		WELLUMS	HERNANDEZ	CUARK	EMPTIST E	CASTILLE	SULLWAN	HEBERT
NELSON SANDENA SANDENA SANDENA SANDENA	0-2890-007831-01 0-2890-000-005-01	0.0200000000000000000000000000000000000	O Province Company	9-54-19-00-00-2-K	254(1800-048)240	10-921900000211	242690-006176-5	200			H2860-008795-01	10-caccoocacco		D481200-00824				

200

se 11-	200	8	9 Docum	ent 19	3	File	d in TXS	SB	on ()2/1	2/1	1	Pa	ge 4
				:		·····								
				•										
				į										
					No. of the Control of									
					, and a									
. 8	, 1	L	* **.fc t* .	2 25 522	-	* g ±	8 page 15	< x	. 2 2	, ž : \$	5. 1		5 . P	, 115 v
and strongeries declarated of the and strongeries of the strongeries of strongeries of the strongeries of a on 12/17/10 i. He had a port of 10/2/11 with Dr. Fern, FT was 10/2/11 with Dr. FT was 10/2/11 wi	Idi. Hill was brought in for evaluation to OMS by CRS. He was digitorised with metamen special mode in transfer workshood ware not. A MATM was seen tomes. He is to follow to with his personal disposition in his lang my further completation. It was trained and whether the completation.	a published on Dage	we see to to Dr. I allow the evaluation. It is completely the transfer of the second o	leaves, the was to dig 1 of 11 in 11 in 1 in the base get a Ma Received to 10 CP. Received to 10 CP. Rec	al Precions, once ever has quilitie	palent hereta (right variational general transform, Cr. Fork, No. 1 errports, Cr. Fork, No. 4 errports, achievit is follow up 12,22,2010. If this review, No. without restriction in 4-	On Core stated the servery just to the pipe and beam the pipe of the pipe and the pipe and the pipe and the content peaked to the an expension of the pipe and the content peaked to the an expension of the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the the pipe and the the pipe and the pipe and the the pipe and the pipe and the the pipe and the the pipe and the pipe and the the pipe and the the pipe and the pipe and the the pipe and the the pipe and the pipe and the pipe and the pipe and the the pipe and the pipe and the pipe and the pipe and the the pipe and the pipe and the the pipe and the the pipe and the the pipe and the the the pipe and the the the pipe and the the the the the the the the	Ther leaves with coupling up blood. A and by Or, Hermanday at Azadam rot able to get see his pulnorologist exhabit. Involves will be	Cotted evading to the olget lesse with vers early in for a resolved evaluation to solves Gairs in Lafayelia. A callar with formed Lift Schillage. He was sent home to erroral physicien. He was not neesed to	ted to RSTA track he has had an upwer constructed to the constitution to the debuggit me by freed or weller. Mr. Use was brought in sudam Occupational for evaluation. He was a distructed that and instructed to follow with newfork. He was not home.	edde Nis claim for Metad by America T. T. An excepted of documents. This reservices provides	a Occupational Clin Sarr has recovered Snow the enedical	of a fraction to the right frog freque paper is a spectral and M.F. Dernick was placed for following and M.F. Dernick was placed on following and his work. of all that there. The cleric his chose to did not work to M. Investrals will be a second-out.	off respect. He was a found to the Child. He was see the control of the child to th
eried, synoredos condosperer of the brea en 127720 (OTOS2011 with I Jones has follow sediment. His nex sediment fredeng Jones was in sur dot release	in for evaluation to an influence bysight. Air All was sen physician if he ha ares released as died. All was peed	It need to down wa	seet to Dr. 51 Ma update for human is a seed to "Mass" is a seed to "Mass" in the property of the pro- top of the property of the mapping was seed to a seeding prior to a seeding prior to the property of the pro- top of th	toy 11 of his high, of states report on saids Die reserve arrest to \$150f. wated earne mile preding with uppe preding with uppe preding due to some here been filed of	the General Medicals	d southerny on hy wouldown with a will care use at a will care use at on 12.17.2010. It surgery. He had a surgery. He had a surgery as one	Sine survey tool in the picture of the pool and the pool	to Arther tenues with taxed by Dr. Herry was not able to pet a submiddled http. In	of swelling to his re- seer in for a me mer Christ in Lathy and MR SA/Stagh.	A STA Dut he he seed hos condin y food or water. In Occapitional is provided and it	Xed the offer to a ted the RRI (comp \$2010 in Houstone state the settlement ingui involute fo	piers to Academ at this time. Mr. I he right thumb. C	d a fracture to the separated and Mr. follow up in two of at that time. The Mr followers out.	had teeling dazy v medambon. He we med better on 12/30 onel Center. He we medically. He will be te will be condition
Mr. Jones underwe medical and atoms microfridate and e econyle of the right recommended. Mr his home town for prototyped given the completed when the	fr. HE was brough the disproved will rebicke were fr y with the persons and the persons watered is article	74s claim will close	if, Thorp poor was use and to a neural ingress was before considered a point with particle, spanish proposed with a proposed and research if, Pacion cerea in and research and disagraphic and disagraph	where the was on where then his PA tookwas in updat tookwas an updat tookwas an updat so brain the that an arrange of the so brain the that an arrange of the stepsifier was one cursely. Medons a paintiff's deport a paintiff's de	waiting the Lafey recessed, this file	colion at Seaber (1) Mutchel report (4) In Mutchel report (4) Variances. His in Anteresson. His in Anteresson His in Ant	boo Crew stables and the total con- contract placed for contract placed for placed for the place of the total placed for and causing bean sected evaluation. These sected evaluation	indey reports to correlesse was best of outse. Endery w	fr. Fauteur reported e- dement. He was se- tradiana Oceanostorial alan which confirmed it olow with the personal	fr. Use reported to most the property of the forms of sent to According to the forms of the form	A. Gongalus acon 7,500 and comple of GATBS) on 128 was undered to tra	Julys medical in wmain outstanding rom the cellulus in	fr. Perrot sustains off. The froger was got day. He has date with a feet blow up at h MS	At. Weether rept ent in for medical telebort in lattacos tradiana Occupal medical and treate medicaldors to en
		ā	<u></u>	1										
OSKER OF MEXICO		O GALLE OF MEXICO	D GULF OF MEXICO	O OULF OF MEXICO	A COLUMN TO THE PERSON	O GULF OF MEXICO							o our of wexico	QUE OF MEXICS
0			0	•										
9		2 8	8	8		8 8		2 5	8 8				8 8	. 8
		3500	995	0051		0000							3	
°			a					,						
90 809	i	200.00	£ - 54	•		C							8	
002				0						-			•	, ė
8		9877	0 223	0		7. XIX			1				122	0 16735
9	į	er 2002	1746.03	•		8 75 88 88			F				2 2	66
o.		•	٠	•	>		•	•			,			٥
8	§	991	98	9		8 8 8 8			3 5					. 8
88			9	•		• 8								
9004		80	8	8										
00002	ş	888		8		§ §								8
88	ş	87	88	8 1		§ §				1				. 8
88		000001	8	8		§ §								8800
USD addedor	9	Probable USD	Probabe USO	Processia USD		den seemaa						000	020	, agen
55				rescored bylasy		recommendation of the second s								1
SEAMANA 20 Perso	TOWNER OF PRESENTING	11/16/10 SEAVANK 20 Personal Injury	SEALKWAR 20 Person	EEAHAWAK 20 Perso		SEALANK 20 Perso								4
11.00ATD 55/2	200000	130 00001	oloonio	All HACENIA		23 0140001 23 0140011							4	
23000	C 4 8	WILLEL	E S	PATRICK		THOMAS SOE							2	RAYTORD
			THOMPSON											

Sephent Drilling, Inc. FAS - S Report 02.01.2011

3 of 5

se 11-20	J08	9	Dc	CUI	me	nt 19	9-3	File	d in	TX	SE	on	0	2/	12/11 F	age	80	1	- 1
			Fray Ashugae us eduanna dreung rc		Ness Gasses Bartock Y Otros vs. Chapa Hegesteds	ANY COMMON TO SEASON TO SE											ZARIYA SOUTEDAN SASS	DI OCTAVIO AVILA V GOMPS, Y MEXICO GRUPO DE ASESORES JUNIDICOS CHICAMUL.	LEANDRO PERALTA «GOMPS Y MEXICO DRILLING LANGOY PRINGE PATEMATOMAL, NC. GRUPO DE ASESORES JURIDICOS - CHI CANVIL. ORGENESIONES MANGES DE CANDANA.
The course specificacy is a full delicited control of the course of the		P.E.1 De second. The control matter has been continued to a later data in 2011. Decovery is all corporing. A new defendant has been added to the case (Superior Wed Service, Inc.). We are etil	Tendo I Goods for I Proposes of Plantacients. III. Reveal west brought in for middle reportibilisms. Propose with the made of the resolution broad and a service from other OUGLES (I for seeding themself and dalars service from other OUGLES (I former flowes). Figure 1. A. Answell was refreshed for independent was applicated on well-before. GSS methods with Promotion and middles and middles and proposed and proposed and proposed and proposed and proposed and proposed		1	At the back to be carebook, Mr. Keyr reported to heavy should be carebook, Mr. Keyr reported to heavy should be carebook, credit the condition, spread to extract the carebook spread to the carebook spread t	At the part has been completed from the good from a through have been completed for the good from the been edged on the TGD. It has proceed it checked at the brokes maked to the request and the CO-314 edded (7227000) and from the CO-314 edded (72270000000000000000000000000000000000	incident (2022/2016 a STATOSI) in white Sashawi, wall recover to miches for repair pick the T/22/2010 due to the separate deboulbes, beself on the times debusidous the major cost to the fig as a read of the T/22/2010 fundant was the STATOSI factory. Therefore, recovery costs for STA status and Audition.	Unident formsteering the processes of the service work on the Sander Across The register or form the service of		14/, Goodfelf has recovered from the cobbied with organization holders. He recovers a published with ordinary and an extraction holders. He recovers a property of the month to work as of 1100/2010 thron Dr. Hermandar, Holders or the Management of the month of the m		Mr. Edinumbern has returned to work at Mal Ady for the researt Nach. The involves for the medical cars will need to that. The processed prior to dowing the the.	Mr. Martog has forwarded his work ralease from No urbigiet. Involces will be processed and this claim will be diseased.	Accordance (Total Control Cont		The Labor Although reports that Mi. Valatiques has submitted a withdrawal of the audion and demand; housewar, the sond beam confirmed or Machael has not retified the methods of the first of the submitted in souther the soft retified the methods of the souther to the submitted to souther the souther than the	There is no none evidence to present in this matter. The event is being dictated. Available resolution at this time.	Aveiding the resolution/season of event.
o our oversion		D QUIT OF MEXICO						000000000000000000000000000000000000000						O GAUFOFMENCO		1	000000000000000000000000000000000000000		
80 N		8						e		<u> </u>			0005	80] *	1 -	
92		0	T THE STATE OF THE	•				c			ac.		2008	0			7.00.00	0 12785208	Highesta 0
		0	2		656		Tired y			0	222		90008				900	0 0	0
62 (ZZ)					9 9		D. PALCO				2817	•	2784.56 0 1984.51	0			9	0	6
0		8 11						•	ş										0 0
° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °				0						0		0	88	٥				0	0
0 00052				• ;					\$		8		8			8			0
20001				ş				<u> </u>	778	5			-	000001		000000	<u> </u>	Т	П
Son Research	Call Springer	Call Speciment		monthylog Probable USD			·	Coort Denice	i	100	constraint Probate USI	Posselle USD	nored blazy Probestee USD			Oson special Livers	1		Implyment/labor Remote MXH
me of waveless outce.				SEAHAWK 20 Pe	200			SOLUTION OF THE PROPERTY OF TH	o de name	SEAUAWK 30 Pe	toning seawayer ag Per	1200m0 SEAVANK 30 Per	12/18/10 SEALKIVK 30 Personal Warr	OUIZH SEAMANK 30 Per		S S S S S S S S S S S S S S S S S S S			
SIRVY	SHRADER	<u></u>	MUSSELL	AUSSELL JAMES	SORBOULA		PROPERTY GALL		MALEN BIT ERWATI	SALNAS	GOGDINIGHT			HARTZOG BENJAMIN	some ·		VELVZGUEZ		PERALTA LEAKORO
10-00-102-00-00	UNIVERSITY OF STATE O				808	-	-	10-20005e-01	MAKE		10-CT1010-04.000	BATISTE MATERIALITA	EOM/			Section Section	AEC.	042090-006874-01	10-512100-0005

404

Seahwel (Nilleg, Inc. FAS - 8 Report 02:01:2011

se	11-20	280	Dod	:
25		089	sim	_
10490		ESCAMELA	TOWART I	١
SOS	i	2 2	\$000 \$000	
GRUPO DE ASESORES JURIDICOS			3 KES JURID	
DEASERO		GARZA TELLO GARZA TELLO	006.488.60	
		QHEZ QHEZ	QRUPC	١
Omps Y Prid		MAKE, SA	35 M 7 Se	
Anges v. G		GANZA TELLC DANIEL ADRWH DAZ VE, PRIDE FORAMER, SAS GARZA TELLC	Тімозіно — воходит завовату до силов —— заят я силор за коне зналія ченен ку	
de in Crud		AN DIAZ VI	Mornies S.	
Entique Augusto Certiga America		ANIEL ADI	Con Bedeate	
	4		4 2 4	
The audience will be carried out on Fabruary 19, 2011. Our superior of proofs will deduce be algorithms and freque pricts of the sure employage. In the provious audience the sor employages repect did not show, Both appart of proofs audience and according to the superior of proofs audience and according to the superior of proofs audience analysis.	The outsit in Caudia delicromes is serioud for serioud but and control in device the but of principles consisting to the sect of serious pix levy due for the past of blazelit provide, the shall proceeding the sect in sectional but my other the last proceeding the sect in control and my other the last perform my their control area. The revolution could be supplied to publish their control area. The section control are proper my their control area to sect to control are performed to the control area of the my their control area to control area of the proceeding the population and and the my their control area control stopped and and control area area to a section of the most and the population and and the most area area to control stopped and and control area to the most area area.	Multitiffied in secesal or in secesary or in secesary and secsion or this case will depend on the section or provided by the plaintiff. The status is set to sent or section.	Actor Cutes Liver process the in bedeeting her been more than twenty to change it in the common than twenty to change it in the common than twenty. In section to control, the common than twenty to the refer to control, the common than twenty to the refer to the mental mental which the west of the refer to the mental mental to the section that the section than the mental than the section than the section than the section that the section than the section than the section that th	Center, Orc
ed out on Fe re the signed a pranfous is show. Both	affect by De of fleet by De tog suit in the fleet by deam on althought on althought on althought and by Public by Comes. If Phi of this match	rnek. Jesion panda e outzome o ed by the pla	Carmen; The of he is the search for the search for the search for the foreign	on Academa
e will be court lofe will deck playee. In the repert clid not	Chidad del C Chidad del C Managal process legal process le	Taketiff flad on second or not. Issuing offering and admission remoded by plaintiff. The outer pon the avidence provided by oramote.	a Lopes report to a mediate to a mediate to a mediate to a mediate to a 1000 a.m. and Arbitrat to case to CA il suchority of a soure find to a source fi	he involce for a mark filter fi
The audience expert of pro of the extern employees	The court in which orders the back of a granted, the perty should may depth the transfer or separation or separati	Paintiffied Hearing offs promoted by upon the and to remote.	Artero Geral moved from challenge be suffernos for 100-W2010; challenge and challenge challenge of the federal Versong to	Whiten on the knotoe from Acadian received and cald. This the can done
25		¥ \$	MAS	Pa,
D MEXICO		0 MEXICO	973	QUIT OF MEXICO
°				٩
•			-	٥
501901			į	881
			12 CONTR. 00 00 00 00	3200
85786.38		2 381	2	٦
۰				۰
		• •	6	12120
		0 0	•	
				714.19
		ь о		-
			•	1945.4E
			•	٥
				-
۰				
		0 0		0000
L.				1500
				3800
				2500
1		1 1	1	100000
1		A T		Probable USD
Sadomental abov		Commercial Dispute Remote		Personal fator
		Comme	1	Person
ļ		a -	,	9
8		904170		סויגוענו
T TOTAL		UETICIA PANIE		Упнаог
No.	ECHAVARRIA		ş	ROE
ARAGON		DW2	SANDRA	LABORD
0.00		10-000-00		

8

	Closed Claims									
	ClaimNum	LastName	FirstName	ClaimName	AccidentDate S	Status	OpenDate	CloseDate	ResTot	
` '	1 042690-008329-01	Young	Daniel	Unresponsive	10/09/08 Closed	losed	10/10/08	01/31/11	\$ 1.	123,918
, 4	2 042690-006791-01	Robinson	John	Racial Discrimination	11/13/07 Closed	losed	01/31/08	01/26/11	\$	16,436
	3 060799-010152-01	Johnson	Ronnie	Redness and Swelling to R Leg	09/08/10 Closed	losed	09/13/10		\$	1
_	4 060799-010186-01	Wilczewski	Joshua	Sore throat/Congestion	10/22/10 Closed	losed	10/22/10	01/17/11	\$	1,154
-,	5 060799-010178-01	Emonet	Rodney	Upper Back/Neck Pain	10/15/10 Closed	losed	10/18/10	01/21/11	\$	ı
	6 060799-010092-01	Stafford	Richard	Right Forearm Swollen	07/07/10 Closed	losed	01/60/10	01/17/11	\$	1,786
	Opened Claims				-					
	ClaimNum	LastName	FirstName	ClaimName	AccidentDate S	Status	OpenDate	CloseDate	ResTot	
	1 060799-010252-01	Hileman	Ricky	Breach of Contract	01/14/11 Open	hen	01/17/11		s	25,000
Ľ	2 060799-010257-01	Paxton	Patrick	Flu Like Symptoms	01/23/11 Open)pen	01/24/11		φ.	3,500
	3 060799-010263-01	Use	Billy	Nausea and Vomiting	01/31/11 Open	ben	01/31/11		٠	5,000
_	4 060799-010256-01	Shrader	John	Flu Like Symptoms	01/22/11 Open)pen	01/24/11		\$	3,500
	5 060799-010249-01	Hartzog	Benjamin	Abdominal Pain	01/12/11 Open)ben	01/13/11		\$	2,000
	6 060799-010261-01	Ramos	Naesar	R Forearm Injury	01/28/11 Open)ben	01/31/11		\$	5,000
	Reserve Changes									
	ClaimNum	LastName	FirstName	AccidentDate	Status					
	1 060799-009417-01	NELSON	DAVID	02/01/09	9 RESERVE INCREA	SE FROM \$	150,000 TO \$250,0	02/01/09 RESERVE INCREASE FROM \$150,000 TO \$250,000 - POTENTIAL LOSS \$250,000	\$\$ \$250,000	0

ANNEX 4.7 FIN48 REPORTS

See attached (Parts 1, 2, and 3)

Ħ	l		(2)					
_		•	(O.	•	٠			
S S S		2,849,871.52	1,033,394.73	270,390.04	614,335.29			
End Balance		2,849,871.52	1,033,394.20	270,390.04	614,335.29	4,767,991.07		
Tax Expense		117,871.52	187,735.85	270,390.04	26,797.54	602,794.97	(26,472.97)	576 321 99
Tax Expense		33,871.52	47,319.85	14,376.45	16,359.93	111,927.76	•	111 027 76
Tax Expense	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	103,000.00	47,834.00	256,013.59	22,613.00	429,460.59		429 ARD 59
Tax Expense 2nd Q	00 1 2 1 2 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1		46,291.00		(71,070,11)	(24,779.11)		
Tax Expense		(19,000.00)	46,291.00		58,894.72	86,185.72	(26,472.97)	59 719 75
Beg Balance		2,732,000.00	845,658.35		587,537.75	4,165,196.10		
enssj	Tax Depreciation	Drillaser	Labor	Transfer Pricing Documentation	Lost Deductions	Total	Correction BB	Total
		Mexico	China	Mexico	Mexico			
UTP (dentifier		2	က	4	co.	Total		
			_					

Seahawk Drilling, Inc. FIN 48 Summary As of December 31, 201

FIN48 FIN48 FIN48 FIN48

Seahawk Drilling Inc FIN 48 Calculation

Tax Identifier:

	2003	2004	2005	2006	TOTAL	Notes
Revenue Billed	942,000	744,000	897,000	880,611	3,463,611	Revenue provided by Houma acctng dept
Deemed Profit Rate (Assume no deducts)	100%	100%	100%	100%	•	Assumes no salary deductions given no PIT filings made
Deemed Profit	942,000	744,000	897,000	880,611	3,463,611	
CIT Rate	33%	33%	33%	33%		
CIT	310,860	245,520	296,010	290,602	1,142,992	
Less: 3.3% Enterprise w/h tax	(31,086)	(24,552)	(29,601)	(29,060)	(114,299)	w/h tax verified by Apache (Roc Oil) Controller
5% Business Tax (Paid)			jiha bergelese san	gab gran <mark>g</mark> t ma	-	
Total Chinese Tax	279,774	220,968	266,409	261,541	1,028,692	
Surcharge (.05%/day)	357,411	241,960	243,098	190,925	1,033,395	Estimate based on days late
Less: Creditable Foreign Tax to SDI	(279,774)	(220,968)	(266,409)	(261,541)	(1,028,692)	100% of CIT amount after withholding
	357,411	241,960	243,098	190,925	1,033,395	
			7			

1,033,395

at 12/31/10 1,028,692 (1,028,692) Remaining Tax after Withholding Foreign Tax Credit Net Tax Surcharges 1,033,395 Total Accrual 1,033,395 Total Accrual at 12/31/2010 1,033,395 986,075 47,320 Total Accrual at 9/30/10

Change in Balance

Note: During the 3rd quarter of 2007, Pride switched over to a foreign tax credit position. This change in methodology generates a 100% FTC benefit versus a 35% foreign tax deduction. Thus, this accrual has been reduced to the amount related to the sur

Note: This exposure no longer exists for years post 2006; however, surcharge accrual continues on daily basis.

Case 11-20089 Document 19-3 Filed in TXSB on 02/12/11 Page 11 of 60 FIN48 Report Annex 4 7 Part 2 China Labor 12-31 v1.xls

Seahawk Drilling Inc FIN 48 Calculation

Tax Identifier:

3

	2003	2004	2005	2006	TOTAL	Notes
Revenue Billed	942,000	744,000	897,000	880,611	3,463,611	Revenue provided by Houma acctng dept
Deemed Profit Rate (Assume no deducts)	100%	100%	100%	100%		Assumes no salary deductions given no PIT filings made
Deemed Profit	942,000	744,000	897,000	880,611	3,463,611	
CIT Rate	33%	33%	33%	33%		
CIT	310,860	245,520	296,010	290,602	1,142,992	
Less: 3.3% Enterprise w/h tax	(31,086)	(24,552)	(29,601)	(29,060)	(114,299)	w/h tax verified by Apache (Roc Oil) Controller
5% Business Tax (Paid)		ymagi di dini Salahari			-	
Total Chinese Tax	279,774	220,968	266,409	261,541	1,028,692	
Surcharge (.05%/day)	357,411	241,960	243,098	190,925	1,033,395	Estimate based on days late
Less: Creditable Foreign Tax to SDI	(279,774)	(220,968)	(266,409)	(261,541)	(1,028,692)	100% of CIT amount after withholding
	357,411	241,960	243,098	190,925	1,033,395	
•						

	1,033,395
1/14/10/24/2010	at 12/31/10
Remaining Tax after Withholding	1,028,692
Foreign Tax Credit	(1,028,692)
Net Tax	-
Surcharges	1,033,395
Total Accrual	1,033,395
Total Accrual at 12/31/2010	1,033,395
Total Accrual at 9/30/10	986,075

Change in Balance

Note: During the 3rd quarter of 2007, Pride switched over to a foreign tax credit position. This change in methodology generates a 100% FTC benefit versus a 35% foreign tax deduction. Thus, this accrual has been reduced to the amount related to the sur

Note: This exposure no longer exists for years post 2006; however, surcharge accrual continues on daily basis.

EXHIBIT 3.3(a)

MASTER BILL OF SALE

THIS MASTER BILL OF SALE (t	this "Bill of Sale") dated as of,
2011, is by and among, a	, and, a
("Assignee"). Capit	talized terms used but not otherwise defined in this
Bill of Sale shall have the respective meani	ings ascribed thereto in the Purchase Agreement (as
defined below).	
WHEREAS, Assignor and Assignee	e are party to that certain Asset Purchase Agreement
dated as of, 2011 (the '	"Purchase Agreement"), by and among Assignor,
Assignee, and the other parties named there	ein, pursuant to which Assignor has agreed to sell to
Assignee, and Assignee has agreed to purch	hase from Assignor, all of Assignor's interest in the
Purchased Assets of Assignee (the "Transfe	Gerred Assets"), and pursuant to which Assignee has
agreed to assume the Assumed Liabilitie	es of Assignee (the "Transferred Liabilities"), in
accordance with the provisions of the Purcha	ase Agreement; and

WHEREAS, simultaneously with the Closing of the transactions contemplated by the Purchase Agreement, the parties hereto mutually desire that Assignor assign all of its interest in the Transferred Assets to Assignee or its designee and that Assignee or its designee assume the Transferred Liabilities, as hereinafter set forth.

NOW, THEREFORE, for and in consideration of the mutual covenants contained herein and in the Purchase Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. Sale of Assets. Assignor does hereby convey, sell, assign, transfer, and deliver to Assignee and its successors and permitted assigns, and Assignee does hereby accept and assume, as of the Closing, all of Assignor's right, title, and interest in, to, and under the Transferred Assets TO HAVE AND TO HOLD the Transferred Assets unto Assignee, its successors and assigns for their own benefit and use forever.
- 2. Assumption of Liabilities. Assignor does hereby assign and transfer unto Assignee, and Assignee hereby accepts, assumes, and agrees to pay, perform, discharge, and satisfy as and when due, the Transferred Liabilities of Assignor in accordance with the provisions of the Purchase Agreement. Other than such Transferred Liabilities, Assignee is not assuming and in no way agrees to undertake to pay, satisfy or discharge, or be liable for, any Liabilities of Assignor (including without limitation the Retained Liabilities, for which Assignor shall remain liable).
- 3. Binding Effect and Amendment. This Bill of Sale is binding upon, inures to the benefit of, and is enforceable by Assignor and Assignee and their respective successors and permitted assigns. This Bill of Sale may be amended, modified, or supplemented only by written agreement of the parties hereto.

- 4. Conflict Between Agreements. This Bill of Sale is entered into pursuant to the Purchase Agreement and is subject and subordinate, in all respects, to all of the terms and conditions of the Purchase Agreement. To the extent any conflict or inconsistency exists between any term or provision of this Bill of Sale and any term or provision of the Purchase Agreement, the latter shall control.
- 5. Attorney-in-Fact. Assignor hereby constitutes and appoints Assignee as its true and lawful attorney-in-fact, with full power of substitution and re-substitution, in the name of Assignor, but on behalf of and for the benefit of Assignee, to (a) institute or prosecute, in the name of Assignor or otherwise, all Proceedings which Assignee may deem necessary, advisable, or convenient in order to realize upon, affirm, or obtain title to or possession of, or to collect, assert, or enforce any property, Claim, right, or title of any kind in or to the Transferred Assets, and (b) defend and compromise any and all actions, suits, or Proceedings in respect of any of the Transferred Assets. Assignor agrees that the foregoing powers are coupled with an interest and are and shall be irrevocable by Assignor for any reason.
- 6. Further Assurances. Assignor and Assignee, for themselves and their respective successors and assigns, each hereby covenant and agree to execute, acknowledge, and deliver, or to cause to be executed, acknowledged, and delivered, all and every further documents or instruments (including assignments and bills of sale) and to do such further acts as any party hereto reasonably may deem necessary or appropriate in order to effect the intent and purposes of this Bill of Sale and the transactions contemplated hereby and by the Purchase Agreement.
- 7. Governing Law. THIS BILL OF SALE, AND THE LEGAL RELATIONS BETWEEN THE PARTIES WITH RESPECT TO THIS BILL OF SALE, SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS WITHOUT REGARD TO RULES CONCERNING CONFLICTS OF LAWS. Assignor and Assignee agree that the Bankruptcy Court shall have exclusive jurisdiction over all disputes and other matters relating to the interpretation and enforcement of this Bill of Sale, and Assignee expressly consents to and agrees not to contest such exclusive jurisdiction. All Claims and Proceedings brought, arising out of, or related to this Bill of Sale or the transactions contemplated hereby shall be brought in the Bankruptcy Court, and the Bankruptcy Court shall retain jurisdiction to determine any and all such Claims and Proceedings. The parties hereto waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the bringing of any such Claim or Proceeding in such jurisdiction.
- 8. Counterparts. This Bill of Sale may be executed in two or more counterparts (including by means of facsimile or other electronic transmission), each of which shall be deemed to be an original, but all of which together shall constitute one and the same agreement.

[Signature Page Follows]

IN WITNESS WHEREOF, each of the parties hereto have caused this Bill of Sale to be executed as of the date first above written.

ASSIGNOR:
By:
Name:
Title:
ASSIGNEE:
By: Name:
Title:
11110.

EXHIBIT 3.3(b)

ASSIGNMENT AND ASSUMPTION OF CONTRACTS

THIS ASSIGNMENT AND ASSUMPTION OF CONTRACTS (this "Assumption
Agreement") dated as of, 2011, is by and among, a
("Assignor"), and, a ("Assignee"). Capitalized
terms used but not otherwise defined in this Assumption Agreement shall have the respective meanings ascribed thereto in the Purchase Agreement (as defined below).
WHEREAS, Assignor is a party to each of the Assigned Contracts listed in <u>Exhibit A</u> hereto (collectively, together with all schedules, exhibits, and addenda thereto, and as amended to the date hereof, the " <i>Transferred Contracts</i> ");
WHEREAS, Assignor and Assignee are party to that certain Asset Purchase Agreement dated as of, 2011 (the "Purchase Agreement"), by and among Assignor, Assignee, and the other parties named therein, pursuant to which Assignor has agreed to assign to Assignee, and Assignee has agreed to assume from Assignor, all of Assignor's interest in the Transferred Contracts, and pursuant to which Assignee has agreed to assume certain Liabilities of Assignor under each of the Transferred Contracts, in accordance with the provisions of the Purchase Agreement; and

WHEREAS, simultaneously with the Closing of the transactions contemplated by the Purchase Agreement, the parties hereto mutually desire that Assignor assign all of its interest in the Transferred Contracts to Assignee or its designee and that Assignee or its designee assume certain Liabilities of Assignor with respect to such Transferred Contracts, as hereinafter set forth.

NOW, THEREFORE, for and in consideration of the mutual covenants contained herein and in the Purchase Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. Assignment. Assignor does hereby delegate, assign, convey, and transfer to Assignee and its successors and assigns, as of the Closing, all of Assignor's right, title, and interest in, to, and under or with respect to each of the Transferred Contracts.
- 2. Assumption. Subject to the conditions, limitations, and exceptions set forth in the Purchase Agreement, Assignor does hereby assign and transfer unto Assignee, and Assignee hereby accepts, assumes, and agrees to pay, perform, discharge, and satisfy as and when due, the Liabilities of Assignor under the terms of each Transferred Contract, but only to the extent that such Liabilities are performable or otherwise attributable to the period from and after the Closing. Other than such assumed liabilities, Assignee is not assuming and in no way agrees to undertake to pay, satisfy or discharge, or be liable for, any Liabilities of Assignor (including without limitation the Retained Liabilities, for which Assignor shall remain liable).

- 3. Binding Effect and Amendment. This Assumption Agreement is binding upon, inures to the benefit of, and is enforceable by Assignor and Assignee and their respective successors and permitted assigns. This Assumption Agreement may be amended, modified, or supplemented only by written agreement of the parties hereto.
- 4. Conflict Between Agreements. This Assumption Agreement is entered into pursuant to the Purchase Agreement and is subject and subordinate, in all respects, to all of the terms and conditions of the Purchase Agreement. To the extent any conflict or inconsistency exists between any term or provision of this Assumption Agreement and any term or provision of the Purchase Agreement, the latter shall control.
- 5. Attorney-in-Fact. Assignor hereby constitutes and appoints Assignee as its true and lawful attorney-in-fact, with full power of substitution and re-substitution, in the name of Assignor, but on behalf of and for the benefit of Assignee, to (a) institute or prosecute, in the name of Assignor or otherwise, all Proceedings which Assignee may deem necessary, advisable, or convenient in order to realize upon, affirm, or obtain title to or possession of, or to collect, assert, or enforce any property, Claim, right, or title of any kind in or to the Transferred Contracts, and (b) defend and compromise any and all actions, suits, or Proceedings in respect of any of the Transferred Contracts. Assignor agrees that the foregoing powers are coupled with an interest and are and shall be irrevocable by Assignor for any reason.
- 6. Further Assurances. Assignor and Assignee, for themselves and their respective successors and assigns, each hereby covenant and agree to execute, acknowledge, and deliver, or to cause to be executed, acknowledged, and delivered, all and every further documents or instruments (including assignments and bills of sale) and to do such further acts as any party hereto reasonably may deem necessary or appropriate in order to effect the intent and purposes of this Assumption Agreement and the transactions contemplated hereby and by the Purchase Agreement.
- 7. Governing Law. THIS ASSUMPTION AGREEMENT, AND THE LEGAL RELATIONS BETWEEN THE PARTIES WITH RESPECT TO THIS ASSUMPTION AGREEMENT, SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS WITHOUT REGARD TO RULES CONCERNING CONFLICTS OF LAWS. Assignor and Assignee agree that the Bankruptcy Court shall have exclusive jurisdiction over all disputes and other matters relating to the interpretation and enforcement of this Assumption Agreement, and Assignee expressly consents to and agrees not to contest such exclusive jurisdiction. All Claims and Proceedings brought, arising out of, or related to this Assumption Agreement or the transactions contemplated hereby shall be brought in the Bankruptcy Court, and the Bankruptcy Court shall retain jurisdiction to determine any and all such Claims and Proceedings. The parties hereto waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the bringing of any such Claim or Proceeding in such jurisdiction.
- 8. Counterparts. This Assumption Agreement may be executed in two or more counterparts (including by means of facsimile or other electronic transmission), each of which

90327509.5 -2-

shall be deemed to be an original, but all of which together shall constitute one and the same agreement.

[Signature Page Follows]

90327509.5 -3-

IN WITNESS WHEREOF, each of the parties hereto have caused this Assumption Agreement to be executed as of the date first above written.

ASSIGNUR:	
[]	
By: Name: Title:	_
ASSIGNEE:	
By: Name: Title:	

EXHIBIT A

TRANSFERRED CONTRACTS

Agreement	Executed Date
1. [TO COME]	[]

EXHIBIT 3.3(c)

ASSIGNMENT AND ASSUMPTION OF INTANGIBLE ASSETS

THIS ASSIGNMENT AND	ASSUMPTION	OF INTANGIBLE	E ASSETS (this
"Assignment") dated as of	, 2011, is by	and among	, a
(" <i>Assignor</i> "), and	, a	("Assigi	iee"). Capitalized
terms used but not otherwise defined	in this Assignme	nt shall have the re	spective meanings
ascribed thereto in the Purchase Agreen	nent (as defined be	low).	
WHEREAS, Assignor and Assigned as of, 2011 (Assignee, and the other parties named to Assignee, and Assignee has agreed interest in the Assigned Intangible Assand	(the "Purchase As therein, pursuant to to accept and ass	greement"), by and o which Assignor has turne from Assignor	among Assignor, as agreed to assign , all of Assignor's

WHEREAS, simultaneously with the Closing of the transactions contemplated by the Purchase Agreement, the parties hereto mutually desire that Assignor assign all of its interest in the Transferred Intangible Assets to Assignee or its designee and that Assignee or its designee accept and assume the same, as hereinafter set forth.

NOW, THEREFORE, for and in consideration of the mutual covenants contained herein and in the Purchase Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. Assignment. Assignor does hereby convey, sell, assign, transfer, and deliver to Assignee and its successors and permitted assigns, and Assignee does hereby accept and assume, as of the Closing, all of Assignor's worldwide right, title, and interest in, to, and under the Transferred Intangible Assets, including, without limitation, the Intangible Personal Property listed in Exhibit A hereto, and all rights to register, maintain, renew, or otherwise apply for new or continuing protection, and to sue for infringement, thereof.
- 2. Binding Effect and Amendment. This Assignment is binding upon, inures to the benefit of, and is enforceable by Assignor and Assignee and their respective successors and permitted assigns. This Assignment may be amended, modified, or supplemented only by written agreement of the parties hereto.
- 3. Conflict Between Agreements. This Assignment is entered into pursuant to the Purchase Agreement and is subject and subordinate, in all respects, to all of the terms and conditions of the Purchase Agreement. To the extent any conflict or inconsistency exists between any term or provision of this Assignment and any term or provision of the Purchase Agreement, the latter shall control.
- 4. Attorney-in-Fact. Assignor hereby constitutes and appoints Assignee as its true and lawful attorney-in-fact, with full power of substitution and re-substitution, in the name of Assignor, but on behalf of and for the benefit of Assignee, to (a) institute or prosecute, in the name of Assignor or otherwise, all Proceedings which Assignee may deem necessary, advisable,

or convenient in order to realize upon, affirm, or obtain title to or possession of, or to collect, assert, or enforce any property, Claim, right, or title of any kind in or to the Transferred Intangible Assets, and (b) defend and compromise any and all actions, suits, or Proceedings in respect of any of the Transferred Intangible Assets. Assignor agrees that the foregoing powers are coupled with an interest and are and shall be irrevocable by Assignor for any reason.

- 5. Further Assurances. Assignor and Assignee, for themselves and their respective successors and assigns, each hereby covenant and agree to execute, acknowledge, and deliver, or to cause to be executed, acknowledged, and delivered, all and every further documents or instruments (including assignments and bills of sale) and to do such further acts as any party hereto reasonably may deem necessary or appropriate in order to effect the intent and purposes of this Assignment and the transactions contemplated hereby and by the Purchase Agreement.
- 6. Governing Law. EXCEPT TO THE EXTENT THAT FEDERAL LAW PRE-EMPTS STATE LAW WITH RESPECT TO MATTERS COVERED HEREBY, THIS ASSIGNMENT, AND THE LEGAL RELATIONS BETWEEN THE PARTIES WITH RESPECT TO THIS ASSIGNMENT, SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS WITHOUT REGARD TO RULES CONCERNING CONFLICTS OF LAWS. Assignor and Assignee agree that the Bankruptcy Court shall have exclusive jurisdiction over all disputes and other matters relating to the interpretation and enforcement of this Assignment, and Assignee expressly consents to and agrees not to contest such exclusive jurisdiction. All Claims and Proceedings brought, arising out of, or related to this Assignment or the transactions contemplated hereby shall be brought in the Bankruptcy Court, and the Bankruptcy Court shall retain jurisdiction to determine any and all such Claims and Proceedings. The parties hereto waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the bringing of any such Claim or Proceeding in such jurisdiction.
- 7. Counterparts. This Assignment may be executed in two or more counterparts (including by means of facsimile or other electronic transmission), each of which shall be deemed to be an original, but all of which together shall constitute one and the same agreement.

[Signature Page Follows]

Case 11-20089 Document 19-3 Filed in TXSB on 02/12/11 Page 22 of 60

IN WITNESS WHEREOF, each of the parties hereto have caused this Assignment to be executed as of the date first above written.

ASSIGNOR:	
	_]
By:	
Name:	
Title:	
ACCIONED	
ASSIGNEE:	
[
By:	
Name:	
Title:	

EXHIBIT A

ASSIGNED INTANGIBLE ASSETS

[TO COME]

EXHIBIT 3.3(g)

BUSINESS CAPABILITIES CERTIFICATE

THIS BUSINESS CAPABILITIES CERTIFICATE (this "Certificate") dated as of _______, 2011, is executed and delivered by SEAHAWK DRILLING, INC., a Delaware corporation ("Seahawk Parent") and HERCULES OFFSHORE, INC., a Delaware corporation ("Hercules Parent"). Capitalized terms used but not otherwise defined in this Certificate shall have the respective meanings ascribed thereto in the Purchase Agreement (as defined below).

WHEREAS, Seahawk Parent and Hercules Parent are party to that certain Asset Purchase , 2011 (the "Purchase Agreement"), by and among Agreement dated as of Seahawk Parent, SEAHAWK GLOBAL HOLDINGS LLC, a Delaware limited liability company ("Seahawk Global Holdings"), SEAHAWK MEXICO HOLDINGS LLC, a Delaware limited liability company ("Seahawk Mexico Holdings"), SEAHAWK DRILLING MANAGEMENT LLC, a Delaware limited liability company ("Seahawk Drilling Management"), SEAHAWK DRILLING LLC, a Delaware limited liability company ("Seahawk Drilling"), SEAHAWK OFFSHORE MANAGEMENT LLC, a Delaware limited liability company ("Seahawk Offshore Management"), ENERGY SUPPLY INTERNATIONAL LLC, a Delaware limited liability company ("Energy Supply International"), SEAHAWK DRILLING USA LLC, a Delaware limited liability company ("Seahawk USA", and, together with Seahawk Parent, Seahawk Global Holdings, Seahawk Mexico Holdings, Seahawk Drilling Management, Seahawk Drilling, Seahawk Offshore Management, and Energy Supply International, each a "Seller" and, collectively, the "Sellers"), Hercules Parent, and SD DRILLING LLC, a Delaware limited liability company ("Newco", and, together with Hercules Parent, each a "Purchaser" and, collectively, the "Purchasers"); and

WHEREAS, this Certificate is being delivered at Closing by the parties to memorialize the business capabilities of the Business and is intended to be solely for informational purposes and to be provided by Purchasers to third parties, as appropriate, in connection with proposed business arrangements arising in Purchasers' operation of the Business following the Closing.

NOW, THEREFORE, Seahawk Parent and Hercules Parent hereby acknowledge and certify the following:

- 1. From and after the Closing Date, Purchasers (or their affiliates) will have the personnel, equipment, knowledge, working history and capability to conduct the Business and operate the Transferred Rigs in a manner substantially consistent with the conduct thereof by Sellers immediately prior to the Closing.
- 2. Pursuant to the transactions contemplated by the Purchase Agreement, Purchasers are acquiring substantially all of the assets, properties, contracts, operational processes, books and records, business relationships, know-how, rights, and other capabilities and resources of Sellers, which Purchasers will use for the operation of the Business.
- 3. Seahawk Parent further agrees to provide to Hercules Parent, upon request and on an as-needed basis, on such terms and conditions as to be agreed to by the parties hereto, such technical, operating, and other counsel, information, and support services relating to the

operational history of the Business as requested by Hercules Parent in order to facilitate the transition of the such resources and capabilities to Purchasers and Purchasers' uninterrupted use and deployment thereof.

[Signature Page Follows]

90327658.6 -2-

Rolling Cash Forecast Seahawk Drilling, Inc.

Seahawk Drilling, Inc.		-												C
(\$ in millions) Week Number	Filing Date	-	74	**		us			œ			‡	ŧ	Cas
Week Ended	2/11/2011	2/18/2011	2/25/2011	F	3/11/2011	3/18/2011	3/25/2011	471/2011	Ħ	415/2011	47222011	4/29/2011	5/6/2011	5/13/2011
Beginning Cash Balance	\$ 11.7	\$ 3.9	\$ 2.0 \$	\$ 2.0 \$	2.0 \$	2.0 \$	\$ 02	\$ 02	\$ 0.2	\$ 0.2	\$ 0.2	2,0	5 20	2.07
Contract Drilling	1.5	0.0	0.7	9.0	53	2.1	eo er	1.4	2.7	1.6	9.5	8. 2.	90	-2
Total Cash Receipts	3.5		2'0	9.0	2.9	2.1	3.8	7	2.7	3.1	6.0	25	970	00
Operating Expenses	<u> </u>	٠	. (0.5)	(2.7)	(0.6)	(6.7)	(0.5)	(0.5)	(50)	(90)	(0.5)	(9)	(9.5)	<mark>89</mark> .
Payroll	(1.5)		(1.5)	0.7	(1.3)	(0.7)	<u> </u>	(2.0)	5	(p. 7)	(1.3)	(a.7)	(1.3)	(0.7)
Instrance Pride Wyoming		(3.5)		. ,	. 60				, ,		, ,	• •		Þς
Ordinary Course Professionals	•	(00)	(0.3)	(0.0)	(0:0)	(0.0)	(0.3)	(0.0)		(0.0)	(0.3)	(0.0)	(0.0)) C
Rent G&A and Other	(0.1)	. 6.0	. (0.1)	6.2	. 6	. 6	9	(0.2)	. 6	. 5	9	' \$	605	ψņ
Total Cash Disbursements	(5.0		(2.4)	(1.7)	(1.5)	(1.5)	(2.2)	₹E	(6.1)	(12)	22	(T)	(50)	(12)
Restructuring Related Sr. Exec Compensation (1)														nt.1
CIC and LTIP Payments Destructiving Defocious LIST 1 HEL/R		(22)	•	•	•	•	•	•	٠ (•	•	•	. :	19-
Total Restructuring Related (1974)	(3.8)			. -	,	. .	, , ,	. .	6.0	. .		1	(0.4)	3.
Financing Fees		· ·												F
Natbds Facility Repayment	• 6	(18.2)	•	•	•	•	•	•		•	•	•	•	ile
DIP Tees	<u>~</u>		• 1	' F	• 1	•	•	• 6	•	•	•	' ê	•	ęd
DP Borrowings (Repayment)(4)		25.8	. 8.	1.4	(4.6)	(0.6)	(1.6)	(c.3)	. 1.9	(1.8)	. 6:	(4.5)	. 6.	. 22 X
Total Financing Payments / Borrowing ⁽⁴⁾	(7.0)	6.5	1,8	1,1	(4.F)	(0.6)	(1.6)	(0:0)	0.1	(1.8)	1,6	(1.8)	1,9	(2.2)
Ending Cash Balance	\$ 3.9	\$ 2.0	\$ 2.0 \$	\$ 20 \$	2.0 \$	2.0 \$	2.0 \$	2.0 \$	2.0 \$	\$ 0.2	2.0 \$	20	5.02	X0.2
Revolver Credit Facility Retorner		!	ć a	, 1	ģ	370	ş	ž	ķ	a K	5	č.	5	SB
Borrowings		25.8	1.8	1.4	7.07	2 '	? '	0.3	3 5	9 1 E	1.6	0.00	19	or
Kepayments Ending Balance		. 25.8	27.5	29.0	27.6	27.0	25.4	25.7	25.8	24.0	25.6	24.1	26.0	23.80
(1) Payments for change of control agreements for 6 senior officers stotals -\$15M and has not been included in this forecast. (2) Assume Professional Fee Order assumes submission of monthly involves, no objection 20 day period and payment of 80% of fees and 100% of expenses following 20 day period. (3) Upon dosing a sale approximately \$9M will need to be funded under the DP to cover Additional Finding at Closing costs. (4) Per the DIP Credit Agreement the following additional amounts in Schedule 1 would be paid at the closing upon the sale to Hercules. For #lustrative purposes, we have assumed the transaction closes 60 days efter filling.	nifor officers totals in of monthly invoice be funded under the standards in School	-\$15M and has not es, no objection 20 e DiP to cover Addi dule 1 would be pai	been included in the day period and pay front Funding at Clional Funding at Clional at the closing up.	is forecast. ment of 80% of fee losing costs. on the sale to Herou	of fees and 100% of expenses following 20 day period. Haroules. For illustrative purposes, we have assumed the	penses following purposes, we ha	20 day period. ave assumed the tra	insaction closes 6) days efter Min	rt.				2/12/11
	Schedule 1 Employee froe	chedule 1 Employee Incentive Payments CIC Payments			5,5 3,3									l Pa
	Final Fee App for Advisors Success Fee - Advisors	for Advisors - Advisors			(0.8) (1.6)									age
-	Subtotal to Employees DIP Lender Exit Fee	Subtotal to Employees and Advisors DIP Lender Exit Fee	sors		(6.3)									26
-	Total Addition	ouboout to United States at Closing Total Additional DIP Utilization at Closing	t Closing	i.	(8.2)									of
ķī.														60

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	§ Chapter 11
	§
Seahawk Drilling Inc.,	§ Case No.: 11
	§
Seahawk Mexico Holdings LLC,	§ Case No.: 11
	§
Seahawk Drilling Management LLC,	§ Case No.: 11
	§
Seahawk Offshore Management LLC,	§ Case No.: 11
	§
Energy Supply International LLC,	§ Case No.: 11
a 1 1 D 'III' TT C	S Cara Na . 11
Seahawk Drilling LLC,	§ Case No.: 11
Good and Clabal Haldings III C	S Caga No. 11
Seahawk Global Holdings LLC,	§ Case No.: 11
Sochavele Deilling LISA LLC	§ Case No.: 11-
Seahawk Drilling USA LLC,	8
Debtors.	§ Motion for Joint Administration Pending
Debtois.	y 1410tion for some reministration remains

DEBTORS' EMERGENCY MOTION, PURSUANT TO SECTIONS 105, 363 AND 365 OF THE BANKRUPTCY CODE AND RULES 2002, 6004, 6006, 9007, AND 9014 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE, FOR AN ORDER (i) SCHEDULING AN EXPEDITED HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS; (ii) SETTING DEADLINES FOR FILING OBJECTIONS TO (1) THE PROPOSED SALE AND (2) THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND THE PAYMENT OF CURE CLAIMS IN CONNECTION THEREWITH; AND (iii) APPROVING THE FORM AND MANNER OF NOTICE THEREOF

NOTICE UNDER BLR 9013(b) AND 9013(i)

THIS MOTION SEEKS AN ORDER THAT MAY ADVERSELY AFFECT YOU. IF YOU OPPOSE THE MOTION, YOU SHOULD IMMEDIATELY CONTACT THE MOVING PARTY TO RESOLVE THE DISPUTE. IF YOU AND THE MOVING PARTY CANNOT AGREE, YOU MUST FILE A RESPONSE AND SEND A COPY TO THE MOVING PARTY. YOU MUST

FILE AND SERVE YOUR RESPONSE WITHIN 21 DAYS OF THE DATE THIS WAS SERVED ON YOU. YOUR RESPONSE MUST STATE WHY THE MOTION SHOULD NOT BE GRANTED. IF YOU DO NOT FILE A TIMELY RESPONSE, THE RELIEF MAY BE GRANTED WITHOUT FURTHER NOTICE TO YOU. IF YOU OPPOSE THE MOTION AND HAVE NOT REACHED AN AGREEMENT, YOU MUST ATTEND THE HEARING. UNLESS THE PARTIES AGREE OTHERWISE, THE COURT MAY CONSIDER EVIDENCE AT THE HEARING AND MAY DECIDE THE MOTION AT THE HEARING.

EMERGENCY RELIEF HAS BEEN REQUESTED. IF THE COURT CONSIDERS THE MOTION ON AN EMERGENCY BASIS, THEN YOU WILL HAVE LESS THAN 21 DAYS TO ANSWER. IF YOU OBJECT TO THE REQUESTED RELIEF OR IF YOU BELIEVE THAT THE EMERGENCY CONSIDERATION IS NOT WARRANTED; YOU SHOULD FILE AN IMMEDIATE RESPONSE.

REPRESENTED PARTIES SHOULD ACT THROUGH THEIR ATTORNEY.

TO THE HONORABLE UNITED STATES BANKRUPTCY JUDGE:

Seahawk Drilling, Inc. ("Seahawk" of the "Company") and its above-captioned affiliated debtors (collectively, the "Debtors"), file this Emergency Motion Pursuant To Sections 105, 363 and 365 of the Bankruptcy Code And Rules 2002, 6004, 6006, 9007, and 9014 of the Federal Rules Of Bankruptcy Procedure, for an Order (i) Scheduling an Expedited Hearing To Approve The Sale of Substantially All of the Debtors' Assets; (ii) Setting Deadlines for Filing Objections To (1) the Proposed Sale and (2) the Proposed Assumption and Assignment of Executory Contracts and the Payment of Cure Claims in Connection Therewith; and (iii) Approving the Form and Manner of Notice Thereof (the "Sale Motion"). In support of this Sale Motion, the Debtors respectfully state:

JURISDICTION AND PROCEDURAL STATUS

- 1. This Court has jurisdiction over this Sale Motion pursuant to 28 U.S.C. §§ 157 and 1334. Venue of the Debtors' chapter 11 cases in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409. This is a core proceeding pursuant to 28 U.S.C. § 157(b).
- 2. On the date hereof (the "<u>Petition Date</u>"), the Debtors filed their respective voluntary petitions for relief under title 11 of the United States Code (the "<u>Bankruptcy Code</u>"). Contemporaneously herewith, the Debtors have sought joint administration of these chapter 11 cases.
- 3. The Debtors continue to operate their businesses and manage their properties as debtors-in-possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee, examiner, or creditors' committee has been appointed in these cases.

FACTUAL BACKGROUND

A. Events Leading To The Debtors' Petitions

- 1. Seahawk is a Delaware corporation formed in 2008 and headquartered in Houston, Texas, with operations throughout the U.S. Gulf of Mexico. Each of the other debtors is a direct or indirect subsidiary of Seahawk.¹ As of the Petition Date, Seahawk's stock is publicly traded on the NASDAQ Exchange under the ticker symbol, "HAWK."
- 2. Seahawk operates a jackup rig business that provides contract drilling services to the oil and natural gas exploration and production industry throughout the U.S. Gulf of Mexico.²

¹ The debtor-subsidiaries of Seahawk are: Seahawk Drilling LLC, Energy Supply International LLC, Seahawk Global Holdings LLC, Seahawk Mexico Holdings LLC, Seahawk Drilling Management LLC, Seahawk Offshore Management LLC and Seahawk Drilling USA LLC. The subsidiaries of Seahawk that are incorporated in Mexico or have branches in Mexico are not a part of this chapter 11 proceeding.

² Jackup rigs are mobile, self-elevating drilling platforms equipped with legs that can be lowered to the ocean floor until a foundation is established to support the drilling platform. The Debtors' rigs work in water depths up to 300 feet and can drill to a depth of 25,000 feet.

The Debtors contract with their customers on a dayrate basis to provide rigs and drilling crews.

The Debtors' fleet of mobile offshore drilling rigs consists of twenty (20) shallow water jackup rigs. Three (3) of the Debtors' six (6) rigs in Texas are located at Matagorda Island 721 in Aransas County. Two rigs are located at Sabine Pass, Texas, and one rig is located at High Island 176 in Galveston County. The remainder of Seahawk's rigs are currently located in Louisiana.

- 3. Seahawk is a former subsidiary of Pride International, Inc. ("<u>Pride</u>"). On August 4, 2009, the Board of Directors of Pride approved a plan to separate Pride into two independent, publicly-traded companies (the "<u>Spin-Off</u>"). The separation occurred through the distribution to Pride stockholders of all of the shares of common stock of Seahawk that held, directly or indirectly, the assets and liabilities of Pride's jackup rig business.
- 4. On August 24, 2009 (the "Spin-Off Date"), each Pride stockholder received one-fifteenth (1/15) of a share of Seahawk's common stock for each share of Pride common stock held at the close of business on August 14, 2009. After the Spin-Off, Seahawk became an independent public company. Pride no longer retains any ownership interest in Seahawk, although the two companies are parties to various agreements entered into in connection with the Spin-Off (collectively, the "Pride Agreements").
- 5. Since mid-2008, the demand for drilling services has declined dramatically, principally as a result of the global financial crisis, declining prices of crude oil and natural gas and deteriorating worldwide economic conditions. The decline in the United States jackup rig market since 2009 has been one of the sharpest downturns for domestic jackup rig activity over the past thirty (30) years. In addition, the regulatory and financial uncertainties regarding a

former customer, PEMEX, have had a significant effect on Seahawk's business.³ Finally, from August 2009 to March 2010, the Debtors' active rig count averaged five (5) to seven (7) working rigs at any given time.

- 6. On April 20, 2010, the demand for offshore drilling services in the Gulf of Mexico was further negatively impacted by the Macondo well blowout, prompting, among other things, the United States Government to issue a moratorium on all U.S. offshore drilling. On May 26, 2010, the moratorium on Gulf of Mexico drilling in waters less than 500 feet deep was lifted. Subsequently, in November 2010, the deep water moratorium was lifted. However, since the Macondo incident, notwithstanding the termination of the Government-imposed drilling prohibitions, the Debtors' customers are experiencing significant delays in the issuance of drilling permits and very few new drilling permits have been issued. In addition, there is an increasingly uncertain regulatory and cost environment which continues to adversely affect the Debtors' business.
- 7. As a result of all of these factors, Debtors' active rig count declined to three (3) working rigs during October 2010. While there has been some recent marginal improvement in market conditions (seven (7) rigs are currently working), the combined impact of all of these events, together with negative cash flows throughout 2009 and 2010, has severely stressed and exhausted the Debtors' liquidity and even the current operating level results continue to produce negative cash flows for the Debtors. Seahawk's Board of Directors (the "Board") and its management have closely monitored the impact of these conditions and evaluated potential alternatives to address and improve these circumstances with the assistance of various professionals and outside advisors, including Alvarez & Marsal North America, LLC ("A&M").

³ Pemex Exploración y Producción ("<u>PEMEX</u>") is the exploration and production subsidiary of the national oil 90303148.17

- 8. In November 2010, Seahawk publicly announced that its Board had initiated a process to explore and consider possible strategic alternatives, including, but not limited to transactions involving additional funding, recapitalizations, sales of assets, or a sale or merger of Seahawk. Seahawk's Board, through its finance committee (the "Finance Committee"), with regular and substantial Board and independent directors' participation, supervised the process and were managed by Simmons & Company International ("Simmons") and other advisors.
- 9. In February 2011, after evaluating the indications of interest and potential offers received from a number of interested parties, the Board authorized the Debtors to enter into an Asset Purchase Agreement (the "APA") with Hercules Offshore, Inc. ("Hercules"). The executed APA contemplates the acquisition by Hercules or one or more of its subsidiaries ("Purchaser") of substantially all of the assets and jackup rigs of the Debtors (the "Purchased Assets") through a sale pursuant to section 363 of the Bankruptcy Code (the "Transaction"). The aggregate consideration for the Purchased Assets (defined in the APA as the "Base Aggregate Consideration") is (a) 22,321,425 shares of Hercules Common Stock (the "Hercules Shares") plus (b) cash in an amount equal to \$25,000,012 (the "Cash Payment"), subject to certain adjustments. Using the closing stock price of Hercules' stock as of February 10, 2011, the Base Aggregate Consideration would be valued at approximately \$105 million before any adjustments. The Base Aggregate Consideration is to be payable at closing by Purchaser to Sellers.
- 10. The Debtors have sought chapter 11 protection in order to protect and preserve their going concern value and to facilitate a prompt sale of substantially all of their assets to Hercules for the benefit of all stakeholders. The Debtors have determined in the prudent exercise

company of Mexico.

⁴ The summary description of the APA set forth herein is qualified in its entirety by reference to the terms of the APA itself.

of their business judgment that the proposed course of action is the best alternative to ensure that the maximum value of Seahawk's assets will inure to the benefit of all of their creditors and interestholders.

11. Additional information regarding the Debtors' business, capital structure, and the circumstances and events leading to these chapter 11 cases is contained in the Declaration of Randall D. Stilley in Support of First Day Motions, filed contemporaneously herewith and incorporated herein by reference.

B. The Proposed Sale

- 12. The Debtors filed these chapter 11 bankruptcy cases to effectuate the terms and provisions of the APA by and between the Debtors and Hercules (collectively, the "Purchaser") under a sale pursuant to section 363 of the Bankruptcy Code. A true and correct copy of the executed APA is attached hereto as **Exhibit A** and incorporated herein for all purposes.
- 13. One of the conditions precedent to the closing of the APA is that this Court enter an order approving the APA. The Debtors have concluded that a chapter 11 filing designed to facilitate a sale of the Debtors' assets to the Purchaser through a section 363 sale would most efficiently consummate the sale and is necessary to prevent continued operating losses and diminution of the Debtors' enterprise value. Under the APA, the Debtors also intend to assume and assign virtually all of their executory contracts, but not unexpired leases, to the Purchaser.
- 14. The Debtors, in consultation with Simmons, and its Board have concluded that the Purchaser's offer was the highest and best offer received. In the opinion of Simmons and the Board, further marketing efforts would not likely result in any offers superior to the Hercules

APA. The Declaration of Simmons & Company International is attached hereto as **Exhibit 1** and incorporated herein by reference for all purposes.

- Pursuant to the terms and conditions of (i) that certain Debtor-In-Possession Credit 15. Agreement (the "DIP Credit Agreement") by and between Seahawk and D.E. Shaw Direct Capital Portfolios, L.L.C. and certain lenders thereto (the "DIP Lenders") and (ii) that certain proposed Interim Order Pursuant to 11 U.S.C. §§ 105, 361, 362, 363, and 364 of the Bankruptcy Code (I) Authorizing the Debtors to Obtain Postpetition Financing and Use Cash Collateral and (II) Granting Liens, Security Interests and Superpriority Claims (the "DIP Order"), any sale of all or substantially all of the Debtors' assets is prohibited unless the sale generates sufficient proceeds to pay all DIP Obligations (as defined in the DIP Order), all DIP Obligations are indefeasibly paid in full in cash, and the commitments under the DIP Credit Agreement are terminated. As a result, the APA provides for a cash portion of the Base Aggregate Consideration (as defined in the APA) sufficient to pay in full in cash the DIP Obligations. Concurrently with the closing of the sale, the cash portion paid by Hercules to the Debtors will be used to indefeasibly pay the DIP Obligations and the commitments to make loans under the DIP Credit Agreement will terminate. The Hercules Shares received by the Debtors will be held by an escrow agent to be distributed to holders of allowed claims against and interests in the Debtors pursuant a confirmed chapter 11 plan or pursuant to further orders of this Court.
- 16. The Debtors believe that because it is anticipated that the purchase price will fully satisfy all claims against the Debtors and Hercules' offer is the highest and best offer the Debtors have been able to obtain, the APA is in the best interests of the Debtors and all parties-in-interest, and accordingly, should be approved.

THE NEED FOR AN EXPEDITED SALE

17. As set forth above, and as further detailed in the First Day Affidavit of Randall D. Stilley filed contemporaneously herewith, the Debtors have been substantially affected by the downturn in the natural gas market, the moratorium on new offshore oil and gas drilling permits and general economic conditions. The Debtors have suffered operating losses prior to the Petition Date and likewise have projected that they will continue to suffer operating losses throughout 2011. Every day the Debtors continue their operations, and even during the pendency of these bankruptcy cases, the Debtors assets continue to experience a further deterioration in value.

18. Furthermore, an expedited sale of the Debtors' assets is a mandatory condition of the APA. Without an expedited sale, the APA will not be consummated and the Debtors' estate and all the Debtors' creditors and shareholders will lose a significant opportunity to achieve what the Debtors believe is the highest and best recovery on their claims and interests.

RELIEF REQUESTED

19. The Debtors request that, at the initial hearing on this Sale Motion (the "Preliminary Sale Hearing"), the Court enter an order substantially in the form attached hereto as Exhibit C, that: (1) schedules an expedited hearing to consider final approval of the Sale Motion; (2) sets deadlines for filing (a) objections to the Sale Motion, and (b) objections to the proposed assumption and assignment of executory contracts in connection with the sale and the payment of cure claims in connection therewith; and (3) approves the form and manner of notice of the final hearing to approve the sale ("Final Sale Hearing").⁵

A. REQUEST FOR AN EXPEDITED HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS.

⁵ The Debtors have filed a separate motion seeking approval of the Termination Fee payable to Purchaser under certain conditions.

- 20. The Debtors submit that the sale of the Purchased Assets is essential to preserve the going concern value of Debtors' business. The Debtors assert that each day the Purchased Assets are subject to the market and the bankruptcy process will cause further losses in operating revenue, diminish the value of the Purchased Assets and threaten the viability of projected creditor and stakeholder recoveries. Accordingly, from the Debtors' perspective, the APA represents a reasonable exercise of the Board's and management's business judgment and an expedited sale is in the best interests of the Debtors and all of their economic claim and interestholders.
- 21. The Debtors and Simmons have expended substantial time and resources soliciting potential purchasers of the Debtors' assets pre-petition and accordingly are of the opinion that the APA with the Purchaser represents the best terms the Debtors can reasonably anticipate to obtain and that the process they have implemented is comprehensive, fulsome and appropriate in every respect.

KEY TERMS OF THE APA

22. A summary of the pertinent provisions of the APA are as follows⁶:

Seller (APA p.1)	The Debtors
Purchasers (APA p.1)	Hercules Offshore, Inc. and SD Drilling LLC, a Delaware limited liability company ("Newco")
Purchase Price (APA § 2.4)	Subject to the other terms of this Agreement, the aggregate consideration for the Purchased Assets shall consist of (a) 22,321,425 shares of Hercules Common Stock (the "Hercules Shares") and (b) cash in an amount equal to \$25,000,012 (the "Cash Payment" and, together with the Hercules Shares, the "Aggregate Consideration")

The summary of the APA is intended only to assist the Court and parties-in-interest in understanding the key aspects of the transaction. In the event of any conflict between this summary and the APA, the APA shall control. The APA can be amended by the parties pursuant to a written amendment, provided that the amendment does not materially reduce the consideration provided to the Debtors' estates. Any capitalized terms not defined herein shall have the meaning set forth in the APA.

Purchased Assets (APA § 2.1(a)– (n))

All of the Debtors' right, title, and interest in and to all of the property, including but not limited to:

Rigs. The jackup Rigs listed on <u>Schedule 2.9</u>, each of which is located in the United States of America (the "Transferred Rigs");

Contracts. All of the interests, rights, Claims, and benefits arising or accruing to any of Sellers under any Contract (including any Lease) to which a Seller is a party or has or may acquire a benefit and that relate to the Purchased Assets or Business (the "Assigned Contracts"), including the Assigned Contracts listed in Schedule 2.3(a), to the extent Sellers' interest in such Contract is transferable;

Equipment. All furniture, equipment, computers, computer equipment, machinery, tools, hand tools, spare parts, test equipment, supplies, inventory, office supplies, telephones, and all other tangible personal property of every kind and description insofar as any of the foregoing relates to the operation of the Purchased Assets or Business (the "Equipment"), including the Equipment listed in Schedule 2.1(c):

Rolling Stock. All automobiles, vans, trucks, trailers, and other motorized and similar vehicles and stock of every kind, whether owned or leased, used in the operation of the Purchased Assets or Business, including as listed in Schedule 2.1(d);

Additional Tangible Assets. All other Tangible Personal Property of every kind used in the operation of the Purchased Assets or Business, together with any rights appurtenant thereto, including any express or implied warranty by the manufacturer, vendor, or lessor of any such Tangible Personal Property;

Accounts and Notes Receivable. All trade accounts receivable, notes receivable, and other rights of Sellers to payment from customers and other third parties and other amounts due from customers and other third parties or that become due, including all assets constituting working capital, that relate to or arise from the operation of the Purchased Assets or Business;

Insurance Benefits. All insurance benefits arising from or related to the Purchased Assets, Business, and Assumed Liabilities;

Cash. All cash and cash equivalents, securities, money on deposit with banks, certificates of deposit, and similar short-term investments (if any) of Sellers;

Prepaid Deposits and Expenses. Any deposits and prepaid expenses (paid

to or by Sellers), claims for refunds, and rights of set off related to the Purchased Assets or Business;

Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities against third parties relating or attributable to the Business, the Purchased Assets, or the Assumed Liabilities, whether choate or inchoate, known or unknown, contingent or non-contingent;

Permits. Any and all Permits, including pending applications or filings therefor and renewals thereof, of every kind under which any of Sellers has or may acquire any benefits or rights, or by which any of Sellers or any of the Purchased Assets may be subject or bound, or that relate to or are used or held for use in the operation of the Purchased Assets or Business (collectively, the "Assigned Permits");

Intangible Assets. Any and all customer and supplier relationships, and other Intangible Personal Property of every kind (other than the Computer Software, Trademark Intellectual Property and Website Intellectual Property, and uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names that constitute Intellectual Property) that relate to or are used in the operation of the Purchased Assets or Business, together with all rights appurtenant thereto (the "Assigned Intangible Assets");

Books and Records. Original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets and photocopies of the other Books and Records (collectively, the "Transferred Books and Records");

Other Assets. To the extent not otherwise enumerated in this Section 2.1 all other tangible or intangible assets, rights, privileges, benefits, Claims, and interests of a Seller, whether real, personal or mixed, of every kind and description and wherever located, that relate to, used in or held for use in the operation of the Purchased Assets or Business.

Excluded Assets (APA § 2.2(a)— (0))

Notwithstanding anything to the contrary in <u>Section 2.1</u> or elsewhere in this Agreement or other Transaction Documents, the Purchased Assets shall not include:

Transaction Rights. All rights of Sellers under this Agreement and the other Transaction Documents, and all cash and non-cash consideration (including the Hercules Shares and the Cash Payment) payable or deliverable to, or on behalf of, Sellers by Purchasers pursuant hereto and thereto;

Rigs. Any Rig that is deemed to be an Excluded Asset pursuant to <u>Section 2.9</u>, together with all of the Equipment located on each such Rig;

Contracts. All rights of Sellers under any Contract that is not an Assigned Contract, including those Contracts listed in Schedule 2.2(c);

Permits. Any and all rights of Sellers under any Permit that is not an Assigned Permit;

Insurance Benefits. Any and all insurance benefits arising from or relating to (i) Excluded Assets and Retained Liabilities, (ii) Claims made by Sellers prior to the Closing with respect to the Purchased Assets and Business, and (iii) the matters set forth in Schedule 2.2(e);

Books and Records. The original copies of all of the Books and Records, other than the original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets (which shall be provided to Purchasers pursuant to Section 2.1(m) and of which Sellers may retain photocopies), and any Books and Records of Sellers (i) that are not permitted to be transferred to Purchasers under applicable Law, (ii) that constitute charters, bylaws, limited liability company agreements, minute books, stock transfer records, and other records related to the corporate governance of Sellers, and (iii) all other books and records of Sellers that do not relate primarily to the Purchased Assets, Transferred Employees, or Business (collectively, the "Retained Books and Records");

Third Party Property. Any improvements, equipment, inventory and any other tangible personal property located at the Leased Real Property or on the Transferred Rigs as of the Closing Date that are not owned by or leased to Sellers; and

Equity Interests. Any Equity Interest in any Seller or any direct or indirect subsidiary of a Seller;

Bankruptcy Claims. Any and all Claims or rights of Sellers arising under chapter 5 of the Bankruptcy Code;

Pride Claims. Any and all Claims or rights of Sellers or any Affiliate thereof against Pride or any Affiliate thereof;

Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities of Sellers, whether choate or inchoate, known or unknown, contingent or non-contingent with respect to the matters set forth in Schedule 2.2(k);

Leased Real Property. All leasehold interests in the real property listed in Schedule 2.2(1) (the "Leased Real Property"), together with all Real Property Rights related thereto and all Improvements and Tangible Personal

Property located at, under, or on such Leased Real Properties or otherwise related thereto;

Intellectual Property. The Intellectual Property constituting: (i) Sellers' software programs, tools, kits, and any content or related documentation or third party or open source code embedded therein, either locally stored on Sellers' computers or remotely accessed by Sellers, that is owned or made available to Sellers and is used in, held for use in, or is necessary for the operation of the Purchased Assets or Business or otherwise, and any upgrades, updates, releases, fixes, enhancements, or modifications thereto, and all written materials and specifications applicable thereto (collectively, the "Computer Software"); (ii) the name "Seahawk Drilling" and the mark "Seahawk Drilling" and any other registered or unregistered trademarks, trade names, service marks, domain names, and email addresses, including the term "Seahawk", used or held for use in the Business (collectively, the "Trademarks"), and any and all goodwill associated with the Business embodied in the Trademarks, and any and all rights of Sellers with respect to the Trademarks (collectively, the "Trademark Intellectual Property"); (iii) Sellers' internet website located at www.seahawkdrilling.com and any derivations thereof (collectively, the "Website") and all intellectual property rights that may exist or arise in connection with the Website, including all Underlying Technology (collectively, the "Website Intellectual Property"); and; (iv) uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names.

Goodwill. Any goodwill and going concern value related to the Business; and

Accounts and Notes Receivable. The trade accounts receivable, notes receivable, and other rights of Sellers to payment from, and other amounts due or that become due, including all assets constituting working capital, that relate to Blake International USA Rigs, LLC or any Affiliate thereof.

Assigned Contracts (APA § 2.3)

Transfer of Contracts and Permits. At least three Business Days prior to the Closing Date, Purchasers may designate any Contracts that would otherwise be an Assigned Contract as a Contract to be excluded from the Purchased Assets for purposes of Section 2.1(b). Any such designation by Purchasers shall not reduce Purchaser's assumption of any working capital liabilities pursuant to Section 2.7(a), unless both (i) such working capital liability was incurred pursuant to such designated Contract and (ii) such working capital liability is not attributable to goods and services provided to the Business prior to the Closing, in which case such working capital liability shall not be assumed under Section 2.7(a).

Notwithstanding anything herein to the contrary, this Agreement shall not constitute an assignment to Purchasers of (and Purchasers shall not assume

any rights or obligations under) any Contract or Permit if an attempted assignment thereof, without the Consent of a Governmental Authority or other third party thereto, would constitute a breach of the Contract or Permit, unless such Consent has been received or such assignment is otherwise authorized and permitted by the Transfer Order or other Order of the Bankruptcy Court. The Transfer Order shall provide that any Assigned Contracts that are executory contracts shall be assumed by Sellers and assigned to Purchasers at the Closing, and Sellers shall pay all pre-Closing cure amounts under Assigned Contracts required to be so paid by the Bankruptcy Court in connection with such assumptions by Sellers (the "Cure Amounts"). The Sellers shall use their commercially reasonable efforts to advise the Purchasers promptly in writing with respect to any Contract or Permit that Sellers' Know or have substantial reason to believe shall not or may not be subject to assignment to Purchasers hereunder. Without in any way limiting Sellers' obligations to obtain all Consents necessary for the sale, transfer, assignment and delivery of the Assigned Contracts and Assigned Permits and the other Purchased Assets to Purchasers hereunder, if any such Consent is not obtained or if such assignment is not permitted irrespective of Consent and the Closing hereunder is consummated, Sellers shall use commercially reasonable efforts to continue to promptly seek Consent or other approval for the sale, transfer, assignment and delivery thereof; provided, that in no circumstance shall any such obligation of Sellers extend beyond the date that Sellers' Plan of Reorganization is confirmed by the Bankruptcy Court and has become effective.

Assumed Liabilities (APA § 2.7(a)— (e))

The Assumed Liabilities include:

Working Capital Liabilities. The current accounts payable of, and accruals for good and services received by, Sellers as of the Closing Date attributable to the Purchased Assets; provided that, for purposes of clarity, such working capital liabilities shall exclude accounts payable to Blake International USA Rigs, LLC or its Affiliates. Nothing in this Section 2.7(a) shall be deemed to effect the assumption of any Contract that is not otherwise an Assigned Contract, but the assumption of liabilities under this Section 2.7(a) shall be subject to Section 2.3(a).

Contracts. The Liabilities of Sellers under the terms of any Assigned Contract to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the closing;

Permits. The Liabilities of Sellers under the terms of any Assigned Permit to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the Closing;

Claims relating to Assumed Liabilities. In the event Claims are filed in the

Sellers' Bankruptcy Cases on account on the Assumed Liabilities, Sellers shall provide Purchasers written notice of such filing within five (5) Business Days after their filing. Purchasers shall have the right to appear, defend and settle those Claims in the Bankruptcy Court. Notwithstanding the foregoing, Sellers shall be solely responsible for any Claims that exceed the amount of current liabilities assumed under Section 2.7(a) and included in the post-Closing adjustment process under Section 2.10; and

Performance Bond. Purchasers shall either assume Sellers' \$2.0 performance bond in favor of the Texas general land office relating to the Seahawk 2008 Rig, or otherwise post their own performance bond such that such \$2.0 performance bond may be terminated by Sellers.

Retained Liabilities (APA § 2.8(a)– (b))

The Excluded Liabilities include:

Subject to the terms of the Transfer Order, Sellers shall retain and be solely liable for and hereby expressly agree to retain any and all of their respective Liabilities not expressly assumed by Purchasers in Section 2.7 (collectively, the "Retained Liabilities"), regardless of whether any such Retained Liability is disclosed herein or in any Schedule hereto, whether known or unknown, absolute or contingent, liquidated or unliquidated, whether due or to become due, and whether Claims with respect thereto are asserted before or after the Closing Date. Notwithstanding the preceding sentence, Retained Liabilities shall include the following:

- Transaction Liabilities. Any and all Liabilities of Sellers under the Transaction Documents;
- Excluded Assets. Any and all Liabilities of Sellers with respect to the Excluded Assets;
- Purchased Assets and Pre-Closing Business. Except as provided in Sections 2.7(a), 2.7(b), and 2.7(c), all Claims and Liabilities arising out of Sellers' ownership, operation, use, or maintenance of the Purchased Assets or Business, as well as any matters arising from events occurring, conditions existing, or costs accruing prior to the Closing and Taxes;
- Contracts. Except as provided in Sections 2.7(a) and 2.7(b), any and all Liabilities of Seller, including Cure Amounts, that relate to the Assigned Contracts to the extent any such Liability is performable or otherwise attributable to the period prior to the Closing;
- Permits. Except as provided in <u>Sections 2.7(a)</u> and <u>2.7(c)</u>, any and all
 Liabilities of Sellers that relate to the Assigned Permits to the extent
 any such Liability is performable or otherwise attributable to the period

prior to the Closing;

- Taxes. Any and all Liabilities of Sellers for Taxes that relate to (A) the ownership, operation, use, or maintenance of the Purchased Assets or Business prior to the Closing Date, or (B) any sales, use, transfer, or other similar Taxes imposed as a result of the consummation of the Contemplated Transactions or performance of the Transaction Documents;
- Legal Proceedings. Any and all Liabilities of Sellers that relate to any Proceeding (A) involving the Purchased Assets or Business, including warranty, personal injury, breach of contract, failure to perform, infringement, noncompliance with Law, and tort Claims, that is (1) pending or threatened as of the Closing, or (2) commenced after the Closing but that arises out of or relate to any event, omission, or occurrence happening as of or prior to the Closing, or (B) set forth in Schedule 2.8(a)(vii);
- Pride Claims. Any and all Liabilities of Sellers under agreements entered into among Seahawk Parent and Pride International, Inc. ("Pride") at the time of Seahawk Parent's spin-off from Pride effective August 4, 2009, including (i) any obligation to indemnify Pride and its Affiliates for Mexican tax liabilities associated with Seahawk Parent's Mexican subsidiaries under that certain Tax Sharing Agreement dated August 4, 2009, between Seahawk Parent and Pride, part of which tax liabilities are supported by letters of credit from Pride, and any settlement with respect to such matters, and (ii) any and all Liabilities of Sellers (if any) to Pride and its Affiliates under such agreements resulting from the Contemplated Transactions;
- Pride Wyoming. Any and all Liabilities of Sellers relating to the loss of the Pride Wyoming Rig, including salvage and wreckage removal costs and any third party damage Claims and any continuing Liability under applicable Law relating to the Pride Wyoming Rig;
- FCPA Investigation. Any and all Liabilities of Sellers, including for purposes of clarity any and all fines and penalties (if any), of Sellers with respect to violations, alleged violations, and investigations for alleged potential or actual violations by Sellers and any of their respective Affiliates of the U.S. Foreign Corrupt Practices Act;
- Environmental Liabilities. Any and all Liabilities of Sellers with respect to any violation of Law including those arising from (A) the release, threatened release, presence, treatment, storage, disposal (including disposal at off site locations), handling, transportation or arrangement for transportation of hazardous substances prior to the Closing, (B) any failure of Sellers to comply in any respect with

- Environmental, Health, and Safety Laws prior to the Closing, and (C) any facts, events, or circumstances in existence prior to the Closing that give rise to Liabilities pursuant to Environmental, Health, and Safety Laws;
- Employee Benefit Plans. Any and all Liabilities of Sellers, fiduciaries or ERISA Affiliate under any Employee Benefit Plan maintained or contributed to by any Seller, fiduciary or any ERISA Affiliate or with respect to which any Seller, fiduciary or ERISA Affiliate has any Liability, whether prior to, on, or after the Closing, including any (a) failure to comply with all applicable Laws, (b) liability with respect to audits, inquiries, Proceedings, or Claims with any Governmental Authority with respect to any Employee Benefit Plan, (c) participation in any "multiemployer" plan (within the meaning of Section 3(37) of ERISA, whether or not governed by the provisions of ERISA) or withdrawal liability with respect to any multiemployer plan, (d) required employer contributions with respect to the Employee Benefit Plans, (e) accumulated funding deficiency, (f) Lien under ERISA or Section 412 of the Code, or (g) the termination of, or intent to terminate, any Plan;
- Employees. Any and all Liabilities of any nature of Sellers or any fiduciary or ERISA Affiliate to Seller Employees, including Liabilities with respect to (a) any Contract, plan or policy, (b) wages, withholdings, overtime pay, minimum wage, employment Tax, vacation, sick pay, bonuses, severance pay, retirement, or other compensation, (c) benefits under Employee Benefit Plans, (d) the Worker Adjustment and Retraining Notification Act (WARN) of August 4, 1988 or equivalent state or local statutory or regulatory requirements, (e) any collective bargaining agreement or obligation or requirement under the National Labor Relations Act, (f) reporting, filing, hiring or other employment obligations with the Office of Federal Contract Compliance Programs, (g) all immigration related obligations, including all requirements of the Immigration Reform and Control Act of 1986, (h) any governmental or administrative proceeding for the enforcement of labor and employment laws and regulations, and (i) all other employment and employment related federal, state and local statutes, regulations, administrative requirements, common laws, and public policies;
- Intercompany Liabilities. Any and all Liabilities of Sellers for intercompany advances, charges, or accounts payable of any kind or nature;
- Bankruptcy Claims. Any and all Claims filed against Sellers in their respective Bankruptcy Cases, except for the Assumed Liabilities, as

	limited in Section 2.7(a);	
	• Broker Fees. Any and all fees, commissions, and other compensation to any broker, finder, or agent retained by any Seller as contemplated by Section 4.22;	
	The Retained Liabilities shall constitute Claims and alleged Claims in the Sellers' Bankruptcy Cases; <i>provided</i> , <i>however</i> , that nothing herein shall create any rights in favor of the holders of such Claims and alleged Claims or create any priority right of payment.	
Releases (APA §9.9)	In consideration for payment of the Cash Payment, issuance of the Hercules Shares and assumption of the Assumed Liabilities by Purchasers, Sellers shall cause the Plan of Reorganization to contain a mutual release provision by Sellers in favor of Purchasers and by Purchasers in favor of Sellers on terms mutually acceptable to the Parties.	
Closing (APA § 3.1)	Provided that this Agreement shall not have been earlier terminated pursuant to Section 11.1, and further provided that all of the conditions set forth in Sections 10.1 and 10.2 to the obligations of the Parties to consummate the Contemplated Transactions (other than conditions with respect to actions each Party will take at the Closing itself) shall have been satisfied or waived, the closing of the Contemplated Transactions (the "Closing") shall take place at the offices of Fulbright & Jaworski L.L.P. at 1301 McKinney, Suite 5100, Houston, Texas 77010 commencing at 11:00 a.m. Central Time on the date that is no earlier than the first Business Day that the Transfer Order has become a Final Order, but no later than the third (3rd) Business Day following such date, or such other date or location as Purchasers and Sellers may mutually agree to in writing (the "Closing Date").	
Termination Provisions (APA § 11.1(a)— (d))	 Mutual Consent. Purchasers and Sellers may terminate this Agreement as to all Parties by mutual written consent at any time prior to the Closing by written instrument authorized by the respective Boards of Directors of Seahawk Parent and Hercules Parent. By Purchasers. Purchasers may terminate this Agreement by giving written notice to Sellers at any time prior to the Closing in the event that (i) Sellers have breached any representation, warranty, or covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.1(a) and 10.1(b) shall not have been satisfied, or cannot be satisfied by August 15, 2011 (the "Termination Date"); provided, that Purchasers shall have provided written notification to Sellers of such breach and the breach shall 	

have continued without cure for a period of ten (10) days after delivery of the notice of such breach; (ii) the Transfer Order shall not have been entered by the Bankruptcy Court and become a Final Order by one hundred twenty (120) days after the date hereof; (iii) any of the Bankruptcy Cases are converted to cases pursuant to Chapter 7 of the Bankruptcy Code; (iv) a Chapter 7 or Chapter 11 Trustee has been appointed; (v) an Examiner with expanded powers has been appointed in any of the Bankruptcy Cases of Sellers; (vi) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement; (vii) the DIP Financing is not approved within twenty-one (21) days of the Petition Date; (viii) Sellers are in breach or default in any material respect of, the DIP Credit Documents, DIP Order or DIP Budget after five days notice by any Person and opportunity to cure; or (ix) the Termination Fee Order shall not have been entered by the Bankruptcy Court within twenty-five (25) days of the date hereof or shall not have become a Final Order by the eleventh day after such entry of the Termination Fee Order. At their sole and absolute discretion, Purchasers have the right to waive termination or agree to extend any deadlines under this Section 11.1(b).

- By Sellers. Sellers may terminate this Agreement by giving written notice to Purchasers at any time prior to the Closing in the event that (i) any Purchaser has breached any representation, warranty, or covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.2(a) and 10.2(b) shall not have been satisfied, or cannot be satisfied by the Termination Date; provided, that Sellers shall have provided written notification to Purchasers of such breach and the breach shall have continued without cure for a period of ten (10) days after delivery of the notice of breach; or (ii) in accordance with the terms and subject to the conditions of Section 6.5; or (iii) Purchasers' failure to timely obtain the Consent as contemplated by Section 6.9.
- By Either Party. Sellers or Purchasers may terminate this Agreement by giving written notice to the other Party if (i) any court of competent jurisdiction or any other Governmental Authority in a suit instituted by a third party or a Governmental Authority shall have issued an Order or shall have taken any other action prior to the Termination Date permanently enjoining, restraining, or otherwise prohibiting the Contemplated Transactions or a material portion thereof, (ii) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement or denies approval of this Agreement or the Contemplated Transactions, or (iii) the Closing has not occurred by the Termination Date, provided, that the Party electing to terminate shall not have caused such failure to close.

- 23. At the Preliminary Sale Hearing, the Debtors request that the Court schedule an expedited hearing on the Sale Motion.
- B. REQUEST FOR THE SETTING OF DEADLINES FOR FILING OBJECTIONS TO (i) THE PROPOSED SALE AND (ii) TO THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND THE PAYMENT OF CURE CLAIMS IN CONNECTION THEREWITH
- 24. The Debtors request that, at the Preliminary Sale Hearing, the Court order that any objections to the Sale Motion (a "Sale Objection") set forth in writing with particularity the factual and legal basis for the objection and be filed by a party with standing and served on counsel for the Debtors, counsel for the Purchaser and all other parties entitled to notice no later than 12:00 p.m. (CST) on the third Business Day prior to the Sale Hearing Date (the "Objection Deadline").
- 25. The Debtors further request the Court to order that, unless an objection to the sale is timely filed and served by the Objection Deadline, all interested parties who have received actual or constructive notice thereof be deemed to have waived and released any right to assert a Sale Objection and shall be deemed to have otherwise consented to the proposed sale and shall be forever barred and estopped from asserting any Sale Objection against the Debtors, or the Purchaser.

ASSUMPTION AND ASSIGNMENT OF ASSIGNED CONTRACTS

26. As part of the sale, the Debtors desire to assume and assign the executory contracts set forth on **Exhibit B** hereto (Schedule 2.1(b) to the APA)(the "Assigned Contracts") to the Purchaser. Pursuant to this Sale Motion, the Debtors seek authority pursuant to sections 105(a) and 365 of the Bankruptcy Code to (i) assume and assign the Assigned Contracts to the Purchaser, and (ii) execute and deliver to the Purchaser, or its assignee, such documents or other instruments

as may be necessary by the Purchaser to assign and transfer the Assigned Contracts at the Final Sale Hearing. The Debtors have set forth on **Exhibit B** the cure amounts they believe to be payable to each of the counterparties under the Assigned Contracts.

- 27. The Debtors request that the Court order, at the Preliminary Sale Hearing, that any objections relating to the proposed assumption and assignment of executory contracts (including, but not limited to, any objections relating to the validity of the cure amount as determined by the Debtors or to otherwise assert that any amounts, defaults, conditions, or pecuniary losses must be cured or satisfied under any of the Assigned Contracts as of the date of the Sale Hearing) (a "Cure Objection") be filed and served so as to be actually received by counsel for both the Debtors and the Purchaser no later than the Objection Deadline.
- 28. The Debtors further request that the Court order that, unless a Cure Objection is timely filed and served by a party to an Assigned Contract by the Objection Deadline, all interested parties who have received actual or constructive notice thereof be deemed to have waived and released any right to assert a Cure Objection and to have otherwise consented to the assignment of the Assigned Contracts with payment of the cure amounts set forth on **Exhibit B** hereto and shall be forever barred and estopped from asserting or claiming against Debtors, the Purchaser or any assignee that any additional amounts are due or that a default exists with respect to such Assumed Contract.
- C. APPROVAL **OF** THE **FORM** AND **OF** NOTICE **TERMINATION** FEES, SALE **OBJECTION** HEARING OF THE DEADLINES.
- 29. At the Preliminary Sale Hearing, the Debtors request approval of the form and manner of notice of the Sale Hearing and of the Objection Deadlines to be served on all parties-in-interest affected by the Sale Motion.

Notice") attached hereto as Exhibit D provides due and adequate notice of the Final Sale Hearing and of the proposed assumption and assignment of the Assigned Contracts and the payment of cure amounts in connection therewith.

D. APPROVAL OF THE DISPOSITION OF SALE PROCEEDS.

- 31. Finally, at the Sale Hearing, the Debtors request, pursuant to sections 105, 363, and 365 of the Bankruptcy Code and Bankruptcy Rules 2002, 6003, 6004, 6006, and 9006, that, at the Final Sale Hearing, the Court will enter an order (the "<u>Transfer Order</u>"):
 - (i) approving the APA by and between the Debtors and the Purchaser, substantially in the form attached hereto as Exhibit A,
 - (ii) authorizing the sale of the Debtors' assets under the APA free and clear of all liens, claims, encumbrances, and other interests;
 - (iii) authorizing the assumption and assignment of substantially all of the Debtors' executory contracts, identified in the attached Exhibit B, to the Purchaser;
 - (iv) (i) authorizing the Debtors to use the cash portion of the Aggregate Consideration to pay the DIP Obligations (the "DIP Loan Payment"), (ii) directing the Debtors to make the DIP Loan Payment concurrently with the Closing; and (iii) stating that, upon receipt of the DIP Loan Payment, (i) the DIP Lenders are authorized to apply the DIP Loan Payment to the DIP Obligations, (ii) such payment shall constitute indefeasible payment of the DIP Obligations so paid and (iii) any commitments of the DIP Lenders to make any loans to the Debtors are terminated;
 - (v) authorizing the Debtors to place the Hercules Shares received as the remainder of the Aggregate Consideration with an Escrow Agent in an account to be held in trust pursuant to the an escrow agreement and to be distributed to holders of allowed claims and interests of the Debtors pursuant a confirmed chapter 11 plan or pursuant to further orders of this Court; and
 - (vi) granting other related and appropriate relief.

BASIS FOR RELIEF

- A. Approval of the Sale of Substantially All of the Debtors' Assets.
- 32. Section 363(b)(1) of the Bankruptcy Code provides, in relevant part, that "[t]he trustee, after notice and a hearing, may use, sell, or lease, other than in the ordinary course of business, property of the estate." 11 U.S.C. § 363(b)(1).
- 33. Although section 363 of the Bankruptcy Code does not set forth a standard for determining when it is appropriate for a court to authorize the sale or disposition of a debtor's assets, in applying this section, courts have required that it be based upon the debtor's business judgment. In re Continental Air Lines, Inc., 780 F.2d 1223, 1226 (5th Cir. 1986) (A debtor must demonstrate "some articulated business justification for using, selling, or leasing property outside of the ordinary course of business."); Comm. Of Equity Security Holders v. Lionel Corp., 722 F.2d 1063, 1071 (2d Cir. 1983) (A sale of a substantial part of a Chapter 11 estate other than in the ordinary course of business may be conducted if a good business reason exists to support it.). "Where the debtor articulates a reasonable basis for its business decisions (as distinct from a decision made arbitrarily or capriciously), courts will generally not entertain objections to the debtor's conduct." Comm. of Asbestos-Related Litigation v. Johns-Manville Corp. (In re Johns-Manville Corp.), 60 B.R. 612, 616 (Bankr. S.D.N.Y. 1986). When a valid business justification exists, the law vests the debtor's decision to use property out of the ordinary course of business with a strong presumption that "in making a business decision the directors of a corporation acted on an informed basis, in good faith and in the honest belief that the action taken was in the best interest of the company." Official Comm. of Subordinated Bondholders v. Integrated Res., Inc. (In re Integrated Res., Inc.), 147 B.R. 650, 656 (S.D.N.Y. 1992) (quoting Smith v. Van Gorkom, 488 A. 2d 858, 872 (Del. 1985)).

- 34. Moreover, section 105(a) of the Bankruptcy Code grants this Court broad authority under its equitable powers to fashion any order or decree that would preserve or protect the value of the debtor's assets. See 11 U.S.C. § 105(a); see also, Adelphia Commc'ns Corp. v. Rigas (In re Adelphia Commc'ns Crop.), 2003 WL 21297258, at *4 (S.D.N.Y. June 4, 2003) ("Section 105 of Title 11 provides the bankruptcy courts with a broad range of equitable powers over cases within its jurisdiction.").
- 35. Courts look to various factors to determine whether to approve a motion under section 363(b), such as: (i) whether a sound business reason exists for the proposed transaction; (ii) whether fair and reasonable consideration is provided; (iii) whether the transaction has been proposed and negotiated in good faith; and (iv) whether adequate and reasonable notice is provided. See In re Condere, 228 B.R. 615, 626 (Bankr. S.D. Miss. 1998); See In re Delaware & Hudson Ry. Co., 124 B.R. 169, 176 (D. Del. 1991). The Debtors respectfully submit that the proposed sale of the Purchased Assets fits squarely within the parameters of the sound business judgment test articulated above.

1. A Sound Business Reason Exists For The Sale Of The Purchased Assets

36. The Debtors' decision to sell property out of the ordinary course of business enjoys a strong presumption "that in making a business decision the Debtors acted on an informed basis, in good faith and in an honest belief that the action taken was in the best interests of the company." *In re Integrated Res., Inc.*, 147 B.R. 650, 656 (S.D.N.Y. 1992). Therefore, parties objecting to the Debtors' proposed sale must make a showing of "bad faith, self-interest, or gross negligence." *Integrated Res.*, 147 B.R. at 656.

- 37. The Debtors have determined, in their considered business judgment, that the sale of the Purchased Assets pursuant to the APA is appropriate and in the best interest of, and represents the best means of preserving and maximizing value for, the Debtors' estates.
- 38. As discussed above, the Debtors have faced pre-petition monthly operating losses for an extended period of time. The Debtors project continued losses if a sale is not effectuated. Without additional funding, the Debtors anticipate that they will soon run out of cash to operate their business. Because of the lack of capital, continued operation of the Debtors' business without a sale of the Purchased Assets is not an option.
- 39. To avoid a chapter 7, which likely would result in unsecured creditors and equity interests receiving little or no recovery or distribution, subject to approval of the Court, the Debtors have agreed to sell the Purchased Assets to the Purchaser pursuant to section 363 of the Bankruptcy Code.
- 40. The sale to the Purchaser will permit the Debtors to distribute cash and Hercules Common Stock generated from the sale to creditors and interestholders pursuant to a chapter 11 plan or pursuant to further orders of this Court. Accordingly, the proposed disposition of assets is born of the Debtors' economic circumstances, and not "bad faith, self-interest or gross negligence."
- 41. The Debtors note that while the sale of the Purchased Assets is not part of its ordinary business, the proposed sale will not substantially impact (and, in fact, will facilitate a plan of liquidation) or significantly alter the rights of the Debtors' creditors. Therefore, the Debtors submit that the proposed sale of the Purchased Assets in no way constitutes a *sub rosa* plan or falls outside the scope of Bankruptcy Code § 363. *C.f. Continental Airline, Inc. v. Braniff Airways, Inc. (In re Braniff Airways, Inc.)*, 700 F.2d 935. 939 (5th Cir. 1983) (noting that the

proposed sale pursuant to section 363 would require "significant restructuring of the rights of Braniff creditors").

2. The Purchaser Is Paying Fair And Reasonable Consideration For The Purchased Assets

42. The aggregate consideration for the Purchased Assets is approximately One Hundred Five Million Dollars (\$105,000,000). Given the difficulty the Debtors have faced trying to sell the Purchased Assets, the Debtors believe that the purchase price is fair and reasonable consideration for the Purchased Assets and is the result of extensive arm's length negotiations between the respective representatives of the Debtors and the Purchaser. The purchase price offered by the Purchaser represents the highest and best bid for the Purchased Assets received to date.

3. The Sale Of The Purchased Assets Has Been Proposed And Negotiated In Good Faith

- 43. The Purchaser should be afforded all protections under Bankruptcy Code Section 363(m) as a good faith purchaser. Section 363(m) of the Bankruptcy Code provides that "the reversal or modification on appeal of an authorization under subsection (b) or (c) of this section of a sale or lease of property does not affect the validity of a sale or lease under such authorization to an entity that purchased or leased such property in good faith." 11 U.S.C. § 363(m). Courts generally determine that parties have acted in good faith with respect to a proposed sale if the purchase price is adequate and reasonable and the terms of the sale are disclosed fully. See In re Abbotts Dairies of Pa., Inc., 788 F.2d 143, 149-50 (3d Cir. 1986).
- 44. The Debtors submit that the APA is not tainted by self-dealing, collusion or manipulation as it was proposed and negotiated in good faith and executed by the Debtors and the Purchaser in an arm's length transaction, and the Debtors and the Purchaser have fully disclosed all material information.

- 45. Similarly, the Purchaser's proposed consideration for the Purchased Assets was made in good faith. Moreover, the Purchaser was not preferred and did not receive special treatment. Though Randall D. Stilley was formerly employed as the Chief Executive Officer of Hercules, his former relationship did not influence Simmons' solicitation process or the negotiation or consummation of the APA. All aspects of Simmons' solicitations and the negotiation and consummation of the APA were comported in good faith and pursuant to reasonable business judgment.
- 46. Accordingly, the Debtors assert that it is appropriate that the Purchaser receives the protection of a good faith finding by the Court in accordance with section 363(m).
 - 4. The Court Should Authorize The Sale Of The Purchased Assets To The Purchaser Free And Clear Of All Liens, Claims, Encumbrances, And Interests
- 47. The Debtors request that the sale and transfer of the Purchased Assets be approved free and clear of all interests, liens, claims, encumbrances and other interests, other than those specifically assumed by the Purchaser. Such relief is consistent with the provisions of section 363(f).
- 48. The Court may, pursuant to section 363(f) of the Bankruptcy Code, authorize a debtor-in-possession to sell property free and clear of any interest, lien, claim or interest of another entity in such property if any of the following circumstances pertain:
 - (i) applicable non-bankruptcy law permits sale of such property free and clear of such interest;
 - (ii) such entity consents;
 - (iii) such interest is a lien and the price at which such property is to be sold is greater than the aggregate value of all liens on such property;
 - (iv) such interest is in bona fide dispute; or
 - (v) such entity could be compelled, in a legal or equitable proceeding, to accept a money satisfaction of such interest.

11 U.S.C. § 363(f).

- 49. As indicated by the use of the disjunctive term "or," satisfaction of any of the five requirements listed in section 363(f) is sufficient to permit the sale of assets free and clear of all interests, liens, claims, encumbrances, pledges, mortgages, security interests, charges, options and other interests, including any right of setoff. *See, e.g., In re C-Power Prods., Inc.*, 230 B.R. 800 (Bankr. N.D. Tex. 1998); *In re Elliott*, 94 B.R. 343, 345 (E.D. Pa. 1988) (stating that section 363(f) is written in the disjunctive; the court may approve a sale "free and clear" provided that at least one of the subsections is met).
- Assets as of the Petition Date are Natixis, New York Branch, as Administrative Agent for the Lenders (the "Pre-Petition Lenders"), and the U.S. Internal Revenue Service. First, the price at which the Purchased Assets is to be sold is greater than the aggregate value of all liens on such property pursuant to section 363(f)(iii) of the Bankruptcy Code. Second, it is anticipated that the DIP Lenders will be substituted in the place of the Pre-Petition Lenders with senior liens on the Purchased Assets as of the Closing Date.
- 51. Therefore, the Debtors submit, to the extent section 363(f) is applicable, at least one requirement of section 363(f)(iii) has been satisfied and the Purchased Assets may be sold and transferred free and clear of all liens, claims, and encumbrances.

5. Approval Of The APA

52. Courts routinely approve asset purchase agreements in connection with the sale of a debtor's assets. See In re Arlco, Inc., 239 B.R. 261, 265 (Bankr. S.D.N.Y. 1999). In this case, the Purchaser's APA was the subject of intense arm's-length negotiations between representatives of the Debtors and the Purchaser, and the Debtors submit that the terms and conditions of the

APA are the best that could be obtained under the circumstances, and that entry into the APA is a sound exercise of the Debtors' business judgment.

- 6. Assumption And Assignment Of The Assigned Contracts Should Be Authorized
- 53. The Debtors intend to assume and assign the Assigned Contracts to the Purchaser. Under section 365 of the Bankruptcy Code, a debtor-in-possession may assume an executory contract "subject to the court's approval" if (a) the debtor-in-possession cures, or provides adequate assurance of a prompt cure, of any default under such contract or lease; (b) compensates, or provides adequate assurance that prompt compensation will be paid to any non-debtor party to such contract or lease for any pecuniary loss to such party resulting from such default; and (c) provides adequate assurance of future performance under such contract. See 11 U.S.C. § 365(b)(1).
- 54. Courts have consistently applied the business judgment test in determining whether to approve a debtor-in-possession's decision to assume an executory contract. *See Richmond Leasing Co. v. Capital Bank, N.A.*, 762 F.2d 1303, 1312 (5th Cir. 1985). The business judgment test is satisfied, when the debtor-in-possession has determined in good faith that assumption of a contract will benefit the estate.
- 55. The Debtors have determined, in the exercise of their sound business judgment, that the assumption and assignment of the Assigned Contracts will benefit the Debtors' estates. The assumption and assignment of the Assigned Contracts is essential to the deal between the Debtors and the Purchaser. Further, the assumption and assignment of the Assigned Contracts will significantly limit administrative expense and rejection damage claims.
- 56. The Debtors also propose to cure defaults and provide adequate assurance of future performance of the Assigned Contracts. The cure amounts that the Debtors believe to be owed in 90303148.17

connection with the assumption and assignment of the Assigned Contracts are set forth on the attached **Exhibit B**. The payment of any cure amounts related to the Assigned Contracts will be in final satisfaction of all obligations to cure defaults and compensate the non-debtor counterparties for any pecuniary losses under section 365(b)(1) of the Bankruptcy Code.

57. At the Final Sale Hearing to approve the sale, the Debtors will provided evidence that illustrates the Purchaser's financial credibility, experience in the industry, and willingness and ability to perform under the Assigned Contracts. The meaning of "adequate assurance of future performance" depends on the facts and circumstances of each case, but should be given "practical, pragmatic construction." *In re Sanshoe Worldwide Corp.*, 139 B.R. 585, 592 (S.D.N.Y. 1992). When an executory contract or unexpired lease is to be assumed and assigned, adequate assurance may be provided by, among other things, demonstrating the financial health of the assignee and its experience and ability in managing the type of enterprise or property assigned. *See In re Bygaph*, 56 B.R. 596, 605-06 (Bankr. S.D.N.Y. 1986).

BANKRUPTCY RULES 6004(H) AND 6006(D) SHOULD BE WAIVED

- 58. Bankruptcy Rule 6004(h) provides that an "order authorizing the use, sale, or lease of property ... is stayed until the expiration of 14 days after the entry of the order, unless the court orders otherwise." Fed. R. Bank. P. 6004(h). Bankruptcy Rule 6006(d) provides that an "order authorizing the trustee to assign an executory contract or unexpired lease ... is stayed until the expiration of 14 days after the entry of an order, unless the court orders otherwise." Fed R. Bank. P. 6006(d).
- 59. The Debtors request that any order entered pursuant to this Sale Motion authorizing the consummation of a transaction that is deemed a sale of assets and/or an assignment of an executory contract be effective immediately by providing that the 14-day stay

under Rules 6004 or 6006, as the case may be, is inapplicable, so that they may proceed to close on the transaction as expeditiously as possible and within the time frames contemplated by the Debtors and the Purchaser.

NOTICE OF THE SALE MOTION

- 60. In accordance with applicable rules, including the United States Bankruptcy Court for the Southern District of Texas Procedures for Complex Chapter 11 Bankruptcy Cases, notice of this Motion has been provided by overnight delivery and e-mail or facsimile to: (a) Natixis, New York Branch, as Administrative Agent for the Lenders; (b) the 30 largest unsecured creditors of the Debtors on a consolidated basis; (c) Pride International, Inc.; (d) Hercules Offshore, Inc.; (e) D.E. Shaw Direct Capital Portfolios, L.L.C.; (f) the U.S. Trustee's Office; (g) the United States Attorney's Office; (h) the Securities and Exchange Commission; (i) the Internal Revenue Service; and (j) parties to executory contracts proposed to be assumed and assigned. The Debtors believe that the notice provided for herein is fair and adequate and no other or further notice is necessary.
- 61. In addition, as soon as approved by the Court, the Debtors will immediately serve the Sale Hearing Notice, substantially in the form of the attached **Exhibit D**, by standard first-class mail, e-mail, or facsimile, to all known parties in interest in the Debtors' bankruptcy cases and on all other parties ordered by the Court.

CONCLUSION

WHEREFORE, PREMISES CONSIDERED, the Debtors respectfully request (A) at the preliminary hearing on the Sale Motion, entry of an order (1) scheduling an expedited hearing to consider final approval of the Sale Motion; (2) setting deadlines and requirements for parties in interest to file (a) objections to the Sale Motion, and (b) objections to the proposed assumption and assignment of executory contracts in connection with the sale and the payment of cure claims

in connection therewith; and (3) approving the form and manner of notice of the Final Sale Hearing; and (B) at the Final Sale Hearing on the Sale Motion, entry of an order (1) approving the Debtors' sale of the Purchased Assets to the Purchaser free and clear of liens, claims, encumbrances and interests to the fullest extent of section 363 of the Bankruptcy Code; (2) approving the proposed assumption and assignment by the Debtors to purchaser of the Assumed Contracts and payment of all cure claims in connection therewith pursuant to section 365 of the Bankruptcy Code; (3) approving the Debtors' use of the cash portion of the consideration received from Purchaser at closing to repay all outstanding principal, interest and fees due under the DIP Loan as of the Closing Date; (4) approving the transfer by Hercules of the Hercules Shares at closing to an escrow agent to be distributed by the escrow agent pursuant to a confirmed chapter 11 plan or pursuant to further orders of this Court; and (C) all such other and further relief as may be just and proper both at law and in equity.

Dated: February 11, 2011.

Respectfully submitted,

FULBRIGHT & JAWORSKI L.L.P.

By: /s/ Berry D. Spears

Berry D. Spears
State Bar No. 18893300
Johnathan C. Bolton
State Bar No. 24025260
1301 McKinney, Suite 5100
Houston, Texas 77010-3095
Telephone: (713) 651-5151
Facsimile: (713) 651-5246
bspears@fulbright.com
jbolton@fulbright.com

and

JORDAN, HYDEN, WOMBLE, CULBRETH & HOLZER P.C.

Shelby A. Jordan
State Bar No. 11016700
Nathanial Peter Holzer
State Bar No. 00793971
500 N. Shoreline Blvd., Suite 900
Corpus Christi, Texas 78401-0341
Telephone: (361) 653-6624
Facsimile: (361) 888-5555
sjordan@jhwclaw.com
pholzer@jhwclaw.com

PROPOSED ATTORNEYS FOR THE DEBTORS AND DEBTORS-IN-POSSESSION

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	§ Chapter 11
Seahawk Drilling Inc.,	§ Case No.: 11
Seahawk Mexico Holdings LLC,	§ Case No.: 11
Seahawk Drilling Management LLC,	§ Case No.: 11
Seahawk Offshore Management LLC,	§ Case No.: 11
Energy Supply International LLC,	§ Case No.: 11
Seahawk Drilling LLC,	§ Case No.: 11
Seahawk Global Holdings LLC,	§ Case No.: 11
Seahawk Drilling USA LLC,	§ Case No.: 11
Debtors.	§ Motion for Joint Administration Pending

DEBTORS' EMERGENCY MOTION, PURSUANT TO SECTIONS 105, 363 AND 365 OF THE BANKRUPTCY CODE AND RULES 2002, 6004, 6006, 9007, AND 9014 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE, FOR AN ORDER (i) SCHEDULING AN EXPEDITED HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS; (ii) SETTING DEADLINES FOR FILING OBJECTIONS TO (1) THE PROPOSED SALE AND (2) THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND THE PAYMENT OF CURE CLAIMS IN CONNECTION THEREWITH; AND (iii) APPROVING THE FORM AND MANNER OF NOTICE THEREOF

NOTICE UNDER BLR 9013(b) AND 9013(i)

THIS MOTION SEEKS AN ORDER THAT MAY ADVERSELY AFFECT YOU. IF YOU OPPOSE THE MOTION, YOU SHOULD IMMEDIATELY CONTACT THE MOVING PARTY TO RESOLVE THE DISPUTE. IF YOU AND THE MOVING PARTY CANNOT AGREE, YOU MUST FILE A RESPONSE AND SEND A COPY TO THE MOVING PARTY. YOU MUST

FILE AND SERVE YOUR RESPONSE WITHIN 21 DAYS OF THE DATE THIS WAS SERVED ON YOU. YOUR RESPONSE MUST STATE WHY THE MOTION SHOULD NOT BE GRANTED. IF YOU DO NOT FILE A TIMELY RESPONSE, THE RELIEF MAY BE GRANTED WITHOUT FURTHER NOTICE TO YOU. IF YOU OPPOSE THE MOTION AND HAVE NOT REACHED AN AGREEMENT, YOU MUST ATTEND THE HEARING. UNLESS THE PARTIES AGREE OTHERWISE, THE COURT MAY CONSIDER EVIDENCE AT THE HEARING AND MAY DECIDE THE MOTION AT THE HEARING.

EMERGENCY RELIEF HAS BEEN REQUESTED. IF THE COURT CONSIDERS THE MOTION ON AN EMERGENCY BASIS, THEN YOU WILL HAVE LESS THAN 21 DAYS TO ANSWER. IF YOU OBJECT TO THE REQUESTED RELIEF OR IF YOU BELIEVE THAT THE EMERGENCY CONSIDERATION IS NOT WARRANTED; YOU SHOULD FILE AN IMMEDIATE RESPONSE.

REPRESENTED PARTIES SHOULD ACT THROUGH THEIR ATTORNEY.

TO THE HONORABLE UNITED STATES BANKRUPTCY JUDGE:

Seahawk Drilling, Inc. ("Seahawk" of the "Company") and its above-captioned affiliated debtors (collectively, the "Debtors"), file this Emergency Motion Pursuant To Sections 105, 363 and 365 of the Bankruptcy Code And Rules 2002, 6004, 6006, 9007, and 9014 of the Federal Rules Of Bankruptcy Procedure, for an Order (i) Scheduling an Expedited Hearing To Approve The Sale of Substantially All of the Debtors' Assets; (ii) Setting Deadlines for Filing Objections To (1) the Proposed Sale and (2) the Proposed Assumption and Assignment of Executory Contracts and the Payment of Cure Claims in Connection Therewith; and (iii) Approving the Form and Manner of Notice Thereof (the "Sale Motion"). In support of this Sale Motion, the Debtors respectfully state:

JURISDICTION AND PROCEDURAL STATUS

- 1. This Court has jurisdiction over this Sale Motion pursuant to 28 U.S.C. §§ 157 and 1334. Venue of the Debtors' chapter 11 cases in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409. This is a core proceeding pursuant to 28 U.S.C. § 157(b).
- 2. On the date hereof (the "<u>Petition Date</u>"), the Debtors filed their respective voluntary petitions for relief under title 11 of the United States Code (the "<u>Bankruptcy Code</u>"). Contemporaneously herewith, the Debtors have sought joint administration of these chapter 11 cases.
- 3. The Debtors continue to operate their businesses and manage their properties as debtors-in-possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee, examiner, or creditors' committee has been appointed in these cases.

FACTUAL BACKGROUND

A. Events Leading To The Debtors' Petitions

- 1. Seahawk is a Delaware corporation formed in 2008 and headquartered in Houston, Texas, with operations throughout the U.S. Gulf of Mexico. Each of the other debtors is a direct or indirect subsidiary of Seahawk.¹ As of the Petition Date, Seahawk's stock is publicly traded on the NASDAO Exchange under the ticker symbol, "HAWK."
- 2. Seahawk operates a jackup rig business that provides contract drilling services to the oil and natural gas exploration and production industry throughout the U.S. Gulf of Mexico.²

¹ The debtor-subsidiaries of Seahawk are: Seahawk Drilling LLC, Energy Supply International LLC, Seahawk Global Holdings LLC, Seahawk Mexico Holdings LLC, Seahawk Drilling Management LLC, Seahawk Offshore Management LLC and Seahawk Drilling USA LLC. The subsidiaries of Seahawk that are incorporated in Mexico or have branches in Mexico are not a part of this chapter 11 proceeding.

² Jackup rigs are mobile, self-elevating drilling platforms equipped with legs that can be lowered to the ocean floor until a foundation is established to support the drilling platform. The Debtors' rigs work in water depths up to 300 feet and can drill to a depth of 25,000 feet.

The Debtors contract with their customers on a dayrate basis to provide rigs and drilling crews. The Debtors' fleet of mobile offshore drilling rigs consists of twenty (20) shallow water jackup rigs. Three (3) of the Debtors' six (6) rigs in Texas are located at Matagorda Island 721 in Aransas County. Two rigs are located at Sabine Pass, Texas, and one rig is located at High Island 176 in Galveston County. The remainder of Seahawk's rigs are currently located in Louisiana.

- 3. Seahawk is a former subsidiary of Pride International, Inc. ("<u>Pride</u>"). On August 4, 2009, the Board of Directors of Pride approved a plan to separate Pride into two independent, publicly-traded companies (the "<u>Spin-Off</u>"). The separation occurred through the distribution to Pride stockholders of all of the shares of common stock of Seahawk that held, directly or indirectly, the assets and liabilities of Pride's jackup rig business.
- 4. On August 24, 2009 (the "Spin-Off Date"), each Pride stockholder received one-fifteenth (1/15) of a share of Seahawk's common stock for each share of Pride common stock held at the close of business on August 14, 2009. After the Spin-Off, Seahawk became an independent public company. Pride no longer retains any ownership interest in Seahawk, although the two companies are parties to various agreements entered into in connection with the Spin-Off (collectively, the "Pride Agreements").
- 5. Since mid-2008, the demand for drilling services has declined dramatically, principally as a result of the global financial crisis, declining prices of crude oil and natural gas and deteriorating worldwide economic conditions. The decline in the United States jackup rig market since 2009 has been one of the sharpest downturns for domestic jackup rig activity over the past thirty (30) years. In addition, the regulatory and financial uncertainties regarding a

former customer, PEMEX, have had a significant effect on Seahawk's business.³ Finally, from August 2009 to March 2010, the Debtors' active rig count averaged five (5) to seven (7) working rigs at any given time.

- 6. On April 20, 2010, the demand for offshore drilling services in the Gulf of Mexico was further negatively impacted by the Macondo well blowout, prompting, among other things, the United States Government to issue a moratorium on all U.S. offshore drilling. On May 26, 2010, the moratorium on Gulf of Mexico drilling in waters less than 500 feet deep was lifted. Subsequently, in November 2010, the deep water moratorium was lifted. However, since the Macondo incident, notwithstanding the termination of the Government-imposed drilling prohibitions, the Debtors' customers are experiencing significant delays in the issuance of drilling permits and very few new drilling permits have been issued. In addition, there is an increasingly uncertain regulatory and cost environment which continues to adversely affect the Debtors' business.
- 7. As a result of all of these factors, Debtors' active rig count declined to three (3) working rigs during October 2010. While there has been some recent marginal improvement in market conditions (seven (7) rigs are currently working), the combined impact of all of these events, together with negative cash flows throughout 2009 and 2010, has severely stressed and exhausted the Debtors' liquidity and even the current operating level results continue to produce negative cash flows for the Debtors. Seahawk's Board of Directors (the "Board") and its management have closely monitored the impact of these conditions and evaluated potential alternatives to address and improve these circumstances with the assistance of various professionals and outside advisors, including Alvarez & Marsal North America, LLC ("A&M").

³ Pemex Exploración y Producción ("<u>PEMEX</u>") is the exploration and production subsidiary of the national oil 90303148.17

- 8. In November 2010, Seahawk publicly announced that its Board had initiated a process to explore and consider possible strategic alternatives, including, but not limited to transactions involving additional funding, recapitalizations, sales of assets, or a sale or merger of Seahawk. Seahawk's Board, through its finance committee (the "Finance Committee"), with regular and substantial Board and independent directors' participation, supervised the process and were managed by Simmons & Company International ("Simmons") and other advisors.
- 9. In February 2011, after evaluating the indications of interest and potential offers received from a number of interested parties, the Board authorized the Debtors to enter into an Asset Purchase Agreement (the "APA") with Hercules Offshore, Inc. ("Hercules"). The executed APA contemplates the acquisition by Hercules or one or more of its subsidiaries ("Purchaser") of substantially all of the assets and jackup rigs of the Debtors (the "Purchased Assets") through a sale pursuant to section 363 of the Bankruptcy Code (the "Transaction"). The aggregate consideration for the Purchased Assets (defined in the APA as the "Base Aggregate Consideration") is (a) 22,321,425 shares of Hercules Common Stock (the "Hercules Shares") plus (b) cash in an amount equal to \$25,000,012 (the "Cash Payment"), subject to certain adjustments. Using the closing stock price of Hercules' stock as of February 10, 2011, the Base Aggregate Consideration would be valued at approximately \$105 million before any adjustments. The Base Aggregate Consideration is to be payable at closing by Purchaser to Sellers.
- 10. The Debtors have sought chapter 11 protection in order to protect and preserve their going concern value and to facilitate a prompt sale of substantially all of their assets to Hercules for the benefit of all stakeholders. The Debtors have determined in the prudent exercise

company of Mexico.

⁴ The summary description of the APA set forth herein is qualified in its entirety by reference to the terms of the APA itself.

of their business judgment that the proposed course of action is the best alternative to ensure that the maximum value of Seahawk's assets will inure to the benefit of all of their creditors and interestholders.

Additional information regarding the Debtors' business, capital structure, and the circumstances and events leading to these chapter 11 cases is contained in the Declaration of Randall D. Stilley in Support of First Day Motions, filed contemporaneously herewith and incorporated herein by reference.

B. The Proposed Sale

- 12. The Debtors filed these chapter 11 bankruptcy cases to effectuate the terms and provisions of the APA by and between the Debtors and Hercules (collectively, the "Purchaser") under a sale pursuant to section 363 of the Bankruptcy Code. A true and correct copy of the executed APA is attached hereto as **Exhibit A** and incorporated herein for all purposes.
- 13. One of the conditions precedent to the closing of the APA is that this Court enter an order approving the APA. The Debtors have concluded that a chapter 11 filing designed to facilitate a sale of the Debtors' assets to the Purchaser through a section 363 sale would most efficiently consummate the sale and is necessary to prevent continued operating losses and diminution of the Debtors' enterprise value. Under the APA, the Debtors also intend to assume and assign virtually all of their executory contracts, but not unexpired leases, to the Purchaser.
- 14. The Debtors, in consultation with Simmons, and its Board have concluded that the Purchaser's offer was the highest and best offer received. In the opinion of Simmons and the Board, further marketing efforts would not likely result in any offers superior to the Hercules

APA. The Declaration of Simmons & Company International is attached hereto as Exhibit 1 and incorporated herein by reference for all purposes.

- 15. Pursuant to the terms and conditions of (i) that certain Debtor-In-Possession Credit Agreement (the "DIP Credit Agreement") by and between Seahawk and D.E. Shaw Direct Capital Portfolios, L.L.C. and certain lenders thereto (the "DIP Lenders") and (ii) that certain proposed Interim Order Pursuant to 11 U.S.C. §§ 105, 361, 362, 363, and 364 of the Bankruptcy Code (I) Authorizing the Debtors to Obtain Postpetition Financing and Use Cash Collateral and (II) Granting Liens, Security Interests and Superpriority Claims (the "DIP Order"), any sale of all or substantially all of the Debtors' assets is prohibited unless the sale generates sufficient proceeds to pay all DIP Obligations (as defined in the DIP Order), all DIP Obligations are indefeasibly paid in full in cash, and the commitments under the DIP Credit Agreement are terminated. As a result, the APA provides for a cash portion of the Base Aggregate Consideration (as defined in the APA) sufficient to pay in full in cash the DIP Obligations. Concurrently with the closing of the sale, the cash portion paid by Hercules to the Debtors will be used to indefeasibly pay the DIP Obligations and the commitments to make loans under the DIP Credit Agreement will terminate. The Hercules Shares received by the Debtors will be held by an escrow agent to be distributed to holders of allowed claims against and interests in the Debtors pursuant a confirmed chapter 11 plan or pursuant to further orders of this Court.
- 16. The Debtors believe that because it is anticipated that the purchase price will fully satisfy all claims against the Debtors and Hercules' offer is the highest and best offer the Debtors have been able to obtain, the APA is in the best interests of the Debtors and all parties-in-interest, and accordingly, should be approved.

THE NEED FOR AN EXPEDITED SALE

- 17. As set forth above, and as further detailed in the First Day Affidavit of Randall D. Stilley filed contemporaneously herewith, the Debtors have been substantially affected by the downturn in the natural gas market, the moratorium on new offshore oil and gas drilling permits and general economic conditions. The Debtors have suffered operating losses prior to the Petition Date and likewise have projected that they will continue to suffer operating losses throughout 2011. Every day the Debtors continue their operations, and even during the pendency of these bankruptcy cases, the Debtors assets continue to experience a further deterioration in value.
- 18. Furthermore, an expedited sale of the Debtors' assets is a mandatory condition of the APA. Without an expedited sale, the APA will not be consummated and the Debtors' estate and all the Debtors' creditors and shareholders will lose a significant opportunity to achieve what the Debtors believe is the highest and best recovery on their claims and interests.

RELIEF REQUESTED

19. The Debtors request that, at the initial hearing on this Sale Motion (the "Preliminary Sale Hearing"), the Court enter an order substantially in the form attached hereto as Exhibit C, that: (1) schedules an expedited hearing to consider final approval of the Sale Motion; (2) sets deadlines for filing (a) objections to the Sale Motion, and (b) objections to the proposed assumption and assignment of executory contracts in connection with the sale and the payment of cure claims in connection therewith; and (3) approves the form and manner of notice of the final hearing to approve the sale ("Final Sale Hearing").⁵

A. REQUEST FOR AN EXPEDITED HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS.

⁵ The Debtors have filed a separate motion seeking approval of the Termination Fee payable to Purchaser under certain conditions.

- 20. The Debtors submit that the sale of the Purchased Assets is essential to preserve the going concern value of Debtors' business. The Debtors assert that each day the Purchased Assets are subject to the market and the bankruptcy process will cause further losses in operating revenue, diminish the value of the Purchased Assets and threaten the viability of projected creditor and stakeholder recoveries. Accordingly, from the Debtors' perspective, the APA represents a reasonable exercise of the Board's and management's business judgment and an expedited sale is in the best interests of the Debtors and all of their economic claim and interestholders.
- 21. The Debtors and Simmons have expended substantial time and resources soliciting potential purchasers of the Debtors' assets pre-petition and accordingly are of the opinion that the APA with the Purchaser represents the best terms the Debtors can reasonably anticipate to obtain and that the process they have implemented is comprehensive, fulsome and appropriate in every respect.

KEY TERMS OF THE APA

22. A summary of the pertinent provisions of the APA are as follows⁶:

Seller	The Debtors
(APA p.1)	
Purchasers (APA p.1)	Hercules Offshore, Inc. and SD Drilling LLC, a Delaware limited liability company ("Newco")
Purchase Price (APA § 2.4)	Subject to the other terms of this Agreement, the aggregate consideration for the Purchased Assets shall consist of (a) 22,321,425 shares of Hercules Common Stock (the "Hercules Shares") and (b) cash in an amount equal to \$25,000,012 (the "Cash Payment" and, together with the Hercules Shares, the "Aggregate Consideration")

The summary of the APA is intended only to assist the Court and parties-in-interest in understanding the key aspects of the transaction. In the event of any conflict between this summary and the APA, the APA shall control. The APA can be amended by the parties pursuant to a written amendment, provided that the amendment does not materially reduce the consideration provided to the Debtors' estates. Any capitalized terms not defined herein shall have the meaning set forth in the APA.

Purchased Assets (APA § 2.1(a)– (n))

All of the Debtors' right, title, and interest in and to all of the property, including but not limited to:

Rigs. The jackup Rigs listed on <u>Schedule 2.9</u>, each of which is located in the United States of America (the "Transferred Rigs");

Contracts. All of the interests, rights, Claims, and benefits arising or accruing to any of Sellers under any Contract (including any Lease) to which a Seller is a party or has or may acquire a benefit and that relate to the Purchased Assets or Business (the "Assigned Contracts"), including the Assigned Contracts listed in Schedule 2.3(a), to the extent Sellers' interest in such Contract is transferable;

Equipment. All furniture, equipment, computers, computer equipment, machinery, tools, hand tools, spare parts, test equipment, supplies, inventory, office supplies, telephones, and all other tangible personal property of every kind and description insofar as any of the foregoing relates to the operation of the Purchased Assets or Business (the "Equipment"), including the Equipment listed in Schedule 2.1(c);

Rolling Stock. All automobiles, vans, trucks, trailers, and other motorized and similar vehicles and stock of every kind, whether owned or leased, used in the operation of the Purchased Assets or Business, including as listed in Schedule 2.1(d);

Additional Tangible Assets. All other Tangible Personal Property of every kind used in the operation of the Purchased Assets or Business, together with any rights appurtenant thereto, including any express or implied warranty by the manufacturer, vendor, or lessor of any such Tangible Personal Property;

Accounts and Notes Receivable. All trade accounts receivable, notes receivable, and other rights of Sellers to payment from customers and other third parties and other amounts due from customers and other third parties or that become due, including all assets constituting working capital, that relate to or arise from the operation of the Purchased Assets or Business;

Insurance Benefits. All insurance benefits arising from or related to the Purchased Assets, Business, and Assumed Liabilities;

Cash. All cash and cash equivalents, securities, money on deposit with banks, certificates of deposit, and similar short-term investments (if any) of Sellers;

Prepaid Deposits and Expenses. Any deposits and prepaid expenses (paid

to or by Sellers), claims for refunds, and rights of set off related to the Purchased Assets or Business;

Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities against third parties relating or attributable to the Business, the Purchased Assets, or the Assumed Liabilities, whether choate or inchoate, known or unknown, contingent or non-contingent;

Permits. Any and all Permits, including pending applications or filings therefor and renewals thereof, of every kind under which any of Sellers has or may acquire any benefits or rights, or by which any of Sellers or any of the Purchased Assets may be subject or bound, or that relate to or are used or held for use in the operation of the Purchased Assets or Business (collectively, the "Assigned Permits");

Intangible Assets. Any and all customer and supplier relationships, and other Intangible Personal Property of every kind (other than the Computer Software, Trademark Intellectual Property and Website Intellectual Property, and uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names that constitute Intellectual Property) that relate to or are used in the operation of the Purchased Assets or Business, together with all rights appurtenant thereto (the "Assigned Intangible Assets");

Books and Records. Original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets and photocopies of the other Books and Records (collectively, the "Transferred Books and Records");

Other Assets. To the extent not otherwise enumerated in this Section 2.1 all other tangible or intangible assets, rights, privileges, benefits, Claims, and interests of a Seller, whether real, personal or mixed, of every kind and description and wherever located, that relate to, used in or held for use in the operation of the Purchased Assets or Business.

Excluded Assets (APA § 2.2(a)— (0))

Notwithstanding anything to the contrary in <u>Section 2.1</u> or elsewhere in this Agreement or other Transaction Documents, the Purchased Assets shall not include:

Transaction Rights. All rights of Sellers under this Agreement and the other Transaction Documents, and all cash and non-cash consideration (including the Hercules Shares and the Cash Payment) payable or deliverable to, or on behalf of, Sellers by Purchasers pursuant hereto and thereto;

Rigs. Any Rig that is deemed to be an Excluded Asset pursuant to Section 2.9, together with all of the Equipment located on each such Rig;

Contracts. All rights of Sellers under any Contract that is not an Assigned Contract, including those Contracts listed in Schedule 2.2(c);

Permits. Any and all rights of Sellers under any Permit that is not an Assigned Permit;

Insurance Benefits. Any and all insurance benefits arising from or relating to (i) Excluded Assets and Retained Liabilities, (ii) Claims made by Sellers prior to the Closing with respect to the Purchased Assets and Business, and (iii) the matters set forth in Schedule 2.2(e);

Books and Records. The original copies of all of the Books and Records, other than the original copies of the Transferred Rig Documents, Assigned Contracts, Assigned Permits, and Assigned Intangible Assets (which shall be provided to Purchasers pursuant to Section 2.1(m) and of which Sellers may retain photocopies), and any Books and Records of Sellers (i) that are not permitted to be transferred to Purchasers under applicable Law, (ii) that constitute charters, bylaws, limited liability company agreements, minute books, stock transfer records, and other records related to the corporate governance of Sellers, and (iii) all other books and records of Sellers that do not relate primarily to the Purchased Assets, Transferred Employees, or Business (collectively, the "Retained Books and Records");

Third Party Property. Any improvements, equipment, inventory and any other tangible personal property located at the Leased Real Property or on the Transferred Rigs as of the Closing Date that are not owned by or leased to Sellers; and

Equity Interests. Any Equity Interest in any Seller or any direct or indirect subsidiary of a Seller;

Bankruptcy Claims. Any and all Claims or rights of Sellers arising under chapter 5 of the Bankruptcy Code;

Pride Claims. Any and all Claims or rights of Sellers or any Affiliate thereof against Pride or any Affiliate thereof;

Claims and Warranties. Any and all Claims, warranties, reimbursements, and indemnities of Sellers, whether choate or inchoate, known or unknown, contingent or non-contingent with respect to the matters set forth in Schedule 2.2(k);

Leased Real Property. All leasehold interests in the real property listed in Schedule 2.2(1) (the "Leased Real Property"), together with all Real Property Rights related thereto and all Improvements and Tangible Personal

Property located at, under, or on such Leased Real Properties or otherwise related thereto;

Intellectual Property. The Intellectual Property constituting: (i) Sellers' software programs, tools, kits, and any content or related documentation or third party or open source code embedded therein, either locally stored on Sellers' computers or remotely accessed by Sellers, that is owned or made available to Sellers and is used in, held for use in, or is necessary for the operation of the Purchased Assets or Business or otherwise, and any upgrades, updates, releases, fixes, enhancements, or modifications thereto, and all written materials and specifications applicable thereto (collectively, the "Computer Software"); (ii) the name "Seahawk Drilling" and the mark "Seahawk Drilling" and any other registered or unregistered trademarks, trade names, service marks, domain names, and email addresses, including the term "Seahawk", used or held for use in the Business (collectively, the "Trademarks"), and any and all goodwill associated with the Business embodied in the Trademarks, and any and all rights of Sellers with respect to the Trademarks (collectively, the "Trademark Intellectual Property"); (iii) Sellers' internet website located at www.seahawkdrilling.com and any derivations thereof (collectively, the "Website") and all intellectual property rights that may exist or arise in connection with the Website, including all Underlying Technology (collectively, the "Website Intellectual Property"); and; (iv) uniform resource locator addresses, trade dress, logos, slogans, symbols and corporate names.

Goodwill. Any goodwill and going concern value related to the Business; and

Accounts and Notes Receivable. The trade accounts receivable, notes receivable, and other rights of Sellers to payment from, and other amounts due or that become due, including all assets constituting working capital, that relate to Blake International USA Rigs, LLC or any Affiliate thereof.

Assigned Contracts (APA § 2.3)

Transfer of Contracts and Permits. At least three Business Days prior to the Closing Date, Purchasers may designate any Contracts that would otherwise be an Assigned Contract as a Contract to be excluded from the Purchased Assets for purposes of Section 2.1(b). Any such designation by Purchasers shall not reduce Purchaser's assumption of any working capital liabilities pursuant to Section 2.7(a), unless both (i) such working capital liability was incurred pursuant to such designated Contract and (ii) such working capital liability is not attributable to goods and services provided to the Business prior to the Closing, in which case such working capital liability shall not be assumed under Section 2.7(a).

Notwithstanding anything herein to the contrary, this Agreement shall not constitute an assignment to Purchasers of (and Purchasers shall not assume

any rights or obligations under) any Contract or Permit if an attempted assignment thereof, without the Consent of a Governmental Authority or other third party thereto, would constitute a breach of the Contract or Permit, unless such Consent has been received or such assignment is otherwise authorized and permitted by the Transfer Order or other Order of the Bankruptcy Court. The Transfer Order shall provide that any Assigned Contracts that are executory contracts shall be assumed by Sellers and assigned to Purchasers at the Closing, and Sellers shall pay all pre-Closing cure amounts under Assigned Contracts required to be so paid by the Bankruptcy Court in connection with such assumptions by Sellers (the "Cure Amounts"). The Sellers shall use their commercially reasonable efforts to advise the Purchasers promptly in writing with respect to any Contract or Permit that Sellers' Know or have substantial reason to believe shall not or may not be subject to assignment to Purchasers hereunder. Without in any way limiting Sellers' obligations to obtain all Consents necessary for the sale, transfer, assignment and delivery of the Assigned Contracts and Assigned Permits and the other Purchased Assets to Purchasers hereunder, if any such Consent is not obtained or if such assignment is not permitted irrespective of Consent and the Closing hereunder is consummated, Sellers shall use commercially reasonable efforts to continue to promptly seek Consent or other approval for the sale, transfer, assignment and delivery thereof; provided, that in no circumstance shall any such obligation of Sellers extend beyond the date that Sellers' Plan of Reorganization is confirmed by the Bankruptcy Court and has become effective.

Assumed Liabilities (APA § 2.7(a)— (e))

The Assumed Liabilities include:

Working Capital Liabilities. The current accounts payable of, and accruals for good and services received by, Sellers as of the Closing Date attributable to the Purchased Assets; provided that, for purposes of clarity, such working capital liabilities shall exclude accounts payable to Blake International USA Rigs, LLC or its Affiliates. Nothing in this Section 2.7(a) shall be deemed to effect the assumption of any Contract that is not otherwise an Assigned Contract, but the assumption of liabilities under this Section 2.7(a) shall be subject to Section 2.3(a).

Contracts. The Liabilities of Sellers under the terms of any Assigned Contract to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the closing;

Permits. The Liabilities of Sellers under the terms of any Assigned Permit to the extent that such Liabilities are performance obligations, or otherwise attributable to the period, from and after the Closing;

Claims relating to Assumed Liabilities. In the event Claims are filed in the

Sellers' Bankruptcy Cases on account on the Assumed Liabilities, Sellers shall provide Purchasers written notice of such filing within five (5) Business Days after their filing. Purchasers shall have the right to appear, defend and settle those Claims in the Bankruptcy Court. Notwithstanding the foregoing, Sellers shall be solely responsible for any Claims that exceed the amount of current liabilities assumed under Section 2.7(a) and included in the post-Closing adjustment process under Section 2.10; and

Performance Bond. Purchasers shall either assume Sellers' \$2.0 performance bond in favor of the Texas general land office relating to the Seahawk 2008 Rig, or otherwise post their own performance bond such that such \$2.0 performance bond may be terminated by Sellers.

Retained Liabilities (APA § 2.8(a)— (b))

The Excluded Liabilities include:

Subject to the terms of the Transfer Order, Sellers shall retain and be solely liable for and hereby expressly agree to retain any and all of their respective Liabilities not expressly assumed by Purchasers in Section 2.7 (collectively, the "Retained Liabilities"), regardless of whether any such Retained Liability is disclosed herein or in any Schedule hereto, whether known or unknown, absolute or contingent, liquidated or unliquidated, whether due or to become due, and whether Claims with respect thereto are asserted before or after the Closing Date. Notwithstanding the preceding sentence, Retained Liabilities shall include the following:

- Transaction Liabilities. Any and all Liabilities of Sellers under the Transaction Documents;
- Excluded Assets. Any and all Liabilities of Sellers with respect to the Excluded Assets;
- Purchased Assets and Pre-Closing Business. Except as provided in Sections 2.7(a), 2.7(b), and 2.7(c), all Claims and Liabilities arising out of Sellers' ownership, operation, use, or maintenance of the Purchased Assets or Business, as well as any matters arising from events occurring, conditions existing, or costs accruing prior to the Closing and Taxes;
- Contracts. Except as provided in Sections 2.7(a) and 2.7(b), any and all Liabilities of Seller, including Cure Amounts, that relate to the Assigned Contracts to the extent any such Liability is performable or otherwise attributable to the period prior to the Closing;
- Permits. Except as provided in <u>Sections 2.7(a)</u> and <u>2.7(c)</u>, any and all Liabilities of Sellers that relate to the Assigned Permits to the extent any such Liability is performable or otherwise attributable to the period

prior to the Closing;

- Taxes. Any and all Liabilities of Sellers for Taxes that relate to (A) the ownership, operation, use, or maintenance of the Purchased Assets or Business prior to the Closing Date, or (B) any sales, use, transfer, or other similar Taxes imposed as a result of the consummation of the Contemplated Transactions or performance of the Transaction Documents;
- Legal Proceedings. Any and all Liabilities of Sellers that relate to any Proceeding (A) involving the Purchased Assets or Business, including warranty, personal injury, breach of contract, failure to perform, infringement, noncompliance with Law, and tort Claims, that is (1) pending or threatened as of the Closing, or (2) commenced after the Closing but that arises out of or relate to any event, omission, or occurrence happening as of or prior to the Closing, or (B) set forth in Schedule 2.8(a)(vii);
- Pride Claims. Any and all Liabilities of Sellers under agreements entered into among Seahawk Parent and Pride International, Inc. ("Pride") at the time of Seahawk Parent's spin-off from Pride effective August 4, 2009, including (i) any obligation to indemnify Pride and its Affiliates for Mexican tax liabilities associated with Seahawk Parent's Mexican subsidiaries under that certain Tax Sharing Agreement dated August 4, 2009, between Seahawk Parent and Pride, part of which tax liabilities are supported by letters of credit from Pride, and any settlement with respect to such matters, and (ii) any and all Liabilities of Sellers (if any) to Pride and its Affiliates under such agreements resulting from the Contemplated Transactions;
- Pride Wyoming. Any and all Liabilities of Sellers relating to the loss of the Pride Wyoming Rig, including salvage and wreckage removal costs and any third party damage Claims and any continuing Liability under applicable Law relating to the Pride Wyoming Rig;
- FCPA Investigation. Any and all Liabilities of Sellers, including for purposes of clarity any and all fines and penalties (if any), of Sellers with respect to violations, alleged violations, and investigations for alleged potential or actual violations by Sellers and any of their respective Affiliates of the U.S. Foreign Corrupt Practices Act;
- Environmental Liabilities. Any and all Liabilities of Sellers with respect to any violation of Law including those arising from (A) the release, threatened release, presence, treatment, storage, disposal (including disposal at off site locations), handling, transportation or arrangement for transportation of hazardous substances prior to the Closing, (B) any failure of Sellers to comply in any respect with

- Environmental, Health, and Safety Laws prior to the Closing, and (C) any facts, events, or circumstances in existence prior to the Closing that give rise to Liabilities pursuant to Environmental, Health, and Safety Laws;
- Employee Benefit Plans. Any and all Liabilities of Sellers, fiduciaries or ERISA Affiliate under any Employee Benefit Plan maintained or contributed to by any Seller, fiduciary or any ERISA Affiliate or with respect to which any Seller, fiduciary or ERISA Affiliate has any Liability, whether prior to, on, or after the Closing, including any (a) failure to comply with all applicable Laws, (b) liability with respect to audits, inquiries, Proceedings, or Claims with any Governmental Authority with respect to any Employee Benefit Plan, (c) participation in any "multiemployer" plan (within the meaning of Section 3(37) of ERISA, whether or not governed by the provisions of ERISA) or withdrawal liability with respect to any multiemployer plan, (d) required employer contributions with respect to the Employee Benefit Plans, (e) accumulated funding deficiency, (f) Lien under ERISA or Section 412 of the Code, or (g) the termination of, or intent to terminate, any Plan;
- Employees. Any and all Liabilities of any nature of Sellers or any fiduciary or ERISA Affiliate to Seller Employees, including Liabilities with respect to (a) any Contract, plan or policy, (b) wages, withholdings, overtime pay, minimum wage, employment Tax, vacation, sick pay, bonuses, severance pay, retirement, or other compensation, (c) benefits under Employee Benefit Plans, (d) the Worker Adjustment and Retraining Notification Act (WARN) of August 4, 1988 or equivalent state or local statutory or regulatory requirements, (e) any collective bargaining agreement or obligation or requirement under the National Labor Relations Act, (f) reporting, filing, hiring or other employment obligations with the Office of Federal Contract Compliance Programs, (g) all immigration related obligations, including all requirements of the Immigration Reform and Control Act of 1986, (h) any governmental or administrative proceeding for the enforcement of labor and employment laws and regulations, and (i) all other employment and employment related federal, state and local statutes, regulations, administrative requirements, common laws, and public policies;
- Intercompany Liabilities. Any and all Liabilities of Sellers for intercompany advances, charges, or accounts payable of any kind or nature;
- Bankruptcy Claims. Any and all Claims filed against Sellers in their respective Bankruptcy Cases, except for the Assumed Liabilities, as

	limited in Section 2.7(a);
	• Broker Fees. Any and all fees, commissions, and other compensation to any broker, finder, or agent retained by any Seller as contemplated by Section 4.22;
	The Retained Liabilities shall constitute Claims and alleged Claims in the Sellers' Bankruptcy Cases; <i>provided</i> , <i>however</i> , that nothing herein shall create any rights in favor of the holders of such Claims and alleged Claims or create any priority right of payment.
Releases (APA §9.9)	In consideration for payment of the Cash Payment, issuance of the Hercules Shares and assumption of the Assumed Liabilities by Purchasers, Sellers shall cause the Plan of Reorganization to contain a mutual release provision by Sellers in favor of Purchasers and by Purchasers in favor of Sellers on terms mutually acceptable to the Parties.
Closing (APA § 3.1)	Provided that this Agreement shall not have been earlier terminated pursuant to Section 11.1, and further provided that all of the conditions set forth in Sections 10.1 and 10.2 to the obligations of the Parties to consummate the Contemplated Transactions (other than conditions with respect to actions each Party will take at the Closing itself) shall have been satisfied or waived, the closing of the Contemplated Transactions (the "Closing") shall take place at the offices of Fulbright & Jaworski L.L.P. at 1301 McKinney, Suite 5100, Houston, Texas 77010 commencing at 11:00 a.m. Central Time on the date that is no earlier than the first Business Day that the Transfer Order has become a Final Order, but no later than the third (3rd) Business Day following such date, or such other date or location as Purchasers and Sellers may mutually agree to in writing (the "Closing Date").
Termination Provisions (APA § 11.1(a)— (d))	 Mutual Consent. Purchasers and Sellers may terminate this Agreement as to all Parties by mutual written consent at any time prior to the Closing by written instrument authorized by the respective Boards of Directors of Seahawk Parent and Hercules Parent. By Purchasers. Purchasers may terminate this Agreement by giving written notice to Sellers at any time prior to the Closing in the event that (i) Sellers have breached any representation, warranty, or
-	covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.1(a) and 10.1(b) shall not have been satisfied, or cannot be satisfied by August 15, 2011 (the "Termination Date"); provided, that Purchasers shall have provided written notification to Sellers of such breach and the breach shall

have continued without cure for a period of ten (10) days after delivery of the notice of such breach; (ii) the Transfer Order shall not have been entered by the Bankruptcy Court and become a Final Order by one hundred twenty (120) days after the date hereof; (iii) any of the Bankruptcy Cases are converted to cases pursuant to Chapter 7 of the Bankruptcy Code; (iv) a Chapter 7 or Chapter 11 Trustee has been appointed; (v) an Examiner with expanded powers has been appointed in any of the Bankruptcy Cases of Sellers; (vi) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement; (vii) the DIP Financing is not approved within twenty-one (21) days of the Petition Date; (viii) Sellers are in breach or default in any material respect of, the DIP Credit Documents, DIP Order or DIP Budget after five days notice by any Person and opportunity to cure; or (ix) the Termination Fee Order shall not have been entered by the Bankruptcy Court within twenty-five (25) days of the date hereof or shall not have become a Final Order by the eleventh day after such entry of the Termination Fee Order. At their sole and absolute discretion, Purchasers have the right to waive termination or agree to extend any deadlines under this Section11.1(b).

- By Sellers. Sellers may terminate this Agreement by giving written notice to Purchasers at any time prior to the Closing in the event that (i) any Purchaser has breached any representation, warranty, or covenant contained in this Agreement to such an extent that the conditions set forth in Sections 10.2(a) and 10.2(b) shall not have been satisfied, or cannot be satisfied by the Termination Date; provided, that Sellers shall have provided written notification to Purchasers of such breach and the breach shall have continued without cure for a period of ten (10) days after delivery of the notice of breach; or (ii) in accordance with the terms and subject to the conditions of Section 6.5; or (iii) Purchasers' failure to timely obtain the Consent as contemplated by Section 6.9.
- By Either Party. Sellers or Purchasers may terminate this Agreement by giving written notice to the other Party if (i) any court of competent jurisdiction or any other Governmental Authority in a suit instituted by a third party or a Governmental Authority shall have issued an Order or shall have taken any other action prior to the Termination Date permanently enjoining, restraining, or otherwise prohibiting the Contemplated Transactions or a material portion thereof, (ii) the Bankruptcy Court enters an Order that materially alters, contravenes or violates this Agreement or denies approval of this Agreement or the Contemplated Transactions, or (iii) the Closing has not occurred by the Termination Date, provided, that the Party electing to terminate shall not have caused such failure to close.

- 23. At the Preliminary Sale Hearing, the Debtors request that the Court schedule an expedited hearing on the Sale Motion.
- B. REQUEST FOR THE SETTING OF DEADLINES FOR FILING OBJECTIONS TO (i) THE PROPOSED SALE AND (ii) TO THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND THE PAYMENT OF CURE CLAIMS IN CONNECTION THEREWITH
- 24. The Debtors request that, at the Preliminary Sale Hearing, the Court order that any objections to the Sale Motion (a "Sale Objection") set forth in writing with particularity the factual and legal basis for the objection and be filed by a party with standing and served on counsel for the Debtors, counsel for the Purchaser and all other parties entitled to notice no later than 12:00 p.m. (CST) on the third Business Day prior to the Sale Hearing Date (the "Objection Deadline").
- 25. The Debtors further request the Court to order that, unless an objection to the sale is timely filed and served by the Objection Deadline, all interested parties who have received actual or constructive notice thereof be deemed to have waived and released any right to assert a Sale Objection and shall be deemed to have otherwise consented to the proposed sale and shall be forever barred and estopped from asserting any Sale Objection against the Debtors, or the Purchaser.

ASSUMPTION AND ASSIGNMENT OF ASSIGNED CONTRACTS

As part of the sale, the Debtors desire to assume and assign the executory contracts set forth on **Exhibit B** hereto (Schedule 2.1(b) to the APA)(the "**Assigned Contracts**") to the Purchaser. Pursuant to this Sale Motion, the Debtors seek authority pursuant to sections 105(a) and 365 of the Bankruptcy Code to (i) assume and assign the Assigned Contracts to the Purchaser, and (ii) execute and deliver to the Purchaser, or its assignee, such documents or other instruments

as may be necessary by the Purchaser to assign and transfer the Assigned Contracts at the Final Sale Hearing. The Debtors have set forth on **Exhibit B** the cure amounts they believe to be payable to each of the counterparties under the Assigned Contracts.

- 27. The Debtors request that the Court order, at the Preliminary Sale Hearing, that any objections relating to the proposed assumption and assignment of executory contracts (including, but not limited to, any objections relating to the validity of the cure amount as determined by the Debtors or to otherwise assert that any amounts, defaults, conditions, or pecuniary losses must be cured or satisfied under any of the Assigned Contracts as of the date of the Sale Hearing) (a "Cure Objection") be filed and served so as to be actually received by counsel for both the Debtors and the Purchaser no later than the Objection Deadline.
- 28. The Debtors further request that the Court order that, unless a Cure Objection is timely filed and served by a party to an Assigned Contract by the Objection Deadline, all interested parties who have received actual or constructive notice thereof be deemed to have waived and released any right to assert a Cure Objection and to have otherwise consented to the assignment of the Assigned Contracts with payment of the cure amounts set forth on **Exhibit B** hereto and shall be forever barred and estopped from asserting or claiming against Debtors, the Purchaser or any assignee that any additional amounts are due or that a default exists with respect to such Assumed Contract.
- C. APPROVAL OF THE FORM AND MANNER OF NOTICE OF THE TERMINATION FEES, SALE HEARING AND OF THE OBJECTION DEADLINES.
- 29. At the Preliminary Sale Hearing, the Debtors request approval of the form and manner of notice of the Sale Hearing and of the Objection Deadlines to be served on all parties-in-interest affected by the Sale Motion.

Notice") attached hereto as Exhibit D provides due and adequate notice of the Final Sale Hearing and of the proposed assumption and assignment of the Assigned Contracts and the payment of cure amounts in connection therewith.

D. APPROVAL OF THE DISPOSITION OF SALE PROCEEDS.

- 31. Finally, at the Sale Hearing, the Debtors request, pursuant to sections 105, 363, and 365 of the Bankruptcy Code and Bankruptcy Rules 2002, 6003, 6004, 6006, and 9006, that, at the Final Sale Hearing, the Court will enter an order (the "<u>Transfer Order</u>"):
 - (i) approving the APA by and between the Debtors and the Purchaser, substantially in the form attached hereto as Exhibit A,
 - (ii) authorizing the sale of the Debtors' assets under the APA free and clear of all liens, claims, encumbrances, and other interests;
 - (iii) authorizing the assumption and assignment of substantially all of the Debtors' executory contracts, identified in the attached Exhibit B, to the Purchaser;
 - (iv) (i) authorizing the Debtors to use the cash portion of the Aggregate Consideration to pay the DIP Obligations (the "DIP Loan Payment"), (ii) directing the Debtors to make the DIP Loan Payment concurrently with the Closing; and (iii) stating that, upon receipt of the DIP Loan Payment, (i) the DIP Lenders are authorized to apply the DIP Loan Payment to the DIP Obligations, (ii) such payment shall constitute indefeasible payment of the DIP Obligations so paid and (iii) any commitments of the DIP Lenders to make any loans to the Debtors are terminated;
 - (v) authorizing the Debtors to place the Hercules Shares received as the remainder of the Aggregate Consideration with an Escrow Agent in an account to be held in trust pursuant to the an escrow agreement and to be distributed to holders of allowed claims and interests of the Debtors pursuant a confirmed chapter 11 plan or pursuant to further orders of this Court; and
 - (vi) granting other related and appropriate relief.

BASIS FOR RELIEF

- A. Approval of the Sale of Substantially All of the Debtors' Assets.
- 32. Section 363(b)(1) of the Bankruptcy Code provides, in relevant part, that "[t]he trustee, after notice and a hearing, may use, sell, or lease, other than in the ordinary course of business, property of the estate." 11 U.S.C. § 363(b)(1).
- Although section 363 of the Bankruptcy Code does not set forth a standard for 33. determining when it is appropriate for a court to authorize the sale or disposition of a debtor's assets, in applying this section, courts have required that it be based upon the debtor's business judgment. In re Continental Air Lines, Inc., 780 F.2d 1223, 1226 (5th Cir. 1986) (A debtor must demonstrate "some articulated business justification for using, selling, or leasing property outside of the ordinary course of business."); Comm. Of Equity Security Holders v. Lionel Corp., 722 F.2d 1063, 1071 (2d Cir. 1983) (A sale of a substantial part of a Chapter 11 estate other than in the ordinary course of business may be conducted if a good business reason exists to support it.). "Where the debtor articulates a reasonable basis for its business decisions (as distinct from a decision made arbitrarily or capriciously), courts will generally not entertain objections to the debtor's conduct." Comm. of Asbestos-Related Litigation v. Johns-Manville Corp. (In re Johns-Manville Corp.), 60 B.R. 612, 616 (Bankr. S.D.N.Y. 1986). When a valid business justification exists, the law vests the debtor's decision to use property out of the ordinary course of business with a strong presumption that "in making a business decision the directors of a corporation acted on an informed basis, in good faith and in the honest belief that the action taken was in the best interest of the company." Official Comm. of Subordinated Bondholders v. Integrated Res., Inc. (In re Integrated Res., Inc.), 147 B.R. 650, 656 (S.D.N.Y. 1992) (quoting Smith v. Van Gorkom, 488 A. 2d 858, 872 (Del. 1985)).

- 34. Moreover, section 105(a) of the Bankruptcy Code grants this Court broad authority under its equitable powers to fashion any order or decree that would preserve or protect the value of the debtor's assets. See 11 U.S.C. § 105(a); see also, Adelphia Commc'ns Corp. v. Rigas (In re Adelphia Commc'ns Crop.), 2003 WL 21297258, at *4 (S.D.N.Y. June 4, 2003) ("Section 105 of Title 11 provides the bankruptcy courts with a broad range of equitable powers over cases within its jurisdiction.").
- Section 363(b), such as: (i) whether a sound business reason exists for the proposed transaction; (ii) whether fair and reasonable consideration is provided; (iii) whether the transaction has been proposed and negotiated in good faith; and (iv) whether adequate and reasonable notice is provided. See In re Condere, 228 B.R. 615, 626 (Bankr. S.D. Miss. 1998); See In re Delaware & Hudson Ry. Co., 124 B.R. 169, 176 (D. Del. 1991). The Debtors respectfully submit that the proposed sale of the Purchased Assets fits squarely within the parameters of the sound business judgment test articulated above.

1. A Sound Business Reason Exists For The Sale Of The Purchased Assets

36. The Debtors' decision to sell property out of the ordinary course of business enjoys a strong presumption "that in making a business decision the Debtors acted on an informed basis, in good faith and in an honest belief that the action taken was in the best interests of the company." *In re Integrated Res., Inc.*, 147 B.R. 650, 656 (S.D.N.Y. 1992). Therefore, parties objecting to the Debtors' proposed sale must make a showing of "bad faith, self-interest, or gross negligence." *Integrated Res.*, 147 B.R. at 656.

- 37. The Debtors have determined, in their considered business judgment, that the sale of the Purchased Assets pursuant to the APA is appropriate and in the best interest of, and represents the best means of preserving and maximizing value for, the Debtors' estates.
- 38. As discussed above, the Debtors have faced pre-petition monthly operating losses for an extended period of time. The Debtors project continued losses if a sale is not effectuated. Without additional funding, the Debtors anticipate that they will soon run out of cash to operate their business. Because of the lack of capital, continued operation of the Debtors' business without a sale of the Purchased Assets is not an option.
- 39. To avoid a chapter 7, which likely would result in unsecured creditors and equity interests receiving little or no recovery or distribution, subject to approval of the Court, the Debtors have agreed to sell the Purchased Assets to the Purchaser pursuant to section 363 of the Bankruptcy Code.
- 40. The sale to the Purchaser will permit the Debtors to distribute cash and Hercules Common Stock generated from the sale to creditors and interestholders pursuant to a chapter 11 plan or pursuant to further orders of this Court. Accordingly, the proposed disposition of assets is born of the Debtors' economic circumstances, and not "bad faith, self-interest or gross negligence."
- 41. The Debtors note that while the sale of the Purchased Assets is not part of its ordinary business, the proposed sale will not substantially impact (and, in fact, will facilitate a plan of liquidation) or significantly alter the rights of the Debtors' creditors. Therefore, the Debtors submit that the proposed sale of the Purchased Assets in no way constitutes a *sub rosa* plan or falls outside the scope of Bankruptcy Code § 363. *C.f. Continental Airline, Inc. v. Braniff Airways, Inc. (In re Braniff Airways, Inc.)*, 700 F.2d 935. 939 (5th Cir. 1983) (noting that the

proposed sale pursuant to section 363 would require "significant restructuring of the rights of Braniff creditors").

2. The Purchaser Is Paying Fair And Reasonable Consideration For The Purchased Assets

42. The aggregate consideration for the Purchased Assets is approximately One Hundred Five Million Dollars (\$105,000,000). Given the difficulty the Debtors have faced trying to sell the Purchased Assets, the Debtors believe that the purchase price is fair and reasonable consideration for the Purchased Assets and is the result of extensive arm's length negotiations between the respective representatives of the Debtors and the Purchaser. The purchase price offered by the Purchaser represents the highest and best bid for the Purchased Assets received to date.

3. The Sale Of The Purchased Assets Has Been Proposed And Negotiated In Good Faith

- 43. The Purchaser should be afforded all protections under Bankruptcy Code Section 363(m) as a good faith purchaser. Section 363(m) of the Bankruptcy Code provides that "the reversal or modification on appeal of an authorization under subsection (b) or (c) of this section of a sale or lease of property does not affect the validity of a sale or lease under such authorization to an entity that purchased or leased such property in good faith." 11 U.S.C. § 363(m). Courts generally determine that parties have acted in good faith with respect to a proposed sale if the purchase price is adequate and reasonable and the terms of the sale are disclosed fully. See In re Abbotts Dairies of Pa., Inc., 788 F.2d 143, 149-50 (3d Cir. 1986).
- 44. The Debtors submit that the APA is not tainted by self-dealing, collusion or manipulation as it was proposed and negotiated in good faith and executed by the Debtors and the Purchaser in an arm's length transaction, and the Debtors and the Purchaser have fully disclosed all material information.

 90303148.17

- 45. Similarly, the Purchaser's proposed consideration for the Purchased Assets was made in good faith. Moreover, the Purchaser was not preferred and did not receive special treatment. Though Randall D. Stilley was formerly employed as the Chief Executive Officer of Hercules, his former relationship did not influence Simmons' solicitation process or the negotiation or consummation of the APA. All aspects of Simmons' solicitations and the negotiation and consummation of the APA were comported in good faith and pursuant to reasonable business judgment.
- 46. Accordingly, the Debtors assert that it is appropriate that the Purchaser receives the protection of a good faith finding by the Court in accordance with section 363(m).
 - 4. The Court Should Authorize The Sale Of The Purchased Assets To The Purchaser Free And Clear Of All Liens, Claims, Encumbrances, And Interests
- 47. The Debtors request that the sale and transfer of the Purchased Assets be approved free and clear of all interests, liens, claims, encumbrances and other interests, other than those specifically assumed by the Purchaser. Such relief is consistent with the provisions of section 363(f).
- 48. The Court may, pursuant to section 363(f) of the Bankruptcy Code, authorize a debtor-in-possession to sell property free and clear of any interest, lien, claim or interest of another entity in such property if any of the following circumstances pertain:
 - (i) applicable non-bankruptcy law permits sale of such property free and clear of such interest;
 - (ii) such entity consents;
 - (iii) such interest is a lien and the price at which such property is to be sold is greater than the aggregate value of all liens on such property;
 - (iv) such interest is in bona fide dispute; or
 - (v) such entity could be compelled, in a legal or equitable proceeding, to accept a money satisfaction of such interest.

11 U.S.C. § 363(f).

- 49. As indicated by the use of the disjunctive term "or," satisfaction of any of the five requirements listed in section 363(f) is sufficient to permit the sale of assets free and clear of all interests, liens, claims, encumbrances, pledges, mortgages, security interests, charges, options and other interests, including any right of setoff. *See, e.g., In re C-Power Prods., Inc.*, 230 B.R. 800 (Bankr. N.D. Tex. 1998); *In re Elliott*, 94 B.R. 343, 345 (E.D. Pa. 1988) (stating that section 363(f) is written in the disjunctive; the court may approve a sale "free and clear" provided that at least one of the subsections is met).
- Assets as of the Petition Date are Natixis, New York Branch, as Administrative Agent for the Lenders (the "Pre-Petition Lenders"), and the U.S. Internal Revenue Service. First, the price at which the Purchased Assets is to be sold is greater than the aggregate value of all liens on such property pursuant to section 363(f)(iii) of the Bankruptcy Code. Second, it is anticipated that the DIP Lenders will be substituted in the place of the Pre-Petition Lenders with senior liens on the Purchased Assets as of the Closing Date.
- 51. Therefore, the Debtors submit, to the extent section 363(f) is applicable, at least one requirement of section 363(f)(iii) has been satisfied and the Purchased Assets may be sold and transferred free and clear of all liens, claims, and encumbrances.

5. Approval Of The APA

52. Courts routinely approve asset purchase agreements in connection with the sale of a debtor's assets. See In re Arlco, Inc., 239 B.R. 261, 265 (Bankr. S.D.N.Y. 1999). In this case, the Purchaser's APA was the subject of intense arm's-length negotiations between representatives of the Debtors and the Purchaser, and the Debtors submit that the terms and conditions of the

APA are the best that could be obtained under the circumstances, and that entry into the APA is a sound exercise of the Debtors' business judgment.

- 6. Assumption And Assignment Of The Assigned Contracts Should Be <u>Authorized</u>
- Under section 365 of the Bankruptcy Code, a debtor-in-possession may assume an executory contract "subject to the court's approval" if (a) the debtor-in-possession cures, or provides adequate assurance of a prompt cure, of any default under such contract or lease; (b) compensates, or provides adequate assurance that prompt compensation will be paid to any non-debtor party to such contract or lease for any pecuniary loss to such party resulting from such default; and (c) provides adequate assurance of future performance under such contract. See 11 U.S.C. § 365(b)(1).
- 54. Courts have consistently applied the business judgment test in determining whether to approve a debtor-in-possession's decision to assume an executory contract. *See Richmond Leasing Co. v. Capital Bank, N.A.*, 762 F.2d 1303, 1312 (5th Cir. 1985). The business judgment test is satisfied, when the debtor-in-possession has determined in good faith that assumption of a contract will benefit the estate.
- 55. The Debtors have determined, in the exercise of their sound business judgment, that the assumption and assignment of the Assigned Contracts will benefit the Debtors' estates. The assumption and assignment of the Assigned Contracts is essential to the deal between the Debtors and the Purchaser. Further, the assumption and assignment of the Assigned Contracts will significantly limit administrative expense and rejection damage claims.
- 56. The Debtors also propose to cure defaults and provide adequate assurance of future performance of the Assigned Contracts. The cure amounts that the Debtors believe to be owed in 90303148.17

connection with the assumption and assignment of the Assigned Contracts are set forth on the attached **Exhibit B**. The payment of any cure amounts related to the Assigned Contracts will be in final satisfaction of all obligations to cure defaults and compensate the non-debtor counterparties for any pecuniary losses under section 365(b)(1) of the Bankruptcy Code.

57. At the Final Sale Hearing to approve the sale, the Debtors will provided evidence that illustrates the Purchaser's financial credibility, experience in the industry, and willingness and ability to perform under the Assigned Contracts. The meaning of "adequate assurance of future performance" depends on the facts and circumstances of each case, but should be given "practical, pragmatic construction." *In re Sanshoe Worldwide Corp.*, 139 B.R. 585, 592 (S.D.N.Y. 1992). When an executory contract or unexpired lease is to be assumed and assigned, adequate assurance may be provided by, among other things, demonstrating the financial health of the assignee and its experience and ability in managing the type of enterprise or property assigned. *See In re Bygaph*, 56 B.R. 596, 605-06 (Bankr. S.D.N.Y. 1986).

BANKRUPTCY RULES 6004(H) AND 6006(D) SHOULD BE WAIVED

- 58. Bankruptcy Rule 6004(h) provides that an "order authorizing the use, sale, or lease of property ... is stayed until the expiration of 14 days after the entry of the order, unless the court orders otherwise." Fed. R. Bank. P. 6004(h). Bankruptcy Rule 6006(d) provides that an "order authorizing the trustee to assign an executory contract or unexpired lease ... is stayed until the expiration of 14 days after the entry of an order, unless the court orders otherwise." Fed R. Bank. P. 6006(d).
- 59. The Debtors request that any order entered pursuant to this Sale Motion authorizing the consummation of a transaction that is deemed a sale of assets and/or an assignment of an executory contract be effective immediately by providing that the 14-day stay

under Rules 6004 or 6006, as the case may be, is inapplicable, so that they may proceed to close on the transaction as expeditiously as possible and within the time frames contemplated by the Debtors and the Purchaser.

NOTICE OF THE SALE MOTION

- 60. In accordance with applicable rules, including the United States Bankruptcy Court for the Southern District of Texas Procedures for Complex Chapter 11 Bankruptcy Cases, notice of this Motion has been provided by overnight delivery and e-mail or facsimile to: (a) Natixis, New York Branch, as Administrative Agent for the Lenders; (b) the 30 largest unsecured creditors of the Debtors on a consolidated basis; (c) Pride International, Inc.; (d) Hercules Offshore, Inc.; (e) D.E. Shaw Direct Capital Portfolios, L.L.C.; (f) the U.S. Trustee's Office; (g) the United States Attorney's Office; (h) the Securities and Exchange Commission; (i) the Internal Revenue Service; and (j) parties to executory contracts proposed to be assumed and assigned. The Debtors believe that the notice provided for herein is fair and adequate and no other or further notice is necessary.
- 61. In addition, as soon as approved by the Court, the Debtors will immediately serve the Sale Hearing Notice, substantially in the form of the attached **Exhibit D**, by standard first-class mail, e-mail, or facsimile, to all known parties in interest in the Debtors' bankruptcy cases and on all other parties ordered by the Court.

CONCLUSION

WHEREFORE, PREMISES CONSIDERED, the Debtors respectfully request (A) at the preliminary hearing on the Sale Motion, entry of an order (1) scheduling an expedited hearing to consider final approval of the Sale Motion; (2) setting deadlines and requirements for parties in interest to file (a) objections to the Sale Motion, and (b) objections to the proposed assumption and assignment of executory contracts in connection with the sale and the payment of cure claims

in connection therewith; and (3) approving the form and manner of notice of the Final Sale Hearing; and (B) at the Final Sale Hearing on the Sale Motion, entry of an order (1) approving the Debtors' sale of the Purchased Assets to the Purchaser free and clear of liens, claims, encumbrances and interests to the fullest extent of section 363 of the Bankruptcy Code; (2) approving the proposed assumption and assignment by the Debtors to purchaser of the Assumed Contracts and payment of all cure claims in connection therewith pursuant to section 365 of the Bankruptcy Code; (3) approving the Debtors' use of the cash portion of the consideration received from Purchaser at closing to repay all outstanding principal, interest and fees due under the DIP Loan as of the Closing Date; (4) approving the transfer by Hercules of the Hercules Shares at closing to an escrow agent to be distributed by the escrow agent pursuant to a confirmed chapter 11 plan or pursuant to further orders of this Court; and (C) all such other and further relief as may be just and proper both at law and in equity.

Dated: February 11, 2011.

Respectfully submitted,

FULBRIGHT & JAWORSKI L.L.P.

By: /s/ Berry D. Spears
Berry D. Spears
State Bar No. 18893300
Johnathan C. Bolton
State Bar No. 24025260
1301 McKinney, Suite 5100
Houston, Texas 77010-3095
Telephone: (713) 651-5151
Facsimile: (713) 651-5246
bspears@fulbright.com
jbolton@fulbright.com

and

JORDAN, HYDEN, WOMBLE, CULBRETH & HOLZER P.C.

Shelby A. Jordan
State Bar No. 11016700
Nathanial Peter Holzer
State Bar No. 00793971
500 N. Shoreline Blvd., Suite 900
Corpus Christi, Texas 78401-0341
Telephone: (361) 653-6624
Facsimile: (361) 888-5555
sjordan@jhwclaw.com
pholzer@jhwclaw.com

PROPOSED ATTORNEYS FOR THE DEBTORS AND DEBTORS-IN-POSSESSION

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	§ Chapter 11
Seahawk Drilling Inc.,	§ Case No.: 11
Seahawk Mexico Holdings LLC,	§ Case No.: 11
Seahawk Drilling Management LLC,	§ Case No.: 11
Seahawk Offshore Management LLC,	§ Case No.: 11
Energy Supply International LLC,	§ Case No.: 11
Seahawk Drilling LLC,	§ Case No.: 11
Seahawk Global Holdings LLC,	§ Case No.: 11
Seahawk Drilling USA LLC,	§ Case No.: 11
Debtors.	§ Motion for Joint Administration Pending

EMERGENCY MOTION FOR AN ORDER APPROVING THE PAYMENT OF CONTINGENT TERMINATION FEE PAYABLE TO HERCULES OFFSHORE, INC. AS AN ADMINISTRATIVE EXPENSE OF THE DEBTORS' ESTATES IN ACCORDANCE WITH THE TERMS OF THE ASSET PURCHASE AGREEMENT

NOTICE UNDER BLR 9013(b) and 9013(i)

THIS MOTION SEEKS AN ORDER THAT MAY ADVERSELY AFFECT YOU. IF YOU OPPOSE THE MOTION, YOU SHOULD IMMEDIATELY CONTACT THE MOVING PARTY TO RESOLVE THE DISPUTE. IF YOU AND THE MOVING PARTY CANNOT AGREE, YOU MUST FILE A RESPONSE AND SEND A COPY TO THE MOVING PARTY. YOU MUST FILE AND SERVE YOUR RESPONSE WITHIN 21 DAYS OF THE DATE THIS WAS SERVED ON YOU. YOUR RESPONSE MUST STATE WHY THE MOTION SHOULD NOT BE GRANTED. IF YOU DO NOT FILE A TIMELY RESPONSE, THE RELIEF MAY BE GRANTED WITHOUT FURTHER NOTICE TO YOU. IF YOU OPPOSE THE MOTION AND HAVE NOT REACHED AN AGREEMENT, YOU MUST ATTEND THE HEARING. UNLESS THE PARTIES AGREE OTHERWISE, THE COURT MAY CONSIDER EVIDENCE AT THE HEARING AND MAY DECIDE THE MOTION AT THE HEARING. EMERGENCY RELIEF HAS BEEN REQUESTED. IF THE COURT CONSIDERS THE MOTION ON AN EMERGENCY BASIS, THEN YOU WILL HAVE LESS THAN 21 DAYS TO ANSWER. IF YOU OBJECT TO THE REQUESTED RELIEF OR IF YOU BELIEVE THAT THE EMERGENCY CONSIDERATION IS NOT WARRANTED; YOU SHOULD FILE AN IMMEDIATE RESPONSE.

REPRESENTED PARTIES SHOULD ACT THROUGH THEIR ATTORNEY.

TO THE HONORABLE UNITED STATES BANKRUPTCY JUDGE:

Seahawk Drilling, Inc. ("Seahawk" or the "Company") and its above-captioned affiliated debtors (collectively, the "Debtors"), file this Emergency Motion For An Order Approving the Payment of a Contingent Termination Fee Payable to Hercules Offshore, Inc. as an Administrative Expense of the Debtors' Estates in accordance with the terms of the Asset Purchase Agreement with Hercules Offshore, Inc. (the "Motion"). In support of this Motion, the Debtors respectfully state as follows:

JURISDICTION AND PROCEDURAL STATUS

- 1. This Court has jurisdiction over this Motion pursuant to 28 U.S.C. §§ 157 and 1334. Venue of the Debtors' chapter 11 cases in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409. This is a core proceeding pursuant to 28 U.S.C. § 157(b).
- 2. On the date hereof (the "<u>Petition Date</u>"), the Debtors filed their respective voluntary petitions for relief under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>"). Contemporaneously herewith, the Debtors have sought joint administration of these chapter 11 cases.
- 3. The Debtors continue to operate their businesses and manage their properties as debtors-in-possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee, examiner, or creditors' committee has been appointed in these cases.

FACTUAL BACKGROUND

A. Events Leading To The Debtors' Petitions

- 4. Seahawk is a Delaware corporation formed in 2008 and headquartered in Houston, Texas, with operations throughout the U.S. Gulf of Mexico. Each of the other debtors is a direct or indirect subsidiary of Seahawk.¹ As of the Petition Date, Seahawk's stock is publicly traded on the NASDAQ Exchange under the ticker symbol, "HAWK."
- 5. Seahawk operates a jackup rig business that provides contract drilling services to the oil and natural gas exploration and production industry throughout the U.S. Gulf of Mexico.² The Debtors contract with their customers on a dayrate basis to provide rigs and drilling crews. The Debtors' fleet of mobile offshore drilling rigs consists of twenty (20) shallow water jackup rigs. Three (3) of the Debtors' six (6) rigs in Texas are located at Matagorda Island 721 in Aransas County. Two rigs are located at Sabine Pass, Texas, and one rig is located at High Island 176 in Galveston County. The remainder of Seahawk's rigs are currently located in Louisiana.
- 6. Seahawk is a former subsidiary of Pride International, Inc. ("<u>Pride</u>"). On August 4, 2009, the Board of Directors of Pride approved a plan to separate Pride into two independent, publicly-traded companies (the "<u>Spin-Off</u>"). The separation occurred through the distribution to Pride stockholders of all of the shares of common stock of Seahawk that held, directly or indirectly, the assets and liabilities of Pride's jackup rig business.
- 7. On August 24, 2009 (the "Spin-Off Date"), each Pride stockholder received one-fifteenth (1/15) of a share of Seahawk's common stock for each share of Pride common stock held

¹ The debtor-subsidiaries of Seahawk are: Seahawk Drilling LLC, Energy Supply International LLC, Seahawk Global Holdings LLC, Seahawk Mexico Holdings LLC, Seahawk Drilling Management LLC, Seahawk Offshore Management LLC and Seahawk Drilling USA LLC. The subsidiaries of Seahawk that are incorporated in Mexico or have branches in Mexico are not a part of this chapter 11 proceeding.

² Jackup rigs are mobile, self-elevating drilling platforms equipped with legs that can be lowered to the ocean floor until a foundation is established to support the drilling platform. The Debtors' rigs work in water depths up to 300 feet and can drill to a depth of 25,000 feet.

at the close of business on August 14, 2009. After the Spin-Off, Seahawk became an independent public company. Pride no longer retains any ownership interest in Seahawk, although the two companies are parties to various agreements entered into in connection with the Spin-Off (collectively, the "<u>Pride Agreements</u>").

- 8. Since mid-2008, the demand for drilling services has declined dramatically, principally as a result of the global financial crisis, declining prices of crude oil and natural gas and deteriorating worldwide economic conditions. The decline in the United States jackup rig market since 2009 has been one of the sharpest downturns for domestic jackup rig activity over the past thirty (30) years. In addition, the regulatory and financial uncertainties regarding a former customer, PEMEX, have had a significant effect on Seahawk's business. Finally, from August 2009 to March 2010, the Debtors' active rig count averaged five (5) to seven (7) working rigs at any given time.
- 9. On April 20, 2010, the demand for offshore drilling services in the Gulf of Mexico was further negatively impacted by the Macondo well blowout, prompting, among other things, the United States Government to issue a moratorium on all U.S. offshore drilling. On May 26, 2010, the moratorium on Gulf of Mexico drilling in waters less than 500 feet deep was lifted. Subsequently, in November 2010, the deep water moratorium was lifted. However, since the Macondo incident, notwithstanding the termination of the Government-imposed drilling prohibitions, the Debtors' customers are experiencing significant delays in the issuance of drilling permits and very few new drilling permits have been issued. In addition, there is an increasingly uncertain regulatory and cost environment which continues to adversely affect the Debtors' business.

³ Pemex Exploración y Producción ("<u>PEMEX</u>") is the exploration and production subsidiary of the national oil company of Mexico.
90364954.6

- working rigs during October 2010. While there has been some recent marginal improvement in market conditions (seven (7) rigs are currently working), the combined impact of all of these events, together with negative cash flows throughout 2009 and 2010, has severely stressed and exhausted the Debtors' liquidity and even the current operating level results continue to produce negative cash flows for the Debtors. Seahawk's Board of Directors (the "Board") and its management have closely monitored the impact of these conditions and evaluated potential alternatives to address and improve these circumstances with the assistance of various professionals and outside advisors, including Alvarez & Marsal North America, LLC ("A&M").
- 11. In November 2010, Seahawk publicly announced that its Board had initiated a process to explore and consider possible strategic alternatives, including, but not limited to transactions involving additional funding, recapitalizations, sales of assets, or a sale or merger of Seahawk. Seahawk's Board, through its finance committee (the "Finance Committee"), with regular and substantial Board and independent directors' participation, supervised the process and were managed by Simmons & Company International ("Simmons") and other advisors.
- 12. In February 2011, after evaluating the indications of interest and potential offers received from a number of interested parties, the Board authorized the Debtors to enter into an Asset Purchase Agreement (the "APA") with Hercules Offshore, Inc. ("Hercules"). The executed APA contemplates the acquisition by Hercules or one or more of its subsidiaries ("Purchaser") of substantially all of the assets and jackup rigs of the Debtors (the "Purchased Assets") through a sale pursuant to section 363 of the Bankruptcy Code (the "Transaction"). The aggregate consideration for the Purchased Assets (defined in the APA as the "Base Aggregate")

⁴ The summary description of the APA set forth herein is qualified in its entirety by reference to the terms of the APA itself.

<u>Consideration</u>") is (a) 22,321,425 shares of Hercules Common Stock (the "<u>Hercules Shares</u>") plus (b) cash in an amount equal to \$25,000,012 (the "<u>Cash Payment</u>"), subject to certain adjustments. Using the closing stock price of Hercules' stock as of February 10, 2011, the Base Aggregate Consideration would be valued at approximately \$105 million before any adjustments. The Base Aggregate Consideration is to be payable at closing by Purchaser to Sellers.

- 13. The Debtors have sought chapter 11 protection in order to protect and preserve their going concern value and to facilitate a prompt sale of substantially all of their assets to Hercules for the benefit of all creditors and stakeholders. The Debtors have determined in the prudent exercise of their business judgment that the proposed course of action is the best alternative to ensure that the maximum value of Seahawk's assets will inure to the benefit of all of their creditors and interestholders.
- 14. Additional information regarding the Debtors' business, capital structure, the circumstances and events leading to these chapter 11 cases, and the reasons why emergency consideration of this Motion is requested are contained in the Declaration of Randall D. Stilley in Support of First Day Motions, filed contemporaneously herewith and incorporated herein by reference.

B. The Proposed Sale

- 15. Contemporaneously herewith, the Debtors filed a motion (the "Sale Motion") seeking to effectuate the terms and provisions of the APA by and between the Debtors and Purchaser.
- 16. The Debtors submit that the sale of the Purchased Assets is essential to preserve the going concern value of Debtors' businesses. The Debtors assert that each day the Purchased Assets are subject to the market, and the bankruptcy process will cause further losses in operating

revenue, diminish the value of the Purchased Assets and threaten the viability of projected creditor and stakeholder recoveries. Accordingly, from the Debtors' perspective, the APA represents a reasonable exercise of the Board's and management's business judgment and an expedited sale is in the best interests of the Debtors and all of their economic claim and interestholders.

- 17. As described in the Sale Motion, the Debtors and their transaction advisors, Simmons, have expended substantial time and resources soliciting potential purchasers of the Debtors' assets pre-petition and, accordingly, are of the opinion that the APA with the Purchaser represents the best terms the Debtors can reasonably anticipate to obtain and that the process they have implemented is comprehensive, fulsome and appropriate in every respect. In the opinion of Simmons and the Board, further marketing efforts would not likely result in any offers superior to the terms embodied in the Hercules APA.
- 18. The Debtors believe that, because Hercules' offer is the highest and best offer the Debtors have been able to obtain, the APA is in the best interests of the Debtors and all parties-in-interest, and accordingly, should ultimately be approved.

RELIEF REQUESTED

- 19. Section 11.2 of the APA provides for the payment of termination fees in the event that the APA is terminated by one of the parties for certain enumerated reasons.
 - 20. With respect to the Purchasers, section 11.2(d) of the APA states:
 - If (i) Sellers terminate this Agreement pursuant to Section 11.1(c)(ii) or (ii) Purchasers terminate this Agreement pursuant to Section 11.1(b)(i) as a result of a breach by Sellers of Section 6.5 or Section 6.6 or (iii) Purchasers terminate this Agreement pursuant to Section 11.1(b)(ii)-(ix), then Sellers shall pay to Purchasers an aggregate amount equal to the Termination Fee and Purchasers' Reimbursable Expenses, which aggregate payment shall be treated in accordance with the Order entered approving the Termination Fee Motion, or if no such Order has been entered, then by wire transfer of immediately available funds no later than the second (2nd) Business Day following such termination of this Agreement. Sellers' payment of the Termination Fee and Purchasers'

Reimbursable Expenses pursuant to this Section 11.2(d) shall be the sole and exclusive remedy of Purchasers with respect to the occurrences giving rise to such payment.

APA at § 11.2(d) (emphasis added).

21. Section 11.2(d) refers to a termination by Sellers under Section 6.5 of the APA, which provides:

Notwithstanding anything herein to the contrary, prior to entering into an agreement in connection with any Competing Transaction, (A) Sellers shall provide prior written notice to Purchasers, at least 48 hours in advance (the "Notice Period"), of its intention to take such action with respect to such Competing Transaction, specifying the material terms and conditions of any such Competing Transaction (including the identity of the party proposing to effect such Competing Transaction) and furnishing to Purchasers a copy of the relevant proposed transaction agreements with the party proposing to effect such Competing Transaction and other material documents and (B) during the Notice Period, and in any event prior to taking such action, Sellers shall negotiate, and shall cause its financial and legal advisors to negotiate, with Purchasers in good faith (to the extent Purchasers desire to negotiate) to make such adjustments in the terms and conditions of this Agreement so that such proposal or offer for a Competing Transaction ceases to constitute a Superior Proposal.

At any time prior to the Transfer Order, Sellers may terminate this Agreement and concurrently with such termination, upon payment to Purchasers of the Termination Fee and Purchasers' Reimbursable Expenses pursuant to Section 11.2, enter into a definitive agreement with respect to a Superior Proposal if the Board of Directors of Seahawk Parent determines in good faith, after consultation with Seahawk Parent's outside legal counsel, that in light of such Superior Proposal such action is required in order for Seahawk Parent's Board of Directors to comply with its fiduciary obligations under applicable Law, provided that Sellers have otherwise strictly complied with all of their obligations in this Section Such agreement to pay the Termination Fee and Purchaser's Reimbursable Expenses is, in part, based on Sellers' recognizing Purchasers' expenditure of time, energy, and resources in connection with this Agreement. The payment of the Termination Fee and Expense Reimbursement shall be governed by the provisions of this Agreement and any related Bankruptcy Court Orders. Except as may be negotiated as part of a Superior Proposal, under no circumstances will a break-up fee, expense reimbursement or other similar bid protections be provided by Sellers to any potential bidder or bidders for any portion of the Business or the Purchased Assets, other than Purchasers.

APA at § 6.5 (d) and (e) (emphasis added).

22. Section 6.3 of the APA requires that, on the Petition Date, the Debtors file a Termination Fee Motion, which is defined in the APA as "the motion to be filed by Sellers pursuant to Section 6.3 seeking to approve the payment of the Termination Fee and Reimbursable Expenses to Purchasers as an administrative expense of the Bankruptcy Cases." Accordingly, by this Motion, the Debtors request the entry of an order approving the payment of the Termination Fee to the Purchasers, if triggered under the APA, as an administrative expense of the Debtors' estates.

APPLICABLE AUTHORITY

- 23. Courts recognize that termination fees are a valuable means of avoiding costs associated with proving damages. See Global Octanes Tex., L.P. v. BP Exploration & Oil Inc., 154 F.3d 518, 521 (5th Cir. 1998) (contracting parties have broad authority to "limit their liability in damages to a specified amount."); Vallance & Co. v. DeAnda, 595 S.W.2d 587 590 (Tex. Civ. App. San Antonio 1980, no writ) ("In the absence of a controlling public policy to the contrary, contracting parties can limit their liability in damages to a specified amount.").
- Accordingly, courts within and outside the Fifth Circuit have approved termination and break-up fees where the proposed fees are consistent with market practices, good faith, and an honest judgment under the "business judgment rule." *In re Redwine Res., Inc.*, No. 10-34041, 2010 WL 5209287 (Bankr. N.D. Tex. Jun. 24, 2010) (approving termination fee equal to 2.5% of the purchase price); *In re Galveston Bay Biodiesel, LP*, No. 10-80278, 2010 WL 5186603, at *1 (Bankr. S.D. Tex. Sept. 10, 2010) (approving termination fee and expenses); *see also Consumer News & Bus. Channel P'ship v. Fin. News Network, Inc. (In re Fin. News Network, Inc.)*, 980 F.2d 165, 167 (2d Cir. 1992) (noting that an \$8.2 million break-up fee had been awarded on a \$149.3 million transaction (5.5%) without discussion); *LTV Aerospace & Defense Co. v. Thomson-CSF*,

- S.A. (In re Chateugay Corp.), 198 B.R. 848, 861 (S.D.N.Y. 1996) (enforcing a \$20 million "reverse breakup fee" payable on a \$450 million offer (4.4%)); In re Datavon, Inc., Case No. 02-38600-SAF-11 (Bankr. N.D. Tex. 2002) [Doc. No. 224] (approving a 3% fee); In re Mirant Corp., Case No. 03-46590 (DML) (Bankr. N.D. Tex. 2004) [Doc. No. 6092] (approving a 3% fee).
- 25. The negotiated Termination Fee payable to the Purchasers under the APA is reasonable and consistent with the Debtors' business judgment. The Purchasers have represented that that the proposed Termination Fee fairly represents the damages the Purchasers will incur upon any breach by the Debtors, and it is within the range of termination and break-up fees approved by courts in other cases, was negotiated in good faith and at arms' length, and will enable the Debtors to secure the Purchasers' cooperation in completing the Transactions.
- 26. The Debtors request that the Court approve the payment of the Termination Fee to Purchasers, if triggered, as an administrative expense under section 503(b) of the Bankruptcy Code. Courts consistently recognize that termination, break-up, and similar fees as necessary costs of an estate entitled to administrative expense status. See, e.g., In re Tridimension Energy, L.P., et al., No. 10-33565, 2010 WL 5209233, at *5 (Bankr. N.D. Tex. Oct. 29, 2010) (approving break-up fee as a section 503(b) administrative expense); In re Lincolnshire Campus, LLC, No. 10-34176, 2010 WL 5269706, at *1 (Bankr. N.D. Tex. July 23, 2010) (approving break-up fee and proposed purchaser's expenses as a section 503(b) administrative expense); In re Redwine Res., Inc., 2010 WL 5209287 at *5 (approving a termination fee as a section 503(b) administrative expense); In re Saint Vincents Catholic Med. Ctrs. of N.Y., No. 10-11963, 2010 WL 3638139, at *2 (Bankr. S.D.N.Y. Aug. 20, 2010) (approving a break-up fee as a section 503(b) administrative expense); In re Ray Realty Fulton, Inc., No. 09-41225, 2009 WL 2600760, at *1 (Bankr. E.D.N.Y. Aug. 21, 2009) (approving a break-up fee as a section 503(b) administrative expense); In re

Chrysler LLC, No. 09-50002, 2009 WL 1360869, at *8 (Bankr. S.D.N.Y. May 7, 2009) (approving a break-up fee as a section 503(b) administrative expense); In re Steve & Barry's Manhattan LLC, No. 08-12579, 2008 WL 8168312, at *2 (Bankr. S.D.N.Y. Aug. 5, 2008) (approving a termination fee and expenses of a stalking horse bidder as section 503(b) administrative expenses).

NOTICE

27. Notice of this Motion has been provided by overnight delivery and e-mail or facsimile to: (a) Natixis, New York Branch, as Administrative Agent for the Lenders; (b) the 30 largest unsecured creditors of the Debtors on a consolidated basis; (c) Pride International, Inc.; (d) Hercules Offshore, Inc.; (e) D.E. Shaw Direct Capital Portfolios, L.L.C.; (f) the U.S. Trustee's Office; (g) the United States Attorney's Office; (h) the Securities and Exchange Commission; and (i) the Internal Revenue Service. The Debtors believe that the notice provided for herein is fair and adequate and no other or further notice is necessary.

CONCLUSION

WHEREFORE, PREMISES CONSIDERED, the Debtors respectfully request the entry of an order approving the payment of the Termination Fee and Purchasers' Reimbursable Expenses, if triggered under the APA, as an administrative expense of the Debtors' estates, and for all such other and further relief as may be just and proper, either at law or in equity.

Dated: February 11, 2011.

Respectfully submitted,

FULBRIGHT & JAWORSKI L.L.P.

By: /s/ Berry D. Spears

Berry D. Spears
State Bar No. 18893300
Johnathan C. Bolton
State Bar No. 24025260
1301 McKinney, Suite 5100
Houston, Texas 77010-3095
Telephone: (713) 651-5151
Facsimile: (713) 651-5246
bspears@fulbright.com
jbolton@fulbright.com

and

JORDAN, HYDEN, WOMBLE, CULBRETH & HOLZER P.C.

Shelby A. Jordan
State Bar No. 11016700
Nathanial Peter Holzer
State Bar No. 00793971
500 N. Shoreline Blvd., Suite 900
Corpus Christi, Texas 78401-0341
Telephone: (361) 653-6624
Facsimile: (361) 888-5555
sjordan@jhwclaw.com
pholzer@jhwclaw.com

PROPOSED ATTORNEYS FOR THE DEBTORS AND DEBTORS-IN-POSSESSION

Form Of Transfer Order To Be Provided To The Bankruptcy Court

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	Chapter 11
Seahawk Drilling, Inc., §	Case No.: 11
Seahawk Drilling LLC, §	Case No.: 11
Energy Supply International LLC, §	Case No.: 11
Seahawk Global Holdings LLC, §	Case No.: 11
Seahawk Mexico Holdings LLC, §	Case No.: 11
Seahawk Drilling Management LLC, §	Case No.: 11
Seahawk Offshore Management LLC, §	Case No.: 11
Seahawk Drilling USA LLC, §	Case No.: 11
Debtors.	Motion for Joint Administration Pending

ORDER APPROVING THE PAYMENT OF CONTINGENT TERMINATION FEE PAYABLE TO HERCULES OFFSHORE, INC. AS AN ADMINISTRATIVE EXPENSE OF THE DEBTORS' ESTATES IN ACCORDANCE WITH THE TERMS OF THE ASSET PURCHASE AGREEMENT

Upon consideration of the Debtors' Emergency Motion for Order Approving the Payment of Contingent Termination Fee Payable to Hercules Offshore, Inc. as an Administrative Expense of the Debtors' Estates in Accordance with the terms of the Asset Purchase Agreement (the "Motion"), and after finding that: (i) the Court has jurisdiction over the Motion pursuant to 28 U.S.C. §§ 157 and 1334, and this matter is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(A); (ii) due and adequate notice of the Motion has been provided under the circumstances and no further notice is required; (iii) good and sufficient business reasons exist for approving the Termination Fee payable to the Purchasers as an administrative expense, as contemplated in the APA; and (iv) the Termination Fees are reasonable and appropriate and in the

best interests of the Debtors, their creditors, their estates and other parties-in-interest. Therefore, it is

ORDERED that, the relief requested in the Motion is granted; and it is further

ORDERED that, to the extent that the APA is terminated by the Debtors pursuant to section 11.1(c)(ii) thereof, the Debtors shall pay the Termination Fee and the Purchasers' Reimbursable Expenses as an administrative expense of the Debtors to Hercules within five (5) days after giving notice of termination; and it is further

ORDERED that, the Debtors' payment of the Termination Fee and Purchasers' Reimbursable Expenses pursuant to Section 11.2(d) of the APA shall be the sole and exclusive remedy of the Purchasers with respect to the occurrences giving rise to such payment; and it is further

ORDERED that the Court shall retain jurisdiction over any matter or dispute arising from or relating to the implementation of this Order.

SIGNED this the	day of	, 2011.
		INITED STATES BANKRIPTCY HIDGE

EXHIBIT B

The Assigned Contracts

Schedule 2.1(b) to Asset Purchase Agreement

Seatured Drilling LLC	ARENA OFFSHORE	REALWANT START DATE: HORSE MESOCURED THE LAND ACRESIANCE OTHER TRAFF DATE: HORSE HORSE THE CONTINUE OF THE PROPERTY OF THE PRO
Seatment Driffing LLC	ANEMA OFFSHORE	BEAUMIN SID PRISE TEMA) DRILING SERVICES AGREEMENT CONTINUE EVANT DATE TEMAS
Seatent Oriting LLC	PADGER OIL	REMAINS AND PROCE TOTAL DELLING SERVICES ACREDMENT CONTRACT DIATE CHESCAPO
Seatment Orilling LLC	BRISTOW US LLC	CONTRACTOR IN TO PROVIDE ARCHAITS, EQUIPMENT AND SERVICES TO THE COMPANY CONTRACT BTART DATE SERVICED.
Seatment Drilling LLC	CASTEX OFFSHORE, INC.	SEAWANN 2807 PRINCE REW MEDOCO) CONTINUES BEAN DATE: 2002200
Seatment Drilling LLC	CHEVRON U.S.A.	BENVINK ZON (PRIZE KANNAS) CONTINUT START DATE-I MIZERIO
Seatheask Drilling LLC	ENERGY XX	BENNAMN 2010 PROCE TEXAS) DRILLING SERVICES AGREEMENT CONTINCT START DATE STREAGN
Seatment Drilling LLC	Na.	BEALWAY ZON PROZE ARKANBAN) DRILLING SERVICES AGREEMENT CONTINCT EINST DATE (172220)
Seatment Drilling LLC	ESSAR OLFIELD SERVICES INDIA	CONTINUENT BALLE OF 2002 CONTINUENT START DATE BEACHAONS
Seatment Drilling LLC	ESSAR ONLFIELD SERVICES INDIA LTD AND BNY MELLON, N.A.	BOCHOW ASSESSMENT FOR BALL OF 2005 CONTRACT START DATE MANAGER
Seatment Drilling LLC	JOMES, ROSE MARY W.	WEST MURENT A MULENT PATURES CONTRACT START DATE SCHOOLS
Seatment Drilling LLC	KEPPEL AMPELS, INC.	VERREL REPART & MICH GREENERT - SEAHAWK 2000 REPARTS AT YARD COMTINACY START DATE STREEZEN
Seatheark Drilling LLC	KEPPEL AMPELS, INC.	VERREL REMAIN & MICO ADMEDIENT - SEAMANN 2000 REPARTS CHROAND COMTRACT START DATE: 1482-200
Seatment Drilling LLC	NABORS DRILLING INTERNATIONAL LIMITED	SALE OF CHIENTING WITH LOCATED IN TEXAS COMPINED START DATE SEMEZON
Seatment Drilling LLC	MATIXOS	MONTHAME FOR 2001,
Seahawk Drilling LLC	MATIXIS	MORTIONAIR FOR 2864 AND 2862 CONTINGE START DATE ABROADES
Sushawk Drilling LLC	MATDUS	JONIQEN AGREEMENT CONTRACT START DATE SINCACION
Summer Driffing LLC	NATDUS	AASIGNAENT OF INSURANCE CONTRACT START DATE SINCE 2009
Seahawk Drilling LLC	NATDUS	SSAMAMY 2001 VERBEL, CREATES SÉCURITY INTEREST ONLY -ASSIGNOR STILL TO PERFORM ALL OBLICATIONS UNDER THE CONTRACT CONTRACT START DATE: INDIRECTOR
Seahuank Drilling LLC	NATDUS	BEANAWK 28DI VERBEL CREATES SECURITY INTEREST ONLY -ASSIGNOR STILL TO PERFORM ALL OBLICATIONS UNDER THE CONTRACT CONTRACT START DATE STRUCTORS
Seathawk Drilling LLC	NATIOS	SSAWING 2808 WESSEL, CREATES SÉCURITY INTEREST ONLY -ASSIGNOR STILL TO PERFORM ALL OBLICATIONS UNDER THE CONTRACT CONTRACT START DATE SIZEDIZIOS
Seathwelt Drilling LLC	MATDUS	BEANAWR 2008 VESSER, CREATES SECURITY INTEREST ONLY -ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS LINDER THE CONTRACT CONTRACT START DATE SECURITY INTEREST ONLY -ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS LINDER THE CONTRACT
Seathwelt Drilling LLC	MATDUS	SEMMENCE START DATE SECURITY MITEREST ONLY -ASSIGNOR STILL TO PERFORM ALL OBLIGATIONS UNDER THE CONTRACT CONTRACT
Seahawk Driffing LLC	OILFIELD EQUIPMENT SALES, INC.	PURCHAIR AGREGAENT FOR EQUIPMENT BILL CONTRACT FOR EQUIPMENT CONTRACT FIRST DATE: 23212011
Seehawk Drilling LLC	PEMINSULA DRILLING LLC	TRANSPER OF BEAVIOR FROM PRIDE LOUISIANA TO SELVAMY DRILLING LLC COMPLACE START ON TEXTORISM
Seahant Drilling LLC	SEAMANK DRILLING DE MEXICO LLC	TRANSPER OF BEANIAMS, 2001 FROM PRIDE ARKANISAS TO SEAMANN ORILLING LLC COMPINAT BYATT DATE: HOZBODOB
Seatheast Drilling LLC	TARPON OFFSHORE VENTURES	SEAWAY 2000 (PRICE ALARIA) CONTRACT START DATE-1/22/2010
Seathawk Drilling LLC	WALTER OIL AND GAS	SEAWANT 2014 (PRICE RESERBATP) CONTINUE STATE LESENZOIO
Seathandt Drilling, Inc.	A & L REPART SERVICES, LLC	SERVICE CONTINCT WATER REPOCE AGRESSIENT
Seatured Drilling, Inc.	ACCOUNTEMPS	TEMPONARY BIRN COMBENCY AGREEMENT AND CONDITIONS OF ASSIGNMENT CONTRACT START DATE: MANISON
Southeask Drilling, Inc.	ACE TRANSPORTATION, LLC	TRAMBROFITATION SERVICES CONTRACT START DATE-SERVICES
Seathank Driffing, Inc.	ACEVEDA, FELICIA	CONTRACT BAPLONIENT AGREEMENT CONTRACT START DATE-INITIZEN
Sanhawk Drilling, Inc.	ACIME TRUCK LINE, INC	SERINCE CONTRACT MASTER BERNCE AGRESSERIT
Seathawk Drilling, Inc.	ACME TRUCK LINE, INC.	TRUCKING SERVICES AGREEMENT AND RELATED AMENDMENT CONTRACT START DATE SERVICED
Seature Drilling, Inc.	AFP ALABA & DETECTION	FINE SHIPESSHOW SYSTEM ALPHARIES PROPERTY OF THE PROPERTY OF
Seahawit Drilling, tho.	AGGREKO, INC.	SERVICE CONTINUE TO THE LANDILLINE SERVICE TO THE LANDILLINE SERVICE TO THE LANDILLINE SERVICE SERVICE TO THE LANDILLINE SERVICE S
Seatment Drilling, Inc.	ALLIS-CHALMERS ENERGY, INC.	PROTECTS INFORMATION PROVIDED FOR FURFICIES OF EVALUATING SEALMANYS MEXICO CEMENTING DIVISION FOR A POTENTIAL TRANSACTION CONTRACT STANT DATE OF STANDARD
Seahank Drilling, Inc.	ALVAREZ & MARSAL BUSINESS CONSULTING, LLC	SERVES TRANSTIONING IT DUTIES FROM PRIDE TO SEAWAWK CONTRACT START DATE-DIZZOXXX
Senhawk Drilling, Inc.	AMERICAN EXPRESS TRAVEL RELATED SERVICES CO., INC.	ENGAGEMENT OF AND TO PROVIDE THE COMPANY WITH TRAVEL AND RELATED SERVICES CONTRACT START DATE: D
Seatment Drilling, Inc.	AMERICAN POLLUTION CONTROL	COMPRACE START DATE ST
Seatherek Drilling, Inc.	ANGEL AIR REPAIR & SPECIALTY COMPANY	PREJMATIC REPARTS SALES AND SERVICE AND RELATED AMENDMENTS COMPINED START DATE SMOTIZED
Seahawk Drilling, Inc.	AP MOLLER	RECIPIOSALI VICEMENTY AGRESSIENT CONTRACT START DATE-1/22/2001
	APA FABRICATION, INC.	SERVACE CONTINACT MASTER SERVACE AGRESSERIT
		בייני בליני על הייני בייני

		The event of the state of the s
Seatment Driffing, Inc.	SHORE SERVICES, LLC	non-date on Energy Services Contracts of Institutional
Seatleant Diffing, Inc.	PARTMERS	A ARREMENT OF EACH MOST. IN THE WITED CAPITOL INSURANCE CO. CASE. ARREMENT AT THE WITED TO CALAIM THE WITED CAPITOL INSURANCE CO. CASE.
Seatheask Drilling, Inc.		CONTINCT START DATE OF DESCRIPTION
Seahent Orlling, Inc.	AT&T	WWILDERS SERVICES SERVICES
Seatment Drilling, Inc.	AXON ENERGY PRODUCTS AS	PAMENT OF CHANCH SERVICES 4 DRELAGE CONTINUES COPPERTIFICATION CONTINUES C
Seahout Driffing, Inc.	BANER ENERGY	PROMEDIC OF CHROMAL SUPPORT FOR MAXIMO CONTRACT STATE DATE CHROMAN
Seahewk Drilling, Inc.	BANCERANO SERVICES, INC.	IMANIO SUPPORT SERVICES APREDIENT CONTINUE SUPPORT SERVICES
Seahaak Drilling, Inc.	BASSOE OFFSHORE, INC.	RANSOL PROPRIED SERVICE SERVICE THE SCHUMMER RID RIG CHARLET STATE THE SERVICE THE SCHUMMER RID RIG THE SERVICE SERVICE SERVICE THE SCHUMMER RID RIGHT
Seatent Orlling, Inc.	BASTION TECHNOLOGIES, INC.	ENGREERING DESIGN A ANLYSIS CONTRACT STATE TATE ON YOUR
Seateset Drilling, Inc.	BAYOU BLACK ELECTRIC SUPPLY, LLC	SERVICE CONTRACT MATTRI SERVICE GREENIT
Seahawk Drilling, Inc.	BEACON MARITME, INC.	FARICATION & WELDING COMMACT STAFF TATE SEXZORS
Seatment Drilling, Inc.	BEACON MARITIME, INC.	SERVICE CONTRACT IMATTRI SERVICE AMERICAN
Seatherek Drilling, Inc.	BECERRA, CHRIS	CONTRACT BANDOMENT AGREBAENT CONTRACT START DATE SON 4200
Seahawk Orlifing, Inc.	BLAKE DRILLING & WORKOVER COMPANY INC.	NOGRATY ACREDMENTS INJULIA WORKEN WAYER OF RECOURSE CONTRACT STATE DATES SHAZOR
Seatewk Drilling, Inc.	BLUE CROSS AND BLUE SHIELD OF TEXAS	BOSS TO ACT AS CLAMIS ADMINISTRATOR IN PROVIDING ADMINISTRATIVE SERVICES, INCLUDING THE DISCLOSLINE OF PERSONAL HEALTH INFORMATION, ALSO SETS FORTH TERMS FOR THE COMPLIANCE OF BOTH PARTIES TO THE HEALTH INSURANCE PORTMANT AND ACCOUNTABILITY AND ACCOUNTABILITY AND ACCOUNTABILITY.
Seatheast Drilling, Inc.	BLUEWATER RUBBER & GASKET CO.	SEMOC CONTRACT MATTR SEMAC AGREGATI
Seahand Oriting, Inc.	BOARDVANTAGE	PURCHASE ARRESTED SERVICES CONTRACT STATE SERVICES
Seathank Drilling, Inc.	BOHEN CRANE AND EQUIPMENT REPAIR, LLC	CONTRACT OF THE CONTRACT OF STORAGE BROKER TO SEAHAWK FOR A POSSIBLE SALE OF TWO CRAMES CONTRACT STATS DESCRIPTIONS
Seatweek Driffing, Inc.	BOLLINGER FOURCHON, LLC	SHEYAO REPARS CONTACT THE DATE OF THE CONTACT OF TH
Seatured Drilling, Inc.	BRACEWELL & GIULANI	LEGAL SERVICES COMPACT STATE DATE DATE DATE DATE DATE DATE DATE
Seehheat Drilling, Inc.	BRACEWELL & GIULANI	TRANSPORTER LE TOTALE CENTRAL MANAGEMENT
Seatent Drilling, Inc.	ROUP	CONTINUE IN THE LEGISLATION ENGINEER SERVICES FEEL ANALOGINEER TON EMPLOYEE PLACEMENT SERVICES
Seatment Oriting, Inc.		NOTIFICATION OF THE PROPERTY O
Seatheast Drilling, Inc.		CONTINUE STATE LOS ESTATES OF THE CONTINUE OF THE CONTINUE SERVICES
Seatment Drilling, Inc.		CONTINUE SIME LATEUREZON AGRIT CONTINUE.
	i evertase	CONTINUE TRATE INTERPRIED BERNAE CONTINUE T
	2	BEAUTE SERVICE LAFEDIENT BEAUTE CHURCHE CANTENDENT BEAUTE CHURCHE CHURCHE CHURCHE BEAUTE CHURCHE CHURCHE CHURCHE CHURCHE BEAUTE CHURCHE CHUR
Seatment Drilling, Inc.	-	MASTRA REPORTS AUTREMENT
Seatment Drilling, Inc.	CAPOENMN U.S., LLC	MATTER SERVICE, MEDICEMENT CONTINUES STATE CARTESTINGSON
Seatment Drilling, Inc.	CAPGERRINI U.S., LLC	SERVER HAST CONTRACT STATE DATE CONTRACTOR
Seahand Drilling, Inc.	CAPGEMM# U.S., LLC	a de blefort Contract statt date essenzio
Seatherek Drilling, Inc.	CAPGEMBR U.S., LLC	SAP RETALLATION & TRANSMIC CONTRACT START DATE, CONTRACT START DATE, CONTRACT START DATE, CONTRACT
Seatment Drilling, Inc.	CHAMBERLAIN HRDLICKA	LEGAL SERVICES LOCATIVACE STATE DATE DATE DATE DATE DATE DATE DATE
Seaheask Drilling, Inc.	CHARTER SUPPLY CO.	SERVICE CONTRACT MATTER SERVICE AGREEMENT
Seahawk Drilling, Inc.	CHUBB GROUP OF INSURANCE COMPANIES	aurent pows Contract synth date de 17200
Seahment Drilling, Inc.	CHURCH ENERGY SERVICES LTD.	SERVICE CONTRACT MATTRI SERVICE AGREDIENT
Seaturek Drilling, Inc.	CMTAS FIRST AID & SAFETY	SEMAC CONTRACT MATTRI SERVICE CAREMENT
Seatenet Drilling, Inc.	CLEANBLAST I & II, LLC	SANGELSTING PARTIES PA
Seahaak Dilling Inc	COGENT COMMUNICATIONS INC.	TECOMALMACINES RAREBERT
Seatment Drilling, Inc.		COMPACE STATION EXPRIZED TO BUSINESS CLASS TELEVISION
Seahart Dilling Inc	ON SERVICES INC	CHARLA BINTELLA BINDER
September 1		MATER SERVICE AGREEMENT MATER SERVICE AGREEMENT
Comment Crimet, No.		CONTINCT START DATE CONTECTION AND ELECTRONIC ASSET WANDERFORT SERVICES TO SEALAWK
Semment Literation Pro.		CONTRACT STATE DAY A CREATER TO THE CONTRACT STATE OF THE CONTRACT STATE S
Seathank Drilling, Inc.		CONTRACT STAT DATE DESIGNO SERVICE CONTRACT
Seahawk Drilling, Inc.	COPPER STATE RUBBER - SPECIALTIES CO.	WASTER SERVICE AGREEMENT

Seaturek Orilling, Inc.		CONTRACT FACT DATE 199/200
Seatment Drilling, Inc.	CROSEY TUGS, LLC	CONTINUE THAT DATE OF THE CONTINUE THE CONTINUE THAT DATE OF THE CONTINUE THAT DATE OF THE CONTINUE THE CONTINUE THE CONTINUE THAT DATE OF THE CONTINUE THE CONTI
Seatment Ording, Inc.	DAHILL (XEROX)	THE
Seathmek Orilling, Inc.	DELL, INC.	AND REPORTED THE WORKEN OF TAKEN OF SECULATION OF USED IN CONNECTION WITH ITS BUSINESS RELATIONSHIP WITH DELL WHICH INCLIDES DESIGN. TESTING, MANUFACTURING, SALE AND SUPPORT OF DELL PRODUCTS.
Seatment Drilling, Inc.	рецопте тах, ще	TAX ADVISIONY BETWICES CONTINUES BY THE DATE SHIFT DATE SHIFT AZODS
Seatuant Drilling, Inc.	DETECTION & SUPPRESSION INTERNATIONAL, LTD	REPORTED AND FIRE CHESTAL M 50 PT FM ZDD SYSTEM . LEVEL ZB FME SUPPRESSION SYSTEM TWO TIMES PER ACREEMENT PERMOD CONTINACT SKINT DATS SHOULD SHOW THE SUPPRESSION SYSTEM TWO TIMES PER ACREEMENT PERMOD
Seatment Drilling, Inc.	DISCOVERY BENEFITS, INC.	DIBICONERT BENEFITS TO ASSIST IN THE ESTABLISHMENT AND OPERATION OF SEAWAWK'S CAFETERIA PLAN, HEALTH FSA AND DEPENDENT CARE FSA. CONTRACT START DATE:OFOLIZION
Seature Oriting, Inc.	DOLPHIN SCAFFOLD SERVICES, LLC	SERVACE CONTRACT CONTRACT RESERVACE CONTRACT
Seatured Drilling, inc.	DOLPHIN SERVICES, LLC	WANTED THAT STATE ARE CALTACT. BETWEEN THE SAFETY CALTACT.
Seahawk Drilling, Inc.	DOLPHIN TOMING	MATTER THE CHATTER ACREDIENT CONTRACT START DATE OF CONTRACT
Seatment Orilling, Inc.	DON ABNEY, INC.	SERVAZ CONTINACT MATTER RETAVIZ AOREBIENT
Seahaeek Ortiling, Inc.	ن	BERVAZ CONTINCT MASTER REPAGE AZREBIENT
Sentensk Driffing, trc. Sentensk Driffing, trc.	EDISON, MCDOWELL & HETHERINGTON ILP EMDSI - HUAT FKA HUNT ENGINE, INC.	BERNZE KAREBIENT LEGAL BERNZE ACREBIENT
Seathank Drilling, Inc.		THE STATE OF WATER THE STATE OF
Sestnavit Driffing, Inc.	Z.A.	PONTAGE OF THE PROPERTY OF THE
Seatewk Orilling, Inc.	VETY AND HEALTH	CONTINUE TENNIC ACTION. MARTINE EXPRISE CONTINUE. MARTINE RESEARCH.
Seatment Drilling, Inc.	ERNST & YOUNG, ILP	CONSULTING SERVICES AGREEMY CONSULTING SERVICES AGREEMY
Seaheask Drilling, Inc.	ESSAR OLFIELD SERVICES INDIA LTD	TCA. CONTRACT START DATE CONTRACTOR
Seahwat Drilling, Inc.	FIANZAS MONTERREY, S.A.	GUARANTEE OF OBLIGATIONS AS TO THE BONDS ISSUED CONTRACT START DATE ON YOUR START OF THE BONDS ISSUED
Seahank Driffing, Inc.	FIRE AND SAFETY SPECIALISTS, INC.	BENVE CONTINCT MATER REMACE AGREEMENT
Seahawk Drilling, Inc.	FOSTER QUAN LLP	LEGAL BÉRNCES CONTRACT START DATE SYMBZOHO
Seahent Drilling, Inc.	FULBRIGHT & JAWORSIQ, LLP	LEGAL BETANCES CONTRACT START DATE:12242008
Sentent Offing, Inc.	GALLOWAY JOHNSON TOMPKINS BURR & SMITH	CONTRACT STATE DE L'AZZONO
Seatheask Drilling, Inc.	GARDERE WYNNE SEWELL, LLP	LEGAL SÉRNCES CONTRACT START DATE DIVISOR
Seatured Drilling, Inc.	GARDERE, ARENA Y ASOCIADOS, S.C.	COMMUNICATION TO THE PROPERTY OF THE PROPERTY
Seatment Drilling, Inc.	GARZA TELLO Y ASOCIADOS, S.C.	LEGAL BETACES CONTRACT START DATE SUMBONO
Sonhundt Orlling, Inc.	GEC CAPITAL GROUP, LP	PELATED TO USE AND DISCLOSURE OF EVALUATION MATERIAL RECEIVED BY GEC CAPITAL FOR PURPOSES OF EVALUATING A PROPIOSED INVESTMENT IN THE COMPANY CONTINGES START DATE CENICATION
Seatured Drilling, Inc.	GENERAL MARINE LEASING, LLC	BERNCE CONTRACT MATTER REPUCE ACREBIENT
Sentransk Drilling, Inc.	GENERAL SERVICES, CO.	MATERS SERVICE CONTROL OF THE SERVICE
Seatharsk Drilling, Inc.	GEOPHYSICAL DATA MANAGEMENT, INC.	SUMPE WORK CONTRACT START DATE SPIN 2000
Seathands Orithing, Inc.	GENMAN, OSCAR A.	BINIONIBIN AGREDARY CONTRACT START DATE: 1/0/2008
Sestimet Drilling, Inc.	GENMAN, OSCAR A.	SEPARATION AGREEMENT CONTRACT START DATE DIZECTOR
Seahenk Driffing, Inc.	GLOBAL CATHODIC PROTECTION, INC.	BERNCE CONTINCT MASTER BERNCE AGREEMENT
Seehank Drilling, Inc.	GLOBAL COMPLIANCE SERVICES, INC.	BRIGHTJAR - ÓNLIKE AND IN-PERSON TRAINING CONTRACT START DATE GENZZOOD
Seatheask Ording, the.	GLOBAL COMPLANCE SERVICES, INC.	GCS TO PROVIDE SEAVANK AN ETHICS AND COMPLIANCE HOTLINE, AS WELL AS A WEB-BASED CASE MANAGEMENT PROGRAM. CONTRACT START DATE SENIOZOGO
Seatheask Drilling, Inc.	GOWAN, INC.	CONTRACT STATE OFFICE ACTIONS OF THE STATE OFFI OFFI OFFI OFFI OFFI OFFI OFFI OFF
Seathands Orllling, Inc.	GRUPO HEGEMONIA SA DE CV	BERNGE CONTINCT MATTER BERNGE AGREBBENT
Seathwelt Drilling, Inc.	GULF COAST INTERNATIONAL, LLC	BARAKE REPARR SOR TESTAR A WANTIFACTURE CONTRACT STAR THE SUBSCISCOR
Seathank Driffing, Inc.	GULF COAST MARINE & ASSOCIATES, INC.	BENACES TO INCLUDE PERFORMANCE OF RIG MOVEMENT, RIG REPAR, SURVEY AND PROJECT MANAGEMENT AS REQUESTED FROM THE TO TIME CONTRACT START DATE:SMIZIZOD
Sanhawk Drilling, Inc.	QULF COPPER & MANUFACTURING CO.	BERNCE CONTRACT MASTER BERNCE AGREDIENT
Seehmok Driffing, Inc.	HANNON HYDRAULICS	SERVICE CONTINACT MASTER SERVICE AGREDIENT
Seathank Drilling, Inc.	HB RENTALS, LC	SERICE CONTRACT MASTER SERIVEE AGREDIENT
Seathansk Drilling, Inc.	HILER OFFSHORE SERVICES, INC.	WANTEWING BERINGS AGREEMENT CONTRACT START DATE OFFICEDOR
Seahawk Drilling, Inc.	HITECH INTEGRATED SOLUTIONS, INC.	FINE AMETE CLUMMENT COMTRACT START DATE CONTRACT

		SERVICE CONTRACT
	TLT, INC.	AND AND REPORT OF THE PROPERTY
Sources Uterry, etc.	LY, INC.	MATER RETICE, AGRESSENT
Seatheast Orliffing, Inc.	HARE SUPPLIES (PTY) LTD.	E CONTRACT IT DATE: 11/160
Seahank Driffing, Inc.	MON FINANCIAL BERVICES	CONTRACT PRINTER AND OTHER BETWEEN PROVIDED CONTRACT DATE OF THE PROVIDED CONTRACT THAT DATE IS CONTRACT TO THE CONTRACT THE PROVIDED CONTRACT TO THE CONTRACT TH
Seatremet Drilling, Inc.	WOM FINANCIAL SERVICES	10 6
Seathank Drilling, Inc.	INFOSTAT SYSTEMS, INC.	SEANAWK WAS GRANTED A LICENSE TO USE THE RAIDSHIL SOFTWARE PROGRAM.
Seathersk Drilling, Inc.	INTEGRICERT LLC	12
Seathank Drilling, Inc.	INTERCALL, INC.	AUDIC COMPENSANG BETWEES CONTRACT STANT DATE SHOULDS
Seatrewit Drilling, Inc.	MTERNATIONAL SATELLITE SERVICES, INC.	SATELLIE TELEMONE BENYCE CONTRACT START DATE SBEGGZD10
Seehawk Drilling, Inc.	JORDAN, HYDEN, WOMBLE CULBRETH & HOLZER	ENGAGENENT LETTER FEATUREMENT LETTER FOUTERING TO THE CHARGES AND THE CHARGES
Seahawk Drilling, Inc.	KAHANEK ADVISORY SERVICES	PROMICIAL LONG THE CHILD THE CONTROL OF THE CHILD THE CH
Seahawk Driffing, Inc.	KEAN MELLER HAWTHORNE D'ARMOND MCCOWAN & JARMAN LLP	·I +
Seehawk Drilling, Inc.	KEANE, INC.	استا⊦
Seahmak Drilling, Inc.	KEANE, INC.	FINANCIA, ADVACIVY SERVICES ACREBIENT CONTRACT START DATE: 1182/2010
Seminant Drilling, Inc.	KEM SUPPLY HOUSE, INC	SERVICE CONTRACT MATTER SERVICE AGRESSMENT
Seatheret Driffing, Inc.	KPMG	AUDIT SERVICES ACRESSENT CONTRACT START DATESTIZACES
Seahmork Driffing, Inc.	KUEHNE + NAGEL, INC.	CONTINUE START DATABASEMENT CONTINUE START DATABASEMENT
Seatrawk Drilling, Inc.	HURTZMAN CARSON CONSULTANTS LLC	SERVICE AMERIARY FOR NOTICING AND SOLICITATION PURPOSES CONTRACT START DATE: 1274-25010
Seatherek Drilling, Inc.	LAL RIG SERVICES, INC.	SERVACE CONTRACT IMASTER SERVICE AGRESMENT
Seatheask Drilling, Inc.	LABORDE AND NEUNER	LEGAL SERVICES CONTINCT START DATES/(2022)0
Seatment Drilling, Inc.	LAFAYETTE STEEL ERECTOR, INC	SERVICE CONTRACT MATTER SERVICE AGREEMENT
Seathmet Orlling, Inc.	LOADMASTER DERRICK & EQUIPMENT, INC	HATTER REPORTED.T.
Seatlank Drilling, Inc.	LOCKE LOND BISSELL & LIDDELL, LLP	AGREEMENT T DATE ORZAG
Seatheask Drilling, Inc.	LOCKE LORD BISSELL & LIDDELL, LLP	LINES IN C
Seatment Orilling, Inc.	LOF INTERNATIONAL HUMAN RESOURCES SOLUTIONS, INC.	HUMAN RESOUNCES BETWEES AGREEMENT CONTINCT STAFT DATESSANGEIG
Seatment Drilling, tnc.	LONGNECKER & ASSOCIATES	PROVIDES DETALS OF ENCAGELENT CONTINCT START DATE-BAZINGS
Seatlank Drilling, Inc.	LOUISMANA ELECTRIC RIG SERVICE, INC.	SERVICE CONTRACT MATTER SERVICE ADREDIENT
Seatland, Orlling, Inc.	LOUISIANA ENVIRONMENTAL MONITORING INC	SERVICE CONTRACT IMATTR BETWEE ARRESHENT
Seathersk Driffing, Inc.	LOUISIANA INTERNATIONAL MARINE, LLC	RIG TOWING BETWICES CONTINCT START DATE: 8629200
Seathannic Drilling, Inc.	LOUISIANA MACHINERY	SERVICE CONTINUCT MASTER SERVICE AGREEMENT
Sembsouk Drilling, Inc.	MAD LTD.	SERWCE CONTRACT MASTER SERVICE AGRESHENT
Seatment Drilling, brc.	MAGNUM CONSTRUCTION SERVICES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seathawk Drilling, Inc.	MANIFOLD VALVE SERVICES, INC.	SERVICE CONTRACT MASTER SERVICE AGREEMENT
Seatment Drilling, Inc.	MANZ, STEVEN A.	RELEASE AGREEMENT CONTRACT START DATE CHUTZD10
Seatment Drilling, Inc.	MARINE SYSTEMS, INC.	DIESEL BROINE REPARE CONTRACT START DATE: I (0)2009
Seatherek Drilling, Inc.	MARITIME INDUSTRIAL SERVICES CO. LIMITED	SECRECY AGREEMENT CONTRACT START DATE 992-2009
Seaharek Drilling, Inc.	MARTIN, DISHERE, JEFFERSON & WISDOM LLP	LEGAL SERVICES CONTRACT START DATECHINZOLO
Seatwark Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	AUTOMATIC SERVICES PROVIDED CONTRACT START DATE:112/42009
Seahawk Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	PROVIDES DETALED OUTLARE OF THE ADMINISTRATIVE SERVICES THAT ARE TO BE PERFORMED UNDER THE AGREEMENT CONTRACT START DATE OF OUTLARE OF THE ADMINISTRATIVE SERVICES THAT ARE TO BE PERFORMED UNDER THE AGREEMENT
Seatheast Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	DISCLOSLINE AND ACKNOWLEDGEMENT RELATED TO AGENT COMMISSIONS UNDER THE BENEFITS PLAN
Seatured Orilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	THE STRENGT SERVICES AND STREN
Seaturest Drilling, Inc.	MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY	PRESCRIBES GLUDELINES FOR SELECTING, MONITORING & EVALUATING INVESTINENT OPTIONS UNDER THE PLAN CONTRACT START DATE EMERGEORY
Seatnewt Drilling, Inc.	MASTER RIG INTERNATIONAL LLC	STORAGE SERVICES AGREBIENT AND RELATED AMENDMENTS CONTRACT START DATE: COORDIN
Seatment Drilling, Inc.	MATTHEWS-DANIEL COMPANY	MARINE WARRANTY SERVICES CONTRACT START DATE DEBESZOD
Seahawk Drilling, Inc.	MCGRIFF, SEIBELS & WILLIAMS OF TEXAS, INC.	BROKERAGE SERVICE AGREEMENT FOR INSURANCE POLICIES CONTRACT START DATE GRAZAZON

		Inchestate survive adeptating to usure and the survivers of the survivers
	TEXANS, INC.	CONTRACTS (SENTEMENTED FOR SENIE AND TAKE FOR SENIE AND THE FIELD RECEITS OF BECKER SECONDS
Secretary Crang, etc.		CONTINET START DATE SERVICES
Seathwest Drilling, Inc.	MELLON INVESTOR SERVICES LLC	A ZIBBARTATOR AND MERINDENT INCLUDING ADMINISTRATION OF IGN-6 PLANS.
Seaturest Drilling, Inc.	MELLON INVESTOR SERVICES LLC	CONTINCE AND STATE OF TWANTER ACENT SERVICES
Seathquek Drilling, Inc.	MELLON INVESTOR SERVICES LLC	SOLICIT PROXES FOR ZNO AMBLAL MEETING CONTRACT START DATE: SMILIGHO
Seathank Drilling, Inc.	MELLON INVESTOR SERVICES LLC	CONTRACT DE URINE DATE AGREEMENT TO AGREEMENT
Seathwesk Drilling, Inc.	MCROSOFT LICENSING, GP	LIVE OF MICHOGOFT MICHOGOFTS CONTRACT START DATE MICHOGOP
Seaheavk Drilling, Inc.	MCROSOFT LICENSING, GP	URE OF MICHOGOTT PRODUCTS CONTRACT SEAT DATE RESPECTO
Seatment Drilling, Inc.	MICROSOFT LICENSING, GP	SHOWN THE ACREMENT FOR THE OTHER MICROSOFT ACREEMENTS AND AMENDMENT TO ENTERPRISE ACREEMENT CONTRACT START DATE GROWN 2009
Seahawk Driffing, Inc.	MONTCO OFFSHORE, INC.	MATTER THE CHATTER ACREDIENT CONTRACT MART DATE SHIPMON
Seathernic Drilling, Inc.	MS LEGAL SEARCH, LLC	LEGAL PEDPHANTER BHANGBRENT CONTRACT CONTRACT START DATE SIZENCE
Seahard Orling, Inc.	NANCE INTERNATIONAL, INC.	SERVE CONTINCT MATER SERVE ASSESSED.
Seatewek Drilling, Inc.	NATIONAL OILFIELD SERVICES, LLC	STRUCTURAL PRING REPLACEMENT CONTRACT START DATE: 294-2009
Seahewk Drilling, Inc.	NATIONAL OILWELL VARCO, LP	SERVE CONTRACT MATER REMOTE ARRESENT
Seaturek Drilling, Inc.	NATIXIS	FINANCIAL AGREEMENT PENCIVING CREDIT AGREEMENT; INCLUDING 1ST, 2ND, 3RD, AND 4TH AMENDMENTS TO AGREEMENT
Seatheast Drilling, Inc.	NATIXIS	CONTINUE STATE LEAVE ASSESSED. CONTINUE STATE LEAVE ASSESSED. CONTINUES OF ASSESSED.
Sestrawk Drilling, Inc.	NATDOS	SIL DOD COOK NOTE:
Seatherek Drilling, Inc.	NATIXUS	SECURITY ACREDIENT
Seathersk Drilling, Inc.	NATIOS	PLEDGE AGREDIENT CONTRAIT START DATE GRECOMS
Seatheask Drilling, Inc.	NATIOUS	CONTRACT STATE OUT TO THE CONTRACT STA
Sembsek Drilling, Inc.	NATIXIS	INTIDES IS GRANTED CONTINOL OVER EACH OF THE GRANTORS ACCOUNTS CONTINAT STANT DATE SHIPLOOD
Seathwak Orlling, Inc.	NATDUS	CONTRACT ETRAIN ETROP TO EDITIVATION CONTRACT SHARE TO EDITIVATION CONTRACT SHARE TO EDITIVATION CONTRACT SHARE ETRAIN CONTRACT SHAR
Seathwest Drilling, Inc.	NATDUS	FINANCIA, AGREDIENT GLAMANTO ON REVOLVING CREDIT AGREDMENT, INCLUDING 18T, 2ND, 3ND, AND 4TH AMENDMENTS TO AGREEMENT
September Drilling, Inc.	MANGIS, LLC	SCHWAE LORINE AND TO THE SAME AND THE SAME A
Seatment Orithns, Inc.	ME CONTRACTORS INC.	CATHER STREET ST
		LECHRE ACCESSEDED. LECHRE
Seatland Orlang, Inc.	OCEANWEATHER INC.	MARTINE REPORT OF A CHIEF OF A CH
		MATERIAL PROPERTIES. INTROGRAM INSPECTION
Seament Drinks, Inc.	OES ON FIELD SERVICES	DROPED OBLECT SURVEY CONTRACT START DATE CONTRACT
Seatmosk Orilling, Inc.	OFFSHORE EQUIPMENT SOLUTIONS	STORAGE SERVICES AGREBIENT AND RELATED AMENDMENTS CONTRACT START DATE: GDOB/2011
Seahwork Drilling, Inc.	OFFSHORE STAFFING SERVICES OF ACADIANA, LLC	SERVACE CONTRACT MANTEN BENVEC ARREEMENT CONTRACT TO ATTEMPT CANTELENT CONTRACT TO ATTEMPT CANTEND ATTEMPT CAN
Seahawk Drilling, Inc.	OFFSHORE TOWING, INC.	RIG TOWNG BENYCES CONTRACT START DATES SAZZIZES
Seathwesk Drilling, Inc.	OFFSHORE TOWING, INC.	TOWING A CAREENIENT CONTRACT STANT DATE CON YOUR CONTRACT STANT DATE CON YOUR STANT DA
Seament Drilling, Inc.	OFFSHORE TOWING, INC.	SENICE CONTRACT MASTER SERVICE AGREEMENT
Sestrawk Drilling, Inc.	OIL AND GAS RENTAL SERVICES INC	SERVCE CONTRACT MASTER SERVCE AGRESSERY
Seatheast Drilling, Inc.	OILFIELD AUDIT SERVICES, INC.	CONTRACT STATE OF THE SECOND STATE OF THE SECO
Seahawk Drilling, Inc.	OILFIELD BEARING INDUSTRIES, INC.	SERVACE CONTRACT MASTER SERVACE AGREEMENT
Seathwelt Drilling, Inc.	ORACLE USA, INC.	ASSOCIATION OF MAY POSE STRICTION OF MAY POSE STRICT AND CERTERCATION OF CONTRACT STRICT DATE (1/1/2/2008)
Seahand Drilling, Inc.	ORACLE USA, INC.	LICENSK AND SERVICES AGREDMENT AND FREATED AMENDMENT TO ORDERING DOCUMENT COMPRACT STATES THAT ITO ORDERING DOCUMENT
Seetheask Drilling, Inc.	ORACLE USA, INC.	ASBOANENT AND CERTIFICATION OF THE PROSESSION OF CONTRACT STATE TATE CONSISTOR
Seaheask Drilling, Inc.	PARTNERS CONSULTING	IT COMBILTING ACREEMENT CONTRACTS FART DATE SECURITIONS
Senhawk Drilling, Inc.	PARTNERS CONSULTING	CONTRACT STATE AND STATE
Seahawk Drilling, Inc.		POWER ON SEAMING ZOLI TOWNS AND THE CONTRACT OF THE CONTRACT O
		CONTINUE STATE DATE STATE STA

		ITMANER OF SEAMAN 2011 FROM PRICE CALIFORNIA TO SEAMAN DIRIL INC. I.C.
		CONTINCT START DATE DICENSION WANK-FACTURE AND RESTAL OF
Seatment Orlling, Inc.	PETRON NIDUSTRIES, INC.	BIG MONITORING METRIABETRATION CONTRACT START DATE SOSZIZZOO
Seatment Drilling, Inc.	PITHEY BOWES	LEARE ARRESMENT CONTINUES START DATE: DISTRICTOR
Seatent Orling, Inc.	PLATINUM EQUITY ADVISORS, LLC	THEOD MATT PROFESSIONAL ACREEMENT THANKACTONAL ACREEMENT TO CONTRACT STAFF TAKENET THE 23772010
Seahwork Driffing, Inc.	POWER PRODUCTS SERVICE CO., INC.	NUM AND COMPRESSION BALES AND SERVICE CONTRACT START DATE SECTIONS
Seahawk Drilling, Inc.	PREFERRED ELECTRIC, INC.	ELECTRO & HANG CONSTRUCTION MAINTENANCE CONTRACT ENANCEMENT CONTRACT ENANCEMENT CONTRACT ENANCEMENT CONTRACT ENANCEMENT CONTRACT
Seahewk Driffing, Inc.	PRIDE INTERNATIONAL, INC.	TAMENTON ESTATEMENTS RETWEEN PRIDE AND SEAHAWK POST SPIN-OFF AND RELATED AMENDMENT OF SPECIAL PROVISIONS TO SCH. E OF AGREEMENT CONTRACT TART TART TART TO BE THE TAMENT OF THE TAMENT OF SPECIAL PROVISIONS TO SCH. E OF AGREEMENT
Seatured Driffing, Inc.	PRIDE INTERNATIONAL, INC.	TERMS AND CONDITIONS OF SEPARATION OF SIX FROM PIL
Seaturesk Drilling, Inc.	PRIDE INTERNATIONAL, INC.	ALLOCATION OF TAXES MADING PARTIES AT DISTRIBUTION CONTINUE STAFT DATE SHOUZDS
Seatheask Drilling, Inc.	PRIDE INTERNATIONAL, INC.	CHEDITAL BEPORT OF USED FOR MEDICAN TAX OBLIGATIONS
Seatured Orlling, Inc.	PRIDE INTERNATIONAL, INC.	ARREDIENT TO EFFECT ONDERLY TRANSITION OF EMPLOYEES, BENEFITS, ETC. CONTINUES HAT DATE SHOUTED
Seahank Drilling, Inc.	PRIDE INTERNATIONAL INC.	AGREDAENT CONCERNING MADE THI ISSUES CONTRACT START DATE CONCERNING
Seatmank Drilling, Inc.	PRIDE INTERNATIONAL, INC.	AGREDIENT CONCERNING PRIDE WISCONSIN ISSUES CONTINUES TART DATE SHOULD BE SH
Seathwelt Drilling, Inc.	PRIDE TENNESSE LLC	ASSIGNMENT OF WITEREST WITHE CHARTERPARTY AND MODIFICATION OF TERMS RELATED TO PAYMENT FOR USE OF RIG CONTINUES START DATE SOGUZOOD
Seathersk Driffing, Inc.	PRIDE WISCONSIN LLC	ASSIGNMENT B. ASSUMPTION NOREBINENT AND AMENDMENT TO CHARTERPARTY AGREEMENT CONTRACT START DATE SEGULATION
Seathank Drilling, Inc.	PRIME ELECTRICAL SERVICES, LLC	SERVACE CONTRACT IMASTER SERVICE AGREEMENT
Seathersk Drilling, Inc.	PSC INDUSTRIAL OUTSOURCING, LP	NORM REPEDATION, VACUUM SERVICES, SALT WATER INJECTION, DOCKSIDE CLEANING CONTINACT STRET DATESTIZAZIO
Seathawk Drilling, Inc.	QUALITY OIL TOOLS, INC.	SERVINE CONTINUCT IMATTER REPAICE AGREEMENT
Seathersk Drilling, Inc.	REID DAVIS ILP	LEGAL SERVICES CONTRACT START DATE SECURIOR
Seahawk Drilling, Inc.	RESOURCES GLOBAL PROFESSIONALS	JOE OFTINGATION CONTINUES START DATE STITE 2009
Seatnessk Driffing, Inc.	RICHET, INC.	DATA CIRCUITS COMMUNICATION SERVICES AGREEMENT AND RELATED ADDENDA POTATALAT TATES TO ATTENDED ADDENDA
Seatheavk Drilling, Inc.	RIMANI STREET, INC.	SUPPORT FOR COALCE STYPES (ID EXPRESS)
Seatrank Drilling, Inc.	REELDRIDGE CONSTRUCTION, INC.	SERVICE CONTRACT MARTER ENEME A ARRESEUR
Seatment Drilling, Inc.	RLI INSURANCE COMPANY	INCOMINY ACREMENT FOR PERFORMANCE BONDS CONTINUES STATE DAYLOWS
Seatherek Drilling, Inc.	ROBALES, EDGAR ALEXANDER	Separation Apparent.
Seathank Drilling, Inc.	ROTOCRAFT LEASING COMPANY, LLC	HELOCHTER BETWICES FOR OFTSHORE CONTINUES TART DATE-SHEAZOID
Seatment Drilling, Inc.	ROTORCRAFT LEASING COMPANY LLC	SERVACE CONTRACT MASTER SERVACE AGREEMENT
Seaturek Drilling, Inc.	SAGE SOFTWARE, INC.	SAME FAS PRODUCT CONTRACT START DATE 1281/2010
Seatrank Orlling, Inc.	SAP AMERICA, INC.	SAP LICENSING AGREEMENT CONTRACT START DATE: 12182000
Seatment Drilling, Inc.	SAP AMERICA, INC.	DATE
Seatment Drilling, Inc.	SAVAGE DESIGN	E DESP
Sentrank Drilling, Inc.	SAVAGE DESIGN	žΑ
Seehawk Orlling, Inc.	SCHLUMBERGER INFORMATION SOLUTIONS	50
Seathank Oriting, Inc.	SCHLUMBERGER INFORMATION SOLUTIONS	≨ 8
Seathanki Orilling, Inc.	SCHELIMBERGER INFORMATION SOLUTIONS	CTIVITY SERI
Seehannk Drilling, Inc.	SEABROKERS LTD	REPRESENTATION/COMMISSION AGREEMENT FOR SALE OF 2001, 2002, 2003,
Seahawk Orilling, Inc.	SEAHAWK DRILLING DE MEXICO LLC	TRANSPER OF SEAWAY 2005 FROM PRIDE ON JAHOMA TO SEAMANN DRILLING LLC CONTRACT START DATE: 00202000
Seathaint Orfling, Inc.	SEAHAWK DRILLING DE MEDICO LLC	TRAMBER OF SEAWAY 2005 FROM PRIDE ON JHOMA TO SEAWAYK DRILLING LLC CONTRACT START DATE 0028/2009
Sembarek Drilling, Inc.	SEAMAR DIVERS, INC.	SERACE CONTRACT MASTER SERVICE AGREBAENT
Seathank Orläng, Inc.	SEATRAX SERVICES, INC.	SERVICE CONTINACT MASTER SERVICE AGREEMENT
Seathank Orlling, Inc.	SEATRAX, INC	STORAGE SERVICES AGREEMENT AND RELATED AMENDMENTS CONTINACT START DATE: COMPACT
Seethersk Drilling, Inc.	SECON	SERVACE CONTRACT WASTER SERVACE AGREEMENT
Seaturet Drilling, Inc.	SHER & BLACKWELL	LEGAL BERINCES CONTRACT START DATE:DITESZOLO

		I EGAL SÉRVICES
	SCORES INTERNATIONAL INC.	SCHING CHIRE DATE OF THE CONTRACTION
		MANTENERY AMERIENT
Seahand Drilling, Inc.	SINOR ENGINE COMPANY, INC.	WATER CONTINUE.
Seahawk Drilling, Inc.	SIRIUS SOLUTIONS, LLLP	REVIEW AND DOCUMENTATION OF INTERNAL CONTROL OVER FINANCIAL REPORTING CONTRACT START DATE SCANZOO
Seatmank Drilling, Inc.	SITRICK AND COMPANY	ENDAGRIENT LETTER SERVICES AMERICALITY CHARACTER TEAT CONTRACTOR
Seatment Drilling, fric.	SMIT SALVAGE B.V.	TERMINATION NOTICE OF SERVICE AGREEMENT COMPACT STATE ON AZONE
Seatent Driffing, Inc.	SOLUTION CRANE SERVICESMEXICO S. DE RL DE CV	CHANE SERVICES AND SUPLIES ON AND OFFSHORE
Seathwesk Driffing, Inc.	SOUTHERN SPECIALTY & SUPPLY, INC.	CONTINUE STATE LATE LATE AND THE EXPONENCE AND T
Seatment Driffing, Inc.	SOUTHERN TECHNOLOGY & SERVICES, INC.	BENEFIC CONTINCT BENEFIC CONT
Seaturest Drilling, Inc.	SOUTHWEST OILFIELD PRODUCTS, INC.	SERVIC CONTINAT MATTHER REMAYE AGREDIENT
Seament Drilling, Inc.	STUART & ASSOCIATES	LEGAL SERVICES COMMACT STATI DATE DATE DATE DATE DATE DATE DATE DATE
Seathersk Drilling, Inc.	SUNGARD AVAILABILITY SERVICES, LP	VPN, Names, SAM houling COMTNACY START DATE OBLICEDS
Seaturesk Drilling, Inc.	SUPERIOR SUPPLY & STEEL	SERVICE CONTRACT MATTER SERVICE AGREGATE
Seatherek Drilling, Inc.	SURVIVAL SYSTEMS INTERNATIONAL, INC.	STORAGE SERVICES ACREDIENT AND RELATED AMENDMENTS CONTRACT START DATE CONTRACT
Seatment Drilling, Inc.	SYNACTIVE, INC.	SAP CORBUTING SERVICES CONTRACT STANT TANTE TO YEAR OF
Seatheast Oriting, Inc.	SYNACTIVE, INC.	UCDRETO USE GLUT SOFTWARE CONTRACT START DATE 2022/2010
Seaheak Drilling, Inc.	T BAKER SMITH, INC.	SERVICE CONTINACT MATTER SERVICE CONTINACT
Seahewk Drilling, Inc.	TO ENERGY SERVICES	SENACE CONTRACT MATTRIERAGE GREEN!
Seathersk Drilling, Inc.	TALEN'S MARINE & FUEL, INC.	SERVICE CONTRACT MASTRE SERVICE ACRESSMENT
Seathersk Drilling, Inc.	TELEMAR USA, LLC	REPAIR MANNE ELECTRONICS COMTACT STAFT DATE SANIZED
Seatheask Drilling, Inc.	TESCO, INC.	SERVICE CONTRACT MATTR SERVICE AGRESSIVET
Seathannik Drilling, Inc.	THE NACHER CORPORATION	SERVICE CONTRACT MATTER SERVICE AGRESSIANT
Seabank Drilling, Inc.	THE STATE OF TEXAS - GENERAL LAND OFFICE	LAND LEAST FOR STORME OF SEAVANK ZOR RIG
Septemble Drilling, Inc.	THE TRITANIUM COMPANY BV	SNOLLS VP DATAMARE CONTINACT START DATE SOFWEXXXX
Seathereck Drilling, Inc.	THE ULTIMATE SOFTWARE GROUP, INC.	COMPACTS INTERNIT DATE AND CONSULTING SERVICES
Seathank Drilling, Inc.	THE ULTIMATE SOFTWARE GROUP, INC.	ULTIPPO BOFTWARE, AND AMENDINENTS RELATING TO DATA EXCHANGE SERVICES AND TESTING ENVIRONMENT SERVICES AGREEMENT CONTRACT START DATE SAGREDOR
Septemble Drilling, Inc.	THOMISON FINANCIAL CORPORATE BERVICES	IS WEST TART TART TART TO TE SOCIETION
Seathannik Drilling, Inc.	THOMISON FINANCIAL CORPORATE SERVICES	THOMBON BOTTHWIRE CONTINCT START DATE 2025/2009
Seathank Oriting, Inc.	T-MOBILE USA, INC.	CELL PHONE BEHINCE CONTINACT START DATE: SUBJACIOD
Seatment Drilling, Inc.	TODAY IN AMERICA	FOX BUBINESS NEWS SEGMENT CONTINACT START DATE START DA
Seathwelt Oriting, Inc.	TOTAL SAFETY U.S., INC	SERVICE CONTRACT MASTER REPLICE AGRESMENT
Seathwesk Drilling, Inc.	TRANSCONTINENTAL GAS PIPE LINE COMPANY, LLC	RELATED TO WYONEWG CLEAN UP CONTRACT START DATE: 97/20/2010
Seathardt Drilling, Inc.	TRANSCONTINENTAL GAS PIPE LINE COMPANY, LLC	SETTURIENT AGREBHENT RE WYCHRING DAMAGES CONTRACT START DATESTIZISZOO
Seathwelt Drilling, Inc.	TRIANGLE WASTE SOLUTIONS, LP	VACUUM AND CLEARING SERVICES CONTRACT START DATE!!!!
Seahawk Drilling, Inc.	TRIDRILL, INC.	NOT / UNBLARR INSPECTION CONTRACT START DATE: SENESZOR
Seatheavit Orilling, Inc.	UBS LOAN FINANCE LLC	REVOLVING CREDIT AGREEMENT CONTRACT START DATE: SIGNAZOR
Seahawk Drilling, Inc.	UHY ADVISORS TX, LLC	VALUEMARITY ASSERBIET CONTRACT STATT DATE: 1122:000
Seathanok Drilling, Inc.	UHY ADVISORS TX, LLC	IT PROCESS DOCUMENTATION SERVICES CONTRACT START DATE SYSTEM
Seathmek Drilling, Inc.	UHY ADVISORS TX, LLC	MTERNAL AUDIT CONTRACT START DATE:0227/2009
Seahawk Drilling, Inc.	UNITED MARINE SHIPYARD, INC.	SERVICE CONTRACT MASTER SERVICE AGREBAENT
Seathwek Drilling, Inc.	US UNDERWATER SERVICES, LP	WICHWATER WELDING DIVING SERVICES
Seathank Delino Inc	NEWTHER TRANSPORT INC	COMPACT STATT DATE SHOTIZOR
Contract County, Inc.	KI, INC.	MASTER REPUCE AGRESHENT ASSIGNMENT OF LUBILITY FROM PRIDE TO SEAMANK
Seminar Vinite in C.		CONTINUE STATE ON EXACTION CONTINUES ADMINISTRATION OF THE STATE OF TH
Seahmek Drilling, Inc.	VERIZON WIRELESS	CONTRACT STATT DATESTICATION

September Drilling, Inc.	WADLEIGH OFFSHORE EQUIPMENT SOLUTIONS	ENTRY ENTRY ARENET AREAN
Seathwark Driffing, Inc.	WAIGHRIYA & WAIGHRIYA	ADVICE ON TECHNICAL SUPPORT DOCUMENTS CONTINCT START DATE SOMEONS
Seatheast Driffing, Inc.	WALZEL TECHNICAL SERVICES, INC.	NOE, INSPECTIONS CONTRACT START DATE SMOOZEGE
Seatheask Driffing, Inc.	WATKINS & EAGER PLLC	LEGAL BERIOCES CONTRACT START DATE DATE DATE DATE DATE DATE DATE DAT
Seahead Oriting, Inc.	WAYPOINT BUSINESS SOLUTIONS, LP	COMBLIT TWO SERVICES AGREEMENT
Seatment Drilling, Inc.	WILD WELL CONTROL, INC.	CONTINUE BLYINGE SERVICES AND ADDENDUM RENEWING SERVICES AGREEMENT CONTINUES THAT TARE REDIED AND ADDENDUM SERVICES AGREEMENT
Seatheask Orilling, Inc.	WILL DUETT	TAMK CLEANING AND COATING CONTINCT START DATE, DAZEJZOD
Seahand Drilling, Inc.	WORKPLACE STAFFING SOLUTIONS, LLC	SERVACE CONTRACT WASTER SERVICE AGREBANT CONTRACT STATE TAXESTICAZOTI
Seathawk Drilling, Inc.	XEROX	EQUIPMENT LEAS AGREBMENT CONTRACT START DATE STIN 42008
Seatheast Orilling, Inc.	XEROX	EQUIPMENT LEASE ARRÉBINENT CONTRACT START DATE DILAYZOS
Seathwark Driffing, Inc.	ХЕРОХ	PRINTER AND FAX MACHINES CONTTACT START DATE: 1201/2008
Seatheask Driffing, Inc.	ZURICH AMERICAN INSURANCE COMPANY	INDEMNTY FOR BONDS RE: MACIENDA CONTRACT START DATE DAZEZODE
Seathment Global Holdings LLC	NATIOS	JORNOR AGREDIENT CONTRACT START DATE SEGRAZIOS
Seathwelt Global Holdings LLC	NATIXIS	ASSIGNMENT OF INSURANCE CONTRACT START DATE-DROBEZOR
Seahawk Global Holdings LLC	NATIXIS	TRINAVIAM, ACREDIBENT GLUNDANTY ON REVOLVING CREDIT AGREDMENT; INCLUDING (1ST, 2ND, 3RD, AND ATH AMENDMENTS TO AGREEMENT CONTRACTS TATHE UNDERVIOLD

EXHIBIT C

Preliminary Order on Sale Motion

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	Chapter 11
Seahawk Drilling, Inc.,	S Case No.: 11
Seahawk Drilling LLC,	S Case No.: 11
Energy Supply International LLC,	S Case No.: 11
Seahawk Global Holdings LLC,	Case No.: 11
Seahawk Mexico Holdings LLC,	Case No.: 11
Seahawk Drilling Management LLC,	S Case No.: 11
Seahawk Offshore Management LLC,	S Case No.: 11
Seahawk Drilling USA LLC,	S Case No.: 11
Debtors.	Motion for Joint Administration Pending

ORDER (i) SCHEDULING AN EXPEDITED HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS; (ii) SETTING DEADLINES FOR FILING OBJECTIONS TO (1) THE PROPOSED SALE AND (2) THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND THE PAYMENT OF CURE CLAIMS IN CONNECTION THEREWITH; AND (iii) APPROVING THE FORM AND MANNER OF NOTICE THEREOF

Upon consideration of the Emergency Motion Pursuant to Sections 105, 363 and 365 of the Bankruptcy Code and Rules 2002, 6004, 6006, 9007, and 9014 of the Federal Rules of Bankruptcy Procedure, for an Order (i) Scheduling an Expedited Hearing to Approve the Sale of Substantially all of the Debtors' Assets; (ii) Setting Deadlines for Filing Objections to (1) the Proposed Sale and (2) the Proposed Assumption and Assignment of Executory Contracts and the Payment of Cure Claims in Connection Therewith; and (iii) Approving the Form and Manner of Notice Thereof (the "Sale Motion"); the evidence presented and the arguments of counsel in support of the Sale Motion, the Court finds that: (i) the Court has jurisdiction over the Sale

Motion pursuant to 28 U.S.C. §§ 157 and 1334, and this matter is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(A); (ii) good and sufficient business reasons exist for setting an expedited hearing to consider approval of the Sale Motion; (iii) good cause exists to shorten the applicable notice periods in Bankruptcy Rules 2002, 6004, and 6006 and the applicable notice periods in the Local Bankruptcy Rules for the Southern District of Texas; and (iv) the Sale Hearing Notice attached to the Sale Motion conforms with all applicable rules, including Federal Rules of Bankruptcy Procedure 1005, 2002(l), 2002(n), and 9008, and is reasonably calculated to provide notice and an opportunity to participate for all affected parties. Therefore, it is

ORDERED that the relief requested in the Sale Motion is granted to the extent set forth herein; and it is further

ORDERED that all objections to the Sale Motion (a "Sale Objection") must set forth in writing, with particularity, the factual and legal basis for the objection and be filed, by a party with standing, on Debtors' counsel, Fulbright & Jaworski L.L.P., Attn: Berry D. Spears, Esq. and Johnathan C. Bolton, Esq., 1301 McKinney, Suite 5100, Houston, Texas 77010 and counsel to the Purchaser, Thompson & Knight L.L.P., Attn: Rhett G. Campbell, Esq., 333 Clay Street, Suite 3300, Houston, Texas 77002 and all other parties entitled to notice by 12:00 p.m. (Central Time) on the third Business Day prior to the Sale Hearing Date (the "Objection Deadline"). The failure to timely file and serve an objection by the Objection Deadline shall be a bar to any further

assertion, at the Sale Hearing or thereafter, of any objection to the Sale Motion, the sale, or the Debtors' consummation and performance of the APA; and its further;

ORDERED that except as set forth herein, unless a Sale Objection is filed and served by the Objection Deadline, all parties who have received actual or constructive notice hereof shall be deemed to have waived and released any right to assert a Sale Objection and to have otherwise consented to the proposed sale and shall be forever barred and estopped from asserting any Sale Objection against the Debtors, the Purchaser or any other assignee of them; and it is further

ORDERED that objections, if any, that relate to the proposed assumption and assignment of executory contracts (including, but not limited to, any objections relating to the validity of the cure amount as determined by the Debtors or to otherwise assert that any amounts, defaults, conditions, or pecuniary losses must be cured or satisfied under any of the assigned executory contracts as of the date of the Sale Hearing) (a "Cure Objection") shall be filed and served so as to be actually received by the Objection Deadline; and it is further

ORDERED that except as set forth herein, unless a Cure Objection is filed and served by a non-Debtor party to an executory contract proposed to be assumed and assigned by the Objection Deadline, all interested parties who have received actual or constructive notice hereof shall be deemed to have waived and released any right to assert a Cure Objection and to have otherwise consented to the assignment of the executory contracts set forth on Exhibit B to the Asset Purchase Agreement and shall be forever barred and estopped from asserting or claiming against Debtors, the Purchaser or any other assignee of them that any additional amounts are due or defaults exist, or conditions to assignment must be satisfied, under such assigned contract for the period prior to the date of the Sale Hearing; and it is further

ORDERED that the form of Sale Hearing Notice attached hereto as Exhibit 1 is hereby

approved; and it is further

ORDERED that the Debtor shall serve the Sale Hearing Notice on (a) Natixis, New York

Branch, as Administrative Agent for the Lenders; (b) the 30 largest unsecured creditors of the

Debtors on a consolidated basis; (c) Pride International, Inc.; (d) Hercules Offshore, Inc.; (e) D.E.

Shaw Direct Capital Portfolios, L.L.C.; (f) the U.S. Trustee's Office; (g) the United States

Attorney's Office; (h) the Securities and Exchange Commission; (i) the Internal Revenue Service;

(i) parties to executory contracts proposed to be assumed and assigned; (k) entities known or

reasonably believed to have expressed an interest in acquiring any of the assets offered for the

sale; (1) entities who hold or assert liens or security interests in the Purchased Assets; and (m)

entities who have requested notice pursuant to Federal Rule of Bankruptcy Procedure 2002; and it

is further

ORDERED that the Court shall retain jurisdiction over any matter or dispute arising from

or relating to the implementation of this Order.

SIGNED this the	day of	, 2011.	

UNITED STATES BANKRUPTCY JUDGE

EXHIBIT D

The Sale Hearing Notice

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	Chapter 11
Seahawk Drilling, Inc.,	Case No.: 11
Seahawk Drilling LLC,	Case No.: 11
Energy Supply International LLC,	Case No.: 11
Seahawk Global Holdings LLC,	Case No.: 11
Seahawk Mexico Holdings LLC,	Case No.: 11
Seahawk Drilling Management LLC,	Case No.: 11
Seahawk Offshore Management LLC,	Case No.: 11
Seahawk Drilling USA LLC,	Case No.: 11
Debtors.	Motion for Joint Administration Pending

NOTICE OF (i) FINAL HEARING TO APPROVE THE SALE OF SUBSTANTIALLY ALL THE DEBTORS' ASSETS TO HERCULES OFFSHORE, INC. PURSUANT TO SECTIONS 105, 363, AND 365 OF THE BANKRUPTCY CODE AND RULES 2002, 6004, 6006, 9007, AND 9014 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE AND (ii) DEADLINES FOR FILING OBJECTIONS TO (1) THE PROPOSED SALE AND (2) THE PROPOSED ASSUMPTION AND ASSIGNMENT OF EXECUTORY CONTRACTS AND PAYMENT OF CURE CLAIMS DUE IN CONNECTION THEREWITH

PLEASE TAKE NOTICE that, on February 11, 2011, Seahawk Drilling, Inc. and the above-captioned affiliated debtors (collectively, the "<u>Debtors</u>"), filed their Emergency Motion Pursuant To Sections 105, 363 and 365 of the Bankruptcy Code And Rules 2002, 6004, 6006, 9007, and 9014 of the Federal Rules Of Bankruptcy Procedure, for an Order (i) Scheduling an Expedited Hearing To Approve The Sale of Substantially All of the Debtors' Assets; (ii) Setting Deadlines for Filing Objections To (1) the Proposed Sale and (2) the Proposed Assumption and

90303148.17

Assignment of Executory Contracts and the Payment of Cure Claims in Connection Therewith; and (iii) Approving the Form and Manner of Notice Thereof (the "Sale Motion").

PLEASE TAKE FURTHER NOTICE that on February _____, 2011, the United States Bankruptcy Court for the Southern District of Texas, Corpus Christi Division (the "Court") entered an order that, among other things, set a date for the final hearing to approve the sale of substantially all of the Debtors' assets.

PLEASE TAKE FURTHER NOTICE that a complete copy of the Sale Motion is being served concurrently with this Sale Notice. The Sale Motion and related case pleadings are also publicly available on a website maintained by the Debtors' Noticing Agent, Kurtzman Carson Consultants LLC, at www.kccllc.net/seahawk and can also be accessed by contacting the Clerk of the Bankruptcy Court.

PLEASE TAKE FURTHER NOTICE that the Court has scheduled February _____, 2011, at __:____.m. (CST) before the Honorable Richard S. Schmidt, United States Bankruptcy Judge, United States Courthouse, 1133 N. Shoreline Blvd., Corpus Christi, Texas 78401, as the date for a hearing (the "Sale Hearing Date") to consider entry of an final order authorizing and approving (i) the sale of substantially all the Debtors' assets to Hercules Offshore, Inc. ("Purchaser") free and clear of all liens, claims, encumbrances and interests; and (ii) authorizing the assumption and assignment of certain executory contracts (the "Assigned Contracts") in connection therewith. The Sale Hearing may, however, be adjourned from time to time by announcement at the Sale Hearing in open Court, without notice.

PLEASE TAKE FURTHER NOTICE that the Debtors propose to sell substantially all of their assets (the "<u>Purchased Assets</u>") free and clear of any liens, claims, encumbrances, or interests pursuant to 363(f) of the United States Bankruptcy Code.

PLEASE TAKE FURTHER NOTICE that objections, if any, that relate to the proposed Assigned Contracts (including, but not limited to, any objections relating to the validity of the cure amount as determined by Debtors or to otherwise assert that any amounts, defaults, conditions, or pecuniary losses must be cured or satisfied under any of the Assigned Contracts as the Sale Hearing Date) (a "Cure Objection") shall be filed and served so as to be actually received by Debtors' counsel, Fulbright & Jaworski L.L.P., by 12:00 p.m. (Central Time) on , 2011 (the "Cure Objection Deadline").

PLEASE TAKE FURTHER NOTICE that except as limited herein, unless a Cure Objection is filed and served by a party to an Assigned Contract or a party interested in an Assigned Contract by the Cure Objection Deadline, all interested parties who have received actual or constructive notice hereof shall be deemed to have waived and released any right to assert a Cure Objection and to have otherwise consented to the assignment of the Assigned Contract and shall be forever barred and estopped from asserting or claiming that any additional amounts are due or defaults exist, or conditions to assignment must be satisfied, under such assigned contract for the period prior to the Sale Hearing Date, against the Debtors, the Purchaser, or any other assignee of the relevant Assigned Contract.

 writing and with particularity the factual and legal basis being asserted as the basis of such objection and be filed by a party with standing.

PLEASE TAKE FURTHER NOTICE that hearings with respect to Cure Objections and Sale Objections may be held (a) at the Sale Hearing; or (b) at such other date as the Court may designate.

PLEASE BE FURTHER ADVISED that all additional requests for information concerning the sale of the Purchased Assets should be in writing directed to Debtors' counsel, Berry D. Spears and Johnathan C. Bolton, Fulbright & Jaworski L.L.P., 1301 McKinney, Suite 5100, Houston, Texas 77010-3095, with a copy to Seahawk Drilling, Inc., Attn: General Counsel, 5 Greenway Plaza, Suite 2700, Houston, Texas 77046.

Dated: February , 2011

Respectfully submitted,

FULBRIGHT & JAWORSKI L.L.P.

By: _____

Berry D. Spears
State Bar No. 18893300
Johnathan C. Bolton
State Bar No. 24025260
1301 McKinney, Suite 5100
Houston, Texas 77010-3095
Telephone: (713) 651-5151

Facsimile: (713) 651-5151 Facsimile: (713) 651-5246 bspears@fulbright.com jbolton@fulbright.com

and

JORDAN, HYDEN, WOMBLE, CULBRETH & HOLZER P.C.

Shelby A. Jordan
State Bar No. 11016700
Nathanial Peter Holzer
State Bar No. 00793971
500 N. Shoreline Blvd., Suite 900
Corpus Christi, Texas 78401-0341
Telephone: (361) 653-6624
Facsimile: (361) 888-5555
sjordan@jhwclaw.com

pholzer@jhwclaw.com

ATTORNEYS FOR THE DEBTORS AND DEBTORS-IN-POSSESSION

90303148.17

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF TEXAS CORPUS CHRISTI DIVISION

In re:	§ Chapter 11
Seahawk Drilling, Inc.,	§ Case No.: 11
C, ,	§
Seahawk Drilling LLC,	§ Case No.: 11
Energy Supply International LLC,	§ Case No.: 11
Seahawk Global Holdings LLC,	§ Case No.: 11
Seahawk Mexico Holdings LLC,	§ Case No.: 11
Seahawk Drilling Management LLC,	§ Case No.: 11
Seahawk Offshore Management LLC,	§ Case No.: 11-
,	§
Seahawk Drilling USA LLC,	§ Case No.: 11
Debtors.	§ Motion for Joint Administration Pending

DECLARATION OF SIMMONS & COMPANY INTERNATIONAL

- I, Matthew G. Pilon, pursuant to 28 U.S.C. § 1746, declare as follows:
- 1. My name is Matthew G. Pilon. I am over the age of 18 and am otherwise fully competent to make this declaration. I am a Managing Director of Simmons & Company International ("Simmons"), which maintains offices at 700 Louisiana Street, Suite 1900, Houston, Texas 77002. Simmons serves as the transaction advisor to Seahawk Drilling, Inc. ("Seahawk" or the "Company") and the other debtor-subsidiaries of Seahawk in the above-captioned chapter 11 cases (together with Seahawk, the "Debtors"). I am authorized to make this Declaration on behalf of Simmons.
- 2. Except as otherwise indicated, all statements set forth in this declaration are based upon: (i) my personal knowledge, (ii) documents and other information prepared by members of

my team at Simmons or the Debtors' management, their employees, or their professionals, (iii) my review of the relevant documents, and/or (iv) my opinion based upon my experience and knowledge of financial transactions of a similar nature, the oil and gas industry, and the Debtors' operations and financial condition. If called to testify, I would testify competently to the facts set forth herein based upon my personal knowledge, my review of relevant documents, or my opinion.

- 3. In November 2010, Seahawk publicly announced that its Board of Directors (the "Board") initiated a process to explore and consider possible strategic alternatives, which included: transactions involving additional funding, recapitalization, sales of assets, or a sale or merger of Seahawk. Seahawk's Board, through its finance committee (the "Finance Committee") with regular and substantial Board and independent directors' participation, oversaw the process along with Simmons and other advisors.
- 4. The solicitation process that Simmons managed included ultimately targeting and contacting significantly more than 100 potential parties, including potential merger partners, strategic acquirers and financial investors.
- 5. Simmons initially contacted a short list of financial groups (the "<u>Tier 1</u> <u>Contacts</u>") prior to expanding the process to a larger group of financial and strategic groups (the "<u>Tier 2 Contacts</u>"). In its discussions with potential interested parties, Simmons informed potential interested parties that Seahawk was open to exploring all strategic alternatives including a sale of the Company, debt and equity investments and asset sales.
- 6. On October 6, 2010, Simmons began by contacting the Tier 1 Contacts, a short list of seventeen (17) high-graded candidates. Of these seventeen (17) candidates, six (6) of

90360459.5 - 2 -

them signed a confidentiality agreement ("<u>CA</u>"), four (4) received management presentations and one (1) submitted an indication of interest/term sheet on November 5, 2010.

- 7. On November 3, 2010, Simmons began contacting a broader group of companies, subsequent to Seahawk announcing its process to review strategic alternatives. Simmons then contacted 115 Tier 2 Contacts, which consisted of twenty-three (23) strategic and ninety-two (92) financial entities. Of this broader group, fifty-one (51) received the summary memorandum and thirty-seven (37) received the CA, of which seventeen (17) CAs were executed. Eleven (11) of these interested parties received management presentations either in-person, electronically or telephonically; eight (8) submitted indications of interest, and four (4) of these indications were summarized and presented to Seahawk's Finance Committee on November 30, 2010. An additional three (3) subsequently submitted term sheets. The group that had submitted a term sheet on November 5, 2010 submitted a revised financing term sheet on November 17, 2010, but declined to submit a new indication of interest on November 29, 2010.
- 8. Subsequent to the receipt of the proposals at the end of November 2010, Seahawk negotiated with one party ("Company A") through mid-December 2010. On December 17, 2010, Seahawk signed a letter of intent ("LOI") and entered into exclusive negotiations with Company A. From early December until January 21, 2011, Seahawk worked exclusively with Company A to effect a transaction during this period including the negotiation of an asset purchase agreement. On January 20, 2011, Company A emailed Seahawk a revised proposal, and upon the Board's consideration of the proposal, which had been substantially revised, Seahawk responded that this new proposal was unacceptable and suggested to Company A that it submit a proposal structured with terms similar to those originally contained in the LOI. On

90360459.5

January 21, 2011, Company A responded to Seahawk with another revised proposal that the Company deemed to be even worse than their revised proposal.

- 9. After receiving this lower proposal, Seahawk's Board immediately terminated the LOI with Company A. The Board then authorized Simmons to contact several parties that had previously expressed interest in acquiring Seahawk. Simmons thereafter sent follow-up information packages to these companies, including updated financials, draft asset purchase agreements and other ancillary documentation and requested that they respond by January 28, 2011. On or before January 28, 2011, Simmons had received proposals from Hercules Offshore, Inc. ("Hercules") and one other bidder.
- 10. In February 2011, after evaluating the indications of interest and potential offers received from the interested parties, the Debtors ultimately entered into an Asset Purchase Agreement (the "APA") with Hercules. The executed APA contemplates the acquisition by Hercules or one or more of its subsidiaries ("Purchaser") of substantially all of the assets and jackup rigs of the Debtors (the "Purchased Assets") through a sale pursuant to section 363 of the Bankruptcy Code (the "Transaction"). The aggregate consideration for the Purchased Assets (defined in the APA as the "Base Aggregate Consideration") is (a) 22,321,425 shares of Hercules Common Stock (the "Hercules Shares") plus (b) cash in an amount equal to \$25,000,012 (the "Cash Payment"), subject to certain adjustments. Using the closing stock price of Hercules' stock as of February 10, 2011, the Base Aggregate Consideration would be valued at approximately \$105 million before any adjustments. The Base Aggregate Consideration is to be payable at closing by Purchaser to Sellers.

90360459.5 - 4 -

¹ The summary description of the APA set forth herein is qualified in its entirety by reference to the terms of the APA itself.

11. Simmons believes that the Debtors proposed course of action to seek the Bankruptcy Court's approval of the APA with Hercules is the best alternative to ensure that the maximum value of their assets will inure to the benefit of all of their creditors and interestholders.

I declare under penalty of perjury that the foregoing is true and correct.

Executed on this the 11^{Tk} day of February, 2011, at Houston, Texas.

Matthew G. Pilon